

Ĺ

National Library of Canada

Canadian Theses Service

Bibliothèque nationale

Service des thèses canadiennes

Ottawa, Canada K1A 0N4

NOTICE

The quality of this microform is heavily dependent upon the quality of the original thesis submitted for microfilming. Every effort has been made to ensure the highest quality of reproduction possible.

If pages are missing, contact the university which granted the degree.

Some pages may have indistinct print especially if the original pages were typed with a poor typewriter ribbon or if the university sent us an inferior photocopy.

Reproduction in full or in part of this microform is governed by the Canadian Copyright Act, R.S.C. 1970, c. C-30, and subsequent amendments.

AVIS

La qualité de cette microforme dépend grandement de la qualité de la thèse soumise au microfilmage. Nous avons tout fait pour assurer une qualité supérieure de reproduction.

S'il manque des pages, veuillez communiquer avec l'université qui a conféré le grade.

La qualité d'impréssion de certaines pages peut laisser à désirer, surtout si les pages originales ont été dactylographiées à l'aide d'un ruban usé ou si l'université nous a fait parvenir une photocopie de qualité inférieure.

La reproduction, même partielle, de cette microforme est soumise à la Loi canadienne sur le droit d'auteur, SRC 1970, c. C-30, et ses amendements subséquents.



INDUSTRY ROTATION IN THE U.S. STOCK MARKET : AN

APPLICATION OF MULTIPERIOD PORTFOLIO THEORY

by

Frederick C. Shen

B.A., Simon Fraser University, 1977

M.A., Simon Fraser University, 1981

THESIS SUBMITTED IN PARTIAL FULFILLMENT OF

THE REQUIREMENTS FOR THE DEGREE OF

DOCTOR OF PHILOSOPHY

in the Department

of

Economics

Frederick C. Shen 1988

SIMON FRASER UNIVERSITY

September 1988

All rights reserved. This work may not be reproduced in whole or in part, by photocopy or other means, without permission of the author. Permission has been granted to the National Library of Canada to microfilm this thesis and to lend or sell copies of the film.

The author (copyright owner) has reserved other publication rights, and neither the thesis nor extensive extracts from it may be printed or otherwise reproduced without his/her written permission. L'autorisation a été accordée à la Bibliothèque nationale du Canada de microfilmer cette thèse et de prêter ou de vendre des exemplaires du film.

L'auteur (titulaire du droit d'auteur) se réserve les autres droits de publication; ni la thèse ni de longs extraits de celle-ci ne doivent être imprimés ou autrement reproduits sans son autorisation écrite.

ISBN 0-315-48867-0

Name: Frederick Chong Shen

Degree: Ph.D. (Economics)

Title of Thesis: Industry Rotation in the U.S. Stock Market: An Application of Multiperiod Portfolio Theory

Examining Committee:

Chairman: John F. Chant

Dr. Robert R. Grauer Professor Senior Supervisor

Dr. Pao L. Cheng Professor

Dr. John P. Herzog Professor

Ør. J. Cheung Associate Professor

Dr. E.H. Neave Professor School of Business Queen's University External Examiner

.

Date Approved: 23 September 1988

I hereby grant to Simon Fraser University the right to lend my thesis, project or extended essay (the title of which is shown below) to users of the Simon Fraser University Library, and to make partial or single copies only for such users or in response to a request from the library of any other university, or other educational institution, on its own behalf or for one of its users. I further agree that permission for multiple copying of this work for scholarly purposes may be granted by me or the Dean of Graduate Studies. It is understood that copying or publication of this work for financial gain shall not be allowed without my written permission.

Title of Thesis/Project/Extended Essay

Industry Rotation in the U.S. Stock Market: An Application

of Multiperiod Portfolio Theory

Author:

(signature)

Frederick Chong Shen

(name)

December 2, 1988

(date)

In this thesis we apply the simple reinvestment version of multiperiod portfolio theory to the construction and rebalancing of portfolios composed of the industry sectors of the U.S. stock market and a risk-free asset. In particular, we use this model to examine portfolio choices of investors with utility functions of the isoelastic class and with beliefs specified by a joint probability distribution of asset returns.

Previous studies of this type employed an investment universe composed of major U.S., as well as international, asset categories. With the exception of one study, a simple probability assessment approach was used to represent investors' beliefs. We present a natural extension of these works. First, we examine the efficacy of multiperiod portfolio theory in asset allocation at a more micro level. This is of interest since industry analyses, without the use of multiperiod portfolio theory, have long been employed by the investment community. Second, as in previous studies, we initially employ a simple probability assessment approach where the joint return distribution is estimated from past realized returns, but we extend this in several ways.

In order to examine the robustness of our results, we examine investment universes of eight, twelve and twenty-four equal- and value-weighted industry groupings of the U.S. stock market. The joint return distributions are estimated using realized returns of the most recent 28, 32 and 40 quarters with each vector of returns given an equal probability of occurrence in the next quarter. Then this approach is extended in several ways. First, we include "all of history" as our estimate of The return distribution. Second, we employ an inflation adapter to adjust the estimated return distribution. Third, we examine a "disaster states" scenario, and fourth, we alter the

iii

probabilities of the joint return distribution giving more weight to the . recent past.

The results show that the portfolio choices performed well in both the full 1934-86 period and in the 1966-86 sub-period, achieving both economically and statistically significant excess returns in several instances. For the most part, the portfolio returns were not statistically different: (1) when 28, 32, or 40 quarters of returns were employed as estimates of the joint return distribution, and (2) when eight, twelve, or twenty-four industry groupings were considered. Finally, the simple probability assessment approach has much to recommend it since it did as well as the other approaches in estimating the return distribution.

iv

ACKNOWLEDGMENTS

The author is grateful to Rob Grauer for his patient assistance and guidance on this thesis. Thanks are also extended to John Herzog and Pao Cheng for providing helpful comments. Last, but not least, I am indebted to Simon Ng for his able assistance in times of need and for many invaluable suggestions. As always, any remaining errors are the property of the author.

TABLE OF CONTENTS

.

| Appr | oval . | |
|-------|---------|---|
| Abst | ract . | |
| Ackn | owledg | ements v |
| List | of Fi | gures |
| List | of Ta | bles vill |
| | Ι. | Introduction 1 |
| | II. | The Multiperiod Portfolio Model 11 |
| | III. | Literature Review 17 |
| | IV. | Model Inputs and Statistical Tests of Investment Performance 25 |
| | ۷. | The Data |
| | VI. | The Results: Portfolio Returns and Compositions |
| | VII. | The Results: Statistical Tests of Investment Performance 102 |
| | VIII. | Summary and Extensions 116 |
| Bibli | lograph | ny 241 |

LIST OF FIGURES

,

wp

| 1a. | . Geometric Means and Standard Deviations of Annual Portfolio Returns for 10 Power Policies with Twelve Equal-Weighted Industry Indices, 1934-86 (Quarterly revision, with leverage, 32 quarter estimating period) | 232 |
|-----|--|-----|
| 1b. | Geometric Means and Standard Devia <u>ti</u> ons of Annual Portfolio Returns for 10 Power Policies with Twelve Equal-Weighted Industry Indices, 1966-86 (Quarterly revision, with leverage, 32 quarter estimating period) | 233 |
| 1c. | Geometric Means and Standard Deviations of Annual Portfolio Returns for 10 Power Policies with Twelve Equal-Weighted Industry Indices, 1934-86 (Quarterly revision, without leverage, 32 quarter estimating period) | 234 |
| 1d. | Geometric Means and Standard Deviations of Annual Portfolio Returns for 10 Power Policies with Twelve Equal-Weighted Industry Indices, 1966-86 (Quarterly revision, without leverage, 32 quarter estimating period) | 235 |
| 2a. | Geometric Means and Standard Deviations of Annual Portfolio Returns for 10 Power Policies with Twelve Value-Weighted Industry Indices, 1934-86 (Quarterly revision, with leverage, 32 quarter estimating period) | 236 |
| 2b. | Geometric Means and Standard Deviations of Annual Portfolio Returns for 10 Power Policies with Twelve Value-Weighted Industry Indices, 1966-86 (Quarterly revision, with leverage, 32 quarter estimating period) | 237 |
| 2c. | Geometric Means and Standard Deviations of Annual Portfolio Returns for 10 Power Policies with Twelve Value-Weighted Industry Indices, 1934-86 (Quarterly revision, without leverage, 32 quarter estimating period) | 238 |
| 2d. | Geometric Means and Standard Deviations of Annual Portfolio Returns for 10 Power Policies with Twelve Value-Weighted Industry Indices, 1966-86 (Quarterly revision, without leverage, 32 quarter estimating period) | 239 |
| 3. | Average Excess Returns and Betas of Quarterly Portfolio Returns for 10 Power Policies With Twelve Equal-Weighted Industry Indices and Equal- and Value-Weighted Benchmarks, 1966-86 (32 quarter estimating period, with leverage) | 240 |

LIST OF TABLES

| la. | Summary Description of the 12 Industry Indices | 122 |
|-------------|--|----------|
| 1b. | Summary Description of the 8 Industry Indices | 123 |
| 1c. | Summary Description of the 24 Industry Indices | 124 |
| 2. | Industry Categories and Benchmark Portfolio Symbols | 126 |
| 3a. | Geometric Means and Standard Deviations of Annual Returns using 12 Equal-Weighted Industry Indices, with Leverage (32 quarter estimating period, simple probability assessment) | 127 |
| 36. | Geometric Means and Standard Deviations of Annual Returns using 12 Equal-Weighted Industry Indices, without Leverage (32 quarter estimating period, simple probability assessment) | 128 |
| 4a. | Geometric Means and Standard Deviations of Annual Returns using 12 Value-Weighted Industry Indices, with Leverage (32 quarter estimating period, simple probability assessment) | ° 129 |
| 4b. | Geometric Means and Standard Deviations of Annual Returns using 12 Value-Weighted Industry Indices, without Leverage (32 quarter estimating period, simple probability assessment) | 130 |
| 5a. | Geometric Means and Standard Deviations of Annual Returns using 8 Equal-Weighted Industry Indices, with Leverage (32 quarter estimating period, simple probability assessment) | 131 |
| 5b. | Geometric Means and Standard Deviations of Annual Returns using 8 Equal-Weighted Industry Indices, without Leverage (32 quarter estimating period, simple probability assessment) | 132 |
| 5c. | Geometric Means and Standard Deviations of Annual Returns using 8 Value-Weighted Industry Indices, with Leverage (32 quarter estimating period, simple probability assessment) | 133 |
| 5d. | Geometric Means and Standard Deviations of Annual Returns using 8 Value-Weighted Industry Indices, without Leverage (32 quarter estimating period, simple probability assessment) | 134 |
| 6 a. | Geometric Means and Standard Deviations of Annual Returns using 24 Equal-Weighted Industry Indices, with Leverage (32 quarter estimating period, simple probability assessment) | 135 |
| 6b. | Geometric Means and Standard Deviations of Annual Returns using 24 Equal-Weighted Industry Indices, without Leverage (32 quarter estimating period, simple probability assessment) | 136 |

| 6c. | Geometric Means and Standard Deviations of Annual Returns using 24 Value-Weighted Industry Indices, with Leverage (32 quarter estimating period, simple probability assessment) | 137 |
|----------------|--|------------|
| . 6d. ' | Geometric Means and Standard Deviations of Annual Returns using 24 Value-Weighted Industry Indices, without Leverage (32 quarter estimating period, simple probability assessment) | 138 |
| 7a: | Geometric Means and Standard Deviations of Annual Returns using -12 Equal-Weighted Industry Indices, with Leverage, 1936-86: 32, 28 and 40 Quarter Estimating Periods (simple probability assessment) | 139 |
| 7b. | Geometric Means and Standard Deviations of Annual Returns using 12 Equal-Weighted Industry Indices, with Leverage, 1966-86: 32, 28 and 40 Quarter Estimating Periods (simple probability assessment) | <u>140</u> |
| 7c. | Geometric Means and Standard Deviations of Annual Returns using 12 Equal-Weighted Industry Indices, without leverage, 1936-86: 32, 28 and 40 Quarter Estimating Periods (simple probability assessment) | 141 |
| 7d. | Geometric Means and Standard Deviations of Annual Returns using 12 Equal-Weighted Industry Indices, without leverage, 1966-86: 32, 28 and 40 Quarter Estimating Periods (simple probability assessment) | 142 |
| 8a. | Geometric Means and Standard Deviations of Annual Returns using 12 Value-Weighted Industry Indices, with Leverage, 1936-86: 32, 28 and 40 Quarter Estimating Periods (simple probability assessment) | 143 |
| 8b. | Geometric Means and Standard Deviations of Annual Returns using 12 Value-Weighted Industry Indices, with Leverage, 1966-86: 32, 28 and 40 Quarter Estimating Periods (simple probability assessment) | 144 |
| 8c. | Geometric Means and Standard Deviations of Annual Returns using 12 Value-Weighted Industry Indices, without leverage, 1936-86: 32, 28 and 40 Quarter Estimating Periods (simple probability assessment) | 145 |
| 8d. | Geometric Means and Standard Deviations of Annual Returns using 12 Value-Weighted Industry Indices, without leverage, 1966-86: 32, 28 and 40 Quarter Estimating Periods (simple probability assessment) | 146 |
| 9a. | Geometric Means and Standard Deviations of Annual Returns_using 12 Equal-Weighted Industry Indices, with Leverage, 1934-86: Inflation Adapter vs. Sum-of-the-Digits vs. All-of-History (32 quarter estimating period) | L47 |

| 9b. | Geometric Means and Standard Deviations of Annual Returns using 12 Equal-Weighted Industry Indices, with Leverage, 1966-86 Inflation Adapter vs. Sum-of-the-Digits vs. All-of-History (32 quarter estimating period) | 148 |
|---------------|---|-----|
| 9c. | Geometric Means and Standard Deviations of Annual Returns using 12 Equal-Weighted Industry Indices, without Leverage, 1934-86: Inflation Adapter vs. Sum-of-the-Digits vs. All-of-History (32 quarter estimating period) | 149 |
| 9d. | Geometric Means and Standard Deviations of Annual Returns using 12 Equal-Weighted Industry Indices, without Leverage, 1966-86: Inflation Adapter vs. Sum-of-the-Digits vs. All-of-History (32 quarter estimating period) | 150 |
| 10 a . | Geometric Means and Standard Deviations of Annual Returns using 12 Value-Weighted Industry Indices, with Leverage, 1934-86: Inflation Adapter vs. Sum-of-the-Digits vs. All-of-History (32 quarter estimating period) | 151 |
| 10b. | Geometric Means and Standard Deviations of Annual Returns using - 12 Value-Weighted Industry Indices, with Leverage, 1966-86: Inflation Adapter vs. Sum-of-the-Digits vs. All-of-History (32 quarter estimating period) | 152 |
| 10c. | Geometric Means and Standard Deviations of Annual Returns using 12 Value-Weighted Industry Indices, without Leverage, 1934-86: Inflation Adapter vs. Sum-of-the-Digits vs. All-of-History (32 quarter estimating period) | 153 |
| 10d. | Geometric Means and Standard Deviations of Annual Returns using 12 Value-Weighted Industry Indices, without Leverage, 1966-86: Inflation Adapter'vs. Sum-of-the-Digits vs. All-of-History (32 quarter estimating period) | 154 |
| 11a. | Geometric Means and Standard Deviations of Annual Returns using 12 Equal-Weighted Industry Indices, with Eeverage, 1934-86: Disaster States Scenario for $J = 1$, 6, and 12 (32 quarter estimating period, simple probability assessment) | 155 |
| 11b. | Geometric Means and Standard Deviations of Annual Returns using 12 Equal-Weighted Industry Indices, with Leverage, 1966-86: Disaster States Scenario for J = 1, 6, and 12 (32 quarter estimating period, simple probability assessment) | 156 |
| 11c. | Geometric Means and Standard Deviations of Annual Returns using 12 Equal-Weighted Industry Indices, without Leverage, 1934-86: Disaster States Scenario for J = 1, 6, and 12 (32 quarter estimating period, simple probability assessment) | 157 |
| | | |

х

,

| 11d. | Geometric Means and Standard Deviations of Annual Returns using 12 Equal-Weighted Industry Indices, without Leverage, 1966-86: Disaster States Scenario for $J = 1$, 6, and 12 (32 quarter estimating period, simple probability assessment) 158 |
|-------|---|
| 12a. | Geometric Means and Standard Deviations of Annual Returns using 12 Value-Weighted Industry Indices, with Leverage, 1934-86: Disaster States Scenario for J = 1, 6, and 12 (32 quarter estimating period, simple probability assessment) |
| [12b. | Geometric Means and Standard Deviations of Annual Returns using 12 Value-Weighted Industry Indices, with Leverage, 1966-86: Disaster States Scenario for J = 1, 6, and 12 (32 quarter estimating period, simple probability assessment) |
| 12c. | Geometric Means and Standard Deviations of Annual Returns using 12 Value-Weighted Industry Indices, without Leverage, 1934-86: Disaster States Scenario for $J = 1$, 6, and 12 (32 quarter estimating period, simple probability assessment) 161 |
| 12d. | Geometric Means and Standard Deviations of Annual Returns using 12 Value-Weighted Industry Indices, without Leverage, 1966-86: Disaster, States Scenario for $J = 1$, 6, and 12 (32 quarter estimating period, simple probability assessment) 162 |
| 13a. | Portfolio Composition and Realized Returns for Power -15 Managing 12 Equal-Weighted Industries, with leverage, 1934-86 (32 quarter estimating period, simple probability assessment approach) |
| 13b. | Portfolio Composition and Realized Returns for Power O Managing 12 Equal-Weighted Industries, with leverage, 1934-86 (32 quarter estimating period, simple probability assessment approach) |
| 14a. | Portfolio Composition and Realized Returns for Power -15 Managing 12 Equal-Weighted Industries, without leverage, 1934-86 (32 quarter estimating period, simple probability * assessment approach) |
| 145. | Portfolio Composition and Realized Returns for Power O Managing 12 Equal-Weighted Industries, without leverage, 1934-86 (32 quarter estimating period, simple probability assessment approach) |
| 15a. | Portfolio Composition and Realized Returns for Power -15 Managing 12 Value-Weighted Industries, with leverage, 1934-86 (32 quarter estimating period, simple probability assessment approach) |

xi

| | · · · · · · · · · · · · · · · · · · · | |
|---------------|---|-----|
| 15b. | Portfolio Composition and Realized Returns for Power O Managing 12 Value-Weighted Industries, with leverage, 1934-86 (32 quarter estimating period, simple probability assessment approach) | 173 |
| 16a. | Portfolio Composition and Realized Returns for Power -15 Managing 12 Value-Weighted Industries, without leverage, 1934-86 (32 quarter estimating period, simple probability assessment approach) | 175 |
| 16b. | Portfolio Composition and Realized Returns for Power O Managing 12 Value-Weighted Industries, without leverage, 1934-86 (32 quarter estimating period, simple probability assessment approach) | 177 |
| 17a. | Summary of Portfolio Compositions of Power Policies Managing 12 Equal-Weighted Industries, With Leverage, 1934-86 (32 quarter estimating period, simple probability assessment approach) | 179 |
| 17b. | Summary of Portfolio Compositions of Power Policies Managing 12 Equal-Weighted Industries, Without Leverage, 1934-86 (32 quarter estimating period, simple probability assessment approach) | 180 |
| 17c. | Summary of Portfolio Compositions of Power Policies Managing 12 Value-Weighted Industries, With Leverage, 1934-86 (32 quarter estimating period, simple probability assessment approach) | 181 |
| 17d. | Summary of Portfolio Compositions of Power Policies Managing 12 Value-Weighted Industries, Without Leverage, 1934-86 (32 quarter estimating period, simple probability assessment approach) | 182 |
| 18 a . | Summary of Portfolio Compositions of Power Policies Managing 8 Equal-Weighted Industries, With Leverage, 1934-86 (32 quarter estimating period, simple probability assessment approach) | 183 |
| 18b. | Summary of Portfolio Compositions of Power Policies Managing 8 Value-Weighted Industries, With Leverage, 1934-86 (32 quarter estimating period, simple probability assessment approach) | 184 |
| 19a. | Summary of Portfolio Compositions of Power Policies Managing 24 Equal-Weighted Industries, With Leverage, 1934-86 (32 quarter estimating period, simple probability assessment approach) | 185 |
| 19b. | Summary of Portfolio Compositions of Power Policies Managing 24 Value-Weighted Industries, With Leverage, 1934-86 (32 quarter estimating period, simple probability assessment approach) | 186 |
| 20a. | Summary of Portfolio Compositions of Power Policies Managing 12 Equal-Weighted Industries, With Leverage, 1936-86 (28 quarter estimating period, simple probability assessment approach) | 187 |
| | | |

)

,

,

| 20b. | Summary of Portfolio Compositions of Power Policies Managing 12 Value-Weighted Industries, With Leverage, 1936-86 (28 quarter estimating period, simple probability assessment approach) | 188 |
|---------------|--|-----|
| 21a. | Summary of Portfolio Compositions of Power Policies Managing 12 Equal-Weighted Industries, With Leverage, 1936-86 (40 quarter estimating period, simple probability assessment approach) | 189 |
| 21b. | Summary of Portfolio Compositions of Power Policies Managing 12 Value-Weighted Industries, With Leverage, 1936-86 (40 quarter estimating period, simple probability assessment approach) | 190 |
| 22a. | Summary of Portfolio Compositions of Power Policies Managing 12 Equal-Weighted Industries, With Leverage, 1934-86 (32 quarter estimating period, inflation adapter method) | 191 |
| 22b. | Summary of Portfolio Compositions of Power Policies Managing 12 Value-Weighted Industries, With Leverage, 1934-86 (32 quarter estimating period, inflation adapter method) | 192 |
| 23a. | Summary of Portfolio Compositions of Power Policies Managing 12 Equal-Weighted Industries, With Leverage, 1934-86 (32 quarter estimating period, sum-of-the-digits probability assessment approach) | 193 |
| 23b. | Summary of Portfolio Compositions of Power Policies Managing 12 Value-Weighted Industries, With Leverage, 1934-86 (32 quarter estimating period, sum-of-the-digits probability assessment approach) | 194 |
| 24a. | Summary of Portfolio Compositions of Power Policies Managing 12 Equal-Weighted Industries, With Leverage, 1934-86 (32 quarter estimating period, all-of-history simple probability assessment approach) | 195 |
| 24b. | Summary of Portfolio Compositions of Power Policies Managing 12 Value-Weighted Industries,' With Leverage, 1934-86 (32 quarter estimating period, all-of-history simple probability assessment approach) | 196 |
| 25a. | Summary of Portfolio Compositions of Power Policies Managing 12 Equal-Weighted Industries, With Leverage, 1934-86 (32 quarter estimating period, disaster state scenario J = 1, simple probability assessment approach) | 197 |
| 25 <u>b</u> . | Summary of Portfolio Compositions of Power Policies Managing 12 Value-Weighted Industries, With Leverage, 1934-86 (32 quarter estimating period, disaster state scenario J = 1, simple probability assessment approach) | 198 |

| 26a. | Summary of Portfolio Compositions of Power Policies Managing 12 Equal-Weighted Industries, With Leverage, 1934-86 (32 quarter estimating period, disaster state scenario $J = 6$, simple probability assessment approach) |
|------|---|
| 26b. | <pre>Summary of Portfolio Compositions of Power Policies Managing 12 Value-Weighted Industries, With Leverage, 1934-86 (32 quarter estimating period, disaster state scenario J = 6, simple probability assessment approach)</pre> |
| 27a. | <pre>Summary of Portfolio Compositions of Power Policles Managing 12 Equal-Weighted Industries, With Leverage, 1934-86 (32 quarter estimating period, disaster state scenario J = 12, simple probability assessment approach)</pre> |
| 27b. | Summary of Portfolio Compositions of Power Policies Managing 12 Value-Weighted Industries, With Leverage, 1934-86 (32 quarter estimating period, disaster state scenario J = 12, simple probability assessment approach) |
| 28. | Paired <i>t</i> -tests : Selected Active Strategies Managing 12 Equal-Weighted Industry Indices vs. Passive and Semi-passive Strategies (32 quarter estimation period, simple probability assessment approach) |
| 29. | Paired <i>t</i> -tests : Selected Active Strategies Managing 12 Value-Weighted Industry Indices vs. Passive Strategies (32 quarter estimation period, simple probability assessment approach) |
| 30a. | Paired t-tests : 12 vs. 8 Industry Universe, Equal-Weighted (32 quarter estimation period, simple probability assessment approach) |
| 30b. | Paired t-tests : 12 vs. 8 Industry Universe, Value-Weighted (32 quarter estimation period, simple probability assessment approach) |
| 31a. | Paired t-tests : 12 vs. 24 Industry Universe, Equal-Weighted (32 quarter estimation period, simple probability assessment approach) |
| 31b. | Paired t-tests : 12 vs. 24 Industry Universe, Value-Weighted (32 quarter estimation period, simple probability assessment approach) |
| 32a. | Paired t-tests : 32 vs. 28 Quarter Estimating Period, 12 Equal-Weighted Industry Indices (simple probability assessment approach) |

xiv

| 32b. | Paired t-tests : 32 vs. 28 Quarter Estimating Period, 12 Value-Weighted Industry Indices (simple probability assessment approach) | 211 |
|-------|---|-----|
| 33a. | Paired t-tests : 32 vs. 40 Quarter Estimating Period, 12 Equal-Weighted Industry Indices (simple probability assessment approach) | 212 |
| 33b. | Paired <i>t</i> -tests : 32 vs. 40 Quarter Estimating Period, 12 Value-Weighted Industry Indices (simple probability assessment approach) | 213 |
| 34a. | Paired t-tests : Simple Probability Assessment Approach vs. Inflation Adapter to Simple Probability Assessment Approach, 12 Equal-Weighted Industry Indices (32 quarter estimating period) | 214 |
| 34b. | Paired t-tests : Simple Probability Assessment Approach vs. Inflation Adapter to Simple Probability Assessment Approach, 12 Value-Weighted Industry Indices (32 quarter estimating period) | 215 |
| 35a. | Paired <i>t</i> -tests : Simple Probability Assessment Approach vs. "Sum-of-the-Digits" Probability Assessment Approach, 12 Equal-Weighted Industry Indices (32 quarter estimating period) | 216 |
| 35b. | Paired t-tests : Simple Probability Assessment Approach vs. "Sum-of-the-Digits" Probability Assessment Approach, 12 Value-Weighted Industry Indices (32 quarter estimating period) | 217 |
| 36a. | Paired t-tests : Simple Probability Assessment Approach vs. "All-of-History" Probability Assessment Approach, 12 Equal-Weighted Industry Indices (32 quarter estimating period) | 218 |
| 36b., | Paired t-tests : Simple Probability Assessment Approach vs. "All-of-History" Probability Assessment Approach, 12 Value-Weighted Industry Indices (32 quarter estimating period) | 219 |
| 37a. | Paired t-tests : Simple Probability Assessment Approach vs. "Disaster States" Scenario When $J = 1$, 12 Equal-Weighted Industry Indices (32 quarter estimating period) | 220 |
| 37b. | Paired t-tests : Simple Probability Assessment Approach vs. "Disaster States" Scenario When J = 1, 12 Value-Weighted Industry Indices (32 quarter estimating period) | 221 |

xv

| 38. | Summary Results of Jensen's Performance Index Regression When the 10 Active Strategies Manage 12 Equal-Weighted Industries (32 quarter estimating period, simple probability assessment) | 222 |
|----------|--|-------|
| 39. | Summary Results of Jensen's Performance Index Regression When the 10 Active Strategies Manage 12 Value-Weighted Industrics (32 quarter estimating period, simple probability assessment) | 223 |
| 40a | Summary Results of Treynor-Mazuy's Market Timing Regression Without a Correction for Heteroscedasticity When the 10 Active Strategies Manage 12 Equal-Weighted Industries (32 quarter estimating period, simple probability assessment) | 224 |
| 40b - | Summary Results of Treynor-Mazuy's Market Timing Regression With a Correction for Heteroscedasticity When the 10 Active Strategies Manage 12 Equal-Weighted Industries (32 quarter estimating period, simple probability assessment) | 2.2.5 |
| 41a | Summary Results of Treynor-Mazuy's Market Timing Regression Without a Correction for Heteroscedasticity When the 10 Active Strategies Manage 12 Value-Weighted Industries (32 quarter estimating period, simple probability assessment) | 226 |
| 41b | Summary Results of Treynor-Mazuy's Market Timing Regression With a Correction for Heteroscedasticity When the 10 Active Strategies Manage 12 Value-Weighted Industries (32 quarter estimating period, simple probability assessment) | 227 |
| 42a | Summary Results of Henriksson-Merton's Market Timing Regression Without a Correction for Heteroscedasticity When the 10 Active Strategies Manage 12 Equal-Weighted Industrics (32 quarter estimating period, simple probability assessment) | 228 |
| 42b | Summary Results of Henriksson-Merton's Market Timing Regression With a Correction for Heteroscedasticity When the 10 Active Strategies Manage 12 Equal-Weighted Industries (32 quarter estimating period, simple probability assessment) | 229 |
| 43a | Summary Results of Henriksson-Merton's Market Timing Regression Without a Correction for Heteroscedasticity When the 10 Active Strategies Manage 12 Value-Weighted Industries (32 quarter estimating period, simple probability assessment) | 230 |
| 43b | Summary Results of Henriksson-Merton's Market Timing Regression With a Correction for Heteroscedasticity When the 10 Active Strategies Manage 12 Value-Weighted Industries (32 quarter estimating period, simple probability assessment) | 231 |

The management of investment portfolios has traditionally consisted of three major activities:¹ (1) asset allocation, (2) group rotation, and (3) security selection. The first stage, asset allocation, can be characterized as the apportioning of funds between major asset categories in an attempt to obtain higher returns at lower risk in the long run. For example, the fund manager 'can invest either domestically in common 'stocks, government bonds. corporate bonds and Treasury bills, or internationally in bond and equity markets of different countries, or some combination of both. An extension of asset allocation is the "pure market timing" strategy where funds are shifted from equities to a riskless asset and vice versa. In group rotation, securities with common characteristics are combined to form various major groupings of equities; for example, the various industry components of the U.S. stock market, or as in Farrell's (1974) classification of growth, cyclical, stable and energy stocks. As in asset allocation, opportunistic shifts between these groupings of securities are made with the objective of higher long run returns and lower risk. Finally, in the security selection stage, the fund manager chooses, from among the many securities that make up the major asset classes or groupings, those that have above average return to risk prospects.

In this thesis we focus on stage two of the investment process. In particular, we apply the multiperiod portfolio theory of Mossin (1968), Hakansson (1971c,1974), Leland (1972), Ross (1974) and Huberman and Ross (1983) to the construction and rebalancing of portfolios composed of the

¹For example, see Sharpe (1985).

industry sectors of the U.S. stock market and a risk-free asset. While there are studies examining the allocation of funds across major asset categories using the model drawn from multiperiod portfolio theory (see Grauer and Hakansson (1981,1982,1984,1985,1986,1987,1988)), there are none applying this model to industry/group rotation. Most studies in industry analysis have concerned themselves with the question of whether there are any benefits to be had from such analysis (see, for example, King (1966), Latané and Tuttle (1968), Brigham and Pappas (1969), Tysseland (1971), Meyers (1973), Reilly and Drzycimski (1974), and Livingston (1977)). Furthermore, while there have been studies reporting results for calculating optimal portfolios using expected utility approaches, they only did so at a point in time rather than over time. These studies include, for example, the pioneering work of Ziemba, Parkan and Brooks-Hill (1974), and later studies by Grauer (1981a,1981b)), and the studies referred to in footnote 2.

Perhaps the reason for the lack of literature on the application of multiperiod portfolio theory to the construction and rebalancing of portfolios for long run investment horizons is the computational requirements needed to effect such a study (even for a single time period). This has been noted in the literature (for example, in Grauer (1981a) and in Kroll, Levy and Markowitz (1984)), so that the emphasis seems to have been to develop and/or evaluate mean-variance approximations to the power utility functions used in multiperiod portfolio theory.²

However, in pathbreaking studies, Grauer and Hakansson (1982,1985,1986) applied multiperiod portfolio theory to explore active asset allocation among

²Studies here include Ziemba, Parkan and Brooks-Hill (1974), Levy and Markowitz (1979), Pulley (1981,1983), Kroll, Levy and Markowitz (1984), Kallberg and Ziemba (1983), and Grauer (1986).

U.S. stocks, corporate bonds, government bonds, and Treasury bills. Borrowing was ruled out in the first article while margin purchases were permitted in the other two. The third article also included small stocks as a separate investment vehicle. The major input required of this model is an estimate of next period's joint return distribution. As these papers were the pioneering studies in the area. Grauer-Hakansson employed a simple probability assessment approach to estimate the joint return distribution. That is, the estimate of the joint distribution is just the past realized returns of the assets being considered, with each joint realization given equal probability of occurring in the next period. Both annual and quarterly rebalancing were used and the results from these initial studies indicated that the gains from active portfolio management were substantial.

Grauer and Hakansson (1987) examined a universe composed of the four major U.S. asset categories and up to fourteen non-U.S. equity and bond categories. Again, the simple probability assessment approach was employed to estimate the joint return distribution. The results from this study indicated that the gains from having non-U.S. assets in the investment universe were especially large (some were statistically significant) and there was strong evidence of market segmentation in that the U.S. assets were (mostly) ignored in the presence of non-U.S. asset categories.

Moving slightly away from the simple probability assessment approach of their earlier studies, Grauer and Hakansson (1988) applied an inflation adapter to the raw joint empirical distribution approach to generating probability assessments and used it to choose portfolios of the value-weighted index of New York Stock Exchange stocks and a risk-free asset. This market timing approach, which is a variation of asset allocation, was particularly successful in the 1966-85 sub-period.

Thus, the Grauer-Hakansson studies have shown that applying multiperiod portfolio theory to asset allocation can be quite successful. There are indications from their conclusions that such active portfolio management can "beat the market". Given that the inputs were generated from historical data, the results call into question the weak-form of the efficient markets hypothesis which states that one cannot use past data to obtain superior investment performance.

As such, this thesis explores several questions. First, we examine whether multiperiod portfolio theory can be successfully applied beyond the asset allocation stage. We extend the work of Grauer-Hakansson to the next level, that of industry rotation, to see if the same general conclusions an be reached. If the approach can be extended, it adds to their evidence that suggests the market may not be weak-form efficient. It will be interesting to see whether industry analysis can be successfully employed in a structured portfolio selection model given its central role in traditional investment analysis. We focus on a base case. The Grauer-Hakansson studies mostly employed a 32 quarter estimating period for the joint return distribution. We follow their lead and establish this as our reference point; that is, we initially employ the most recent 32 quarters of joint realized returns, each with an equi-probable chance of occurring in the next quarter. Furthermore, we use a twelve industry investment universe, which follows an industry grouping procedure employed by Breeden, Gibbons and Litzenberger (1986), as our base case. Given this base case, we have structured the thesis to explore a second question: whether the results from the base case are data specific. Thus, we examine three sets of investment universes: an eight, twelve and twenty-four equal- and value-weighted industry universe. The success of the Grauer-Hakansson studies was based on a naive simple probability assessment

approach to estimating next period's joint return distribution. The third major question we examine is the efficacy of this technique. We do so by experimenting with different ways of estimating the joint return distribution.

To be more specific, from the established reference point described. above, we make logical extensions to examine the robustness of the simple probability assessment technique and the Grauer-Hakansson conclusions. First, we conduct industry rotation by expanding the investment universe from twelve industries (the base universe) to twenty-four and contracting it to eight.³ Second, we examine the effects of using different amounts of realized returns to estimate the joint return distribution. We extend the estimation of the joint return distribution from the most recent 32 quarters (the base case) to the most recent 28 and 40 quarters, and lastly, we use all the historic returns available.⁴ This latter approach is an "all-of-history" method as opposed to the simple probability assessment approach, which is a "moving window" method. Following these extensions to estimating the joint return distribution, we adopt the inflation adapter method from the Grauer-Hakansson "market timing" study as the next logical step. We apply this approach to our base case. These simple extensions and replication of the methods from 💼 Grauer-Hakansson studies to industry rotation will explore the question of robustness of the model in general and of the estimating techniques in t particular.

On a more innovative level, we make some changes to the simple probability assessment approach. We deviate from the equi-probable joint

³In this analysis we employ the most recent 32 quarters of realized returns as the estimate of the joint return distribution.

^{&#}x27;In this case we hold constant the base case twelve industry investment universe.

realization method to employ a simple "sum-of-the-digits" rule in assigning probabilities to the outcome for the next period. This is applied to our base case. Basically, this rule assigns the largest probability of occurrence in the upcoming period to the most recent joint realization, and assigns the smallest to the joint realization the furthest away. The motivation for using this exponential decay-type technique of assigning probabilities comes from univariate time series modelling. The appeal is that the most recent past should be a better guide to the next period than the more distant past. This contrasts with the simple probability assessment approach, which assigns equal probabilities to each outcome in the moving window.

Finally, we examine a "disaster state" scenario for the base case by appending the worst states for each industry to the estimate of the joint returns for the next period. These "disaster states" are assigned varying degrees of likelihood of occurrence. The motivation for this method is to keep the investor aware that such states may occur in any period, but at the same time to assign these states a probability so as not to drive the investor completely out of the equity markets.

Given the intent of the thesis, we examine the portfolio rates of return and portfolio compositions over time as a gauge of the success of the active strategies. We present tables of geometric mean rates of return and standard deviations of returns, and portfolio compositions over time, of the active strategies for this purpose.⁵ To further judge the success of the active strategies, we employ several well known measures of investment performance. First, we evaluate the success of the strategies using several traditional

[°]This follows the Grauer-Hakansson studies.

academic measures of abnormal performance. With the CRSP value-weighted index as the benchmark market portfolio, we report the Jensen (1968) performance index, and the Treynor-Mazuy (1966) and Henriksson-Merton (1981) tests of market timing ability. Second, we perform a paired t-test that compares the portfolio returns from the active strategies to three sets of benchmarks: (i) the (completely) passive strategy of holding any of one the value-weighted industry indices; (ii) a passive strategy of holding the Center for Research in Security Prices (CRSP) value-weighted index with varying degrees of leverage; and, (iii) a set of "semi-passive" strategies which up- or down-lever an equal-weighted index of the equally weighted industry indices. Third, we compare the portfolio returns of the active strategies among themselves when different methods are used to estimate the joint return distribution. Again, the test for differences in portfolio returns is the paired t-statistic.

The results show that the portfolio choices performed well in both the full 1934-86 period and in the 1966-86 sub-period, achieving both economically and statistically significant excess returns in several instances. Thus, the success of multiperiod portfolio theory applied to major asset categories also appears to hold at a more micro level. We conclude that the simple empirical probability assessment approach is not without merit. This evidence of superior performance from the use of a naive technique reinforces the conclusions of the Grauer-Hakansson studies, and deals a damaging blow to the weak-form efficient markets hypothesis.

Also known as Jensen's alpha; it is the intercept from a characteristic line regression.

These benchmarks and the reason for their use are described in more detail in Chapter V.

For the most part, the portfolio returns were not statistically different: (1) when 28, 32, or 40 quarters of equi-probable realized returns are employed as the empirical probability distribution, and (2) when eight, twelve, or twenty-four industry groupings are the investment universes. Moreover, the "all-of-history" method in estimating the joint return distribution showed statistical inferiority over the 32 quarter simple probability assessment approach, particularly for the more risk averse investors. This occurred in both the 1934-86 and 1966-86 periods, when equal-weighted industries were the universe, but only in the full period, when value-weighted industries were the universe.

Comparing returns from the twelve industry universe employing a 32 quarter simple probability assessment approach to the returns generated when various adjustments were made to the estimating distribution reveal the following results. (1) For the period 1966-86 with the value-weighted universe, none of the adjustments made any difference (statistically) to the portfolio returns. (2) For the 1954-86 and 1966-86 periods, regardless of leverage opportunities for both equal- and value-weighted industries, there is no statistical difference between the portfolio returns using the simple historical raw joint estimates as the joint' return distribution versus the inflation adapter method to adjust the distribution. (3)Both the "all-of-history" and the "disaster states" scenario approaches to estimating the return distribution showed statistical inferiority to the simple historical raw estimate approach as well as to the inflation adapter method in the 1934-86 and 1966-86 periods, especially for the more risk averse "sum-of-the-digits" approach to adjusting investors. (4) The the probabilities, at least for the more risk averse investors, was found to be

superior to all methods in the 1934-86 period while it made no difference in the 1966-86 sub-period. Again, these results serve to reinforce the conclusions of the Grauer-Hakansson studies, particularly on the efficacy of using just the simple empirical probability assessment approach to estimating the joint return distribution.

The evidence from the performance tests of the active strategies reveal that: (1) according to the Jensen performance index, there are mostly positive α 's with some statistical significance at the 5% and 1% levels, indicating some positive abnormal performance, and, (2) positive market timing ability existed with respect to the Treynor-Mazuy and Henriksson-Merton performance measures for the period 1966-86, while the reverse is true for the 1934-86 period. However, most of the timing ability, positive or negative, disappeared when a statistical correction for heteroscedasticity was made.

The thesis proceeds as follows. Chapter II describes multiperiod portfolio theory and the model to operationalize it. Chapter III presents a review of the literature. Chapter IV describes the inputs required of the model, and details the various extensions and modifications made to these inputs. Chapter V contains a description of the data used, and details the construction of the investment universes and the benchmark portfolios. Chapter VI presents the portfolio returns and compositions of the active strategies. Chapter VII examines the results from the performance measurement tests. Finally, Chapter VIII summarizes the thesis, states the conclusions and discusses possible extensions of the research.

II. The Multiperiod Portfolio Model

Consider the simple reinvestment problem, where we assume markets are perfect and returns are independent, but not necessarily stationary, over time. Let the investor have a preference function U_0 (with $U'_0 > 0$, $U''_0 < 0$) defined on terminal wealth w_0 (time 0). If we let w_n denote the investor's wealth with n periods to go, r_{in} the return on asset i in period n, z_{in} the amount invested in asset i in period n (with i - 1 being the riskless asset), and $U_n(w_n)$ the relevant, but unknown, utility of wealth with n periods to go, then the investor's wealth at the end of period n is

$$w_{n-1}(\underline{z}_n) = \sum_{i=2}^{M} (r_{in} - r_{1n}) z_{in} + w_n(1 + r_{1n}^*),$$

where $\underline{z}'_n - (z_{2n}, \dots, z_{Mn})$, with the prime denoting transposition, and M is the number of securities.

Now consider the portfolio problem with one period to go. The investor, with wealth w_1 to invest, must solve

$$\max_{\underline{z}_1 | w_1} \mathbb{E} \left[U_0(w_0(\underline{z}_1)) \right] = U_1(w_1).$$

Clearly, $U_1(w_1)$ is the highest attainable expected utility given w_1 and thus is the induced utility of wealth⁸ with one period to go. The portfolio problem with two periods to go is

$$\max_{\underline{z}_2 \mid \underline{w}_2} E \left[U_1(w_1(\underline{z}_2)) \right] = U_2(w_2).$$

⁸It is known as a derived utility function since it is derived from a specific $U_0(w_0)$.

The induced utility of wealth with n periods to go is given by the recursive equation

$$U_{n}(w_{n}) - \max_{\underline{z}_{n} \mid w_{n}} E \begin{bmatrix} U_{n-1}(w_{n-1}(\underline{z}_{n})) \end{bmatrix}, \quad n=1,2,\ldots$$

Thus, $U_n(w_n)$ is the expected utility of following an optimal policy from period n to 0 (terminal date) given that w_n dollars are available at n. We can see from the above that the induced utility of current wealth, $U_n(w_n)$, generally depends on "everything", namely, the utility function of terminal wealth, $U_0(w_0)$, the joint distribution functions of future returns, and future interest rates. Furthermore, to derive $U_{n-1}(w_{n-1})$, the investor must solve a portfolio problem in each period for all levels of wealth. Moreover, $U_{n-1}(w_{n-1})$ may not have the same functional form as $U_0(w_0)$.

However, Mossin (1968) showed that $U_n(w_n)$ depends only on U_0 if and only if $U_0(w_0)$ is isoelastic, i.e.,

$$U_0(w_0) - \frac{1}{\gamma} w^{\gamma}, \qquad \gamma < 1.$$

(Note that for $\gamma = 0$, $U_0(w_0) = \ln w_0$.) We can now write

$$U_{n}(w_{n}) = \frac{1}{\gamma} w_{n}^{\gamma} . \tag{1}$$

For these utility functions, the optimal investment policy, \underline{z}_n^* , is proportional to wealth; that is,

$$z_{in}^{\star}(w_{n}) - x_{inn}^{\star}, \quad \text{all } \gamma, \gamma < 1, \text{ all } i, \qquad (2)$$

where the x_{in}^{*} are constants. In addition, these preferences are also completely myopic since they depend only on U₀ and the return structure for the current period (and not on future periods). Finally, only functions of

family (1) exhibit constant relative risk aversion.⁹ From (2), we see that

$$U_{n}(w_{n}) = \frac{1}{\gamma} w_{n}^{\gamma} \iff V_{n}(1 + r_{n}) = \frac{1}{\gamma} (1 + r_{n})^{\gamma};$$

that is, the utility sof wealth relatives, $V_n(1 + r_n)$, is of the same form as the utility of wealth. This holds only for family (1).

We note that these properties, while interesting, are clearly restricted to utility functions of class (1). However, Leland (1972), Hakansson (1974), Ross (1974), and Huberman and Ross (1983), showed the important result

$$U_n(w_n) \longrightarrow \frac{1}{\gamma} w_n^{\gamma}$$
 for some $\gamma < 1$, (3)

holds for a very broad class of terminal utility functions $U_0(w_0)$; that is, the induced utility function $U_n(w_n)$ converges to an isoelastic function. Furthermore, Hakansson (1974) has shown that (3) is usually accompanied by a convergence in policy; that is¹⁰

 $\underline{z}_{n}^{\star} \xrightarrow{} \underline{x}_{n}^{\star} w_{n} \qquad \text{for some } \gamma < 1$

Thus the objectives given by (1) encompass a broad variety of different goal formulations for investors with intermediate- to long-term investment horizons.¹¹ In addition, since the relative risk aversion function is 1- γ , the family (1) incorporates a full range of risk attitudes, ranging from risk neutrality ($\gamma = 1$) to infinite risk aversion ($\gamma = -\infty$).

⁹Relative risk aversion is defined as -wU"(w)/U'(w) and equals 1- γ for the family (1)

¹⁰Convergence to an isoelastic policy at n does not imply the isoelastic policy will be optimal to the end. In other words, if Interstate 80 is a good route from Boston to San Francisco, then it is an equally good route to Los Angeles or Seattle, but only as far as Salt Lake City. See Hakansson (1974). ¹¹However, the simple reinvestment formulation ignores consumption.

Summarizing, the several noteworthy properties of the isoelastic class of utility functions are: (i) they are consistent with multiperiod expected utility maximization whenever returns are independent over time (although this is not required when $\gamma = 0$). Moreover, the investment objective is quite robust, encompassing a wide variety of tastes when the investment horizon is intermediate- to long-run. (ii) They are myopic in that the return structure beyond the current period is not required. (iii) They are the only class of Enctions for which we can formulate the investment problem in a rate of return setting as opposed to wealth for multiperiod horizons. (iv) They exhibit decreasing absolute risk aversion, which implies that risky assets are normal goods. (v) They span a continuum of risk attitudes, from risk neutrality to infinite risk aversion. Moreover, they are aptly suited to the multiperiod reinvestment problem, particularly for the institutional investors, since the functions (except for $\gamma = 1$) display an aversion to pegative returns and bankruptcy that increases as γ decreases.

We now operationalize multiperiod portfolio theory. At the beginning of each period t, the investor chooses a portfolio, \underline{x}_t , on the basis of some member, γ , of the family of utility functions for returns given by

$$V(1+r_p) = \frac{1}{\gamma} (1+r_p)^{\gamma}.$$
 (4)

This is equivalent to solving the following problem in each period:

$$\max_{\underline{X}_{t}} E\left[\frac{1}{\gamma}\left(1+\tilde{r}_{pt}\right)^{\gamma}\right] - \max_{\underline{X}_{t}} \sum_{s} \pi_{ts} \frac{1}{\gamma}\left(1+r_{pts}\right)^{\gamma}$$
(5)

subject to:

$$x_{it} \ge 0, x_{Lt} \ge 0, x_{Bt} \le 0,$$
 all i, t (6)

$$\sum_{i} x_{it} + x_{Lt} + x_{Bt} = 1, \quad \text{all t}, \quad (7)$$

$$\sum_{i} m_{it} x_{it} \leq 1, \quad \text{all t}, \quad (8)$$

$$pr(1 + \tilde{r}_{pt} \ge 0) - 1,$$
 (9)

where

$$r_{pts} = \sum_{it} x_{it}r_{its} + x_{Lt}r_{Lt} + x_{Bt}r_{Bt}^{d}$$
, is the ex-ante return on the portfolio

at time t in state s,

 $\gamma \leq 1$ = a parameter that remains fixed over time,

x = the amount invested in risky asset i in period t as a fraction of own
capital,

 $\underline{x}'_{t} = (x_{1t}, \dots, x_{Mt}, x_{Lt}, x_{Bt})$, where \underline{x}_{t} is a column vector and a prime denotes transposition,

 \tilde{r}_{it} = the random return expected on asset i in period t,

 r_{1t} = the return on the riskless asset in period t,

 r_{Bt}^{d} - the borrowing rate at the time of the decision at the beginning of period t,

 m_{it} = the initial margin requirement for asset i in period t expressed as a fraction, and

 π_{ts} = the probability of state s in period t, in which case the random return \tilde{r}_{it} will assume the value r_{its} .

Constraint (6) rules out short positions¹² and (7) is the budget

¹²While the program can solve (5) to find the optimal portfolio with or without this constraint, we have it in for two reasons: (1) many institutional investors are constrained by law to hold only long positions, and (2) so that this study is comparable to previous ones which explicitly considered this constraint.

constraint. Constraint (8) serves to limit borrowing (when desired) to the maximum permissible under the margin requirements that apply to each of the various assets. Finally, (9) is the solvency constraint which rules out any (ex-ante) probability of bankruptcy. Note that with finite return distributions, this constraint is not binding for $\gamma < 1$ because of the power functions' aversion towards negative returns and bankruptcy. However, we explicitly consider this constraint in formulating the maximization problem to prevent the algorithm from considering infeasible solutions as it searches for the optimum.

Several inputs are required to operate this model. First, we need an estimate of the joint return distribution of the investment universe for the next period.¹³ Second, we need to specify the return on a riskless asset and a *decision* borrowing rate (when leverage opportunities are permitted) at the beginning of the holding period considered. Finally, the margin requirements must be given when borrowing is allowed. A detailed discussion of these inputs will be given in Chapter IV.

¹³See Bawa, Brown, and Klein (1979) for a comprehensive look at the issues and problems involved with the estimation of return distributions.

III. Literature Review

This chapter reviews the literature relating to: (1) applications of multiperiod investment theory (the multiperiod theory itself was reviewed in Chapter II); (2) the relationship between applications of multiperiod portfolio theory and mean-variance analysis; and (3) studies relating to industry analysis.

A. Applications of Multiperiod Portfolio Theory

As noted, the multiperiod theory was reviewed in Chapter 11. This section reviews the application of multiperiod portfolio theory to the rebalancing of portfolios over time. Grauer and Hakansson (1982) was the first study applying multiperiod portfolio theory to the rebalancing of portfolios over time.¹⁴ Since that time, three more articles from the same authors have made their appearance in the finance literature, viz. Grauer and Hakansson (1985,1986,1987). The first three articles explored active asset allocation strategies in a domestic setting composed of U.S. common stocks, corporate bonds, government bonds and a riskless asset. Borrowing was ruled out in the first article while margin purchases were permitted in the other two. In addition, the third article included small stocks as a separate investment vehicle. Thus, the first three articles in this area considered only major U.S. asset categories as the investment universe.

The model given in Chapter II, equations (5)-(9), was the one that

¹⁴This study is actually the second (chronologically) to appear in the finance literature, but the first in a major journal. Grauer and Hakansson (1981) appeared in a German publication.

Grauer-Hakansson used in their studies. In essence, they maximized the expected utility of wealth for utility functions of the isoelastic class for each period. The constraints in the model were: (1) no short sales were permitted, (2) the budget constraint, i.e., the asset weights, including borrowing and lending, must sum to one, (3) leverage, when permitted, must not exceed the margin requirements of each asset class, and (4) insolvency, i.e., the ex-ante probability of bankruptcy, was ruled out.

The major input to their model was an estimate of the next period's joint return distribution. In their studies, Grauer-Hakansson used the so-called simple probability assessment approach. The idea was the following: at the beginning of period t, the realized returns of the most recent N periods were recorded; each of the N joint realizations in periods t-N through to t-1 was assumed to have probability 1/N of occurring in the upcoming period. Thus, estimates were obtained on a moving basis and used in raw form without adjustment of any kind.

Turning first to the 1982 and 1985 articles, Grauer-Hakansson considered an investment universe composed of U.S. stocks, corporate bonds, government bonds, and a riskless asset. The first article did not permit leverage while the second did, and both employed annual and quarterly revisions. The results revealed that there were substantial gains to be had from applying long-run investment theory to portfolio selection. The gains from diversification were especially sharp, particularly for the more risk averse investors. However, there seemed to be only very minor differences when portfolios were revised once a year versus quarterly. Finally, a comparison between levered and unlevered portfolios produced results that were to be expected. The highly risk averse investors never borrowed, so the availability of margin purchases
made no difference to their portfolio returns, while the more risk tolerant investors were fairly liberal in their use of leverage, occasionally employing up to the maximum allowed. As to be expected, leverage increased the variability of portfolio returns, and in the case of most investors, also increased their realized returns.

When small stocks were included in the investment universe, Grauer and Hakansson (1986) found the conclusions from their previous studies to be generally unaltered. Small stocks, when chosen, tended to replace common stocks in the portfolios. This had a notably positive effect on realized returns for all investors, which is not surprising.¹⁵ Moreover, the performances of the active strategies, when compared with fixed weight portfolios of similar riskiness, were statistically significantly higher in some cases.

So far, the studies reviewed have employed a "domestic" universe; that is, only U.S. assets were considered. In Grauer and Hakansson (1987), the investment universe was expanded to include the four principal U.S. asset groups and up to fourteen non-U.S. equity and bond categories.¹⁶ The principal

¹⁵Many investigators, for example Banz (1981) and Reinganum (1981), have noted that small stocks have produced excess risk-adjusted returns. This is now commonly known as the "small firm effect".

commonly known as the "small firm effect". ¹⁶When we examine the international arena, we find many studies of international diversification, but most were based upon the mean-variance model of portfolio choice at a point in time. One of the earliest was by Grubel (1968) who showed the benefits of international diversification for eleven countries from 1959-66. The (expected) result was that a portfolio of international assets dominated (in the mean-variance sense) a portfolio of U.S. stocks only. Subsequently, Levy and Sarnat (1970) put forth a study examining the gains from international diversification of twenty-eight countries. The conclusions reached were generally similar to those of Grubel. Again, these and other studies (see the references cited in Adler and Dumas (1983) for a comprehensive list of studies) in this area only looked at portfolio choice at a point in time.

findings were: (1) the gains from including non-U.S. asset categories were dramatic, especially so for the more risk averse investors, (2) there were large gains from removing the no leverage constraint compared with when only U.S. assets were the investment universe, and, (3) investment in the U.S. assets were mostly non-existent in the presence of non-U.S. categories, which provided strong evidence of market segmentation.¹⁷

Overall, the Grauer and Hakansson studies have shown that applying multiperiod portfolio theory to the construction and rebalancing of portfolios composed of major U.S. and international asset categories can be remarkably successful. The results were all the more remarkable when we consider that only the simple probability assessment approach was used to form an estimate of the next period's joint return distribution. This certainly does not bode well for the weak-form of the efficient markets hypothesis.

More recently, Grauer and Hakansson (1988) refined the simple probability assessment approach by including an inflation adapter to adjust the raw estimates of next period's joint return distribution. At the beginning of period t, the following regression is run:

 $r_{ij} = \alpha_i + \beta_i r_{ij} + \epsilon_{ij}$, j=t-1 to t-N,

where r_{ij} is the return on asset i in period j, and r_{Ij} is the inflation rate in period j. This regression is run for data contained in the same time frame as that used in estimating the joint return distribution employing the simple probability assessment approach. In other words, the realized returns of the most recent N periods are regressed against the inflation rates of the most

¹⁷Agmon (1973) and Lessard (1973) also provided evidence of market segmentation.

recent N periods. The simple probability assessment approach to estimating the joint return distribution in period t is then adjusted by the following equation:

 $\hat{r}_{ij} = r_{ij} + \hat{\beta}_i (r_{Lt} - \bar{r}_{Lj}), \quad j=t-1 \text{ to } t-N,$

where β_i is the estimated regression slope coefficient, r_{Lt} is the return on the riskless asset for period t, \bar{r}_{Lj} is the (arithmetic) average return of the riskless asset in the most recent N periods, r_{ij} is the realized rate of return on asset i in period j for j-t-l to t-N. Presumably, if $r_{Lt} - \bar{r}_{Lj}$ is positive, then what the "market" is telling us is that (unless real interest rates have gone up) the inflation rate in period t is expected to be higher than the average inflation rate over the preceding N periods. Thus, it is only "rational" to assume that, if investors expect a higher inflation rate in period t, then the expected return on asset i should be affected through the inflation rate and $\hat{\beta}_i$, since $\hat{\beta}_i$ reflects the relationship between asset i's returns and inflation in the past N periods.

This inflation adapter method to refining the raw joint empirical distribution approach was applied to choose portfolios composed of the value-weighted market index of New York Stock Exchange stocks and Treasury Bills. Basically, Grauer and Hakansson applied this technique to determine the "right" time to move out of the stock market and into Treasury Bills and vice versa. Traditionally, this has been the approach taken by investors who feel that they do not have micro-selectivity,¹⁸ but by analysing broad macro-economic variables are able to "time the market". This approach to

¹⁸Micro-selectivity is the ability to identify under-priced securities in the stock market and thereby include them in the portfolio.

"timing the market" was suprisingly successful, especially in the 1966-85 period.¹⁹

B. A Comparison of Power Utility and Mean-Variance Portfolio Choice

Even though multiperiod portfolio theory has been around since the early 1970s, it has been largely ignored in portfolio selection applications. This is so despite the fact that the multiperiod model has a strong theoretical foundation and is particularly well suited to the problem of rebalancing portfolios over many periods. However, a disadvantage of these models may be in the computational problems and costs that arise when applying theory to practice i^{20}

It is well known that finding the set of mean-variance efficient portfolios is typically less computationally burdensome than finding an optimal portfolio through expected utility maximization. Thus, there have been many studies examining how closely portfolios chosen on the basis of functions of means and variances approximate those picked by expected utility maximization.²¹ However, these studies just examine the portfolio choices at a point in time and none went so far as to construct and rebalance the portfolios over many periods.

C. Industry Studies

¹⁹It can be seen that if inflation is low or non-existent, as it was in the thirties through to the mid-sixties, this approach will have no significant impact on the raw joint distribution.

²⁰See, for example, Kroll, Levy and Markowitz (1984), and Grauer (1981a).

²¹See, for example, Ziemba, Parkan, and Brooks-Hill (1974), Pulley (1981, 1983), Kallberg and Ziemba (1983), Kroll, Levy and Markowitz (1984), and Grauer (1986).

There have been no studies applying multiperiod portfolio theory to the industry sectors of the U.S. stock market. This is not surprising since the Grauer-Hakansson studies reviewed in *Section A* were the first to employ the multiperiod model. However, there are several studies focussing on industry analysis and the need to do such analyses.

King (1966) was one of the earliest studies on industry analysis. He was concerned with whether industry factors were present in individual stock price movements. He found that approximately 10-15% of the variance of individual stocks were related to the industry component after taking into account the market factor. Later studies by Meyers (1973) and Livingston (1977) had results that were consistent with King's. They concluded that there existed a pervasive industry influence on the performance of stocks over time. This conclusion emphasized the need for portfolio managers to perform such analyses so as to obtain higher returns and lower risks in the long run.

Studies examining the performance of industries over time include Latané and Tuttle (1968), Brigham and Pappas (1969), and Reilly and Drzycimski (1974). These studies showed that differences in performance between industries were substantial. For example, Latané-Tuttle found that the market increased by five times in the period 1950-67 while the industries had varying changes; one industry declined while another had an increase of about forty times over the same time period. Moreover, these studies, plus one by Tysseland (1971), showed that there was almost no association (i.e., correlation) in industry performance over time, which lent support to the weak-form efficient markets hypothesis. However, this does not imply that industry analysis is useless. On the contrary, these studies do support the

concept of industry analysis and of an industry influence on individual stock performance. The evidence showed that industries do not perform in the same way over time, and therefore this makes industry analysis all the more important to the portfolio manager.

On a related topic, Farrell (1974,1975) investigated the clustering of companies along other lines. In particular, he used a broader than industry classification (stable, cyclical, growth and energy) to see if the price action of stocks conformed to this classification. He did find that the four stock groupings were homogeneous; stocks within each group were highly correlated, and the inter-group relationship showed near independence. Moreover, Farrell, (1975) showed that there were substantial benefits to be had from an ex-post group rotation strategy among these four classifications.

More recently, Sorensen and Burke (1986) examined portfolio returns from active industry group rotation. They considered forty-three industries from 1972-84 and applied a "relative strength" trading rule, rather than any portfolio selection models, to rotate in and out of these industries.²² The measures used to judge the performance of the portfolios formed by this "technique were the Sharpe index, the Treynor index, and Jensen's alpha.²³

²²For each industry, the relative strength index is calculated. This index is formed by the ratio of an industry's current price to its average price over the previous six months. Portfolios are formed based on this index. The rule is: sell an industry when its index falls below a specific rank (e.g., below 30%), and replace it with the next highest ranking industry not in the portfolio. Each industry in the portfolio is equally weighted. For more details, see Sorensen and Burke (1986).

²³The Sharpe index is defined as $(\bar{r}_p - r_L)/\sigma_p$ where \bar{r}_p is the average return on a portfolio, r_L is the riskless return, and σ_p is the standard deviation of returns of portfolio p. The Treynor index is defined as $(\bar{r}_p - r_L)/\beta_p$ where β_p is the beta of portfolio p. Jensen's alpha is the intercept term in the characteristic line regression; that is, when we regress the excess returns of portfolio p against the excess returns of the market.

Based on such measures, the overall conclusion was that that some strategy of group rotation may be of use to portfolio managers. In fact, the strategies that did the best were those that were not very active, However, in the full period considered, 1972-82, this "relative strength" method of group rotation did not lead to superior performance, while in the sub-period 1977-82 it did. Thus, this technique has severe limitations when employed in highly cyclical markets (which 1972-82 was).

IV. Model Inputs and Statistical Tests of Investment Performance

We now describe the inputs required when applying the multiperiod portfolio model described in Chapter II to industry groupings of the U.S. stock market. The investment universe considered is either an eight, twelve or twenty-four industry grouping of the New York Stock Exchange.²⁴ Both equaland value-weighted industry indices are used. Details of these industry groupings and their formation will be given in Chapter V.

A. Joint Return Distribution Estimation

1. The Simple Probability Assessment Approach

The major input to the model is an estimate of the joint return distribution of the industries for the next period.²⁵ We initially follow previous multiperiod studies²⁶ by employing the so-called simple probability assessment approach. As described in Chapter III, this approach works in the following manner. Suppose quarterly revision is used. Then, at the beginning of quarter t, the realized (industry) returns of the most recent N quarters are recorded; each of the N joint realizations in quarters t-N through to t-1 is assumed to have probability 1/N of occurring in the coming quarter t. Thus, estimates are obtained on a moving basis and used in raw form without

²⁴The eight industry breakdown comes from Sharpe (1982); the twelve industry grouping follows that of Breeden, Gibbons and Litzenberger (1986), while the twenty-four industry classification is basically derived from the twelve industry grouping by breaking each of the twelve into more specific industries. See Chapter V for more specific details. ²⁵See Bawa, Brown, and Klein (1979) for a comprehensive look at the issues and

²⁶See Bawa, Brown, and Klein (1979) for a comprehensive look at the issues and problems involved with the estimation of return distributions. ²⁶See Grauer and Hakansson (1982,1985,1986,1987).

adjustment of any kind.²⁷ Note that there is no information loss involved here because the whole joint distribution is specified; all moments and correlations are implicitly taken into account.

Having estimated the joint return distribution for quarter t, the only other inputs required are the (observable) riskless return, r_{Lt} , and the (observable) decision borrowing rate, r_{Bt}^d , for quarter t. The former is assumed to be the 91-day U.S. Treasury bill maturing at the end of quarter, while the latter is assumed to be the call money rate plus 1% at the beginning of quarter.

We initially let N = 32 quarters for each of the equal- and value-weighted eight, twelve, and twenty-four industry investment universes; that is, we use the realized returns of the most recent 32 quarters as our estimate of the joint return distribution.²⁸ In addition, we examine the effect on portfolio policies and their returns when we allow N to be 28 and 40 quarters for the twelve equal- and value-weighted industries.²⁹ Thus we replicate the Grauer-Hakansson methodology on a new and different data set.

2. The Simple Probability Assessment Approach With An Inflation Adapter

Recently, Grauer and Hakansson (1988) employed an inflation adapter method to refine the simple probability assessment approach. Again, we replicate their inflation adapter method for the twelve equal- and

²⁷If the investor has no information about the form and parameters of the true distribution, but believes that this distribution went into effect N quarters ago, then it is optimal to use the empirical distribution of the past N quarters. See Bawa, Brown, and Klein (1979), p. 100. ⁴ ²⁸This follows the cited Grauer and Hakansson studies where they mostly used a

^{2°}This follows the cited Grauer and Hakansson studies where they mostly used a 32 quarter estimating period when quarterly portfolio revisions were considered. They also experimented with 40 quarters as the estimating period. ²⁹The twelve industry universe with the most recent 32 quarters as the estimate of the joint return distribution is our base case. Thus, we examine deviations from the base case when we employ the last 28 and 40 quarters as the estimate of the return distribution, ceteris paribus.

value-weighted industry investment universe (our base case) to see if the same general conclusions reached by Grauer-Hakansson can be supported. As described earlier in Chapter III, the following regression is run for each industry at the beginning of quarter t:

$$r_{ij} - \alpha_i + \beta_i r_{ij} + \epsilon_i$$
, j-t-l to t-N,

where r_{ij} is the return on industry i in quarter j and r_{Ij} is the inflation rate in quarter j. Suppose N - 32 quarters are employed in the simple probability assessment approach. Then, the preceding 32 quarters of returns on industry i are regressed on the same 32 quarters of inflation rates. We now adjust the raw estimates of the joint distribution to be

$$\hat{r}_{ij} = r_{ij} + \hat{\beta}_i (r_{Lt} - \bar{r}_{Lt}), \quad j=t-1 \text{ to } t-N,$$

where $\hat{\beta}_{i}$ is the regression slope coefficient, r_{Lt} is the riskless return for quarter t, \bar{r}_{Lt} is the (arithmetic) average return on the riskless asset for the most recent N quarters, and r_{ij} is the realized rate of return on industry i in quarter j, for j - t-l to t-N. Note that while we adjust the magnitude of the returns in the joint distribution, we still give a probability of 1/N to each joint realization. The rationale for this approach is to consider the impact of inflationary expectations on our estimates of the joint return distribution. If changes in the U.S. Treasury bill rate reflect investors' expectations about inflation in the coming period (given that the real interest rate has not changed), then our estimates of next quarter's returns should be affected by the right hand side_ of the above equation.

3. A "Sum-of-the-Digits" Probability Assessment Approach

The simple probability assessment approach assigns equal probabilities

to each joint realization in the estimate of the return distribution. We now introduce a completely new way to assign these probabilities to the joint realizations of our base case. These probabilities are assigned by a "sum-of-the-digits" rule which operates as follows. Let

$$K = \sum_{k=1}^{N} k$$

Then, at the beginning of quarter t, the joint realization of quarter t-1 is given probability N/K; the joint realization of quarter t-2 has probability (N-1)/K; and so on until the joint realization of quarter t-N has probability 1/K of occurrence. Thus, we have assigned more weight to the recent observations. The "sum-of-the-digits" method is an exponential decay-type function and the idea is borrowed from the exponentially weighted moving average (EWMA) model of univariate time series modelling. Consider the simple probability assessment approach: we are saying that the forecast for the next period is the (simple) average of the past N realized return vectors. The impetus behind the "sum-of-the-digits" notion is a "common-sense appeal" that the recent past would be a better guide to the next period than the more distant past.³⁰

4. An "All-of-History" Simple Probability Assessment Approach

As noted, the Grauer-Hakansson studies have employed a simple probability assessment approach, which is a "moving window" method to estimating next period's joint distribution. That is, only the most recent N periods of realized returns are used, and as we move sequentially forward in time, the earliest observation is dropped while the most recent one is added.

³⁰The method is arbitrary, but perhaps no more or less than assigning equal probabilities to each joint realization.

Thus, implicit in their approach is that the return distribution is only stationary for the last N quarters. We now modify this by assuming that the world is stationary by considering an "all-of-history" method³¹ whereby all past returns are used in estimating the raw joint distribution. In other words, at the beginning of quarter t, returns from quarter 1 (the first available data) to quarter t-1 are used. At the beginning of quarter t+1, returns from quarter 1 to quarter t will now be the estimate of the joint distribution, and so on. Each joint realization is given an equal probability 1/(t-1) of occurring in quarter t. Note that the probabilities become smaller as we move forward in time, unlike the ("moving window") simple probability assessment approach. This technique is applied to our base case.

5. A "Disaster States" Scenario Probability Assessment Approach

Finally, we introduce a "disaster states" scenario in estimating the return distribution of our base case. This idea is a modification of the simple probability assessment approach, and is as follows. Let there be a universe of M industries and let N be the number of quarters of realized returns used in the simple probability assessment approach to estimate the next quarter's return distribution (M - 12 and N - 32 in our base case). At the beginning of quarter t, the worst state for each asset i from quarter 1 '(first observation) to quarter t-1 (most recent observation) is recorded. Thus we should obtain M such states (one for each asset in the universe). Then we append these M worst states to the most recent N states to obtain

⁻

³¹Notice that this method contrasts with the "sum-of-digits" approach. In this case, we would like to use every shred of information available to estimate the joint return distribution.

M + N states as our estimate of the joint return distribution for quarter t. 32

The probabilities we assign to these states are now determined by the following rule:

| Probabilities of | | Probabilities of |
|--------------------|---|------------------|
| each of the N most | | each of the M |
| recent states | | worst states |
| | n | - |
| 1 | | J |

(N + J) M

 $\overline{N + J}$

where J is an integer and $1 \le J \le M$. For example, if we let J = M, then all we are doing is assigning each of the N + M states a probability of 1/(N+M); that is, equal probability: We examine three variations around our base case here; we let J = 1, 6, and 12, and N = 32 quarters. Suppose J = 1. Then we see that each of the N (= 32) "normal" states has probability 1/33 of occurrence, and each of the M (= 12) "disaster" states has probability $1/(33\cdot12)$ of occurrence. When J = 12, we are assigning the "disaster" states a probability equal to each of the "normal" states; when J = 6, a lower probability is assigned to each of the "disaster" states than to the "normal" states; and when J = 1, we are assigning the lowest probability to the "disaster" states.

The motivation for the use of a "disaster states" scenario is to keep the investor aware that such states may occur at any time, but not to give it a probability of occurrence such that it will completely drive the investor out of the equity markets. Thus, we have made the simple probability assessment estimate of the return distribution more conservative, and by letting J vary from 1 to M, have varied the degree of conservatism.

³²The "moving window" method places a zero probability on a previous disaster state (e.g., the fourth quarter of 1929) if it does not fall within the last N quarters (i.e., it is out of the window). This provides the motivation for the "disaster states" approach.

Given these inputs, the portfolio weights for the various assets and the proportion of assets borrowed (if permitted) are calculated by solving system (5) - (9) in Chapter II via non-linear programming methods.³³ At the end of quarter t, the realized return on the portfolio is determined, using the weights selected at the beginning of quarter t. This cycle is repeated in all subsequent quarters.

We report returns gross of transactions costs and taxes, and assume that the investor cannot influence prices. This is in line with the assumptions made in previous multiperiod studies. The reasoning is, first, the data series used as inputs also exclude transactions costs (for reinvestment of dividends) and taxes. Furthermore, a set of benchmark portfolios is constructed from this data series for comparison with the active strategies and these benchmark portfolios also exclude transactions costs and taxes. Second, many institutional investors are tax-exempt, and there are techniques available for keeping transactions costs low.³⁴

Finally, we present the portfolio compositions of selected active strategies to detect the differences, if any, between the various return distribution estimation techniques.

B. Statistical Tests of Investment Performance

To judge the performance of the active strategies, we test whether there is any statistical difference in the returns of the active strategies with quarterly revision, both with and without leverage opportunities, under the

³³The non-linear programming algorithm employed is described in Best (1975).
³⁴Furthermore, the proper treatment of transactions costs and taxes are nontrivial.

following scenarios. (1) The twelve industry universe versus the eight and twenty-four industry universe (both value- and equal-weighted), for N = 32quarters under the simple probability assessment Approach to estimating the joint return distribution. (2) The use of N = 32 versus 28 versus 40 quarters under the simple probability assessment approach for the twelve equal- and value-weighted industry investment universe. (3) The different methods for estimating the joint return distribution for the twelve industry universe when N = 32 quarters; that is, the simple probability assessment approach the inflation adapter (our base case) versus approach versus the "sum-of-the-digits" approach versus the "disaster state" scenario (for J = 1, 6, and 12) versus the "all-of-history" approach. (4) The comparison of selected base case active strategies against the passive and semi-passive benchmark portfolios.³⁵ We report paired t-statistics for the four cases outlined above.³⁶

The paired t-test statistic is constructed as follows. The terminal wealth w_{Ω} is given by

$$w_0 = w_n (1 + r_n)(1 + r_{n-1}) \dots (1 + r_1)$$

 $= \underset{t=1}{\overset{\text{w}}{\longrightarrow}} \exp \left[\sum_{t=1}^{n} \ln(1 + r_{t}) \right],$

³⁵The passive benchmarks are the CRSP value-weighted index levered up and down, and the twelve individual value-weighted industry indices themselves. The semi-passive benchmarks are the equally weighted portfolios of the twelve equal-weighted industry indices levered up and down, plus the individual equally weighted industry indices. Chapter V contains a more detailed discussion of the construction of these benchmark portfolios.

³⁶This test is also used in Fama and MacBeth (1974) and Grauer and Hakansson (1986,1987,1988).

where w_n is current wealth, and r_t is the portfolio return³⁷ in period t for a particular γ . Note that the returns compound multiplicatively, so we employ the paired t-test for dependent observations to the quarterly (and additive) variables $\ln(1 + r_t)$. Thus, to compare the return series r_1^1, \ldots, r_n^1 with the return series r_1^2, \ldots, r_n^2 for two different strategies under different scenarios, we calculate the statistic

$$t = \frac{\overline{d}}{\sigma(d) / \sqrt{n}}$$

where

$$\overline{d} = \sum_{t=1}^{n} \frac{\ln(1+r_t^1) - \ln(1+r_t^2)}{n}$$

and $\sigma(d)$ is the standard deviation of $\left[\ln(1+r_t^1) - \ln(1+r_t^2) \right]$. In each case, the null hypothesis is that

$$E \left[\ln (1 + r_t^1) \right] - E \left[\ln (1 + r_t^2) \right]$$

while the alternative hypothesis is that

$$E \left[\ln (1 + r_t^1) \right] > E \left[\ln (1 + r_t^2) \right].$$

C. Performance Measures

Finally, we present tests of abnormal performance for the active strategies: (1) with and without leverage opportunities present, (2) when managing the twelve equal- and value-weighted industry investment universes under the various return distribution estimation methods, and (3) when managing the eight and twenty-four industry investment universes (both equaland value-weighted) when N - 32 quarters. The standard academic measures of 3^{37} Note we have dropped the subscript p for convenience.

performance are the ex-post characteristic line α (also known as Jensen's (1968) performance index) and the Treynor-Mazuy (1966) and Henriksson-Merton (1981) tests of market timing.³⁸ We now discuss each of these performance tests in turn.³⁹

1. The Jensen Performance Index

For the Jensen performance index, we run the following characteristic line regression for each portfolio p of the active strategies:

$$r_{pt} - r_{Lt} - \alpha_p + \beta_p (r_{mt} - r_{Lt}) + \epsilon_{pt},$$

where r_{mt} is the return on the CRSP value-weighted index, r_{Lt} is the return on 91-day U.S. Treasury Bills, and r_{pt} is the portfolio return on an active strategy. The intercept, α_p , is the measure of abnormal investment performance, where a positive (negative) value indicates superior (inferior) performance. The null hypothesis is $\hat{\alpha}_p = 0$ (no superior or inferior, performance), and the alternative hypothesis is $\hat{\alpha}_p > 0$ (there is superior performance). Thus, the results of one-tailed t-tests are reported.

2. The Treynor-Mazuy Test For Market Timing

Treynor and Mazuy postulated the following test for a fund manager's market timing ability:

$$r_{pt} - r_{Lt} - \alpha_p + \beta_{1p} (r_{mt} - r_{Lt}) + \beta_{2p} (r_{mt} - r_{Lt})^2 + \epsilon_{pt}$$

 $^{^{38}}_{__}$ We also run these tests with a correction for heteroscedasticity.

³⁹Note that the application of multiperiod portfolio theory to group rotation is not exactly a market timing strategy (in the Henriksson-Merton or Treynor-Mazuy sense), nor is it simply a portfolio selection strategy (in the Jensen sense). Thus one could argue that none of these measures will give a completely accurate measure of the investment performance of the multiperiod strategies. On the other hand, these performance measures are the benchmarks for measuring the performance of professional money managers and they could make exactly the same arguments. Therefore, we judge the multiperiod strategies against these same commonly accepted benchmarks.

where the notation is as before. If the manager has (positive) market timing ability, then $\hat{\beta}_{2p} > 0$. The idea behind this test is quite intuitive. Basically, a superior fund manager will increase a fund's target beta to η_2 if he assesses correctly that $r_m - r_L > 0$ (an "up" market) in the upcoming period, and will target the fund's beta to a low η_1 if the assessment is $r_m - r_L < 0$ (a "down" market), where $\eta_2 > \eta_1$. Thus, the manager will have a very low beta fund in "down" markets, and a high fund beta in "up" markets. Thus, we expect to find $\hat{\beta}_{2p} > 0$ for a successful market timer.

3. The Henriksson-Merton Test For Market Timing

Merton (1981) also postulated, in a Capital Asset Pricing Model framework, the two-target-beta strategy. As in the Treynor-Mazuy test, the fund manager sets the fund's beta to η_1 if the forecast in the coming period is $r_m - r_L < 0$, and increases the fund's beta to η_2 if $r_m - r_L > 0$, where $\eta_2 > \eta_1$. Henriksson and Merton (1981) developed the following test for market timing:

 $r_{pt} - r_{Lt} - \alpha_p + \beta_{1p} (r_{mt} - r_{Lt}) + \beta_{2p} y_t + \epsilon_{pt}$,

where $y_t^{\mu} - \max(0, r_{Lt} - r_{mt})$ is equal to the payoff associated with a put option on the market portfolio with excercise price r_{Lt} . The interpretation in the Henriksson-Merton test is that $\hat{\beta}_{1p}$ measures the fund's average beta in "up" markets (i.e., $r_{mt} - r_{Lt} > 0$), while $\hat{\beta}_{2p}$ measure the average decrease from $\hat{\beta}_{1p}$ of the fund's beta in "down" markets.⁴⁰ Thus, we can test the null

⁴⁰Note that these are just estimates of average betas due to the imperfection of the manager's forecasts. That is, the manager will target a low beta for the fund when in fact the next period turns out to be an "up" market, and vice versa. This creates a problem of heteroscedasticity in the regression estimation where the absolute value of the error term is linearly related to the absolute value of $r_m - r_L$.

hypothesis $\beta_{2p} = 0$ where (if not rejected) we would conclude that the manager does not have market timing ability, or does not act on his forecasts.⁴¹

Thus, we present three measures of investment performance that are CAPM related. The measures are not without their critics.⁴² Nevertheless, they are the most well known and used tests in the academic and applied literature. Therefore, we subject the multiperiod strategies to these commonly accepted benchmarks.

⁴¹We focus on $\hat{\beta}_{2p}$ as the measure of investment performance. The estimate, $\hat{\alpha}_p$, is a measure of the manager's micro-selectivity; that is, can the fund manager select individual securities with success. However, the $\hat{\alpha}_p$'s are strongly negatively correlated with the $\hat{\beta}_{2p}$'s as evidenced in Henriksson (1984) and Grauer and Hakansson (1988). This means that if the active strategies show positive market timing (i.e., $\hat{\beta}_{2p}$ positive and significant), then they would more than likely also show negative micro-selectivity. One explanation for this can be found in Jagamathan and Korajczyk (1984). ⁴²See Roll (1978), Dybvig and Ross (1985), Grauer (1987) for criticisms of using SML (security market line) measures (Jensen's test), and see Jagannathan and Korajczyk (1984) for a criticism of the Henriksson-Merton measure.

The primary source of our data is the 1986 monthly returns tape of the Center for Research in Security Prices of the University of Chicago (CRSP). This tape contains the relevant monthly price, share and return information of all firms on the New York Stock Exchange from December 1925 to December 1986. Furthermore, we used the Standard Industrial Classification (SIC) handbook (1967) as the guide to the industrial classifications of the firms.

A. The Industry Universes

First, we replicate the twelve industry indices of Breeden, Gibbons and Litzenberger (1986),⁴³ which in turn were an adaptation of Sharpe's (1982) eight industry indices. Basically, firms on the monthly CRSP returns file are grouped into the twelve broad industry categories according to their (first) two-digit SIC code. Table la gives the Ewelve industries and their respective SIC codes. The eight industry universe is patterned after Sharpe (1982). Table 1b gives the eight industries and their respective SIC codes. Finally, industry classification. Table 1c twenty-four This presents the categorization is based upon the twelve industry classification and is formed each of the twelve industries from Breeden, Gibbons as follows: and Litzenberger is examined and further broken down into smaller industries until twenty-four industries are formed. Care was taken to ensure that each

⁴³Note that Breeden, Gibbons and Litzenberger (1986) omitted industries with two-digit SIC codes of 0,2,39,76 and 99. Thus, we followed Sharpe (1982) in assigning these SIC codes to the following industries: 0,39,99 to Consumer Durables, 2 to Food, and 76 to Services.

industry existed from 1926 on.44

We now present the details of the construction of the equal- and value-weighted industry indices. Essentially, the methodology used in constructing both types of indices is virtually identical to the CRSP construction of their value-weighted index. First, firms whose shares trade as American Depository Receipts (ADRs) are excluded.⁴⁵ Second, we exclude firms with missing price data; that is, the firm is dropped from the industry if its price information is missing, but we include it back as soon as all the information is available. Third, and this is a departure from the CRSP methodology, we exclude firms with missing share data.⁴⁶ Furthermore, we construct both the equal- and value-weighted indices from the same universe of firms.⁴⁷

A particular value-weighted industry index is constructed as follows: first all firms with the industry's two-digit SIC code(s) are recorded; then the industry return for a particular month t is just the sum of the value-weighted individual security returns⁴⁸ for t. The value-weight of a firm

⁴⁴While the eight and twelve industry indices are drawn from the same universe of firms, we dropped two SIC classifications in the construction of the twenty-four industry indices. Specifically, two digit codes 00 and 99 are excluded as they represent firms with missing SIC codes on the CRSP database and non-classifiable firms respectively. Thus the twenty-four industry universe has the same or fewer firms than the other two universes. However, there was no difference in the count of firms at the beginning, while the later years only saw a few firms dropped. See Table 2 and footnote 1 as well. ⁴⁵Beginning in 1986, CRSP excluded ADRs in the construction of its yalue-weighted index.

⁴⁶This missing share data is recorded as zero so that it does not affect the CRSP computation of its value-weighted index. However, we choose to exclude the firm explicitly, so that our count of firms in the universe at any point in time may be smaller than CRSP's.

⁴⁷This is a departure from CRSP's construction of its equal-weighted index. For example, CRSP includes ADRs and firms with missing share data in its equal-weighted index.

"These security returns are contained in the RET1 vector on the CRSP database; i.e., they include dividends.

for month t is its beginning (of month t) price times its beginning (of month t) shares outstanding divided by the total equity value of all firms in the industry at the beginning of month t. The quarterly rate of return for the industry is now just the compounded rate of return for the three months in the quarter. Employing the same firms that are used in constructing a particular value-weighted industry index, an equal-weighted industry index is also constructed as the arithmetic average of the returns of all the firms in that industry.

Tables la-c give more details on the industries. The tables identify the twelve, eight, and twenty-four industry classifications (plus their SIC codes) respectively, the total number of firms and the total market values in the sample, and the percentage of firms and of market values in each industry at four points in time.⁴⁹ These four points in time are chosen to represent the focal points of the dataset and of the portfolio selection problem. Specifically, we choose January 1926 and December 1986 since these are the start and end dates of the CRSP database respectively; January 1934 and January 1966 are chosen because they represent the starting points of our investment horizon — the full period from the first quarter of 1934 to the fourth quarter of 1986, and with the first quarter of 1966 to the last quarter of 1986 as an important sub-period.⁵⁰

B. Other Data Requirments

⁴⁹From the percentages given, we can calculate the number of firms in each industry at the four points of time.

We choose this sub-period for two reasons: (1) it represents the most recent two decades, and therefore is of interest to investors and portfolio managers, and (2) it is consistent with the previous Grauer and Hakansson studies.

The next piece of information required is the return on a riskless asset; in the portfolio selection problem with quarterly revision, we use the 91-day U.S. Treasury Bill maturing at the end of the quarter. The source of this information is the Survey of Current Business and The Wall Street Journal. The borrowing rate for decision purposes, r_{Bt}^d , is assumed to be the call money rate plus 1%. This rate is viewed as persisting throughout the quarter and therefore riskfree. The sources for this rate are the Survey of Current Business for the period 1934-76, and The Wall Street Journal for later periods. The borrowing rate used for rate of return calculations, the realized rate, r_{Bt}^r , is calculated as a monthly average. Finally, the margin requirements are obtained from the Federal Reserve Bulletin. These are the initial margin requirements; there was no pratical way to take maintenance margin requirements into consideration in the programs. In any event, such requirements would have come into play for the more risk tolerant strategles, and then only occasionally.

Finally, for the inflation adapter method to adjust the joint return distribution, we obtain the inflation rates from the 1988 Yearbook of Ibbotson Associates. The rates are derived from the Consumer Price Index for All Urban Consumers, not seasonally adjusted (CPI-U, NSA). Prior to 1978, the rates are derived from the CPI (as opposed to the CPI-U). The quarterly rates of inflation are just the compounded monthly rates given in the Yearbook.

C. Investment Policies and Their Benchmarks

Next, we describe the portfolios whose returns serve as the benchmarks against which we compare the returns of the active strategies derived from multiperiod portfolio theory. The active strategies have previously been

discussed, and therefore we proceed to address the benchmark portfolios. Since the construction of the benchmark portfolios are identical whether we use an eight, twelve, or twenty-four investment universe, we will confine the discussion to the twelve industry universe with the understanding that it also applies to the other universes. Exceptions, when they occur, will be noted.

First, consider the twelve value-weighted industry indices plus borrowing and lending as the investment universe. Thus, the first set of benchmark portfolios is the individual industry sectors themselves. These are shown in Table 2 together with their abbreviated names. Note that these are just pure buy-and-hold strategies.

A second set of benchmark portfolios is the holding of the CRSP value-weighted index with different degrees of leverage. Note that holding this index is akin to holding each of the twelve industries described here in proportion to their respective market values; that is, holding a value-weighted index of the value-weighted industry indices.⁵¹ These benchmarks are shown in Table 2, and are labelled V2-V20. The labels are mnemonic. For example, V2 represents 20% invested in the CRSP value-weighted index and 80% invested in the riskfree asset. Thus, V10 referse to the CRSP value-weighted index, and V20 is now a portfolio with 200% invested in V10 with 100% borrowed. Note that the margin requirements may have made any (or all) of the levered benchmarks, V12-V20, not feasible at a given point in

⁵¹This is true for the twelve and eight value-weighted industry indices. However, the twenty-four industry universe has two SIC classifications excluded, i.e., the two digit SIC codes 0 and 99, which represents a missing value and a non-classifiable firm on the CRSP tape. Thus, there may be less firms in this universe than the other two. See Tables la-c, and footnotes 1 and 2.

time. The current margin requirement is 50% so that all levered benchmarks are feasible, but there was a 100% margin requirement from the second quarter of 1946 to the first quarter of 1947 so that none of the levered benchmarks were feasible during that time. Hence, we lever the portfolios V12-V20 to either their stated borrowing limits, or to the highest feasible value consistent with the margin requirments.

ø

Second, consider the benchmark portfolios associated with the twelve equal-weighted industry indices plus borrowing and lending as the investment universe. Again, as with the value-weighted industry indices, the individual industry indices are themselves benchmarks. However, note one important difference here. Equally weighting the firms within an industry is not a completely passive strategy as with the value-weighted industry indices. The reason is that equally weighting a portfolio requires an investor to rebalance the portfolio in each period. This rebalancing is very specific; it requires the investor, in each period, to sell the securities that have risen in value and buy those that have fallen so as to maintain the equal weighting scheme of the securities in the portfolio. Thus, we expect the returns on the equal-weighted industry indices to be higher and more volatile than their value-weighted counterparts. There are two reasons for this. Firstly, equally weighting the firms in an index gives more weight to the small firms, and it is a documented anomaly that small firms are characterized by higher returns than large firms (see, for example, Banz (1981) and Reinganum (1981)). Secondly, it reflects the gains from the semi-active equal-weighting scheme over the completely passive value-weighting scheme.

The second set of benchmarks is exactly the same as when the universe is the value-weighted industry indices; that is, they are the portfolios denoted

V2-V20. Finally, the third benchmark considered for this universe is an equally weighted portfolio of the equal-weighted industry indices⁵² levered up and down. These are what we call the semi-passive strategies. We label these benchmarks E2-E20 and are also shown in Table 2. As with the portfolios V2-V20, the benchmarks E2-E20 have the same mnemonic feature, and the margin restrictions applicable to V12-V20 also apply equally to E12-E20.

⁵²Equally weighting the equal-weighted industry indices will not give us the CRSP equal-weighted index. The reason for this is that our criteria for the inclusion of a stock in the index is stricter than what CRSP uses. CRSP includes ADRs and firms with missing share values while we exclude them so that both our equally weighted and value-weighted industry indices will be constructed from an identical universe of firms.

VI. The Results: Portfolio Returns and Compositions

This Chapter presents the results when multiperiod portfolio theory is applied to industry rotation for the period 1934-86. We describe the portfolio returns and compositions of a base case. This base case was described in Chapters I and IV, and is briefly restated below. We then make comparisons and note the differences between the base case and the other cases described in Chapter IV. This Chapter is divided into five major sections, A - D. Section A examines the portfolio returns of the active strategies of the base case, and compares these returns with the returns of benchmark portfolios. Section B compares the portfolio returns of the active strategies of the base case with those of the other cases. Section G describes the portfolio compositions of the active strategies of the base case. Finally, Section D contains a comparison of portfolio compositions of the base case versus the other cases. Each of the Sections A - D is further divided into several sub-sections, that include concluding remarks that summarize the results of each section.

A. Portfolio Returns: The Base Case

The base case is when the power utility functions, with and without leverage opportunities, manage either the twelve equal- or value-weighted industry investment universe using a 32 quarter simple probability assessment approach to estimate the joint return distribution. We present this base case scenario for the periods 1934-86 and 1966-86 in two tables and four figures, a description of which now follows.

Tables 3a-b present the geometric means and standard deviations 53 of $\ln(1+r_{1})$ for each of the twelve equal-weighted industries and riskless lending for 10 active strategies corresponding to the γ 's in equation (4). Chapter II, ranging from -75 (extremely risk averse) to 1 (risk neutral) when leverage opportunities are present (Table 3a) and when thev are not for the semi-passive benchmark portfolios which (Table 3b): upand down-lever an equally weighted portfolio of the equally weighted industries (E2-E20); as well as for the passive benchmark portfolios which up- and down-lever the CRSP value-weighted index (V2-V20) for the full 1934-86 period and the 1966-86 sub-period.⁵⁴ Figures la and lb are the graphical equivalents of Table 3a, columns 1-2 and 3-4 respectively, while Figures 1c and 1d portray the first two and last two columns, respectively, of Table .3b. The following symbols are employed in Figures la-d: (i) the industries and riskless lending - squares; (ii) the 10 active strategies - circles; (iii) the semi-passive benchmark portfolios - diamonds; and, (iv) the passive benchmark portfolios - triangles. We also include the U.S. inflation rate (square) as a horizontal dashed line. Table 2 presents the notation used in labelling the industries and the semi-passive and passive benchmark portfolios contained in all the figures.

Tables 4a-b are the equivalents of Tables 3a-b when the investment universe is the value-weighted industries. Figures 2a-b portray the data contained in Table 4a and Figures 2c-d portray the data contained in Table 4b in the same fashion that Figures 1a-b and 1c-d represent the data contained

⁵³This measure is approximately the same as the standard deviation of rates of return for levels below 25%.

⁵⁴Chapter V detailed the construction of these semi-passive and passive benchmark portfolios.

in Tables 3a and 3b respectively. The notation and symbols used are identical in all figures. The only difference is the absence of the semi-passive benchmark portfolios in both Tables 4a-b and Figures 2a-d.

We now proceed to examine the portfolio returns of the active strategies of the base case, making comparisons with the benchmark pertfolios when necessary. This section contains six sub-sections.

1. The Investment Universes

A comparison of the industry returns shows that, for both the 1934-86 and 1966-86 periods, the mean returns from equal-weighting the industries (Table 3a) are consistently higher than the mean returns obtained by value-weighting them (Table 4a). This phenomenon almost surely reflects (i) the small firm effect, and, (ii) the "active" nature of equal-weighting versus passive value-weighting.⁵⁵ Among the equally weighted industries in the 1934-86 period, Services had the highest geometric mean (16.40%) and standard deviation (32.39%) while Utilities had the lowest geometric mean (13.22%). Among the value-weighted industries, Services again topped the list with a mean of 12.75% and standard deviation of 31.39%, while Transportation provided the lowest mean return (9.31%). During the 1966-86 sub-period, of the equal-weighted industries, we find Food and Tobacco had the highest mean return (16.02%) and Transportation had the lowest at 10.97%. For this same period, among the value-weighted industries, Food and Tobacco again came out on top with 12.96% while Transportation earned a meagre 7.03%; even Treasury ·bills earned 7.49%, and with much less variability.

2. The Leverage Case: 1934-86

 55 Chapter V contains a discussion of this phenomenon.

` ~ 46

Comparing the active strategies contained in the first two columns of Table 3a with those in the first two columns of Table 4a, we find very different geometric mean returns when the powers manage the equal- vis-a-vis the value-weighted industries. The highest geometric mean return for the active strategies was 18.28% attained by the logarithmic investor when managing the equally weighted industries (see Table 3a) compared to 13.95% for the power 0.5 investor managing the value-weighted universe (see Table 4a). Except for the very risk averse investors (powers -75 to -30), the strategies employing the equal-weighted industries always outperformed their counterparts managing the value-weighted industries with about the same variability.⁵⁶

Turning to the benchmark portfolios in the first two columns of Table 3a, we find that in moving from a passive strategy (portfolios V2-V20) to a semi-passive one (portfolios E2-E20), there was a dramatic improvement in geometric mean returns at the expense of only a slight increase in variability. Comparing the active strategies to these benchmarks, we find the following results: (i) when the universe was the equal-weighted industries (see columns 1-2 of Table 3a and Figure 1a), the more risk averse active strategies did slightly worse than the passive and semi-passive strategies; while the results for the more risk tolerant active strategies were mixed. Powers 0.5 and 1 clearly outperformed the passive benchmarks (albeit with a higher variability), while powers 0, -2 and -5 had higher geometric mean returns than the comparable passive. strategies. The active strategies at

⁵⁶While these highly risk averse powers had slightly lower returns when managing the equal- versus the value-weighted industries, they compensated by having lower standard deviations.

times outperformed the semi-passive strategies, and vice versa. (ii) When the value-weighted industries were the universe, the active strategies performed slightly worse than the passive benchmarks (see columns 1-2 of Table 4a and Figure 2a).

3. The Leverage Case: 1966-86

the Table 3a contain Columns 3-4 in results from " managing the . equal-weighted industries for the 1966-86 sub-period. The logarithmic policy, with a geometric mean return of 19.05%, was clearly ahead of the other active the benchmark portfolios. With approximately the strategies and same variability, the semi-passive El4 and El6 benchmark portfolios earned only 13.95% and 14.10% respectively; the passive strategy V20 earned only 7.15%; the Services industry realized 14.22%. Moreover, all the active and strategies outperformed the corresponding passive benchmark portfolios. Not only that, the semi-passive strategies clearly outdistanced their passive counterparts, but not to the same extent as the active strategies. The more risk averse powers (-75 to -5) performed about the same as the corresponding semi-passive benchmark portfolios while the less risk averse powers surpassed the comparable semi-passive results. These results are shown in a graphic form in Figure 1b.

Turning to the last two columns of Table 4a and Figure 2b, the results of the active strategies managing the value-weighted industries, while more muted, are still relatively good. The -2 power had the highest geometric mean return (11.31%), which surpassed the corresponding (V12) passive benchmark portfolio (8.77%). Moreover, all the active strategies, with the exception of the risk neutral investor, had higher mean returns than the corresponding passive strategies with equal variability. However, these active strategies

did not all exceed the industry categories.

4. The No Leverage Case: 1934-86

Turning to the first two columns of Tables 3b and the corresponding graph, Figure 1c, we find that when the universe was the equal-weighted industries, the active strategies had mixed results. The lower powers (-75 to -30) slightly underperformed their passive and semi-passive counterparts; the higher powers (-15 to 1) did about the same or better than the passive benchmarks, and were only slightly worse than the semi-passive benchmarks.

When we examine columns 1-2 of Table 4b and Figure 2c, we find that the comparative passive strategies outperformed the active strategies managing the value-weighted industries. Only the Construction and Transportation industries had a lower geometric mean return than the active strategies. 5. The No Leverage Case: 1966-86

During this time period, the active strategies performed relatively well. The logarithmic investor had a geometric mean of 15.14% (highest) when the industries were equally weighted while the power -2 investor did the best (10.61%) managing the value-weighted industries.

Examining columns 3-4 of Table 3b and Figure 1d, we find that all the active strategies, when managing the equal-weighted industries, outperformed the comparative passive benchmarks, while only the less risk averse investors did better than the corresponding semi-passive strategies; the lower powers performed about the same as the equivalent semi-passive portfolios. With the exception of only the Food industry, the active strategies were distinctly north-west of the industry categories in return-standard deviation space.

Turning now to Table 4b (columns 3-4) and Figure 2d, we find the results relatively good. The -2 power managing the value-weighted industries earned

10.61% (highest) compared to the passive benchmark V10's 9.38% with about the same standard deviation. Overall, the powers from -75 to -2 had geometric means greater than their passive benchmark counterparts, while the risk neutral to logarithmic investors performed substantially worse than the passive strategies.

6. Concluding Remarks

When we compare the no leverage results with those when leverage opportunities were present, we find a salutary effect when margin purchases were permitted. For example, when the equal-weighted industries were the universe, the logarithmic investor earned "only" 13.84% for the 1934-86 period compared with 18.28% when margin purchases were permitted. This effect was evident, in both the 1934-86 and 1966-86 periods, and whether the industries were equal- or value-weighted. However, this was evident only for the more risk tolerant strategies because for the highly risk averse investors (powers -75 to -30), the existence of leverage opportunities made no difference to their portfolio choices.

The principal findings for the base case scenario then are: (1) there are substantial gains to be had from applying the multiperiod portfolio model to active industry rotation, particularly when managing an equal-weighted industry universe. This was especially evident during the 1966-86 sub-period. (ii) The presence of leverage opportunities generally enhanced the portfolio returns of the more risk tolerant investors. (iii) The simple probability assessment approach is not without merit, producing good results when compared to the passive and semi-passive benchmark portfolios. The use of (only) historical returns to forecast next period's joint return distribution, in conjunction with the multiperiod portfolio model, obtained

results that deal a damaging blow to the existence of weak-form efficient markets. (iv) The multiperiod portfolio model appears to be robust when extended beyond the asset allocation stage to a more micro level. Overall, the conclusions reached here with respect to the twelve-industry universe are consistent with the results of the Grauer-Hakansson studies on active asset allocation.

B. Portfolio Returns: The Base Case versus The Other Cases

We now examine the portfolio returns of the active strategies under various joint return distribution estimation techniques and with different sized industry investment universes. In particular, we focus on the differences in portfolio returns between these other cases and our base case and the benchmark portfolios. This section contains seven sub-sections.

1. A Simple Probability Assessment Approach: 8 and 24 Industries

We examine the effect that investment universe size has on portfolio returns. Specifically, we consider a contraction from twelve industries to eight and an expansion to twenty-four. For these investment universes, we estimate the joint return distribution by the simple probability assessment approach using the last 32 quarters of realized returns.

Tables 5a-d present the geometric means and standard deviations of the portfolio returns of the active strategies managing the eight industry universe; of the benchmark portfolios; and of the industries themselves for the periods 1934-86 and 1966-86. Tables 5a-b (5c-d) contain the results when the industries are equal-weighted (value-weighted), with and without leverage opportunities, respectively. Each of the Tables 5a-d have the results for the 1934-86 period in columns 1-2 and for the 1966-86 sub-period in columns 3-4.

- 51

Tables 6a-d, which are constructed in a parallel fashion to Tables 5a-d, present the results when the universe is twenty-four industries.

la. The Eight Industry Universe

Consider the equally weighted eight industry universe when leverage opportunities are present (Table 5a). For the period 1934-86 (columns 1-2), we find mixed results for the active strategies when compared with the benchmarks. The lower powers (-75 to -10) did slightly worse than their passive and semi-passive counterparts; the middle powers (-5 to 0) performed as well as or better than the benchmarks, while the high powers clearly did worse than the semi-passive strategies. A comparison with columns 1-2 of Table 3a (the base case) shows that the active strategies managing an eight industry universe had lower geometric mean returns, but this was compensated for by lower variability. On the whole, there was practically no difference for the more risk averse investors. On the other hand, the logarithmic policy earned 16.93% managing eight industries versus 18.28% managing twelve industries; the risk neutral had geometric means of 14.07% versus 16.91%; the power 0.5 investor earned 15.86% versus 18.01%. Surely such differences in mean returns over a 53 year period have economic significance. For example, in following the logarithmic policy, an 18.28% annual return implies that \$1 invested at the beginning of 1934 grows to \$7316 by the end of 1986, while an annual return of 16.93% results in \$3782.⁵⁷

For the same universe, the 1966-86 period (Table 5a, columns 3-4) produced about the same results as in the 1934-86 period. All the active strategies managing the twelve industry universe had higher mean returns than from

⁵⁷Whether the difference is statistically significant will be examined in Chapter VII.

managing the eight industry universe. Again, there were only marginal differences offer the lower powers, while the higher powers displayed less difference than in the full period. With this eight industry universe, all the active strategies outperformed their corresponding passive benchmark portfolios, and with the exception of powers 0.5 and 1, also had higher mean returns than the semi-passive strategies.

· · · · · ·

On removing the leverage opportunities, the active strategies managing the eight equal-weighted industries (Table 5b) had virtually identical results as when they managed the twelve equal-weighted industries (Table 3b) for both periods.

Now consider the value-weighted eight industry universe with leverage opportunities present (Table 5c). The more risk averse investors (powers -75 to -2) in the 1934-86 period (columns 1-2) slightly underperformed the corresponding passive strategies, while the rest of the active strategies did much worse. Comparing this to the twelve value-weighted industry universe (Table 4a, columns 1-2), we find that the geometric mean returns were about the same for the lower powers. The "superiority" of managing the twelve industry universe was apparent for the logarithmic to the risk neutral investors. For example, the power 0.5 investor managing the twelve value-weighted industries earned 13.95% per annum over the 53 year period compared to 10.91% when the universe is eight industries. This is equivalent to investing \$1 in 1934 and having \$1014 versus \$242 at the end of 1986. Surely this is economically significant.

For the 1966-86 period with the same investment universe (Table 5c, columns 3-4), we find the less risk averse active strategies being outperformed by the passive strategies while the other active strategies
managed to earn about the same as the passive benchmarks. However, when we compare these results to those of the active strategies managing twelve industries (Tables 4a, columns 3-4), we find fairly large differences, especially for the less risk averse investors (except for the risk neutral investor).

In the no leverage case for the value-weighted industries, we find the active strategies had lower geometric means than their corresponding passive benchmarks for both the full 1934-86 period (Table 5d, columns 1-2) and 1966-86 sub-period (Table 5d, columns 3-4). All the active strategies managing twelve industries, except for the risk-neutral in the 1966-86 sub-period, outperformed their active counterparts managing eight industries (Table 4b versus Table 5d).

1b. The Twenty-four Industry Universe

Tables 6a-d present the results when the active strategies manage the twenty-four industry universe. Consider first the equal-weighted industry universe when leverage is available. For the 1934-86 period (Table 6a, columns 1-2), we see that the active strategies had lower geometric means, than the corresponding semi-passive strategies, except for powers -2 and 0 which performed about as well. For the 1966-86 sub-period, (see Table 6a, columns 3-4), all the powers from -75 to -2 did well relative to the passive and semi-passive benchmarks. Only the Tobacco industry managed to generate a higher geometric mean than the best the active strategies could muster. The higher powers did relatively poorly, with the risk neutral losing money.

When we compare with the active strategies managing twelve industries, we find that powers -75 to -2 managing twenty-four industries had higher returns than when managing twelve industries for both the 1934-86 and 1966-86 periods

(Tables 6a versus 3a). However, the reverse was true for powers 0 to 1, with quite large differences. There is certainly economic significance here.

The results with no leverage opportunities, for both periods, are given in Table 6b. The more risk averse investors (powers -75 to -10) performed about the same whether leverage was available or not. The higher powers were clearly liberal users of leverage when available, so that removing such opportunities lowered their mean returns, except for the risk neutral investor in both periods and the power 0.5 investor in the 1966-86 period. Comparing Table 6b with Table 3b (the twelve industry universe), we find the full period mean returns (columns 1-2) to be comparable while for the sub-period 966-86 (columns 3-4), the higher powers managing the twelve industry universe clearly outperformed their counterparts managing twenty-four industries. For example, power 0 earned 15.14% versus 9.27% with comparable variability; power 0.5 earned 14.88% versus 5.41%; and 14.42% versus 2.57% for the risk neutral investor.

Turning to the twenty-four value-weighted industries, the active strategies, regardless of whether leverage was available or not, performed relatively poorly in the 1934-86 period compared to the passive benchmark portfolios (see columns 1-2 of Tables 6c and 6d). For the lower powers, the size of the universe did not matter. The higher powers (1 to -2) had higher geometric means managing the twelve industry universe (Tables 4a-b, columns 1-2) than the twenty-four industry universe (Tables 6c-d, columns 1-2); the differences were particularly acute in the leverage case. For example, power 0 earned 13.61% with leverage while managing twelve industries compared to 7.51% managing twenty-four industries, and at the same time had lower variability.

For the 1966-86 period (Tables 6c-d, columns 3-4), only the lower powers managed to do as well as the passive benchmarks (leverage did not matter for these powers). Nevertheless, the active strategies managing twelve industries (Tables 4a-b, columns 3-4) did outperform their counterparts managing the twenty-four industry universe (Tables 6c-d, columns 3-4).

Summarizing, we find that: (i) the active strategies did better when managing equal- rather than value-weighted strategies, regardless of the size of the investment universe; (ii) the results for the sub-period 1966-86 were usually better than for the full period, especially when managing the equal-weighted industries; (iii) the use of leverage, when available, tended to enhance the portfolio returns of the less risk averse investors, but at the expense of higher variability; (iv) there we're economically significant differences, especially for the higher powers, when managing different sized investment universes. In particular, the less risk averse active strategies managing twelve industries almost always did better than their counterparts managing either the eight or twenty-four industries. The differences were especially sharp when leverage opportunities were available. For the lower powers, there was almost no difference whatever the size of the investment universe. An explanation for this is that these investors hardly ever employed leverage when it was available, and they always had a large proportion of their funds in the riskfree asset. Thus, the variation in their portfolio returns is not (mostly) attributable to the variation in the industries invested, but rather primarily from the variability of the riskfree rate over time.

2. 12 Industries: A 28 and 40 Quarter Simple Probability Assessment Approach We now examine the portfolio returns when different amounts of realized

returns are used to estimate the joint return distribution for the next period. For the base case, which employed the most recent 32 quarters of realized returns to estimate the joint return distribution, we found the results to be most respectable. Now we examine the effects on portfolio returns when the estimating period is 28 and 40 quarters.

Tables 7a-b give the results for the active strategies managing the twelve equal-weighted industries, with leverage opportunities, for the 1936-86 and 1966-86 periods,⁵⁸ respectively, when the most recent 32, 28 and 40 quarters (columns 1-2, 3-4 and 5-6 respectively) of realized returns are employed to estimate the joint return distribution. Tables 7c-d present the results for the 1936-86 and 1966-86 periods when leverage is not permitted, and are constructed to parallel Tables 7a-b. Finally, Tables 8a-d are the equivalents of Tables 7a-d when the twelve value-weighted industries are the investment universe.

2a. The **J**8 Quarter Estimating Period

When the equal-weighted industries were the universe, we find that in moving from a 32 to a 28 quarter estimating period, the geometric means of all the active strategies decreased in the 1966-86 period (except for the risk neutral investor with leverage present), whether leverage was present or not (columns 1-2 versus 3-4 in each of Tables 7b and 7d). Furthermore, most of the active strategies had higher variability; with no leverage (see

⁵⁸Note that if 40 quarters of realized returns are use to estimate the joint return distribution, then the first quarter for which a portfolio can be selected is the first quarter of 1936. This is because the period 1926-35 inclusive is required to estimate the joint return distribution. Note that the earliest date for which we have data is the first quarter of 1926. Thus, for comparison purposes, we report portfolio returns only the period from, 1936 on for the 28 and 32 quarter simple probability assessment approaches.

Table 7d), all standard deviations except for power 0 went up, and with leverage (see Table 7b), only powers 0.5 and 1 had lower variability. For the 1936-86 period, all the standard deviations increased when using a 28 quarter vis-a-vis a 32 quarter estimating period (see Tables 7a and 7c), regardless of leverage opportunities. The direction of change in the geometric means were mixed. Generally, the lower powers (-75 to -5) and the risk neutral investor attained higher mean returns to compensate for the increased variability.

Turning to Tables 8a-d, columns 3-4 (28 quarter) compared with columns 1-2 of each table reveals that, with a universe of the twelve value-weighted industries, the standard deviations of the lower powers (-75 to -5) Increased while they decreased for powers 0 and up. This occurred for both periods, irrespective of whether leverage was permitted or not. The geometric mean returns, however, did not have the same consistency. They went up, usually for the lower powers, in the full 1936-86 period, but decreased, again for the lower powers, in the 1966-86 sub-period. For the more risk tolerant strategies (powers 0 to 1), the mean returns increased, when margin purchases were not permitted, in both periods (see Tables 8c-d). When leverage was allowed, these powers, except for the risk neutral, had their returns go down (see Tables 8a-b) in both periods.

To summarize, we generally found that going from a 32 to a 28 quarter estimating period for both the equal- and value-weighted industries, with or without leverage, resulted in: (i) higher standard deviations (usually) for the lower powers in both periods, and mixed changes for the others; and, (¶i) higher geometric means for the more risk averse strategies in the 1936-86 period, and lower mean returns for most powers in the 1966-86 sub-period.

Thus, decreasing from a 32 quarter to a 28 quarter estimating period produced consistent changes (i.e., up or down) in the geometric means and standard deviations for the more risk averse strategies, while the higher powers had more erratic results.

2b. The 40 Quarter Estimating Period

The use of 40, rather than 32, quarters of realized returns to estimate the joint return distribution produced some consistent results. Comparing columns 5-6 with columns 1-2 of each of the Tables 7a, 7c and 8a, 8c show that for the full period 1936-86, all the active strategies attained lower standard deviations accompanied by lower geometric means. The exception was the risk neutral investor. In the 1966-86 sub-period (see Tables 7b, 7d and 8b, 8d, columns 5-6 versus columns 1-2), we found lower geometric means for all investors, except for the risk neutral investor. However, the variability of portfolio returns deciined as well to compensate. The exception occurred when the equal-weighted industries were the universe; for the risk neutral investor regardless of leverage, and when the power 0.5 investor employed leverage . Thus, changing from a 32 to a 40 quarter estimating period produced less erratic changes in the geometric means and standard deviations of the active strategies when compared with changing to a 28 quarter estimating period.

In summary, the differences for the more risk averse strategies were relatively minor, while the more risk tolerant strategies produced economically, if not statistically, significant differences. For example, in the 1936-86 period, the logarithmic investor managing the equal-weighted industries with leverage opportunities available earned 18.07%, 15.85% and 16.01% with about the same variability when '32, 28 and 40 quarters were the

estimating periods respectively. The difference over this 51 year period is equivalent to investing \$1 at the beginning and realizing \$4777, \$1814 and \$1947 respectively at the end of 1986. This must be economically significant, and bodes well for the 32 quarter estimating period.

3. 12 Industries: A Simple Probability Assessment Approach with an Inflation Adapter

The first two columns of Tables 9a-d give the results when the simple probability assessment approach with an inflation adapter is used to estimate the joint return distribution. The estimation period is 32 quarters and the universe is the twelve equal-weighted industries. The leverage case for the periods 1934-86 and 1966-86 is shown in Tables 9a and 9b respectively, while Tables 9c and 9d shows the no leverage case for the periods 1934-86 and 1966-86, respectively. Tables 10a-d, columns 1-2, present the results when the twelve value-weighted industries are the universe and parallel columns 1-2 of Tables 9a-d.

3a. The Equal-Weighted 'Industries

The results for the 1934-86 period when leverage was permitted were mixed; the first two columns of Tables 9a and 3a show that for the low to middle powers, the inflation adapter method clearly enhanced geometric mean returns and reduced variability at the same time. For example, the -2 power earned 13.86% with a standard deviation of 21.32% under the simple probability assessment approach compared with the 14.24% and a lower variability (19.67%) with the inflation adapter. The higher powers (0 to 1) performed slightly worse with this method, but they compensated by having lower variability. Thus, the power policies with the inflation adapter, will shift slightly in the north-westerly direction from their present positions

in Figure 1a.

Turning to the 1966-86 sub-period with leverage permitted, columns 1-2 of Table 9b show respectable results for the active strategies. They clearly earned higher mean returns than their passive and semi-passive counterparts. For example, the logarithmic investor earned 15.84% compared to V14's 8.43% and V16's 8.12%, and to E10's 13.75%. The industries themselves as portfolios did not fare any better. Only the Food industry managed to outperform the active strategies. Comparing with Table 3a (columns 3-4), we find the inflation adapter method reduced the variability of the active strategies quite significantly over the simple assessment approach, at expense of a slight reduction of geometric means. The exception is the logarithmic investor. The standard deviation for this investor was reduced from 33.36% to 25.37%, with a drop in geometric mean from 19.05% to 15.84%.

With no leverage opportunities, the results of periods 1934-86 and 1966-86 (see Tables 9c and 9d respectively) were quite similar to the results from the simple probability assessment approach (Table 3b, columns 1-2 and 3-4 respectively). The recurring result was the reduction in variability of the portfolio returns of the active strategies with the inflation adapter method at the (mostly) slight expense of lower geometric mean returns.

3b. The Value-Weighted Industries

Turning to Tables 10a-d (columns 1-2) and comparing with Tables 4a-b (all columns), we find the following with respect to the inflation adapter method versus the simple probability assessment approach when the investment universe is the twelve value-weighted industries: (i) the variability of the portfolio returns of all active strategies decreased in both periods regardless of leverage opportunities. The comparison is columns 1-2 of

Tables 10a, 10b, 10c, and 10d against Table 4a, columns 1-2 and 3-4, and Table 4b, columns 1-2 and 3-4, respectively. (ii) In the 1934-86 period, the geometric mean returns of the lower powers decreased slightly, but the less risk averse investors (powers -5 and up) experienced increased mean returns whether leverage was available or not (see Tables 10a and 10c, columns 1-2 versus Tables 4a and 4b, columns 1-2). (iii) The 1966-86 period saw dramatic increases in portfolio mean returns for all the active strategies, especially for the higher powers (see Tables 10b and 10d, columns 1-2 versus Tables 4a and 4b, columns 3-4). For example, with leverage (no leverage), the geometric mean of the logarithmic investor increased from 9.34% (8.25%) to 12.35% (10.41%) while decreasing the variability from 30.29% (23.87%) to 25.39% (18.67%); the risk neutral investor had the mean rise from 0.02% (3.85%) to 32.07%(19.16%).

To summarize, the inflation adapter method appears to slightly enhance the results of the simple probability assessment approach when managing the equal-weighted industries, and thus supporting the conclusions reached in Section A. A graphical representation of Tables 9a-d (columns 1-2) will be identical to Figures la-d (which represent the simple probability assessment approach) except that the power policies will now plot slightly more to the left. However, the active strategies managing the twelve value-weighted industries, with or without leverage, showed sharp increases in geometric mean returns during the 1966-86 sub-period, especially for the higher powers, while experiencing lower variability. Thus, the power policies in Figures 2c-d will show a significant north-westerly shift from their present position.

4. 12 Industries: A "Sum-of-the-Digits" Probability Assessment Approach

Columns 3-4 of Tables 9a-d and Tables 10a-d give the geometric mean returns and standard deviations of the "sum-of-the-digits" method to estimating the joint probabilities of the simple assessment approach. We utilize the last 32 quarters of realized returns and assign the probabilities to each joint realization according to the "sum-of-the-digits" rule, with the most recent observations having the higher probabilities of occurrence than the earlier observations.⁵⁹

Specifically, columns 3-4 of Tables 9a-b and 9c-d present the results when the active strategies manage the twelve equal-weighted industries for the 1934-86 and 1966-86 periods, with and without leverage, respectively. Columns 3-4 of Tables 10a-b and 10c-d parallel columns 3-4 of Tables 9a-b and 9c-d except that the universe is now the twelve value-weighted industries.

4a. The Equal-Weighted Industries

Comparing columns 3-4 of Tables 9a-b and 9c-d with Tables 3 (columns 1-2 and 3-4) and Table'3b (columns 1-2 and 3-4) respectively, reveal that the standard deviations of portfolio returns of the active strategies employing this "sum-of-the-digits" probability assignment increased (marginally) over the simple probability assessment approach. The exceptions were the risk neutral investor, and occasionally the 0.5 power and logarithmic investor. This occurred for both the 1934-86 and 1966-86 time periods, and for when leverage opportunities were present and when they were not. However, accompanying these increases in variability were increases in the geometric mean returns, but not always. Specifically, during 1934-86 and with leverage.

⁵⁹This technique was described in Chapter IV.

powers 0 and 0.5 had lower geometric means, but higher variabilities (see Table 9a, columns 3-4 versus Table 3a, columns 1-2); similarly for 1966-86 and with leverage, powers -2 to 0.5 also had lower means (see Table 9b, columns 3-4 versus Table 3a, columns 3-4). When leverage was not present, powers -2 and 0 earned lower returns for the 1934-86 period (see Table 9c, columns 3-4 versus Table 3b, columns 1-2), while during 1966-86, powers -2 to 0.5 had decreased geometric means (see Table 9d, columns 3-4 versus Table 3b, columns 3-4).

4b. The Value-Weighted Industries

The results for the active strategies managing value-weighted industries were mixed. Turning to Tables 10a-d and Tables 4a-b, we find the results almost identical as when the equal-weighted industries were the universe. Specifically, for the 1934-86 with leverage opportunities present, all the standard deviations of the active strategies were slightly higher than when the simple probability assessment approach was used (columns 3-4 of Table 10a versus Table 4a, columns 1-2). Furthermore, only the geometric mean returns of powers -75 to -2 increased to offset the increased variability. For the 1966-86 sub-period with leverage, all standard deviations, except for the risk-neutral investor, increased while the geometric means changes were mixed; powers -75 to -10, 0.5 and 1 increased (see columns 3-4 of Table 10b versus Table 4a, columns 3-4). The no leverage results almost mirror the leverage ones. For the 1934-86 period, all active strategies except for powers 0.5 and 1 had higher variability, while the geometric means only decreased for powers -2 and 0 (columns 3-4 of Table 10c versus columns 1-2 of Table 4b). Finally, only the geometric means of powers -5 and -2 decreased in the 1966-86 period when all the moderate to low powers (-2 to -75) had higher

standard deviations (Table 10d, columns 3-4 versus Table 4b, columns 3-4).

In summary, the "sum-of-the-digits" method to assigning probabilities in the simple assessment approach had marginal effects on the active strategies' portfolio geometric means and variabilities. Most of the time, the standard deviations increased compared to the simple probability assessment approach, but they were usually tempered by increases in geometric means as well. Moreover, for the low powers, the increases in standard deviations were slight while their geometric means increased marginally more.

5. 12 Industries: An "All-of-History" Probability Assessment Approach

Columns 5 and 6 of Tables 9a-d and 10a-d present the "all-of-history" approach to estimating next period's joint return distribution. The initial number of realized returns used to begin this estimation is the first 32 quarters of data (this is the same as the start up point of the base case). We assign equal probabilities to each joint realization so that as we move forward in time, each realized state is assigned a lower and lower probability of occurrence in the next quarter.

Columns 5-6 of Tables 9a and 9b present the results when the active strategies manage the twelve equal-weighted industry universe, with margin purchases permitted, for the periods 1934-86 and 1966-86 respectively. The no leverage case is shown in columns 5-6 of Tables 9c and 9d for the same respective time periods. Columns 5-6 of Tables 10a-d parallel Tables 9a-d, columns 5-6, except that they show results when the universe is the twelve value-weighted industries.

5a. The Equal-Weighted Industries

Comparing Tables 3a-b with Tables 9a-d (columns 5 and 6), we find that employing the "all-of-history" method made the active strategies more.

conservative in their portfolio selection relative to the simple probability assessment approach as evidenced by comparing the geometric mean returns and standard deviations in the Tables. When the equal-weighted industries were the universe (columns 5-6 of Tables 9a-d), all the active strategies had lower mean returns than with the simple probability assessment approach, but this was offset by lower standard deviations. This occurred in both the 1934-86 (columns 5-6 of Tables 9a and 9c) and 1966-86 (columns 5-6 of Tables 9b and 9d) periods regardless of whether leverage opportunities were present or not. Furthermore, the powers -75 to -2 had standard deviations below 10%, but the variability for the other powers start above 20% (see Tables 9a-d, columns 5-6). These lower powers, regardless of leverage opportunities, performed about the same as the semi-passive, and slightly better than the passive strategies in the 1934-86 period (Tables 9a and 9c, columns 5-6). For the 1966-86 period, these same active strategies clearly outperformed the passive benchmarks and barely outearned the semi-passive strategies (see Tables 9b and 9d, columns 56 5b. The Value-Weighted Industries

For this universe (see Tables 10a-d, columns 5-6), we find about the same effects on the active strategies as in the equal-weighted universe case. Specifically, the standard deviations of all active strategies decreased, but not all geometric means declined at the same time. Only for the 1934-86 period with no margin purchases permitted did all mean returns also decrease (see columns 5-6 of Table 10c versus columns 1-2 of Table 4b), while with leverage, only the risk-neutral had a higher geometric mean than before (see Table 10a, columns 5-6 versus columns 1-2 of Table 4a). For the 1966-86 sub-period, powers 0 and 1 increased their geometric means when leverage was

available (see Table 10b, columns 5-6 versus columns 3-4 of Table 4a), together with power 0.5 when it was not (see Table 10d, columns 5-6 versus columns 3-4 of Table 4b). Thus, we see that for the 1934-86 period, the lower powers (-75 to -2) performed about the same as the passive benchmarks whereas before it underperformed. Moreover, the rest of the powers, while still performing worse than the benchmarks, did better than before; i.e., underperformed less. The results for the 1966-86 sub-period show that these lower powers still did better than the corresponding passive strategies, but the higher powers now did worse when leverage was present. Without leverage during the same time frame, the higher powers were more respectable (i.e., did not underperform the benchmarks as much as before) while the lower powers managed to outearn their respective passive benchmark portfolios.

By way of summary, there may be some merit to the "all-of-history" method of estimating the joint return distribution. It certainly is a means of reducing the standard deviation of portfolio returns, but at the expense of a lower geometric mean return. For the less risk averse investors, the method is an "improvement" over the simple probability assessment approach in that their returns went up while their variability went down when managing the value-weighted industries.

6. 12 Industries: A "Disaster States" Probability Assessment Approach

We present the "disaster states scenario" in Tables lla-d and l2a-d. We quickly review this scenario: basically we use the 32 quarter estimating period simple probability assessment approach as the initial starting point; then we add in the worst previous state of joint returns of each industry to the joint distribution of realized returns. Thus we add in an extra 12 states since we considered the twelve industry investment universe. The way that the

probabilities are assigned to the states was given in Chapter IV. We review it quickly here.

There are 32 "normal" states consisting of the most recent 32 quarters of joint realized returns. There are also 12 "disaster" states. The parameter J is an integer that varies between 1 and 12; i.e., $1 \le J \le 12$ and is used to vary the probabilities assigned to each "normal" state and each "disaster" state. The formulas are:

Probability of each normal state - 1/(32+J), and

Probability of each disaster state - $J/(32+J) \cdot 12$,

for $1 \le J \le 12$. We examine three cases: we set J-1,6, and 12. Notice that this formula is quite general. If we let J-0, then we are back to the simple probability assessment approach since the "disaster states" are assigned a zero probability of occurrence. When J-12, then each "normal" state and each "disaster state" has a probability of 1/44 of occurrence. When J-6, then each "normal" state is assigned 1/38 probability and each "disaster state" has 6/(12.38) probability of occurrence. Finally, when J-1, the "normal" states each has probability 1/33 while ¹ the "disaster states" are each given 1/(12.38). Thus, the "disaster states" are assigned decreasing probabilities of occurrence when J goes from 12 to 1.

Tables 11a and 11b present the results when the active strategies manage the twelve equal-weighted industries with leverage permitted for the periods 1934-86 and 1966-86 respectively. Tables 11c and 11d are like Tables 11a and 11b, except that leverage is not permitted. Columns 1-2, 3-4 and 5-6 of each of the Tables 11a-d describe the case when J-1,6, and 12 respectively. Finally, Tables 12a-d present the results when the universe is the twelve value-weighted industries, and is constructed to parallel Tables 11a-d.

'6a. "Disaster States" Scenario: J = 1

This scenario is shown in columns 1 and 2 of Tables 11a-d and 12a-d. Comparing with Tables 3a and 4a, we find lower expected returns and lower standard deviations for most active strategies for this scenario relative to the simple probability assessment approach. This is irrespective of leverage opportunities, whether the industries were equal- or value-weighted, and for both 1934-86 and 1966-86 periods. (Compare columns 1-2 of Tables 11a, 11b, 11c, and 11d with columns 1-2 and 3-4 of Table 3a and columns 1-2 and 3-4 of Table 3b, respectively. Make a similar respective comparison for Tables 12a-d and Tables 4a-b.)

More specifically, when the industries were equal-weighted, power 0.5 achieved higher returns in all cases (see Tables 11a-d columns 1-2 versus Table 3a, columns 1-2, 3-4 and Table 3b, columns 1-2, 3-4 respectively); power 1 had higher returns but also higher standard deviations in only the no leverage case (see Table 11c and 11d, columns 1-2 versus Table 3b columns 1-2 and 3-4 respectively). Thus, in this case, the lower powers consistently achieved lower geometric means and standard deviations, while the results for the higher powers were mixed (i.e., no consistent direction of change of the means and the standard deviations).

For the value-weighted industries, the results appear very similar to managing the equal-weighted industries. In general, most active strategies achieved lower returns vis-a-vis the simple probability assessment approach, while all active strategies had lower variability (compare columns 1-2 of Tables 12a, 12b, 12c and 12d with columns 1-2 and 3-4 of Table 4a and columns 1-2 and 3-4 of Table 4b, respectively). The risk neutral investor managed to attain higher returns in all cases, while the logarithmic investor had higher

returns only during the 1966-86 period.

In summary, for the equal-weighted industries with leverage present, the results for the 1934-86 period were about the same as with the simple probability assessment approach, except that the power 0.5 managed to fore substantially better while the other powers were just marginally better. During the 1966-86 sub-period, the most dramatic improvement again came from the power 0.5, while the lower powers all had slightly better results. When leverage was not permitted, the lower powers achieved about the same results as the base case in the 1934-86 period. Again, only power 0.5 showed any substantial benefit from this disaster state scenario. For the 1966-86 sub-period, the logarithmic and power 0.5 investor achieved the most benefit from this scenario. When the value-weighted industries were the universe, there was hardly any change from the base case for the 1934-86 period with leverage. The only exception was the risk-neutral investor. Even then, this investor managed to close the underperformance gap only a little relative to the passive strategies. During 1966-86, only the risk-neutral and logarithmic investors showed any substantial improvement over their counterparts in the base case. The geometric mean for power 1 went up from 0.02% to 13.56% with a moderate decrease in variability. Surely there is economic significance, at least for this investor. Finally, when no leverage opportunities were present, improvements from the base case were almost non-existent for the active strategies during the 1934-86 period. Overall, the powers 0 to 1 mostly showed significant improvements from the base case; these three investors had higher geometric means and variability as a result of this "disaster state" scenario. The major inference drawn from this is that the less risk averse investors stand to benefit more than their more risk averse

counterparts. This is because appending disaster states to the estimation distribution makes the distribution more conservative and hence contains the less risk averse investors' desire to take on risk.

6b. "Disaster States" Scenario: J = 6

This "disaster states" scenario is shown in Tables 11a-d and 12a-d, columns 3-4. Because we are assigning higher probabilities to each of the "disaster states", we expect to see more conservative results. This is borne "out when we compare with Tables 3a-b and 4a-b.

Comparing columns 3-4 of Tables 11a, 11b, 11c and 11d with columns 1-2 and 3-4 of Table 3a and columns 1-2 and 3-4 of Table 3b, respectively, we find the following results. First, all the active strategies attained lower scandard deviation, regardless of the time period, whether leverage was permitted or not, and irrespective of how the industries were weighted. Second, for the equal-weighted industries, all the powers had lower geometric o mean returns. The only exception occurred in the 1966-86 sub-period with leverage for the risk neutral investor. Third, when the universe was the value-weighted industries, the more risk averse powers (-2 to -75) all experienced lower mean returns. Only the risk neutral investor managed to earn more in both periods, with and without leverage. The other higher powers attained mixed results.

Comparing the results with the passive and semi-passive benchmark portfolios, we find mixed results. Take the equal-weighted industries first. For the 1934-86 period with margin purchases permitted (Table 11a, columns 3-4), this method actually made the active strategies worse than with the simple probability assessment approach and therefore worse than the semi-passive and passive strategies. In the 1966-86 period, the relative

performance was still good in that the active strategies certainly outperformed both benchmarks (see Table 11b, columns 3-4). Only the Food industry did better than the powers. When margin was not allowed, the active strategies in the 1934-86 period still underperformed the benchmarks (see Table 11c, columns 3-4), while for the 1966-86 sub-period they managed to do better (see Table 11d, columns 3-4).

When the value-weighted industries were the universe and leverage was available, the full period results had the powers being outperformed by all the benchmarks (see Table 12a, columns 3-4). The 1965-86 results (see Table 12b, columns 3-4) were better in the sense that (i) the powers had higher geometric returns than comparative passive benchmarks, and (ii) the powers plotted to the north-west of all the industries, with the possible exception of Food. When no leverage was allowed, the full period results remained similarly unchanged; i.e., comparative passive portfolios had higher geometric means (see Table 12c, columns 3-4). For the 1966-86 sub-period, the active portfolios plotted distinctly north-west of the industries and the passive benchmark portfolios (see Table 12d, columns 3-4). 6c. "Disaster States" Scenario: J = 12

The results for this scenario are shown in the last two columns of Tables 11a-d and 12a-d. Comparable results for the simple probability assessment case are in Table 3a-b and 4a-b. Since we are assigning each "disaster" state a probability equal to each "normal" state, we expect even more conservative performances than the previous "disaster states" scenarios.

A summary of the findings are: (i) all active strategies had lower standard deviations than the base case; and (ii) the geometric means were similarly decreased for the powers, save for power 1 in the 1966-86 period

with no leverage. Just by looking at columns 5-6 of Tables lla-d and l2a-d, this method assigned too high a probability to each of the "disaster states" in that there was little difference in the geometric means of the more and less risk averse strategies. This was beyond our original intention and motivation for introducing this type of scenario into our estimation process. Thus we can eliminate this particular scenario from any serious consideration in estimating the joint return distribution.

7. Concluding Remarks

The robustness of applying the multiperiod portfolio model to the reinvestment problem was examined. As a reference point, we chose the twelve industry universe with a 32 quarter estimating period simple probability assessment approach as our base case. We varied the size of the investment universe considered, from the twelve industry base to an eight and twenty-four industry universe while keeping the 32 quarter simple probability assessment approach to estimating period while retaining the twelve industry universe and the simple probability assessment approach. In all cases, we considered an equal- and a value-weighting construction of the industries, and with and without leverage opportunities present.

Our principal findings are (i) the more risk averse strategies performed about the same as the base case; (ii) the higher powers had a slight edge when the twelve industries were the investment universe; (iii) the higher powers managed slightly better performance with a 32 quarter estimating period. Novertheless, the overall results were quite respectable regardless of which of the three estimating periods were utilized and whether the universe was eight, twelve or twenty-four industries; (iv) there were

significant improvements when the no leverage constraint was removed, especially for the more risk tolerant investors during the 1966-86 sub-period; and (v) investors did better when managing an equal- weighted universe. Thus the simple probability assessment approach has much to recommend it. This concurs with, and reinforces, the conclusions found in the Grauer-Hakansson studies. Furthermore, the success of the simple probability assessment approach, which uses the past to (naively) forecast the future, in combination with the multiperiod portfolio model points to the non-existence of (even) weak-form efficient markets.

We further examined the efficacy of the simple probability assessment approach by presenting several methods by which to estimate next period's joint return distribution. As the first step, we replicated the inflation adapter to the simple probability assessment approach in Grauer-Hakansson (1988). Next, we drew upon the "common-sense" appeal found in univariate time series modelling. The idea here is that the future is more likely to be better modelled by the recent rather than the distant past. Hence, a first step in this direction is to assign more weight, by a "sum-of-the-digits" method, to the recent observations in the estimation of the joint return distribution. Noting that the methods described here so far represent a "moving window" approach to estimating the joint return distribution, we extend the analysis to consider an "all-of-history" approach. In other words, rather than just using the most recent N quarters, we supply all the realized returns to the model as the estimate of the return distribution. There is a different underlying assumption behind this approach. The "moving window" method assumes that the joint return distribution is not stationary while the "all-of-history" approach assumes it is. Finally, we note that the "moving

window" approach completely ignores all returns "out of the window". Thus, an investor using this approach "pretends" that the fourth quarter of 1929 never happened if this period is out of the "window". However, the disaster of 1929 did occur and disasters like it may occur again. Thus, we modify the simple probability assessment approach by including "disaster states" to the estimated joint return distribution. We considered three scenarios, each assigning a different probability to the "disaster states". The purpose behind this method is to keep, in the investor's mind, the fact that disaster states did happen and may occur again.

The primary findings are as follows. (i) The inflation adapter to the simple probability assessment approach revealed that this method did not produce the same remarkable results of Grauer-Hakansson (1988). Specifically, when the equal-weighted industries were the universe, the inflation adapter made the more risk tolerant investors perform slightly worse than the simple probability assessment approach i.e., lower geometric means, but they also had correspondingly lower variability. The more risk averse strategies, however, showed improvements in both geometric means (higher) and standard deviations (lower). For the value-weighted industries, there were significant improvements, particularly in the 1966-86 period. The more risk averse active strategies performed about the same, but the higher powers clearly benefitted from this approach. However, these results are still not as dramatic as those in Grauer-Hakansson (1988). (ii) With the "sum-of-the-digits" approach, the results were generally mixed. For the equal- and value-weighted industries, the lower powers showed higher geometric means, but also higher variability, while the higher powers usually performed worse, i.e., lower returns and higher variability. (iii) The "all-of-history" approach had a sobering effect

on the active strategies' standard deviations; they decreased, at times guite dramatically, but this was tempered by a decline in the geometric means. The less risk averse investors managing the value-weighted industries saw their mean returns rise and variability fall. Thus, this approach may have some merit, (iv) The "disaster states" scenario achieved the effect (intended or not) of making all investors more conservative in their portfolio selection. This was particularly evident when J-12 (i.e., assigning the "disaster states" the same probability as the "normal states"). The final judgement was that the J-12 scenario should be eliminated as an estimation approach. However, there may be some merit when J=1 in the "disaster states" scenario. This time the approach had the intended effect in that it subdued the investors desire for risk, but not to the extent of eliminating the equity market as an investment vehicle (unlike the case J-12). The striking effect the (occasionally substantial) decline in the variability of the was portfolio returns of the investors, trading off with a modest drop in the mean returns. This was particularly evident for the more risk tolerant strategies. Finally, in the case when J-6, we naturally find the results from this to be (mostly) in-between that of J-1 and J-12. We did find dramatic decreases in the variability with more modest declines in the geometric particularly evident when the twelve value-weighted means. This was industries were the universe and only for the 1966-86 period.

From an overall viewpoint, we have to return to the initial conclusion that the simple probability assessment approach has much to recommend it. However, each of the approaches considered here have their merits at times. For example, the inflation adapter approach performed quite well in the 1966-86 period. This is understandable as inflation only became a serious

concern from the 1960's on. We did not expect this approach to have much effect in the earlier periods, and the results bear this out.

To complement the results from examining portfolio returns and standard deviations, we present the portfolio compositions of selected powers of the base case. We note the striking features of the power policies under each of the other cases, comparing with the base case compositions whenever necessary.

C. Portfolio Compositions: The Base Case

The base case consists of two sets of twelve industries (equal- and value-weighted), 10 active strategies, and 212 decision points in the period 1934-86. Moreover, there are both leverage and no leverage cases to consider. Thus we cannot report the investment policies in great detail. Instead, we focus on the policies of two active strategies, powers -15 and 0, noting the major trends contained in them,⁶⁰ and relate the investment policies of these two investors to the other eight.

Tables 13a and 13b report the investment policies of the active strategies for powers -15 and 0 respectively, for the full period 1934-86 when they manage the twelve equal-weighted industries with leverage opportunities present. For each power, column 1 shows the period, the second column reports the quarterly portfolio return, columns 3 and 4 show the

⁶⁰Powers -15 and 0 were chosen for two reasons: (i) they are in the middle of the 10 powers; -15 is sufficiently risk averse for us to relate his policies with those of his more risk averse counterparts (-30 to -75) as well as to the -10 power. Power 0 is sufficiently risk tolerant so we can relate his policies to those around him, viz., power 0.5, -2 and -5. Besides, the logarithmic policy is interesting in its own right. (ii) The Grauer-Hakansson studies usually presented the portfolio compositions of these powers. This facilitates comparisons with their work.

proportion invested in the risk-free asset and the fraction in borrowing respectively, and the next twelve columns show the proportions invested in - each of the twelve industry categories. The notation for the industries shown in these tables was previously presented in Table 2. Tables 14a-b are identical to Tables 13a-b except that margin purchases are not permitted. Finally, Tables 15a-b and 16a-b are the value-weighted industry universe counterparts of Tables 13a-b and 14a-b, respectively. All the Tables 13a-b through 16a-b are constructed in a parallel fashion.⁶¹ Finally, summary statistics for the investment policies of all 10 powers are contained in Tables 17a-d. These tables show: the number of times an industry entered into the investors' portfolio (rows 1-12), the number of times the risk-free asset was an investment outlet (next row), how often margin purchases were made (next row), and the number of times the investor was completely in the equity market (next row). The next three rows give the maximum and minimum proportions invested in the risk-free asset as well as the number of quarters the portfolio earned negative returns. Finally, the last twelve rows depict the diversification of each investor, i.e., the number of times the investor invested in one industry, two industries, and so on.

1. The Equal-Weighted Industries

Tables 13a-b show the policies of two investors, powers -15 and 0, in the presence of leverage opportunities, for the 1934-86 period. The case with margin purchases not allowed is shown in Tables 14a-b. Summary statistics of all 10 power policies for this universe is contained in Tables 17a (leverage

 $^{^{61}}$ That is, the tables are numbered NNa-b, where NN is 13, 14, 15 or 16. The "a" following NN always refers to the power -15 investor and the logarithmic investor is always reported in "b".

case) and 17b (no leverage case).

Turning first to Table 13a, the leverage case, the power -15 investor made heavy use of the risk-free asset in his portfolio, lending continuously from 1934-1 to 1956-3 (i.e., from the first quarter of 1934 through to the third quarter of 1956), and again from 1962-3 to 1986-4. His investment in the risk-free asset started at 97% and he slowly reduced this proportion through the mid-60s, not dropping below 60% until 1954-4. From 1962-3 on, he averaged about 80% invested in the risk-free asset. However, never once was he 100% invested in the risk-free asset. This investor made use of the leverage opportunity when allowed to borrow in the period 1956-4 to 1962-2, and again in 1988-2 to 1988-4, thus interrupting his continuous lending policy. The margin purchases were occasionally to the maximum permitted (10 quarters out of 23).

Turning to diversification, we find this investor held each industry at least once (see Table 17a, rows 1-12). He held from one to three industries most times, with an occasional four-industry portfolio, and only once did he hold five (in 1951-3). The most popular investment outlet was Petroleum with holdings in 111 quarters, with Utilities next, which he held 78 times. The least popular industry was Consumer Durables which he held only once (a 4% holding in 1972-3).

Comparing this investor with his more risk averse counterparts, we find the following. (i) The -75 to -30 power investors never borrowed. (ii) The -75 and -50 investors had a continuous presence in the risk-free asset, while the -30 power investor held it in 207 quarters. The -75 investor never held less than 90% of his portfolio in the risk-free asset from 1962-3 on, except for the last quarter of 1986 when he held 89%, while the -50 investor always

had 80% or more invested in the riskless asset during this time period. (iii) The policies of the -10 and -5 investor were similar to the -15, except that they made use of leverage more frequently and more often to the limits allowed. (iv) The industries that the -15 investor favoured were likewise favoured by these investors; i.e., Petroleum was the favourite outlet and Consumer Durables was the least popular.

- F...

Turning to the logarithmic investor (see Table 13b and also 17a), we find a complete turnabout for this investor compared to power -15. First, the logarithmic investor only lent in 30 quarters, continuously from 1934-1 till 1938-4 and again in the second and third quarters of 1939. He staxed away from the risk-free asset until 1974-3. From this point on, he lent till 1976-4, with the exception of two quarters, after which time the risk-free asset was never in the portfolio. With the exception of one quarter (1938-2), the proportion invested in the risk-free asset never exceeded 40%. Second, this investor made liberal use of margin, employing it continuously from 1939-4 to 1974-1, except for five quarters in the 1940s, four of which (1946-2 to 1947-1) had 100% margin requirements which made borrowing impossible. After 1977-4, leverage was again employed continuously save for six quarters. Usually, this investor borrowed to the margin limits in effect at the time.

From Table 17a, we find that the two most popular investment outlets were Petroleum (100 quarters) and Services (69 quarters), while his least favourite was Consumer Durables (2 quarters). Finally, the portfolio of this investor had at most three industries at any one time, and this was achieved but 19 times out of 212 (see Table 17a). Of note is the fact that this investor lost money in 81 quarters vis-a-vis 44 times for the -15 investor.

Comparing the logarithmic policy with the other powers, we note the following. (i) The power -2 investor behaved similarly to the power 0, except that he borrowed less and only occasionally to the limit. Furthermore, he lent more often and diversified into five industries once. (ii) The 0.5 and risk neutral investors never lent, and borrowed in 198 and 206 quarters respectively, most times to the margin limit. Of the six quarters that the power 1 investor did not borrow, only two were by choice (1974-3 and 1974-4). This investor diversified into two industries seven times, from 1938-1 to 1939-3.⁶² At other times, he restricted his portfolio to holding only one industry, i.e., in 205 out of 212 quarters, this investor held a one-industry portfolio. (iii) The Petroleum industry remained heavily favoured by these investors.

Turning to the case where margin purchases were not permitted, we find that the investment policies of the power -15 investor were not much, different than when leverage was permitted (see Table 14a versus 13a and Table 17a versus Table 17b). This is not surprising. The investment policies were the same for the periods 1934-1 to 1956-3 and from 1962-3 on. In these periods, the power -15 investor never borrowed even when allowed to do so, hence the margin restrictions were not binding. The only difference in the policies appeared in 1956-4 to 1962-2, but the industries invested in were identical except for 1958-4; the proportions held in each were just slightly less for the no leverage case. For the even more risk averse strategies (-30 to -75), the policies were identical.

Examining the policies of the logarithmic investor (Table 14b), we find

⁶²This happened because the solvency constraints of the model given by equation (9) in Chapter II were binding.

more differences between the leverage and no leverage cases. Without leverage, the Petroleum industry continued to be the favourite outlet with 88 quarters of investment, closely followed by Services (with 71 quarters). The least popular industry was now Construction (4 quarters) whereas before it was Consumer Durables. This time however, the logarithmic investor held up to five industries (albeit only once) compared to only a maximum of three in the leverage case.

With respect to the other powers, we see from Table 17b that the risk neutral concentrated in one industry all the time. The industries invested in were virtually identical to the ones held when leverage was permitted. The power 0.5 concentrated his investment to one industry even more than when leverage was allowed. Furthermore, he reduced the maximum number of industries held at any point in time to three (versus five when leverage was present). When we remove the leverage opportunities, the portfolios of the higher powers (-5 to 1) which were invested in only one industry increased, from 65 to 68 for power -5 and from 205 to 212 for the risk neutral investor.

There are definite trends evident when we examine the policies of all the active strategies. A striking feature is that each industry was present in the portfolio of the investors at least once. However, not once did any investor have more than five industries in his portfolio. Only seven of the ten powers, with leverage present, diversified into five industries, and they did this but once, while six powers did likewise when the leverage opportunities were removed. The risk-free asset became less important an investment outlet as risk aversion decreased, to the point where it was completely ignored by powers 0.5 and 1. Likewise, a one-industry portfolio became more of the norm as risk aversion decreased. Finally, we note that the

investors had about the same number of loss quarters with or without leverage present. The effect of leverage is best displayed in terms of returns (see the first two columns of Tables 3a and 3b) With leverage, the more risk tolerant strategies certainly earned more, but at the expense of greater variability.

2. The Value-Weighted Industries

Tables 15a, 16a and 15b, 16b present the portfolio compositions of the power -15 and logarithmic investors, respectively, when the value-weighted industries are the investment universe. Tables #15a-b contain the leverage cases while Tables 16a-b are when leverage (borrowing) is not permitted. Summary statistics for all the powers are contained in Tables 17c (with leverage permitted) and 17d (with no leverage permitted).

Turning first to Table 15a, when leverage opportunities are, present, the power -15 investor's portfolios consisted primarily of the risk-free asset. He lent continuously from 1934-1 to 1955-1, beginning with 97% invested in the risk-free asset and decreasing ever so gradually to 59% in 1954-3. He remained out of the market for this asset until the end of the second quarter of 1962, at which time the risk-free asset again had a strong presence in his portfolio to the end of 1986. In the interim period from 1955-2 to 1962-2, this investor made heavy use of leverage beginning in 1956-4 until 1962-2, most times up to the margin limits (17 quarters out of 23). Only once did this investor stay away from the equity market, and that was in 1974-4.

Looking at Table 17c, we find that the power -15 investor diversified into seven industries once (1956-4), but invested mostly in one or two industries at a time. The most popular outlet was (again) Petroleum, held for 100 quarters, with Services a close second at 83 quarters while the least

popular was Finance & Real Estate (13 quarters) and Transportation (tie). There were some substantial changes in which industries were held compared with when the universe was equal-weighted. For example, Consumer Durables was held 26 times here versus just once, and Capital Goods for 59 quarters here versus 11.

Comparing with the other investors, we find the following. (i) The powers -30 to -75 never made use of leverage. (ii) Powers -50 and -75 had a continuous presence in the risk-free asset, while the -30 investor held it for 195 quarters. The -75 power had over 90% of his portfolio in the riskless asset from 1934-1 to 1954-3, and from 1962-3 to 1985-4. This represents 177 quarters out of 212. (iii) The =10 and -5 investors had policies not unlike that of power -15, except that they made slightly more frequent use of leverage. (v) The industries favoured by the -15 investor were similarly looked upon by these two powers.

For the logarithmic investor (see Table 15b), we find him to be a frequent and heavy user of leverage, beginning in 1940-1 through 1973-3, with the exception of three short interruptions - the fourth quarter of 1939, the second and third quarters of 1942, and from 1946-2 to 1947-1 when margin was not permitted. From 1973-4 on, he resorted to borrowing in only three quarters over the next nine years, becoming fully levered from 1982-4 on Most times he was at the margin limits. Compared with when he managed the equal-weighted universe, this investor resorted to leverage in 14 less quarters.

Turning to diversification, this investor (see Table 17c) never held more than four industries, holding four only twice - in the first and second quarters of 1943. More than half the time, he held only one industry. Like

[•] 84

the power -15 investor, he stayed away from the equity market in 1974-4. His favourite outlet was Services (63 quarters), with Petroleum and Capital Goods a close second and third (63 and 62 quarters respectively). The least popular was Finance & Real Estate, held only twice.

Comparing the logarithmic policy with other less risk averse investors, we find the following. (i) The power 0.5 investor lent 12 times, and even the the risk neutral lent (in 1974-4). (ii) The risk neutral diversified into two industries nine times, while concentrating his holding in one industry in 202 quarters. Thus, the diversification across industries declined as the investors became more risk tolerant. (iii) Finally, we see that the risk neutral investor ignored the category Basic Industries, while the rest of the powers invested in each industry at least once.

Turning to Table 16a, the no leverage case, the -15 investor performed much as he did when leverage opportunities were available. Again the investor held seven industries once (in 1956-4), while one and two industry portfolios remained his favourite choice (see Table 17c). The same industries, as in the leverage case, remained popular and unpopular.

The logarithmic investor's (see Table 16b) portfolio choices were more affected by leverage opportunities than were the -15 power. Nevertheless, the portfolio choices were similar, not in fractions of wealth invested in the industries, but in which industries made up the portfolio. The favoured industries remained Services, Capital Goods and Petroleum, while the least popular continued to be Finance & Real Estate. However, this investor had a one-industry portfolio in 140 quarters compared to 111 quarters when leverage was permitted. Moreover, he only had up to three industries in the portfolio at any time compared to up to four with leverage.

The more risk averse investors (-30 to -75) made no use of leverage even when it was available. At the other end of the risk spectrum, the risk neutral investor always had a portfolio of one industry (except in 1974-4 when he lent); i.e., for 211 quarters, whereas in the leverage case he was "forced" to diversify by the solvency constraint (see footnote 10). Moreover, this investor now completely ignored two of the twelve industries - Basic Industries as in the leverage case and Utilities - and invested in one industry twice (Finance & Real Estate and Transportation, as he did when leverage was available).

Summarizing, we can see some definite trends when the industries are value-weighted. All the powers, except the risk neutral, invested in each industry at least once. Both the -15 powers (leverage and no leverage cases) had up to seven industries in their portfolio once. The more risk averse powers invested in up to six industries, and the diversification dropped as the risk aversion declined, culminating with the risk neutral having 211 quarters in which he had a one-industry portfolio. Suprisingly, all the powers lent at least once in contrast with when the universe was equally weighted. While the more risk averse investors (-30 to -75) had several quarters in which they were fully invested in the riskless asset, 1974-4 was the only quarter in which all the investors stayed away from equities. Finally, we see that the number of quarters in which the portfolio earned negative returns were about the same whether leverage was employed or not. As with the equal-weighted industry universe, leverage, when employed, tended to enhance the portfolio returns but increase the variability.

D. Portfolio Compositions: The Base Case versus The Other Cases

We now proceed to describe the salient features of the portfolio compositions of the investors under the various cases considered in the thesis. We have to consider: (i) the twenty-four and eight industry universes, (ii) the 28 and 40 quarter simple probability assessment approach, (iii) the 32 quarter simple probability assessment approach with an inflation adapter, and ,(iv) the "sum-of-the-digits" probability assessment approach, (v) the "all-of-history" simple probability assessment approach, and (vi) the three "disaster states" scenarios to the 32 quarter simple probability assessment approach. Note that (ii)-(vi) above utilize the twelve industry universe only. The above cases are summarized in Tables 18a-b through to Tables 27a-b respectively.

For each case, we have 10 active strategies, equal- and value-weighted industry universes, and with leverage and no leverage present. Given the tremendous number of portfolio policies generated, we consider the -15 power and the logarithmic investors' portfolio policies, and remark only on the differences, if any, with our base case equivalents. Furthermore, the analyses of the portfolio compositions in the base case have shown that the leverage and no leverage cases are quite similar in terms of industries invested (number and type, but in smaller proportions) so that we report on the results of only the leverage cases.

1. 8 and 24 Industry Universes

We turn our attention to a description of the salient points when the 32 quarter estimating period simple probability assessment approach is applied to the eight and twenty-four industry universes. A summary is given in Tables 18a-b and 19a-b respectively. However, we will describe the policies in more detail than with the other cases since this is the only time we use a

universe other than the twelve industry base case.

la. The Eight Industry Universe

Consider the eight equal-weighted industry universe first (see Table 18a). The -15 power was in the risk-free asset continuously from 1934-1 to 1956-3, and again from 1962-3 on, while the logarithmic investor lent from 1934-1 to 1938-4, and from 1974-3 to 1976-1. Borrowing for the -15 power was sporadic in the non-lending quarters and only very occasionally was he at the margin limits. The power 0 investor was very different, employing margin to the limit with great regularity.

In terms of diversification, the -15 and 0 investors invested in all industries at least once, with Energy the favourite outlet (124 and 105 quarters respectively) and Construction the most unpopular (once and 14 times respectively). Construction was completely ignored by the -75 and -50 powers. The maximum invested in industries was three for both investors, although the other powers occasionally went to four. Thus, in comparison with our twelve industry base case (see Table 17a), these powers found the same outlet to be the most attractive (Petroleum and Energy); invested in the risk-free asset about the same number of quarters (190 and 29 quarters for -15 and 0 here versus 189 and 30 quarters in the base (ase); and had about the same number of loss quarters (44 versus 43 for power -15 and 80 versus 81 for power 0).

When the universe was value-weighted (Table 18b), the -15 power lent from 1934-1 to 1955-1, and again from 1962-3 on. Power 0 lent from 1934-1 to 1939-3 and 1973-3 to 1976-1, and sporadically thereafter. Leverage was employed to the limit by the logarithmic investor most times, while the -15 power was more judicious in his use of leverage.

Like the equal-weighted universe, all industries were chosen at one time

or another. The most popular investment vehicle was (again) Energy while Construction (again) seemed to be out of favour, for all investors. This time however, five industries were the most in any portfolio. A key difference for the powers 0.5 and risk ineutral investors (for the value- versus equal weighted universe) was that thow they invested in the risk-free asset as well, forsaking completely the equity market at times. This also occurred in our twelve industry base case. Finally, for both the -15 and 0 powers, the number of losing quarters were about the same; and they lent and borrowed for about the same number of quarters as they did with the twelve industry case.

1b. The Twenty-four Industry Universe

Turning to the twenty-four equal-weighted industry universe (see Table 19a), we find some interesting results. The -15 investor lent from 1934-1 to 1954-3, and again from 1962-3 on, while the power 0 investor lent only to 1939-3 (in four quarters prior to 1939-3, he did not hold the risk-free asset) and thereafter only twice, both in the mid-1970s. Leverage was employed by power -15 continuously/from 1956-4 to 1962-2 (about half the time to the limit). The 0 power investor employed borrowed funds almost continually from 1939-4 on, save for twelve quarters in the 1970s, and usually to the margin limits.

The investment policies were quite different (see Table 19a) from the twelve industry case. The -15 investor completely ignored three industries (Banks & Financial Services; Chemicals, Rubber & Plastic; and Fabricated Metal Products), while the logarithmic investor ignored the same thre plus Food, Textiles and Utilities. In general, all investors ignored at least one industry with the twenty-four industry universe. The most industries held in a portfolio was six, and most powers preferred two to four. The popularity of
Tobacco, Mining and Oil & Gas Exploration was evident.

Looking at the value-weighted industries (Table 19b), the lending policies of the -15 power, from 1934-1 to 1954-3 and from 1962-3 on, were identical with when the universe was equal-weighted. The power O policies, however, were radically different; here the risk-free asset was in the portfolio only eight times - three in the mid 1930s and five in the late 1970s to early 1980s. This time the -15 investor borrowed to the maximum allowed about 75% of the time, while the power O again usually margined to the limit.

Considering diversification, not all industries were in the investors' portfolios (see Table 19b). Banks & Financial Services and Metals were ignored by the -15 investor, while power 0 did not entertain Banks & Financial Services, Fabricated Metal Products, and Railroads & Trucking. Again, Tobacco and Oil & Gas Exploration were the favourite investment outlets. A surprising result was that the risk neutral investor diversified in up to four industries because of the solvency constraint (see footnote 10), albeit only once, while the most any investor held was eight industries (for the -50 and -75 powers).

Comparing the value- with the equal-weighted universe for the -15 and 0 powers, we find very different industries favoured. For example, Chemicals, Rubber & Plastics now entered the portfolios over twenty times while it was ignored previously.

Summarizing, we find both differences and similarities between the base case twelve and the twenty-four industry universes. The lending policies of the -15 investor were generally alike in that he lent and borrowed in the same quarters, while those of the power 0 investor were much more varied. One

facet that did not change was that these investors lost money about the same number of times (about 80 for the power 0 and about 45 for the -15 power), regardless of leverage, the size of the investment universe and the weighting scheme. In general, none of the investors diversified widely, with a portfolio of two to four industries being the most frequent.

2. The 28 and 40 Quarter Simple Probability Assessment Approaches

We examine the policies of the -15 and 0 power investors for the twelve industry universe, noting only the differences, if any, with respect to the 32 quarter base case. A summary of the 28 quarter and 40 quarter estimating period simple probability assessment approach to managing the twelve industry universe (equal- and value-weighted) is given in Tables 20a-b and 21a-b respectively. Note that we present the results from 1936 on since this is the first year -in which the 40 quarter estimating period approach makes a portfolio selection.

2a. The 28 Quarter Estimating Period

3.9

Considering first the equal-weighted industries (see Table 20a), the power -15 investor ignored the risk-free asset beginning in 1955-4 (one year earlier than in the base case).⁶³ In the interim period, he borrowed, usually to the margin limits and more than in the base case, except for the second quarter of 1960. The power 0 investor, on the other hand, lent from 1936-1 to 1936-4 (two years less), and only three times thereafter until 1974-3. This lasted until 1976-2, and except for 1979-1, he did not have the risk-free asset in his portfolio again. He almost had a continual levered position from

 $^{^{63}}$ Note that all comments are in comparison with the base case investor, i.e., the power -15 investor in the other cases with the power -15 investor of the base case.

1937-1 on, levering just five quarters more than in the 32 quarter base case. Thus, the lending and borrowing policies appear somewhat similar to the base case, except that the investors began levering earlier and a little more often.

No industry was ignored by the investors (see Table 20a). In terms of diversification, the most industries held was four, achieved five times by power -15 (and the middle powers). This compares with the 32 quarter base case (Table 17a). Furthermore, the same industries were favoured by the investors here as were in the base case. A small difference was that the investors here usually lost in slightly fewer quarters. A striking difference was that only the -75 investor in the 32 quarter base case completely ignored the equity markets (three times) while all investors here ignored equities at least once and up to eight times for the power -75. Moreover, in this case, all investors lent whereas powers 0.5 and 1 never did (Table 17a).

When the findustries were value-weighted, the power -15 began levering a year and a half earlier. This also occurred for the power 0 investor. Thus investors with a 28 quarter estimating period borrowed more times (and therefore lent less times) than in the base case. We see that investors here were slightly more inclined to be 100% invested in the risk-free asset (see Table 20b versus 17c).

Where diversification was concerned, we note that in the base case the -15 power held a portfolio of seven industries (once) whereas the investor in the 28 quarter estimating period case managed to hold only six (three times). The same investment outlets remained popular, but all industries were held at least once vis-a-vis the 32 quarter base case where the risk neutral investor ignored at least one industry. Moreover, the 'investors here lost

money in fewer quarters.

Summarizing, we find that while there were differences in the policies of the investors when the 28 quarter estimating period was compared with the 32 quarter base case, they were not large. Nevertheless, the differences did have an effect on the geometric mean returns and variability (see Table 7a). 2b. The 40 Quarter Estimating Period

For the equal-weighted industries, a key difference of the powers -15 and 0, with a 40 quarter estimating period compared to the 32 quarter base case; was that here they levered less frequently, i.e., were in the risk-free asset more often, and to a lesser extent. With respect to diversification, we find (see Table 21a versus 17a) the same industries were favoured (and out of favour). Moreover, five industries were the most held in any portfolio, although this occurred more often here than in the 32 quarter base case. Finally, we note that the powers here had less quarters of negative portfolio returns than in the base case. However this did not translate to higher geometric mean returns overall (see Table 7a).

Turning now to the value-weighted industries in Table 21b and comparing this table with Table 17c, we find that the powers, with the 40 quarter estimating period, lent more often (see the lending row).⁶⁴ The next row shows the the powers levered themselves less frequently.

The industries chosen generally matched in terms of relative popularity, but we see, for example, that Petroleum was more favoured with a 40 quarter estimating period (chosen more than half the time) than when the 32 quarter

⁶⁴Note that Table 21b starts in 1936-1 while Table 17c begins with 1934-1, for an 8 quarter difference. But most powers lent in these 8 quarters so just subtract 8 quarters from the lending row in Table 17c to get a direct comparison.

estimating period was used. A notable difference is that the risk neutral ignored four industries completely this time, versus only one in the base case. Furthermore, this investor was forced to diversify in 15 quarters. Finally, a significant difference is the number of times each investor was 100% invested in the risk-free asset.

Thus, while the differences were not major, they did contribute to the more risk tolerant powers having a lower geometric mean than the base case (see Table 7a).

3. A Simple Probability Assessment Approach with an Inflation Adapter

Tables 22a-b present a summary of the portfolio choices of the investors when the 32 quarter estimating period simple probability assessment approach with an inflation adapter is used to estimate the joint return distribution.

When the industries were equally weighted, we find notable differences in the investors' policies when the inflation adapter and the simple probability assessment approaches are employed. First, the inflation adapter caused the all the powers to lend more often (see Table 22a versus 17a). Even the risk neutral investor was fully committed to the risk-free asset in fourteen quarters, whereas previously it was fully held but once. We see that all investors stayed out of equities for at least fourteen quarters (up to twenty-two quarters for the power -75), whereas only power -75 did so previously, and only for three quarters. This in turn shows up in the use of leverage; it was employed less frequently and in lesser amounts. Second, the approach dampened the investors' enthusiasm for all industries; for example, the Petroleum industry was held for less than 100 quarters by all investors, in contrast to the simple probability assessment base case. Third, not more

than four industries were held compared with up to five previously. Finally, the investment policies were not much different up to 1966; thereafter, investors here tended to use more of the risk-free asset than in the base case (and hence used less leverage).

Examining the value-weighted industries (Table 22b versus 170%), we find the same striking differences as in the equal-weighted industries case. The investors lent more often with the inflation adapter, but more notable is the fact that they completely ignored equities in at least twenty-one quarters and up to thirty-three quarters for power -75, whereas with the simple probability assessment approach most powers were fully invested in the risk-free asset for only one quarter, and up to eleven quarters for power -75. Finally, there was lesser enthusiasm for the industries, i.e., the investors usually had the industries in their portfolio less often, e.g., the power -15 investor held Petroleum in 17 less quarters.

Summarizing, we see some striking differences in the policies of the investors when the inflation adapter is used to adjust the simple probability assessment approach to estimating the return distribution. However, when we compare Tables 3a (columns 1-2) and 9a (columns 1-2), we find higher geometric means (with slightly more variability) in the base case when the universe is equal-weighted. Thus the differences in policies did not show up as strongly as differences in geometric means and variability of returns. Even the period 1966-86 showed little difference (see Tables 3a, columns 3-4 and 9b, columns 1-2). When we examine Tables 4a (columns 1-2) and 10a (columns 1-2) we see marked differences for the 1934-86 period when the universe is value-weighted. There were higher geometric means and lower variability for the higher powers (-2 and up) with the inflation adapter.

Furthermore, these differences show up for all powers in the 1966-86 period (see Tables 4a columns 3-4 and 10b, columns 1-2). Thus, the inflation adapter method enhanced the returns of investors managing a value-weighted universe, particularly for the later years.

4. A "Sum-of-the-Digits" Probability Assessment Approach

The results for this case are presented in Tables 23a and 23b for the equal- and value-weighted industries respectively. Turning first to the equal-weighted industries (see Table 23a) and comparing with the simple probability assessment approach base case (Table 17a), we find slight differences in the investors' policies. First, this method caused the powers 0.5 and 1 to lend. Moreover, all the investors were 100% invested in the risk-free asset for at least three quarters whereas only the power -75 in the base case did the same. Second, the investors placed slightly less reliance on a one-industry portfolio and yet never held more than five industries; only the middle powers invested in five industries and they did it more than once. The base case powers held a five-industry only in one quarter, but more investors did it. Third, the risk neutral investor ignored Capital Goods, whereas all industries were held in the base case. The previously favorite industries (Petroleum and Services) continue to remain an important investment outlet, but to a lesser extent. This method "enticed" the investors to hold more of the Finance & Real Estate, Food & Tobacco and Leisure industries.

Turning to the value-weighted industries (Tables 23b and 17c), we find similar results. Specifically, the investors were fully in the risk-free asset more times than with the base case. As well, they made more frequent use of leverage. However, the effect on which industries were held were

almost reversed. Petroleum and Food & Tobacco increased in importance while Leisure continued to be held about the same number of times. A significant difference is that the base case investors held up to seven industries, while six was the most in this case. Furthermore, the lower powers diversified a little more than the base case powers; there were more portfolios containing two to four industries.

While there were some differences when compared with the base case, they were not large. Comparing returns, we findthe lower powers had higher mean returns than in the base case (see Tables 4a and 5a versus 9a and 10a). This could be due to the lower powers holding less one-industry portfolios than in the base case.

5. An "All-of-History" Probability Assessment Approach

Tables 24a-b is a summary of the portfolio compositions when investors employ an "all-of-history" probability assessment approach to estimating the joint return distribution. We previously noted that this approach did not do as well in terms of returns as the base case 32 quarter simple probability assessment approach (see Tables 3a, columns 1-2 versus 9a, columns 5-6 and Tables 4a, columns 1-2 versus 10a, columns 5-6).

For the equal-weighted industry universe (see Table 24a versus 17a), we find large differences in the portfolio compositions. First, many industries never entered into the investors' portfolios with the "all-of-history" approach whereas they were all held, at one time or another, in the 32 quarter base case. Second, even the moderately risk averse investors (up to power -2) never made use of leverage compared with the base case investors. These investors always had the risk-free asset in their portfolios in relatively high proportions; for example the -75 power never had more than 3%

of his portfolio in equities, and the -5 power had a minimum of 68% invested in the risk-free asset all the time. Third, the middle powers held a three-industry portfolio more than half the time versus about a quarter of the time in the 32 quarter base case. Fourth, the lower powers had fewer quarters in which their portfolio earned negative returns.

When we examine the value-weighted industry universe (Tables 24b versus 17c) case, we find similar marked differences. A minimum of five industries were never held by the investors. The low to middle powers always lent, and had more quarters in which they were fully invested in the risk-free asset. Even the risk neutral had four quarters in which he only lent. The conservatism of the middle to lower powers shows up when we compare loss quarters. Surprisingly. the power -75 investor had a one-industry portfolio more than half the time, while the other powers tended to choose a two- to four-industry portfolio.

Summarizing, we find marked differences between the policies of the investors under the "all-of-history" approach and the 32 quarter simple probability assessment approach. The lower powers were definitely more conservative in their portfolio selection, and this showed in their mean returns (Tables 3a versus 9a, and 4a versus 10a).

6. A "Disaster States" Probability Assessment Approach

The summaries of the three "disaster states" scenarios are presented in Tables 25a-b through 27a-b. Table 25a-b contains the results when investors face a "J-1" "disaster states" scenario. Of the three scenarios, this one assigns the lowest probability of occurrence to each of the "disaster" states. Table 26a-b contains results for the "J-6" "disaster states" scenario, and Tables 27a-b presents the "worst" case scenario results; for

"J=12" which assigns each "disaster" state a probability equal to each of the "normal" states.

Turning first to Table 25a, the "J-1" "disaster states" scenario, we find that it almost compares with the base case (Table 17a). A difference was that these investors were slightly more conservative, i.e., they lent more (and hence borrowed less often). However, they tended to hold one-industry portfolios. Furthermore, Consumer Durables never entered into the five most risk averse investors' portfolios, whereas all industries were held in the base case. Clearly, the addition of these "disaster states" made the investors more conservative; even the power 0.5 lent (for 8 quarters).

Comparing Tables 25b with 17c (the value-weighted universe), we again find only minor differences. Specifically, both cases had investors holding up to seven industries, but we notice that the "disaster states" scenario had the investors fully in the risk-free asset more times; at least for ten quarters, and up to 34 quarters for the power -75. Clearly the method affected the high powers more; they had fewer quarters in which they levered their portfolios.

Turning to the "J=6" "disaster states" case (equal-weighted industries), we see from Tables 26a versus 17a that this "higher probability of disaster" scenario had a pronounced effect on portfolio policies. An outstanding difference was that the investors in this case had at least 66 quarters in which they ignored equities, rising to 117 quarters for the power -75 investor. Contrast this to the base case, and we find that this "middle" probability assignment to the "disaster states" had a sobering effect on investor policies. Petroleum, Utilities and Services continued to be their favourite investment outlets whenever they decided to venture into the equity

markets. Capital Goods were ignored all the time. We note that, despite the existence of such "disasater states" in the joint return distribution, the powers -15 to 1 investors still levered, albeit less frequently than in the base case. The logarithmic investor, for example, borrowed in 51 quarters; from 1954-4 for 31 consecutive quarters up to the end of 1962-2, and from 1986-2 to the end, he was fully levered to the limits allowed.

Comparing Tables 26b with 17c (the value-weighted industries), we find the same effects on the portfolio policies of the investors as in the equal-weighted universe. The investors were fully invested in the risk-free asset in more than 100 quarters, even the risk neutral investor. However, these investors ignored less industries than when the universe was equally weighted.

Turning to the "J-12" "disaster states" scenario (Table 27a) when the industries were equal-weighted, we (naturally) find that this scenario had an even more sobering effect on the investors. A minimum of 156 quarters were devoted to portfolios of the risk-free asset only. Five industries were ignored altogether, with Utilities and Services being the popular choices whenever the investors were in the equity markets. Nevertheless, the powers -15 and up investors still levered themselves. For example, the logarithmic investor was levered from 1954-4 through to 1962-2 (only), employing margin to the limit in all but the first quarter.

Finally, looking at Tables 27b versus 17c (the value-weighted universe), we basically find the same results as above. A risk-free asset portfolio was the norm for at least 160 quarters; the powers -15 to 1 investors still managed to employ leverage in the face of such states; and two industries were completely ignored by all investors. The existence of such states in the

joint return distribution certainly made all investors wary of the equity markets. No doubt this is the reason the loss quarters dropped dramatically.

Summarizing the "disaster states" scenarios, we find both intended and unintended consequences of introducing such states into the joint return distribution. For example, we expected the policies to be more conservative as J went from 1 to 12, but the effects of J-12 and J-6 were certainly extreme. The results of such dramatic shifts in investment policies are clearly demonstrated in the mean returns earned by the investors. While the geometric means certainly decreased, the variabilities did not appear to fall sufficiently to compensate. Certainly the motivation behind this approach cannot be faulted; it is imperative in using a "moving window" approach for the investor to be constantly aware that "disasters" can happen. For example, in the bull market of the 1980s, an investor might have easily forgotton what could happen, as was demonstrated by the October 1987 decline of the equity markets around the world. The smaller the "window" that is used to estimate the Joint return distribution, the more important such an approach is to avoiding draasters in an investment portfolio.

VII. The Results: Statistical Tests of Investment Performance

In view of the differences between the returns earned by some of the active strategies and their corresponding passive and semi-passive benchmark portfolios (see Tables 3a-b and 4a-b), it is desirable to conduct some tests of investment performance for these active strategies. We noted in Chapter VI that these differences were economically significant; in this Chapter, we test to see if they are statistically significant using the paired t-test. This test was described in Chapter IV.

We also examined the robustness of the Grauer-Hakansson 32 quarter simple probability assessment approach by: (i) varying the size of the investment universe, while using the base case 32 quarter simple probability assessment approach; and, (ii) keeping the base case universe of twelve industries, while employing several techniques to estimate the joint return distribution. The comparison of the geometric means of the active strategies under (i) and (ii) with the base case revealed that the simple probability assessment approach is not without merit. At times, this approach produced portfolio returns that were economically significantly better than the other methods used to estimate the joint return distribution. We use the paired t-test to see whether these results are statistically significant as well.

Finally, we test for abnormal investment performance of the base case active strategies using the well known Jensen performance test and the Treynor-Mazuy and Henriksson-Merton tests of market timing. These measures were described in Chapter IV.

A. Paired t-tests

Tables 28 and 29 report the results of selected (one-tailed) paired t-tests between the return series of the active strategies in the base case and the return series of comparable passive and semi-passive benchmark portfolios. Table 28 contains the results when the investment universe is equal-weighted, while Table 29 is for the value-weighted universe.

There are three panels in Table 28. Panel A reports the t-statistic for the full period 1934-86 when leverage is permitted. Panel B is the same as Panel A, except that it is for the sub-period 1966-86. Finally, Panel C reports the comparisons for the sub-period 1966-86 when leverage is not permitted. Note that the table contains only selected comparisons. We compare those active strategies which had higher geometric mean returns than passive and semi-passive benchmark portfolios having comparable standard deviations as is shown in Figures la-d. Thus, we ignore those active strategies that underperformed their comparable benchmarks. For example, we make no comparisons for the active strategies in Figure 1c. The comparisons in Table 29 is chosen in the same spirit. Thus, Table 29 only reports the sub-period 1966-86, when margin purchases are permitted (Panel A), and when they are not (Panel B).

We report for each active strategy, the *t*-statistic on the differences of the return series under the various universes, estimation periods, and return distribution estimation techniques with the comparable base case strategy (e.g., the logarithmic investor in the base case versus the logarithmic investor when one of the base case parameters is varied). The results are given in Tables 30a-b through Tables 37a-b. All these tables are similarly constructed. Each table has two parts, "a" and "b". Part "a" always

contains the equal-weighted investment universe, while "b" contains the value-weighted universe. Each part reports the with leverage and without leverage cases for two periods, 1934-86 and 1966-86.

We briefly describe each table. Tables 30a-b make the comparison between two universes - the twelve industry universe versus the eight - while using a 32 quarter simple probability assessment approach. Tables 31a-b compare the twelve industry universe the twenty-four industry versus universe. Tables 32a-b compare two estimating periods; the base case 32 quarter estimating period with the 28 quarter one, while using a simple probability assessment approach and a twelve industry universe. Similarly, Tables 33a-b report the comparisons of the 32 quarter estimating period with the 40 quarter one. Tables 34a-b through to Tables 37a-b compare the base case with the other estimation techniques. The base case uses the twelve industry universe and a 32 quarter estimating period simple probability assessment approach. Tables 34a-b report the comparisons between the base case and the inflation adapter to the simple probability assessment approach. Tables 35a-b compare the base case with the "sum-of-the-digits" approach to assigning the probabilities of the return distribution. Tables 36a-b contain the comparisons between the base case and the "all-of-history" probability assessment approach. Finally, Tables 37a-b compare the "disaster states" scenario with the base case.⁶⁵

1. The Base Case: Active Strategies versus The Benchmark Portfolios

Turning to Table 28, we find mixed results. A notable feature in all

104

(

 $^{^{65}}$ We report only the "disaster states" approach for J=1. As the results of Chapter VI revealed the J=6 and 12 approach do not merit further consideration.

three Panels A-C is that the portfolio returns of the highly risk averse investors (powers -75 to -30) showed statistically significant differences when compared to the returns on the risk-free asset. The logarithmic investor, who earned 19.05% per year in the 1966-86 sub-period, did manage to statistically beat the passive benchmark V20 which earned only 7.15% in the same period. However, the paired t-test did not show any difference between this investor and the E14 benchmark (13.95%) which had comparable variability. Thus, on the basis of the paired t-tests, it appears that most of the active strategies managing an equally weighted industry investment statistically outperform universe did not their comparable benchmark portfolios, though even there is little doubt about the econom1c significance. The results for investors managing the value-weighted universe are not any better (see Table 29). In fact, none of the active strategies outperformed their benchmark portfolios.

2. The Base Case versus The Eight and Twenty-four Industry Universes

Considering the eight industry universe first (see Tables 30a-b), we find there is hardly any difference (statistically) whether investors manage a twelve or eight industry universe. Only when the industries were value-weighted (Table 30b), did the very risk averse investors managing twelve industries outperform their counterparts managing eight (and only in the full period 1934-86).

When we compare the portfolio returns from managing the twelve versus the twenty-four industry universe (see Tables 31a-b), we find the more risk tolerant investors managing twelve industries outperforming their counterparts in the 1966-86 period, but not always (see Table 31a, Panels A and B, and Panel A of Table 31b). Conversely, the very risk averse investors

did better in the 1934-86 period when managing the twenty-four equal-weighted industries. On the whole, there appears to be no consistent difference when the universe is eight, twelve or twenty-four industries.

3. The Base Case Universe: 32 Versus 28 versus 40 Quarter Estimating Periods Consider the 32 versus 28 quarter estimating period presented in Tables 32a-b. First, no differences showed up in the sub-period 1966-86. Second, the most statistically significant differences are when investors manage the equal-weighted universe in the 1936-86 period (see Table 32a, Panels A and B). The low powers (-75 to -15) performed much better with the 28 quarter estimating period, whereas it did not matter for the other investors.

Turning to Tables 33a-b, we find virtually no difference using the 32 or 40 quarter estimating period when the industries are equally weighted. On the other hand, the low powers did better with a 32 quarter estimating period during the 1936-86 period when managing the value-weighted industries. A general observation is that the length of the estimating periods did not matter for the middle to high powers. The low powers, at times, did much better with different estimating periods, but only for the full period 1936-86.

4. The Base Case versus The Simple Probability Assessment Approach with an Inflation Adapter a

The results of the inflation adapter to the simple probability assessment approach are shown in Tables 34a-b. The paired t-tests showed no difference of one approach over the other. This is in contrast to the Grauer-Hakansson (1988) study utilizing the same approach. This does not augur well for this Grauer-Hakansson method to adjusting the return

106

distribution at the more micro level. Perhaps the method is data specific. 5. The Base Case versus The "Sum-of-the-Digits" Probability Assessment Approach

Tables 35a-b show that the low powers did better with the "sum-of-the-digits" method of assigning_probabilities to the joint return distribution, at least for the full 1934-86 period. This occurred whether equal- or value-weighted industries were the universe and whether leverage was available or not. The period 1966-86 showed that there was no difference which assessment approach was used used; i.e., whether we weighted the realized returns equally or otherwise. This was particularly true for the higher powers.

6. The Base Case versus The "All-of-History" Probability Assessment Approach

Table 36a, Panels A and B, show some very significant differences between the "moving window" approach and the "all-of-history" approach. For the full 1934-86 period, the portfolio returns of the low to middle powers (-75 to -2) show marked improvement when all the historical realized returns were used to estimate the joint return distribution. The differences were less dramatic for the 1966-86 period, however. Turning to Table 36b, the same powers again did better with the "all-of-history" approach for the period 1934-86, but all differences disappeared during 1966-86. Thus, there appears to be some value to this approach.

7. The Base Case versus A "Disaster States" Scenario When J = 1

We see some very statistically significant differences in the returns for the full 1934-86 period in Tables 37a and 37b, Panels A and B. The low to middle powers' portfolio returns were significantly lower than comparative portfolio returns of the base case. Again, the sub-period paired *t*-statistics

showed no perceived differences of portfolio returns under these two approaches. Thus, the motivation behind the "disaster states" approach, while logically valid especially when using a "moving window" approach, leads to very much lower returns as the results of the paired t-tests show. This is despite the fact that the J = 1 setting assigns a relatively low probability of occurrence to the "disaster states".

8. Concluding Remarks

A common thread runs through the results of the paired *t*-tests. In all the comparisons made, we used a base case twelve industry universe with a 32 quarter simple probability assessment approach. We then "tweaked" the estimation method by introducing ways of (hopefully) enhancing the results. We noticed that in many cases for the full period, the lower powers had significantly different (good and bad) returns from the base case, whereas the high powers hardly ever showed any difference from one method to another.

For the base case, there were some statistically significant differences in returns for the active strategies when compared to the passive benchmarks, and none when compared to the semi-passive benchmark portfolios. This adjurs well for the multiperiod portfolio model applied to industry rotation.

When we examined the robustness of the joint return distribution estimation method of Grauer-Hakansson, we come the overall conclusion that the simple probability assessment approach performs well. The addition of an inflation adapter to the simple probability assessment approach did not enhance portfolio returns. This is in contrast to the Grauer-Hakansson (1988) study where they showed that this method, combined with the multiperiod portfolio model, exhibited an ability to "time the market". We continued with our extensions of the Grauer-Hakansson studies by using different amounts of

realized returns to estimate the joint return distribution. In most cases, it did not matter whether we used a longer or shorter (than 32 quarters) term. This is consistent with the previous studies.

When we introduced new and innovative estimation techniques based on "common sense" and logic, we found mixed results. The "disaster states" scenario, while logical and sound, produced results that were statistically inferior to the base case method. The most affected were the low powers. The high powers practically ignored the "disaster states" appended to the estimated return distribution for the case when J = 1, while cases J = 6 and 12 were too extreme. The "all-of-history" method also did not fare well in comparison with the simple probability assessment approach. As expected, this method produced an estimate of next period's joint return distribution that was too conservative resulting in the lower powers having statistically inferior portfolio returns. Finally, there seems to be some merit in the "sum-of-the-digits" approach to assigning probabilities to the joint return estimates. This "common sense" appeal approach drawn from univariate time series modelling benefitted the low to middle powers (statistically). particularly over the full time period 1934-86 considered here. This is interesting because the univariate time series literature is replete with examples indicating that a differenced series of stock prices follows a random walk, i.e., a zero order or "white noise" model. Thus the "sum-of-the-digits" method, which assigns the more recent returns a higher probability of occurrence in the next period, in conjunction with the multiperiod portfolio model, implies (possibly) that there is more information in the recent rather than the distant past. The univariate literature implies that there is no information contained in past prices or

price changes (as does the weak-form efficient markets hypothesis). Therefore, it is difficult to explain why the Grauer-Hakansson simple probability assessment approach and this "sum-of-the-digits" technique appear to do so well.

B. Investment Performance Measures

We now proceed with measures commonly used to compare portfolio performance. Specifically, we present the Jensen Performance Index test in Tables 38 and 39; the Treynor-Mazuy market timing test in Tables 40a-b and 41a-b; and, the Henriksson-Merton test of market timing ability in Tables 42a-b and 43a-b. Note that we present the results of these performance measures only for the base case; i.e., only when investors manage the twelve industry universe, with and without leverage, for the two periods 1934-86 and 1966-86.⁶⁶

1. The Jensen Performance Index

Tables 38 and 39 present a summary of the Jensen measure. This measure is just the intercept, α , from a characteristic line regression. The benchmark or "market" portfolio used in constructing this test is the CRSP value-weighted index.

The format of the tables requires some explanation. Table 38 contains the regression results for the 10 active strategies managing the twelve equal-weighted industries. In Table 38, Panels A and B show the results when leverage opportunities are available, while Panels C and D show the results

⁶⁶This is for economy of space reasons. Besides, the paired t-tests have shown the base case to be as good as or better than the other cases. However, for completeness, we did calculate all the performance measures for the other cases.

when margin purchases are not permitted. Panels A and C are for the period 1934-86, and Panels B and D are for the sub-period 1966-86.

Each of the panels in Table 38 reports the the mean, standard deviation, minimum and maximum values of: (i) the R^2 of the 10 regressions; (ii) the estimated regression intercept, $\alpha_{\rm p}$, which is also the measure of performance with the t-statistic in parentheses; and, (iii) the estimated regression slope, $\beta_{_{
m D}}$, with t-values shown in parentheses. These summary measures are shown in the four leftmost columns. The six rightmost columns contain summary counts of the Jensen measure, α_{p} . Notice that these are given on the lines containing the α_{D} statistics. First, we note if the α_{D} for each investor's portfolio is positive (superior performance) or negative (inferior performance). This count of positive and negative α_n is given on the first line. Second, are these estimates statistically different from zero at the 5%and 1% levels? The number of significant positive and negative estimates are given on the second line. Thus, there should be four numbers on this line. Note that we use the symbol "..." to denote a zero value. Table 39 is constructed to parallel Table 38, except that it contains the results when investors manage the twelve value-weighted industries.

from Tables 38 and 39 are encouraging. The results When the equal-weighted industries were the universe (see Table 38), the active all attained positive values for $\alpha_{\rm p}$ indicating superior strategies performance in both the 1934-86 and 1966-86 time periods. Statistically, the 1966-86 sub-period showed the active strategies to have performed rather well. For example, in Panel B (1966-86 period with leverage present), we find nine out, of the ten active strategies had $\alpha_{\rm p}$ significant at the 5% level, with one at the 1% level. The average α_n in this time period was 1-108% per

quarter, with a minimum of 0.071% and a maximum of 2.731% per quarter. However, for the full time period, 1934-86, only about half the active strategies showed any significant $\hat{\alpha}$ values (see Panels A and C).

The results shown in Table 39 (the value-weighted universe) are not quite as strong as in Table 38, with some active strategies exhibiting negative values for $\hat{\alpha}_p$. We note that none of the strategies showed any significant abnormal performance in the 1966-86 sub-period, while only three out of ten were significant at the 5% level during the 1934-86 period.

Thus the Jensen measure seems to indicate (mostly) superior investment performance on the part of the active strategies when compared with the CRSP value-weighted benchmark.

2. The Treynor-Mazuy Market Timing Test

Tables 40a-b and 41a-b give the results of the Treynor-Mazuy regression described in Chapter IV. The construction of these tables parallel Tables 38 and 39, with two exceptions. The first is the addition of a second term in the regression (the quadratic term) to measure market timing ability which necessitates adding two more lines to the tables. The second is the addition of Tables 40b and 41b. These two extra tables replicate Tables 40a and 41a respectively, except that we make a correction for heteroscedasticity⁶⁷ present in the regression (see also Henriksson and Merton (1981) and Henriksson (1984) for more details). The coefficient of interest is $\hat{\beta}_{2p}$. This measures the the ability of the manager to "time" the market, i.e., move funds into equities in an up-market and out of equities in a down-market.

Turning to Tables 40a and 41a, the results show evidence of market

⁶⁷The correction is described in White (1980).

timing by the active strategies managing the equal-weighted industries. For example, in Table 40a for the 1966-86 period (Panels B and D), all the power policies attained a positive β_{2n} . While not all the coefficients were significant, more than half were statistically significant at the 5% level. However, the results for the full period 1934-86 (Panels A and C) show seven out of ten policies exhibiting negative timing ability, with most significant at the 5% level. Turning Table 40b. when correct to we for heteroscedasticity, all the significant negative timing ability disappeared, but some significant positive timing ability still remained (Panels A and C).

When the industries were value-weighted (Table 41a), we still find strong evidence of market timing ability; mostly positive in the 1966-86 sub-period and mostly negative in the full 1934-86 period. Note that all significant timing ability (positive and negative) disappeared upon correcting for heteroscedasticity (see Table 41b).

3. The Henriksson-Merton Market Timing Test

Finally, Tables 42a-b and 43a-b present the summary results of the Henriksson-Merton tests for market timing ability. The test was described in These tables are constructed exactly like those for Chapter IV. the Treynor-Mazuy test (Tables 40a-b and 41a-b). Table 42a contains the results equal-weighted, when the universe while Table 43a for is is the value-weighted universe. Lastly, Tables 42b and 43b contain the results when the heteroscedastic correction is made. In all cases, the market portfolio was the CRSP value-weighted index.

The results for the active strategies managing the equal-weighted universe are like the Treynor-Mazuy results. First, the 1966-86 sub-period showed the active strategies able to time the market (see Panels B and D),

with over half the strategies displaying significance at the 5% level. Furthermore, the changes in the up- and down-market betas, or the timing ability, captured in the β_{2p} -term are quite large compared to the β_{2p} -values found in Henriksson's (1984) mutual fund study. Notice the results also show some negative market timing, although none were significant. Turning to Table 42b, we find that with the correction for heteroscedasticity the timing during 1966-86 was no longer significant. Again, it is interesting to compare these results with Henriksson's results for U.S. mutual funds. Basically, his results were unaffected by the correction for heteroscedasticity. The results for the 1934-86 full period were mixed. There was little significant timing ability of the active strategies shown by $\hat{\beta}_{2p}$.

Finally, an examination of Table 43a shows that positive timing ability was achieved mostly during the 1966-86 sub-period, while negative timing ability was evident in the 1934-86 period. Correcting for heteroscedasticity, we see virtually no timing ability, positive or negative. This is comparable with the Treynor-Mazuy test results.

C. Concluding Remarks

The results from the investment performance measures are encouraging. All the three measures we calculated show that the active policies managing the twelve industry universe exhibit abnormal investment performance. Specifically, the Jensen Performance Index indicates that the power policies are able to earn abnormal excess returns in many cases, especially when managing the equal-weighted industries. These excess returns reached a maximum of 2.731% per quarter during the 1966-86 period. Even when the industries were value-weighted, abnormal excess returns were earned by most

powers.

Turning to the question of market timing, we performed the Treynor-Mazuy and Henriksson-Merton tests of market timing ability, with and without a correction for heteroscedasticity. The uncorrected measures from both tests indicate that some timing ability existed with respect to the powers policies managing the twelve industries during the 1966-86 sub-period. As noted, this contrasts with studies that show no timing ability for mutual fund managers. However, these tests also indicated that some active strategies achieved negative timing. When we corrected for heteroscedasticity, the statistical significance of the powers' market timing ability, both positive and negative, all but disappeared.

A. Summary of the Methodology

This thesis explored several questions. First, we examined the question of whether multiperiod portfolio theory can be successfully applied beyond the asset allocation stage. We extended the pioneering studies of Grauer and Hakansson to a more micro level, that of industry rotation. We focussed on a base case. This consisted of an investment universe of twelve industries, which followed an industry grouping procedure employed by Breeden, Gibbons and Litzenberger. Furthermore, we followed the Grauer-Hakansson studies and used a 32 quarter simple probability assessment approach to estimate next period's joint return distribution.

Given this base case, we explored a second question: whether the results from the base case are data specific. For this purpose, we employed three sets of investment universes: the base case twelve industry grouping, plus eight and twenty-four industry groupings. Both equal- and value-weighted industries were considered. For these industry universes, we employed the 32 quarter simple probability assessment approach as the estimate of the joint return distribution.

The Grauer-Hakansson studies employed the (naive) simple probability assessment approach to estimating the joint return distribution, and found the approach to be very successful at the asset allocation level. Thus, a third major question examined in the thesis is the efficacy of this simple probability assessment approach. We explored this question by experimenting with different ways of estimating the joint return distribution.

Specifically, we made logical extensions from the base case to examine

the questions posed. First, we conducted industry rotation by expanding the investment universe from the twelve industry base case to twenty-four industries, and contracting it to eight. Second, we extended the estimation of the joint return distribution from the most recent 32 quarters of realized returns (the base case) to using the most recent 28 and 40 quarters of past returns. Third, as a further extension of the simple probability assessment approach, which is a "moving window" method, we employed all the historic returns available as the estimate of the joint return distribution. Fourth, we adopted the inflation adapter to the simple probability assessment approach from the Grauer-Hakansson market timing study and applied it to our base case.

 \mathbb{D}

Fifth. from these simple extensions and replication of the Grauer-Hakansson methods, we proceeded on a more innovative level. We deviated from the assignment of equal probabilities to the joint return distribution employing a "sum-of-the-digits" rule in assigning the probabilities. This rule assigned larger probabilities to the more recent past realizations than the more distant ones. The idea was drawn from univariate time series modelling. The appeal behind this approach was that the recent past should be a better guide to the next period than the more distant past. We applied this rule to our base case.

Finally, we examined a "disaster states" scenario applied to our base case. Disaster states were appended to the estimates of the joint return distribution and assigned varying probabilities of occurrence. The motivation for this method came from the fact that disasters do occur in the equity markets (e.g., the fourth quarter of 1929, or more recently, the fourth quarter of 1987). Thus, the intention was to keep the investor aware of such

states when making his portfolio selection for the next period.

B. Summary of the Results

We examined portfolio rates of return and portfolio compositions over time as a gauge of the success of the active strategies in industry rotation. To further judge the success of these active strategies, we employed several well known measures of investment performance. First, we evaluated the portfolio returns of the active strategies using three standard academic measures of abnormal performance. With the CRSP value-weighted index as the market portfolio, we calculated the Jensen Performance Index, and the Treynor-Mazuy and Henriksson-Merton tests of market timing ability. These investment performance measures were calculated for the base case.

Second, we performed a paired *t*-test that compared the portfolio returns of the active strategies to three sets of benchmarks. These benchmarks were: (i) the passive strategy of holding any one of the value-weighted industry indices; this benchmark was applicable only when the investment universe was a value-weighted one. (ii) A passive strategy that up- and down-levered the CRSP value-weighted index. (iii) A set of semi-passive strategies which upand down-levered an equally weighted portfolio of the equal-weighted industry indices. This benchmark was applicable only when the investment universe consisted of the equal-weighted industries.

Third, we compared the portfolio returns of the active strategies among themselves when the various techniques were used to estimate the joint return distribution. The test for differences in portfolio returns between two active strategies was the paired t-test.

The results of the base case (twelve industries with a 32 quarter simple

probability assessment approach) showed that the active strategies performed well in both the full 1934-86 period and in the sub-period 1966-86, achieving economically and statistically significant excess returns in many cases. This was evidenced by the Jensen performance measure in particular. The market timing tests showed that the active strategies, in most cases, displayed an ability to (positively) time the market, although there was little statistical significance when the tests were corrected for heteroscedasticity. Thus, as a first step, the success of multiperiod portfolio theory applied to major asset categories also extends down to a more micro level. This reinforces the conclusions of the earlier[^] Grauer-Hakansson studies. The success in using a naive method to estimate the joint return distribution, in conjunction with the model drawn from multiperiod portfolio theory, questions the existence of (even) weak-form efficient markets.

The robustness of the simple probability assessment approach and of the model applied to portfolio selection is evidenced by the fact that the portfolio returns of the active strategies were, for the most part, not statistically different: (i) when 28, 32 or 40 quarters of realized returns were used to estimate the joint return distribution; and, (ii) when the investment universe was the eight, twelve or twenty-four industry groupings. The efficacy of the simple probability assessment approach is, furthermore, reinforced with the "all-of-history" approach to estimating the joint return distribution. The portfolio returns of the active strategies employing this latter approach showed statistical inferiority over the portfolio returns obtained by the simple probability assessment approach. This was particularly evident for the more risk averse strategies.

The portfolio returns of the active strategies were statistically the same whether the simple probability assessment approach or the inflation adapter approach was employed to estimate the joint return distribution. This is in contrast to the Grauer and Hakansson results which showed the inflation adapter method to be quite successful in timing the market.

The "sum-of-the-digits" method of probability assessment was the only technique found superior (statistically) to the simple probability assessment approach over the full 1934-86 period, especially for the more risk averse investors.

Finally, in the most cases the "disaster states" scenario was not a success, although it did tend to make the choices of the more risk tolerant investors more conservative. While the motivation behind this approach seemed promising, the results did not bear this out. In particular, the rule in assigning probabilities to the "disaster states" was, perhaps, too crude.

C. Extensions for Further Research

The estimation of the joint return distribution clearly offers the most room for future research. Α first step would be combine the to "sum-of-the-digits" assigning approach to probabilities with the "all-of-history" approach. This would incorporate the "disaster states" idea and motivation with the success "sum-of-the-digits" of the into one estimation technique.

A major step in the estimation area would be to attempt to begin reducing estimation risk. The logical step would be to employ the Stein estimator and its variants, e.g., Bayes-Stein estimation (see, for example, "Jorion (1986)) to the simple probability assessment approach.

A different approach to estimating the joint return distribution is Capital Asset Pricing (CAPM) based. Using the simple probability assessment approach, we could adjust the raw estimates of the distribution using the CAPM. For example, we could estimate the beta of the asset by performing a characteristic line regression on the most recent 32 quarters. An estimate of the market return for the next quarter could be given by the average market return of the last 32 quarters. Because we know the risk-free return for the next quarter, an estimate of next quarter's asset return can be made. Thus, we can shift that asset's realized returns of the past 32 quarters to the new mean return. We could modify the adjustment one step further by adding some multiple of the alpha value from the characteristic line regression to the new mean to reflect the historical under- or over-pricing of the asset.

We have presented three possible extensions to the current study. It is clear that much more can be done to estimate the joint return distribution than have been given here. A completely different extension to all the studies in multiperiod portfolio theory application is to begin considering transactions costs and maintenance margins in the computer programs. It should be interesting to replicate the Grauer-Hakansson studies including these two items to see if the same results are attained and whether it would affect the portfolio choices to any great extent.

| | Percent Share of Value (Row 1) Firms (Row 2) Dates | | | |
|--------------------------------|---|-------|--------|----------------------|
| Industry (SIC Codes) | | | | |
| | 1/26 | 1/34 | 1/66 | 12/86 [·] _ |
| Petroleum | 14.77 | 13.20 | 13.71 | 9.11 |
| (13,29) | 9.04 | 6.10 | 3.95 | 4.52 |
| Finance & Real Estate | 1.32 | 2.77 | 2.68 | 16.13 |
| (60-69) | , 3.21 | 5.53 | 6.30 | 19.81 |
| Consumer Durables | 14.33 | 13.73 | 16.81 | 12.64 |
| (0,25,30,36-37,39,50,55,57,99) | 14.66 | 14.47 | 14.69 | 14.18 |
| Basic Industries | 14.77 | 20.05 | 19.32 | 15.13 |
| (10-12,14,24,26,28,33) | 18.67 | 20.00 | 17.68 | 12.65 |
| Food & Tobacco | 9.64 | 12.83 | 5.72 | 7.10 |
| (1-2,20-21,54) | 12.85 | 11.77 | 8.80 | 4.30 |
| Construction | 0.40 | 1.47 | 1.68 | 1.20 |
| (15-17,32,52) | 1.00 | 2.70 | 4.12 | 3.75 |
| Capital Goods | 4.94 | 5.74 | 12.52 | 11.06 |
| (34-35,38) | 7.83 | 10.07 | 14.77 | 11.26 |
| Transportation | ~ 19.98 | 8.72 | 3.22 | 1.73 |
| (40-42,44-45,47) | 15.66 | 11.06 | 5.08 | 2.71 |
| Utilities | 11.39 | 14.08 | 18.09 | 15.05 |
| (46,48-49) | 4.82 | 3.69 | 10.82 | 11.12 |
| Textiles & Trade | 7.45 | 6.54 | 4.98 | 5.12 |
| (22-23,31,51,53,56,59) | 9.24 | 10.50 | 9.28 | 6.81 |
| Services | 0.17 | 0.16 | 0.23 | 1.95 |
| (72-73,75-76,80,82,89) | 0.60 | 0.71 | 0.97 | 4.24 |
| Leisure | 0.85 | 0.72 | 1.03 | 3.77 |
| (27,58,70,78-79) | 2.41 | 3.40 | 3.55 | 4.59 |
| Total Value (\$ Billion) | [•] 27.33 | 27.98 | 521.31 | 2101.61 |
| Total Number of Firms | 498 | 705 | 1239 | 1439 |

Summary Description of the 12 Industry Indices

Table la

| · · · | Percent Share of Value (Row 1) Firms (Row 2) Dates 1/26 1/34 1/66 12/86 | | | | |
|---|---|----------------|----------------|------------------------------|--|
| Industry (SIC Codes) | | | | | |
| Basic Industries (1000-1299,1400-1499,2600-2699, 2800-2829,2870-2899,3300-3399) | 13.90 16.67 | 18.00 16.88 | 13.91 13.48 | 7.38 8.62 | |
| Capital Goods (3400-3419,3440-3599,3670-3699, 3800-3849,5080-5089,5100-5129, 7300-7399) | 8.29 9.24 | 7.89 10.92 | 9.77 15.25 | 11.11 14.45 | |
| Construction (1500-1999,2400-2499,3220-3299, 3430-3439,5160-5219) | 0.77 1.00 | 1.28 2.84 | 1.70 4.36 | 1.55 4.17 | |
| Consumer Goods (0000-0999,2000-2399,2500-2599, 2700-2799,2830-2869,3000-3219, 3420-3429,3600-3669,3700-3719, 3850-3999,4830-4899,5000-5079, 5090-5099,5130-5159,5220-5999, 7000-7299,7400-9999) | 27.24 37.95 ₽ | 32.30 40.43 | 35.84 38.98 | 36.84 33.43 | |
| Energy (1300-1399,2900-2999) | 14.77 9.04 | 13.20 6.10 | 13.71 3.95 | 9.11 4.52 | |
| Finance (6000-6999) | 1.32 3.21 | 2.77 5.53 | 2.68 6.30 | 16.13 19 _{,9} 81 | |
| Transportation (3720-3799,4000-4799) | 22.33 18.27 | 10.49 13.76 | 4.73 7.67 | 3.37 4.38 | |
| Utilities (4800-4829,4900-4999) | 11.38 4.62 | 14.07 3.55 | 17.65 10.01 | 14.50 10.63 | |
| Total Value (\$ Billion) Total Number of Firms | 27.33 498 | 27.98 705 | 521.31 1239 | 2101.61 1439 | |

Table 1b

52

Summary Description of the 8 Industry Indices

| Tai | ble | 1c |
|-----|-----|----|
| | | |

;

| | | Percent Value Firms | Share of (Row 1) (Row 2) | |
|--------------------------------|-------|---------------------------|--------------------------------|--------|
| | Dates | | | |
| Industry (SIC Codes) | 1/26 | 1/34 | 1/66 | 12/86 |
| Banks & Financial Services | 0.63 | 2.37 | 2.16 | 13.71 |
| (60-62,67) | 1.81 | 4.40 | 5.41 | 17.10 |
| Chemicals, Rubber & Plastics | 3.87 | 10.33 | 11.90 | 11.39 |
| (28,30) | 6.02 | 6.67 | 8.08 | 6.62 |
| Construction | 0.06— | 0.09 | 0.06 | 0.22 |
| (15-17) | 0.20 | 0.71 | 0.65 | |
| Fabricated Metal Products (34) | 1.34 | 2.27 | 1.11 | 1.31 |
| | 2.21 | 2.98 | 3.23 | 2.74 - |
| Food | 6.06 | 8.02 | 4.19 | 5.03 |
| (1-2,20) | 8.63 | 7.80 | 6.30 | 2.96 |
| Insurance | 0.34 | 0.26 | 0.43 | 2.37 |
| (63-64) | 0.60 | 0.57 | 0.48 | 1.69 |
| Machinery | 8.40 | 7.14 | 18.04 | 15.43 |
| (35-36,38-39) | 8.84 | 10.78 | 18.50 | 15.06 |
| _ Metals | 8.89 | 8.93 | 5.67 | 1.90 |
| (32-33) | 10.04 | 10.21 | 8.32 | 4.86 |
| Mining | 2.64 | 2.43 | 1.59 | 0.63 |
| (10-12,14) | | 4.11 | 2.67 | 2.04 |
| Oil & Gas Extraction | 0.68 | 0.20 | °0.95 | 3.01 |
| (13) | 1.41 | 0.71 | 1.45 | 3.31 |
| Paper & Wood | 0.51 | 0.58 | 3.41 | 5.02 |
| (24-27) | 1.41 | 3.55 | 5.01 | 5.00 |
| Petroleum & Coal Products | 14.09 | 13.01 | 12.76 | 6.16 |
| (29) | 7.63 | 5.39 | 2.50 | 1.27 |

Summary Description of the 24 Industry Indices

Table 1c (continued)

| | Percent Share of Value (Row 1) Firms (Row 2) Dates | | | |
|--------------------------|---|-------|--------|---------|
| | | | | |
| Industry (SIC Codes) | 1/26 | 1/34 | 1/66 | 12/86 |
| Railroads & Trucking | 19.28 | 8.41 | 1.99 | 0.67 |
| (40,42) | 12.85 | 8.37 | 3.23 | 1.06 |
| Real Estate | 0.35 | 0.14 | 0.09 | 0.14 |
| (65-66) | 0.80 | 0.57 | 0.40 | 1.27 |
| Retail Trade, Food | 0.40 | 0.77 | 0.86 | 1.43 |
| (54,58) | 1.81 | 2.27 | 2.10 | 1.83 |
| Retail Trade, Others | 6.22 | 5.44 | 3.88 | 4 34 |
| (52-53,55-57,59) | 4.62 | 5.96 | 4.68 | 4.15 |
| Services Leisure | 0.62 | 0 37 | 0 27 | 1 07 |
| (70, 78-79) | 1.20 | 1.28 | 1.37 | 1.83 |
| Services. Others | 0.17 | 0.16 | 0.23 | 1.96 |
| (72-73,75-76,80,82,89) | 0.60 | 0.71 | 0.97 | 4.29 |
| Textiles | 1.29 | 1.10 | 1.12 | 0.83 |
| (22-23,31) | 5.22 | 4.96 | 4.85 | 2.81 |
| Tobacco | 3.33 | 4.13 | 0,78 | 1.40 |
| (21) | 3.21 | 2.70 | 0.97 | 0.35 |
| Transportation | 0.70 | 0.32 | 1.23 | 1.07 |
| (41,44-47) | 2.81 | 2.70 | 1.86 | 1.69 |
| Transportation Equipment | 8.74 | 9.40 | 9.03 | 4.69 |
| (37) | 8.63 | 7.94 | 5.17 | 3.10 |
| Utilities | 11.39 | 14.08 | 18.10 | 15.13 |
| (48-49) | 4.82 | 3.69 | 10.82 | 11.20 |
| Wholesale Trade | 0.02 | 0.05 | 0.17 | 1.08 |
| (50-51) | 0.40 | 0.99 | 0.97 | 2.39 |
| Total Value (\$ Billion) | 27.33 | 27.98 | 521.16 | 2090.26 |
| Total Number of Firms | 498 | 705 | 1238 | 1421 |

Summary Description of the 24 Industry Indices
| | 12 Industry | Univer | se | | |
|-------------|------------------------------|---------|--------------------------|--|--|
| PETR | Petroleum | CAPG | Capital Goods | | |
| FINA | Finance & Real Estate | TRAN | Transportation | | |
| CDUR | Consumer Durables | UTIL | Utilities | | |
| BASI | Basic Industries | TEXT | Textiles & Trade | | |
| FTOB | Food & Tobacco | SERV | Services | | |
| CONS | Construction | LEIS | Leisure | | |
| | 8 Industry | Univers | e | | |
| BIND | Basic Industries | ENER | Energy | | |
| CAGD | Capital Goods | FINA | Finance | | |
| CSTR | Construction | TRNS | Transportation | | |
| CGDS | Consumer Goods | UTLY | Utilities | | |
| | 24 Industry | Univers | 5e | | |
| BANK | Banks & Financial Services | RAIL | Railroads & Trucking | | |
| CHEM | Chemicals, Rubber & Plastics | REAL | Real Estate | | |
| CNST | Construction | RTFD | Retail Trade, Food | | |
| FMPS | Fabricated Metal Products | RTOT | Retail Trade, Others | | |
| FOOD | Food | SLEI | Services, Leisure | | |
| INSU | Insurance | SOTH | d Services, Others | | |
| 1ACH | Machinery | TXTL | . Textiles | | |
| 1ETA | Metals . | TOBA | Tobacco | | |
| 11NE | Mining | TRSP | Transportation | | |
| | Dil & Gas Exploration | IEQP | Iransportation Equipment | | |
| PCPS | Petroleum & Coal Products | WHOL | OL Wholesale Trade | | |
| | Other Syr | nbols | | | |
| RL | Risk-free lending (quarterly | IN | U.S. inflation | | |
| | U.S. Treasury bills) | В | Borrowing | | |
| | Benchmark Po | rtfolio | DS | | |
| <i>v</i> 10 | CRSP value-weighted index | V12 | 120% in V10, 20% in B | | |
| 12 | 20% in V10, 80% in RL | V14 | 140% in V10, 40% in B | | |
| 14 | 40% in V10, 60% in RL | V16 | 160% in V10, 60% in B | | |
| 16 | 60% in V10, 40% in RL | V18 | 180% in V10, 80% in B | | |
| /8 | 80% in V10, 20% in RL | V20 | 200% in V10, 100% in B | | |
| E10 | Equal-weighted portfolio of | | | | |
| | risky assets | E12 | 120% in E10, 20% in B | | |
| E2 | 20% in ElO, 80% in RL | E14 | 140% in E10, 40% in B | | |
| 24 | 40% in E10, 60% in RL | E16 | 160% in ElO, 60% in B | | |
| | | | | | |
| 56 | 60% in E10, 40% in RL | E18 | 180% in E10, 80% in B | | |

Table 2

Geometric Means and Standard Deviations of Annual Returns using 12 Equal-Weighted Industry Indices, with Leverage (32 quarter estimating period, simple probability assessment)

| Portfolio | 1934 G. Mean | 1934 - 1986 6 Magn S Day | | 1986 |
|--------------------------|-----------------|-----------------------------|--------|-------|
| | | 3, Dev | G.Mean | S.Uev |
| Petroleum | 14.98 | 24.21 | 11.82 | 27.24 |
| Finance & Real Estate | 14.76 | 25.22 | 12,45 | 26.25 |
| Consumer Durables | 15.12 | 28.04 | 13,42 | 28.12 |
| Basic Industries | 14.60 | 21.38 | 12.69 | 17.41 |
| Food & Tobacco | 14.58 | 19.78 | 16.02 | 20,99 |
| <pre>*Construction</pre> | 13.70 | 26.80 | 12.97 | 24.84 |
| Capital Goods | 14.58 | 25.54 | 11.88 | 25.02 |
| Transportation | 14.22 | 29.43 | 10.97 | 24.40 |
| Utilities | 13.22 | 23.05 | 12.13 | 17.33 |
| Textiles & Trade | 14.81 | 28.26 | 14.18 | 31.28 |
| Services | 16.40 | 32.39 | 14.22 | 32.55 |
| Leisure | 15.64 | 30.40 | 15.50 | 30.15 |
| Risk-Free | 3.76 | 3.47 | 7,49 | 2.71 |
| Inflation | 4.09 | 3.80 | 6.12 | 3.15 |
| Power -75 | 4.70 | 4.18 | 7.93 | 2,88 |
| Power -50 | 5.14 | 5.06 | 8.14 | 3.12 |
| Power -30 | 6,01 | 6.98 | 8.55 | 3.76 |
| Power -15 | 7.38 | 9.72 | 9.51 | 5.82 |
| Power -10 | 8.65 | 11.50 | 10.37 | 7 98 |
| Power -5 | 10.80 | 15.08 | 12 32 | 13 78 |
| Power -2 | 13.86 | 21 32 | 15 27 | 22 27 |
| Power 0 | 18 28 | 33 36 | 19.05 | 33 96 |
| Power 0.5 | 18.01 | 44.78 | 15 44 | 39 57 |
| Power 1 | 16.91 | 54.86 | 14.97 | 46.38 |
| Portfolio E2 | 6.47 | 5.40 | 9.20 | 5.11 |
| Portfolio E4 | 8.99 | 9.72 | 10.69 | 9.38 |
| Portfolio E6 | 11.31 | 14.39 | 11.95 | 13.89 |
| Portfolio E8 | 13.42 | 19.14 | 12.97 | 18.49 |
| Portfolio E10 | 15.31 | 23.96 | 13.75 | 23.15 |
| Portfolio El2 | 16.56 | 28.80 | 13.67 | 27.95 |
| Portfolio E14 | 17.77 | 33.41 | 13.95 | 32.15 |
| Portfolio El6 | 18.61 | 37.35 | 14.10 | 35.44 |
| Portfolio E18 | 19.29 | 41.08 | 14.26 | 37.98 |
| Portfolio E20 | 20.18 | 43.74 | 14.27 | 40.71 |
| Portfolio V2 | 5.47 | 4.24 | 8.12 | 3.99 |
| Portfolio V4 | 7.07 | 6.99 | 8.62 | 6.87 |
| Portfolio V6 | 8.56 | 10.24 | 9.01 | 10.08 |
| Portfolio V8 | 9.94 | 13.66 | 9.26 | 13.42 |
| Portfolio V10 | 11 21 | 17.16 | 9.38 | 16.84 |
| Portfolio V12 | 11 93 | 20.69 | 8 77 | 20 39 |
| Portfolio V14 | 12 65 | 24 15 | 8 43 ° | 23 79 |
| Portfolio VIA | 12.05 | 27,13 | £ 12 · | 26 87 |
| Portfolio V19 | 13.10 | 30 00 | 7 74 | 20.07 |
| Portfolio V20 | 14 04 | 20.00 | 7 15 | 32 00 |
| FOLCEOLIO VZU | 14.04 | 52,54 | 1.15 | 32.90 |

*Standard Deviation is for the variable ln(l+r).

- 1

Table 3b

| | 1934 - 1986 | | 1966 | - 1986 , | | |
|-----------------------|-------------|----------------|--------|----------|--|--|
| Portfolio | G.Mean | S.Dev | G.Mean | S.Dev | | |
| Petroleum | 14.98 | 24.21 | 11.82 | 27.24 | | |
| Finance & Real Estate | 14.76 | 25.22 | 12.45 | 26.25 | | |
| Consumer Durables | 15.12 | 28.04 | 13.42 | 28.12 | | |
| Basic Industries | 14.60 | 21.38 | 12.69 | 17.41 | | |
| Food & Tobacco | 14.58 | 19.78 | 16.02 | 20.99 | | |
| Construction | 13.70 | 26.80 | 12.97 | 24.84 | | |
| Capital Goods | 14.58 | 25.54 | 11.88 | 25.02 | | |
| Transportation | 14.22 | 29.43 | 10.97 | 24.40 | | |
| Utilities | 13.22 | 23.05 | 12.13 | 17.33 | | |
| Textiles & Trade | 14.81 | 28.26 | 14.18 | 31.28 | | |
| Services | 16.40 | 32.39 | 14.22 | 32.55 | | |
| Leisure | 15.64 | 30.40 | 15.50 | 30.15 | | |
| Risk-Free | 3.76 | 3.47 | .7.49 | 2.71 | | |
| Inflation | 4.09 | 3.80 | 6.12 | 3.15 | | |
| Power -75 | 4.70 | 4.18 | 7.93 | 2.88 | | |
| Power -50 | 5.14 | 5.06 | 8.14 | 3.12 | | |
| Power -30 | 6.01 | 6.98 | 8.55 | 3.76 | | |
| Power -15 | 7.31 | 8.12 | 9.51 | 5.82 | | |
| Power -10 | 8.26 | 9.31 | 10.37 | 7.98 | | |
| Power -5 | 10.15 | 13.03 | 12.34 | 13.79 | | |
| Power -2 | 12.23 | 17.24 | 14.46 | 19.05 | | |
| Power O | 13.84 | 24.89 | 15.14 | 26.05 | | |
| Power 0.5 | 13.88 | 26.77 | 14.88 | 26.19 | | |
| Power 1 | 13.18 | 28.67 | 14.42 | 26.17 | | |
| Portfolio E2 | 6.47 | 5.40 | 9.20 | 5.11 | | |
| Portfolio E4 | 8.99 | 9.72 | 10.69 | 9.38 | | |
| Portfolio E6 | 11.31 | 14.39 | 11.95 | 13.89 | | |
| Portfolio E8 | 13.42 | 19.14 | 12.97 | 18.49 | | |
| Portfolio E10 | 15.31 | 23.96 | 13.75 | 23.15 | | |
| Portfolio V2 | 5.47 | 4.24 | 8.12 | 3.99 | | |
| Portfolio V4 | 7.07 | 6,99 | 8.62 | 6.87 | | |
| Portfolio V6 | 8.56 | 10.24 | 9.01 | 10.08 | | |
| Portfolio V8 | 9.94 | 13.66 | 9.26 | 13.42 | | |
| Portfolio V10 | 11.21 | ´ 17.16 | 9.38 | 16.84 | | |

Geometric Means and Standard Deviations of Annual Returns using 12 Equal-Weighted Industry Indices, without Leverage (32 quarter estimating period, simple probability assessment)

*Standard Deviation is for the variable $ln(1+r_t)$.

| | 1934 | - 1986 | 1966 · | - 1986 | |
|-----------------------|--------|--------------------|--------|--------|--|
| Portfoliø | G.Mean | S.Dev [*] | G.Mean | S.Dev | |
| Petroleum | 12.20 | 18.96 | 10.93 | 21.21 | |
| Finance & Real Estate | 12.08 | 20.29 | 10.68 | 18.75 | |
| Consumer Durables | 11.82 | 23.87 | 8.17 | 23,83 | |
| Basic Industries | 10.95 | 17.67 | 8.58 | 15.15 | |
| Food & Tobacco | 11.44 | 16,17 | 12,96 | 17.39 | |
| Construction | 9.61 | 21.91 | 7.99 | 22.18 | |
| Capital Goods | 11.73 | 20.74 | 7.92 | 20.54 | |
| Transportation | 9.31 | 22.73 | 7.03 | 20,85 | |
| Utilities | 10.33 | 15.50 | 9:68 | 13.69 | |
| Textiles & Trade | 11.28 | 22.98 | 9.60 | 26.96 | |
| Services | 12.75 | 31.39 | 10.67 | 28.10 | |
| Leisure | 12.64 | 28.62 | 12.32 | 30,18 | |
| Risk-Free | 3.76 | 3.47 | 7,49 | 2.71 | |
| Inflation . | 4.09 | 3.80 | 6.12 | 3.15 | |
| Power -75 | 4.87 | 4.65 | 7.81 | 2.90 | |
| Power -50 | 5.40 | 5.92 | 7.96 | 3.12 | |
| Power -30 | 6.12 | 7.87 | 8.25 | 3.73 | |
| Power -15 | 7.26 | 10.96 | 8.93 | 5.62 | |
| Power -10 | 7.82 | 12.91 | 9.52 | 7.59 | |
| Power -5 | 9.29 | 16.29 | 10.39 | 12.16 | |
| Power -2 | 11.63 | 21.74 | 11.31 | 19.52 | |
| Power 0 - | 13.61 | 32.77 | 9.34 | 30,29 | |
| Power 0.5 | 13.95 | 43.48 | 8.87 | 35.97 | |
| Power 1 | 11.75 | 56.15 | 0.02 | 49.37 | |
| Portfolio V2 | 5,47 | 4,24 | 8.12 | 3.99 | |
| Portfolio V4 | 7.07 | 6.99 | 8.62 | 6.87 | |
| Portfolio V6 | 8.56 | 10.24 | 9.01 | 10.08 | |
| Portfolio V8 | 9.94 | 13.66 | 9.26 | 13.42 | |
| Portfolio V10 | 11.21 | 17.16 | 9.38 | 16.84 | |
| Portfolio V12 | 11.93 | 20.69 | 8.77 | 20.39 | |
| Portfolio V14 | 12.65 | 24.15 | 8.43 | 23.79 | |
| Portfolio V16 | 13.18 | 27.11 | 8.12 | 26.87 | |
| Portfolio V18 | 13.57 | 30.00 | 7.74 | 29.78 | |
| Portfolio V20 | 14.04 | 32.34 | 7.15 | 32.90 | |

Geometric Means and Standard Deviations of Annual Returns using 12 Value-Weighted Industry Indices, with Leverage (32 quarter estimating period, simple probability assessment)

* Standard Deviation is for the variable $ln(1+r_t)$.

Table 4b

| | 1934 | - 1986 _ | 1966 - | 1986 |
|-----------------------|--------|--------------------|----------|--------------------|
| Portfolio | G.Mean | S.Dev [*] | G.Mean | S.Dev [°] |
| Petroleum | 12.20 | 18.96 | 10.93 | 21.21 |
| Finance & Real Estate | 12.08 | 20.29 | 10.68 | 18.75 |
| Consumer Durables | 11.82 | 23.87 | 8.17 | 23.83 |
| Basic Industries | 10.95 | 17.67 | 8.58 | 15.15 |
| Food & Tobacco | 11.44 | 16,17 | 12.96 | 17.39 |
| Construction | 9.61 | 21.91 | 7.99 | 22.18 |
| Capital Goods | 11.73 | 20.74 | 7.92 | 20.54 |
| Transportation | 9.31 | 22.73 | 7.03 | 20.85 |
| Utilities | 10.33 | 15.50 | 9.68 | 13.69 |
| Textiles & Trade | 11.28 | 22.98 | 9.60 | 26.96 |
| Services | 12.75 | 31.39 | 10.67 | 28.10 |
| Leisure | 12.64 | 28.62 | 12.32 | 30.18 |
| Risk-Free | 3.76 | 3.47 | 7.49 | 2.71 |
| Inflation | 4.09 | 3.80 | 6.12 | 3.15 |
| Power -75 | 4.87 | 4.65 | 7.81 | 2.90 |
| Power -50 | 5.40 | 5.92 | 7.96 | 3.12 |
| Power -30 | 6.12 | 7.87 | 8.25 | 3.73 |
| Power -15 | 6.81 | 9.07 | 8.93 | 5.62 |
| Power -10 | 7.45 | 10.33 | 9.52 🥄 | 7.59 |
| Power -5 | 8.87 | 13.41 | <u> </u> | 12.01 |
| Power -2 | 9.82 | 16.93 | 10.61 | 16.92 |
| Power 0 | 10.59 | 25.15 | 8.23 | 23.87 |
| Power 0.5 | 10.29 | 28.37 | 6.64 | 28.39 |
| Power 1 | 10.22 | 31.24 | 3.85 | 33.66 |
| Portfolio V2 | 5.47 | 4.24 | 8.12 | 3.99 |
| Portfolio V4 | 7.07 | 6.99 | 8.62 | 6.87 |
| Portfolio V6 | 8.56 | 10.24 | 9.01 | 10.08 |
| Portfolio V8 | 9.94 | 13.66 | 9.26 | 13.42 |
| Portfolio V10 | 11.21 | 17.16 | 9.38 | 16.84 |

Geometric Means and Standard Deviations of Annual Returns using 12 Value-Weighted Industry Indices, without Leverage (32 quarter estimating period, simple probability assessment)

*Standard Deviation is for the variable $ln(1+r_t)$.

| | 1934 | - 1986 | 1966 | - 1986 |
|------------------|---------------------------|---------|---------|--------------------|
| Portfolio | G.Mean S.Dev [*] | | G.Mean | S.Dev [*] |
| Basic Industries | 14.33 | 22.29 | 12.14 , | 18.59 |
| Capital Goods | 14.49 | 25.63 | 11.87 | 25.74 |
| Construction | 13.44 | 28.93 | 12.17 | 25.03 |
| Consumer Goods | 15.22 | 24.83 | 14.79 | 26.01 |
| Energy | 14.98 | 24.21 | 11, 82 | 27.24 |
| Finance | 14.76 | 25.22- | 12.45 | 26.25 |
| Transportation | 14.77 | 28.33 | 11.92 | 26.08 |
| Utilities | 12.89 | 23.16 | 11.59 | 16.69 |
| Risk-Free | 3.76 | 3.47 | 7.49 | 2.71 |
| Inflation | 4.09 | 3.80 | 6.12 | 3.15 |
| Power -75 | 4.56 | 3.75 | 7.89 | 2.78 |
| Power -50 | 4.94 | 4.22 | 8.09 | 2.96 |
| Power -30 | 5.67 | 5.51 | 8.47 | 3.52 |
| Power -15 | 7.20 | 8.18 | 9.36 | 5,42 |
| Power -10 | 8.39 | 10.32 | 10.17 | 747 |
| Power -5 | 10.70 | 14.20 | 12.06 | 13,11 |
| Power -2 | 13.60 | 19.67 | 14.48 | 20.82 |
| Power 0. | 16.93 | 29.54 | 18.33 | 32,51 |
| Power 0.5 | 15.86 | 40.06 | 13.17 | 39.24 |
| Power 1 | 14.07 | 56.51 , | 13.36 | 47.46 |
| Portfolio E2 | 6.39 | 5.20 | 8.98 | 4.75 |
| Portfolio E4 | 8,81 | 9.36 | 10.28 | 8.67 |
| Portfolio E6 | 11.04 | 13.88 | 11.37 | 12.85 |
| Portfolic E8 | 13.07 | 18.49 | 12.25 | 17.10 |
| Portfolio E10 | 14.87 | 23.14 | 12.92 | 21.41 |
| Portfolio El2 | 16.06 | 27.80 | 12.76 | 25.82 |
| Portfolio El4 | 17.19 | 32.24 | 12.95 | 29.62 |
| Portfolio El6 | 17.94 | 36.08 | 13.02 | 32.69 |
| Portfolio El8 | 18.49 | 39.81 | 13.03 | 35.33 |
| Portfolio E20 | 19.22 | 42.51 | 12.87 | 3 38.16 |
| Portfolio V2 | 5.47 | 4.24 | 8.12 | 3.99 |
| Portfolio V4 | 7.07 | 6.99 | 8.62 | 6.87 |
| Portfolio V6 | 8,56 | 10.24 | 9.01 | 10.08 |
| Portfolio V8 | 9.94 | 13.66 | 9.26 | 13.42 |
| Portfolio V10 | 11.21 | 17.16 | 9.38 | 16.84 |
| Portfolio V12 | 11.93 | 20.69 | 8.77 | 20.39 |
| Portfolio Vl4 | 12.65 | 24.15 | 8.43 | 23.7 9 |
| Portfolio V16 | 13.18 | 27.11 | 8.12 | 26,87 |
| Portfolio V18 | 13.57 | 30.00 | 7.74 | 29.78 |
| Portfolio V20 | 14.04 | 32.34 | 7.15 | 32.90 |

Geometric Means and Standard Deviations of Annual Returns using 8 Equal-Weighted Industry Indices, with Leverage (32 quarter estimating period, simple probability assessment)

* Standard Deviation is for the variable $\ln(1+r_t)$.

| a , | 1934 | - 1986 | • 1966 | - 1986 |
|------------------|--------|--------------------|--------|--------------------|
| Portfolio | G.Mean | S.Dev [*] | G.Mean | S.Dev [°] |
| Basic Industries | 14.33 | 22.29 · | 12.14 | 18.59 |
| Capital Goods | 14.49 | 25.63 | 11.87 | 25.74 |
| Construction | 13.44 | 28.93 | 12.17 | 25.03 |
| Consumer Goods | 15.22 | 24.83 | 14.79 | 26.01 |
| Energy | 14.98 | 24.21 | 11.82 | 27.24 |
| Finance | 14.76 | 25.22 | 12.45 | 26.25 |
| Transportation | 14.77 | 28.33 | 11.92 | 26.08 |
| Utilities | 12.89 | 23.16 | 11.59 | 16.69 |
| Risk-Free | 3.76 | 3.47 | 7.49 | 2.71 |
| Inflation | 4.09 | 3.80 | 6.12 | 3.15 |
| Power -75 | 4.56 | 3.75 | 7.89 | 2.78 |
| Power -50 | 4.94 | 4.22 | 8.09 | 2.96 |
| Power -30 | 5.67 | 5.51 | 8.47 | 3.52 |
| Power -15 | 7.27 | 7.75 | 9.36 | 5.42 |
| Power -10 | 8.19 | 8.80 | 10.17 | 7.47 |
| Power -5 | 10.08 | 12.39 | 12.06 | 13.10 |
| Power -2 | 11.78 | 15.84 | 13.98 | 18.59 |
| Power 0 | 13.08 | 22.50 | 14.31 | 24.81 |
| Power 0.5 | 13.64 | 25.54 | 13.75 | 25.12 |
| Power 1 | 13.05 | 28.67 | 13.35 | 26.38 [°] |
| Portfolio E2 | 6.39 | 5.20 | 8.98 | 4.75 |
| Portfolio E4 | 8.81 | 9.36 | 10.28 | 8.67 |
| Portfolio E6 | 11.04 | 13.88 | 11.37 | 12.85 |
| Portfolio E8 | 13.07 | 18.49 | 12.25 | 17.10 |
| Portfolio E10 | 14.87 | 23.14 | 12.92 | 21.41 |
| Portfolio V2 | 5.47 | 4.24 | 8.12 | 3.99 |
| Portfolio V4 | 7.07 | 6.99 | 8.62 | 6.87 |
| Portfolio V6 | 8.56 | 10.24 | 9.01 | 10.08 |
| Portfolio V8 | 9.94 | 13.66 | 9.26 | 13.42 |
| Portfolio V10 | 11.21 | 17.16 | 9.38 | 16.84 |

Geometric Means and Standard Deviations of Annual Returns using 8 Equal-Weighted Industry Indices, without Leverage (32 quarter estimating period, simple probability assessment)

* 6

Standard Deviation is for the variable $ln(1+r_t)$.

1

Ą,

Geometric Means and Standard Deviations of Annual Returns using 8 Value-Weighted Industry Indices, with Leverage (32 quarter estimating period, simple probability assessment)

| | 1934 | - 1986 | 1966 | - 1986 |
|------------------|--------------|--------------------|--------|--------|
| Portfolio | G.Mean | S.Dev [*] | G.Mean | S.Dev |
| Basic Industries | 10.37 | 19.04 | 7.54 | 17.58 |
| Capital Goods | 11.66 | 19.63 | 8.56 | 20.06 |
| Construction | 9.30 | 24.52 | 7,44 | 21.51 |
| Consumer Goods | 11.66 | 19.47 | 9.71 | 20.12 |
| Energy | - 12.30 | 18.96 | 10.93 | 21.21 |
| Finance | 12.08 | 20.29 | 10.68 | 18.75 |
| Transportation | 9.97 | 23.10 | 8.07 | 23.07 |
| Utilities | 10.23 / | 15.40 | 9.52 | 13.47 |
| Risk-Free | 3.76 | 3.47 | 7.49 | 2,71 |
| Inflation | 4.09 | 3.80 | 6.12 | 3.15 |
| Power -75 | ۰ ۲۰۶۸ | 3 • 0 1 | 7 64 | 2 78 |
| Power -50 | 4,54 6 01 | J. 58 | 7.04 | 2.70 |
| Power -30 | 5 62 | 6 32 | 7.86 | 3 21 |
| Power -15 | 6 75 | 8 84 | 8 19 | 4 36 |
| Power -10 | 7 62 | 11 09 | 8 47 | 5 69 |
| Power -5 | 9.27 | 14 12 | 8 99 | 9.07 |
| Power -2 | 10.94 | 18 42 | 8 76 | 14 56 |
| Power () | 10,54 | 25 26 | 6 82 | 24 92 |
| Power 0 5 | 10 91 | 31 25 | 7 59 | 29 21 |
| Power 1 | 8.77 | 47.42 | 5.30 | 32.22 |
| Portfolio V2 | 5.47 | 4,24 | 8.12 | 3,99 |
| Portfolio V4 | 7.07 | 6.99 | 8,62 | 6.87 |
| Portfolio V6 | 8.56 | 10.24 | 9.01 | 10.08 |
| Portfolio V8 | 9.94 | 13.66 | 9,26 | 13.42 |
| Portfolio V10 | 11.21 | 17.16 | · 9,38 | 16.84 |
| Portfolio V12 | 11,93 | 20.69 | 8.77 | 20.39 |
| Portfolio Vl4 | 12.65 | 24.15 | . 8,43 | 23.79 |
| Portfolio V16 | 13.18 | 27.11 | 8.12 | 26.87 |
| Portfolio V18 | 13.57 | 30.00 | 7.74 | 29.78 |
| Portfolio V20 | 14.04 | 32.34 | 7.15 | 32.90 |

*Standard Deviation is for the variable ln(1+r).

| | 1934 | 1934 - 1986 | | - 1986 |
|------------------|-------------|--------------------|--------|--------|
| Portfolio | G.Mean | S.Dev [*] | G.Mean | S.Dev |
| Basic Industries | 10.37 | 19.04 | 7.54 | 17.58 |
| Capital Goods | 11.66 | 19.63 | 8,56 | 20.06 |
| Construction | 9.30 | 24.52 | 7.44 | 21.51 |
| Consumer Goods | 11.66 | 19.47 | ° 9.71 | 20.12 |
| Energy | 12.30 | 18.96 | 10.93 | 21.21 |
| Finance | 12.08 | 20.29 | 10.68 | 18.75 |
| Transportation | 9.97 | 23.10 | 8.07 | 23.07 |
| Utilities | 10.23 | 15.40 | 9.52 | 13.47 |
| Risk-Free | 3,76 | 3.47 | 7.49 | 2.71 |
| Inflation 💦 | 4.09 | 3.80 | 6.12 | 3.15 |
| Power -75 | 4.54 | 3.91 | 7.64 | 2.78 |
| Power -50 | 4,91 | 4.58 | *7.72 | 2.89 |
| Power - 30 | 5.62 | 6.32 | 7.86 | 3.21 |
| Power -15 | 6.72 | 8.10 | 8.19 | 4.36 |
| Power -10 | 7.35 | 9.07 | .8.47 | 5.69 |
| Power -5 | 8.50 | 11.40 | 8.97 | 9.32 |
| Power -2 | 9.25 | 13.86 | 8.74 | 13.59 |
| Power 0 | 8.60 | 18.40 | 6.22 | 19.92 |
| Power 0.5 | 8.37 | 21.84 | 6.46 | 22.18 |
| Power 1 | 8.92 | 24.82 | 6.46 | 23.15 |
| Portfolio V2 | 5,47 | 4.24 | 8.12 | 3,99 |
| Portfolio V4 | 7.07 | 6.99 | 8.62 | 6.87 |
| Portfolió V6 | 8.56 | 10.24 | 9.01 | 10.08 |
| Portfolio V8 | 9.94 | 13.66 | 9.26 | 13.42 |
| Portfolio V10 | 11.21 | 17.16 | 9 38 | 16 84 |

Geometric Means and Standard Deviations of Annual Returns using 8 Value-Weighted Industry Indices, without Leverage (32 quarter estimating period, simple probability assessment)

* Standard Deviation is for the variable $ln(1+r_t)$.

Table 6a

| Geometri | .c Means | and Stand | ard Devi | ations o | f Annual | Returns | using | 24 |
|----------|----------|-----------|----------|----------|-----------|----------|--------|----|
| | Equal | -Weighted | Industry | Indices | , with Le | verage | • - | |
| (32 av | uarter e | stimating | period. | simple p | probabili | tv asses | sment) | |

| | 1934 | - 1986 | 1966 - | 1086 |
|----------------------------|---------|-------------------|---------------|--|
| Portfolio | G.Mean | S.Dev* | G.Mean | S. Dev |
| | | | | and a state of the |
| Banks & Financial Survices | 14.49 | 25.48 | <u>11.96</u> | 25.85 |
| Construction | S 14.01 | 20.11 | 13.90 | 17.75 |
| Entricated Matel Brochaste | 14.00 | 41.8/ ~~ 05.22 | 15,90 | 30.50 |
| Food | 14.00 | 23.33 | 14,08 | 23.51 |
| Insurance | 14.70 | 21.00 | 16.08 | 20.78 |
| Machinery | 14.11 | 26.25 | 10,20 | 25.59 |
| Metale | 12 07 | 20.23 | 10.27 | 20,03 |
| Mining | 14 22 | 24.19 | 10.57 | 20,03 |
| 011 & Cas Exploration | 16 04 | 30 17 | 11.74 | 20.79 |
| Paper & Wood | 16 35 | 26 g3 | 14 04 | JL, JL 31 07 |
| Petroleum & Coal Products | 15 73 | 20,00 | 14.04 | 21.67 |
| Railroads & Trucking | 14 19 | 30 94 | 13 40 | 24.01 |
| Real Fetate | 14.22 | 12 67 | 15.40 | 24,99 |
| Retail Trade Food | 13.86 | 26 35 | 11.45 | 33.31 |
| Potail Trade Others | 15.00 | 20,33 | 16 99 | 27.42 |
| Services Leisure | 16 00 | 29.12 | 19.07 | 20.12 |
| Services, Delsure | 16.40 | 34.47 | 16.15 | 30.13 |
| Toytilor | 13 81 | 20 01 | 14.22 | 32.33 |
| Tobacco | 15.01 | 29.03 | 13,11 | 21 02 |
| Transportation | 14 40 | 20.20 | 17.07 | 21.02 |
| Transportation Fauipment | 14,40 | 20.47 2 | 0,JZ 11,74 | 27.37 |
| litilition | 13.11 | 29.07 | 10.74 | 29.00 |
| Wholegele Trade | 16 61 | 23.03 | 14 70 | 27.05 |
| Rick-Froo | 3 76 | 3 47 | 7 49 | 27.05 |
| Inflation | 4 09 | 3,80 | 6 12 | 2.71 |
| | ••••• | 5.00 | 0,12 | J.1.) |
| Power -75 | 5.00 | 5.32 | 8.21 | . 3.07 |
| Power -50 | 5.55 | 6.93 | 8,56 | 3.42 |
| Power -30 | 6.47 | 8.57 | 9.23 | 4.32 |
| Power -15 | 8.01 | 12.23 | 10.80 | 6.89 |
| Power -10 | 9.21 | 14.15 | 12.20 | 9;46 |
| Power -5 | 11.56 | 17.64 | 14.47 | 14,88 |
| Power -2 | 15.27 | 23.22 | 17.42 | 22,89 |
| Power 0 | 16.09 | 33.09 | 10.15 | 31.,41 |
| Power 0.5 | 12.95 | 47.14 | 2.65 | 43,12 |
| Power 1 | 9.46 | 58.19 | - 6 , 88 | 54.92 |
| Portfolio E2 | 6.47 | 5.40 | 9 20 | 5.11 |
| Portfolio E4 | 8.99 | 9.72 | 10.69 | 9.38 |
| Portfolio E6 | 11 31 | 14 39 | 11.95 | 13.89 |
| Portfolio E8 | 13.42 | 19.14 | 12.97 | 18.49 |
| Portfolio ElO | 15.31 | 23.96 | 13.75 | 23.15 |
| Portfolio El2 | 16.56 | 28.80 | 13.67 | 27.95 |
| Portfolio E14 | 17.77 | 33.41 | 13.95 | 32.15 |
| Portfolio E16 | 18.61 | 37.35 | 14.10 | 35.44 |
| Portfolio E18 | 19.29 | 41.08 | 14.26 | 37,98 |
| Portfolio E20 | 20.18 | 43.74 | 14.27 | 40.71 |

* Standard Deviation is for the variable ln(l+r).

٠,

•

Table 6b

| | 1934 - 1986 🔸 | | 1966 | - 1986 |
|-----------------------------|---------------|--------------------|---------------|--------|
| Portfolio | G.Mean | S.Dev [°] | G.Mean | S.Dev |
| Banks & Financial Services | 14.49 | 25.48 | 11.96 | 25.85 |
| Chemicals, Rubber & Plastic | s 14.61 | 20.11 | 13.90 | 17.75 |
| Construction | 13.78 | 41.87 | 15.90 | 30.50 |
| Fabricated Metal Products | 14.00 | 25.33 | 14.08 | 23.51 |
| Food | 14.70 | 19.92 | 16.08 | 20.78 |
| Insurance | 14.11 | 21.00 | 16.26 | 25.59 |
| Machinery | 14.71 | 26.25 | 11.79 | 26.85 |
| Metals | 12.97 | 24.19 | 10.37 | 20.03 |
| Mining | 14.22 | 25.41 | 11.74 | 26.79 |
| Oil & Gas Exploration | 16.04 | 30.17 | 11.03 | 31.31 |
| Paper & Wood | 16.35 | 24.83 | 14.04 | 21.87 |
| Petroleum & Coal Products | 15.73 | 22.11 | 14,49 | 24.01 |
| Railroads & Trucking | 14,19 | 30,94 | 13.40 | 24.99 |
| Real Estate | 14.22 | 32.67 | 11.45 | 35.91 |
| Retail Trade. Food | 13.86 | 26.35 | 13.15 | 27.42 |
| Retail Trade. Others | 15.49 | 29.12 | 14.89 | 31 88 |
| Services, leisure | 16 99 | 34 47 | 18 15 | 38 13 |
| Services, Others | 16.40 | 32 39 | 14 22 | 32 55 |
| Textiles | 13 81 | 29 03 | 13 11 | 32.55 |
| Tabacco | 15 03 | 20,00 | 19.11 | 21 02 |
| Fransnártetion | 14 40 | 30.47 | 17.07 9.32 | 21.02 |
| Fransportation Equipment | 15 11 | 20.47 | 13 74 | 27.37 |
| Itilities | 13 22 | 23.05 | 12.14 | 17 34 |
| Jholesale Trade | 16 61 | 28.89 | 16 70 | 27 05 |
| lisk-Froo | 3 76 | 3 47 | 7 / 0 | 27.05 |
| Inflation | 4.09 | 3.80 | 6.12 | 3.15 |
| ?ower -75 | 5.00 | 5.32 | 8.21 | 3.07 |
| ?ower -50 | 5.55 | 6.93 | 8.56 | 3.42 |
| 'ower -30 | 6.44 | 8.46 | 9.23 | 4.32 |
| ?ower -15 | 7.72 | 9.90 | 10.80 | 6.89 |
| 'ower -10 | 8.77 | 11.32 | 12.20 | 9.46 |
| 'ower -5 | 10.73 | 14.15 | 13.54 | 13.77 |
| 'ower -2 | 12.07 | 17 01 | 12 93 | 17 22 |
| 'ower O | 12 60 | 24 28 | 9 27 | 25 16 |
| $r_{\rm ower} = 0.5$ | 11 53 | 28 91 | 5 41 | 27 60 |
| Power 1 | 11 11 | 30.87 | 2 57 | 30.05 |
| | | 50.07 | 2.57 | 50.05 |
| ortfolio E2 | 6.47 | 5.40 | 9.20 | 5.11 |
| ortfolio E4 | 8.99 | 9.72 | 10.69 | 9.38 |
| 'ortfolio E6 | 11.31 | 14.39 | 11.95 | 13.89 |
| 'ortfolio E8 | 13.42 | 19.14 | 12.97 | 18.49 |
| Cortfolio ElO | 15.31 | 23.96 | 13.75 | 23.15 |

Geometric Means and Standard Deviations of Annual Returns using 24 Equal-Weighted Industry Indices, without Leverage (32 quarter estimating period, simple probability assessment)

*Standard Deviation is for the variable $ln(1+r_t)$.

Table 6c

Geometric Means and Standard Deviations of Annual Returns using 24 Value-Weighted Industry Indices, with Leverage (32 quarter estimating period, simple probability assessment)

| | 1934 | - 1986 | 1966 | 1086 |
|------------------------------|--------|---------|--------|--------|
| Portfolio | G.Mean | S.Dev* | G.Mean | S.Dev* |
| Banks & Financial Services | 12.12 | 21.54 | 10,41 | 19.11 |
| Chemicals, Rubber & Plastics | 11.15 | 16.38 | 9,18 | 15.75 |
| Construction | 10,97 | 35.19 | 8,24 | 32.90 |
| Fabricated Metal Products | 9.68 | 21.19 | 9,68 | 18.53 |
| Food | 11.58 | 16.74 | 12.66 | 19,29 |
| Insurance | 11.49 | 16.88 | 11.92 | 19.41 |
| Machinery | 11,93 | 20.69 | 7.84 | 21.06 |
| Metals | 9.57 | 21.69 x | 6.03 | 17.25 |
| Mining | 8,86 | 20.55 | 4.70 | 22.94 |
| Oil & Gas Exploration | 13.78 | 24,93 | 11.23 | 25,55 |
| Paper & Wood | 12.87 | 21.97 | 10.30 | 18.06 |
| Petroleum & Coal Products | 12.64 | 18,80 | 11.79 | 20.97 |
| Railroads & Trucking | 10.04 | 22.44 | 10,02 | 20.78 |
| Real Estate | 11.84 | 30.56 | 9.70 | 37.99 |
| Retail Trade, Food | 11.96 | 22.92 | 10,99 | 23.97 |
| Retail Trade. Others | 11.40 | 23.06 | 9.42 | 27.04 |
| Services. Leisure | 13.86 | 33.08 | 15.18 | 38.91 |
| Services. Others | 12.75 | 31.39 | 10.67 | 28.10 |
| Textiles | 10.52 | 26.16 | 10.15 | 31.59 |
| Tobacco | 12 01 | 17.70 | 16 45 | 14 20 |
| Transportation | 10 21 | 27 77 | 4 24 | 27 38 |
| Transportation Faulpment | 12 02 | 26 71 | 7 96 | 26 36 |
| litilities | 10 33 | 15 50 | 9.68 | 13 70 |
| Wholesale Trade | 12 52 | 25.09 | 12 64 | 25 78 |
| Rick-Free | 3 76 | 3 47 | 7 49 | 2 71 |
| Inflation | 4.09 | 3.80 | 6.12 | 3.15 |
| Power -75 | 5.35 | 6.94 | 7.76 | 2.90 |
| Power -50 | 5.67 | 7.88 | 7.89 | 3.13 |
| Power -30 | 6.17 | 9.92 | 8.13 | 3.78 |
| Power -15 | 7.27 | 13.54 | · 8.65 | 5.78 |
| Power -10 | 7.97 | 15.09 | 9.06 | 7,78 |
| Power -5 | 9.05 | 17.52 | 9.32 | 12.17 |
| Power -2 | 9,95 | 23.05 | 7.80 | 19.76 |
| Power 0 | 7.51 | 39.74 | 2.13 | 42.01 |
| Power 0.5 | 6.73 | 52.02 | -0.95 | 55.22 |
| Power 1 | 2.98 | 65.22 | -11.21 | 72.81 |
| Portfolio V2 | 5.47 | 4.24 | 8.12 | 3.99 |
| Portfolio V4 | 7.07 | 6.99 | 8.62 | 6.87 |
| Portfolio V6 | 8.56 | 10.24 | 9.01 | 10.08 |
| Portfolio V8 | 9.94 | 13.66 | 9.26 | 13.42 |
| Portfolio V10 | 11.21 | 17.16 | 9,38 | 16.84 |
| Portfolio V12 | 11.93 | 20.69 | - 8,77 | 20.39 |
| Portfolio Vl4 | 12.65 | 24.15 | 8.43 | 23.79 |
| Portfolio V16 | 13.18 | 27.11 | 8.12 | 26.87 |
| Portfolio V18 | 13.57 | 30.00 | 7.74 | 29,78 |
| Portfolio V20 | 14.04 | 32.34 | 7.15 | 32.90 |

*Standard Deviation is for the variable ln(1+r).

Table 6d

1934 - 1986 1966 - 1986 Portfolio G.Mean S.Dev G.Mean S.Dev Banks & Financial Services 12.12 21.54 10.41 19.11 Chemicals, Rubber & Plastics 11.15 16.38 9.18 15.75 Construction 10.97 35.19 8.24 32.90 Fabricated Metal Products 9.68 21.19 9.68 18.53 Food 11.58 16.74 12.66 19.29 11.49 16.88 11.92 Insurance 19.41[·] Machinery 11.93 20.69 7.84 21.06 Metals 9.57 21.69 6.03 17.25 Mining 8.86 20.55 4.70 22.94 13.78 24.93 011 & Gas Exploration 11.23 25.55 Paper & Wood 12.87 21.97 10.30 18.06 Petroleum & Coal Products 12.64 18.80 11.79 20.97 10.04 Railroads & Trucking 22.44 10.02 20.78 Real Estate 11.84 30.56 9.70 37.99 Retail Trade, Food 11.96 22.92 10.99 23.97 Retail Trade, Others 11.40 23.06 9.42 27.04 13.86 Services, Leisure 33.08 15.18 38.91 Services, Others 12.75 31.39 10.67 28.10 Textiles 10.52 26.16 10.15 31.59 Tobacco 12.01 17.70 16.45 14.20 Transportation 10.21 27.77 4.24 27.38 Transfortation Equipment 12.02 26.71 7.96 26.36 Utilities 10.33 15.50 9.68 13.70 12.52 25.09 Wholesale Trade 12.64 25.78 **Risk-Free** 3.76 3.47 7.49 2.71 Inflation 4.09 3.80 6.12 3.15 Power -75 6.94 5.35 7.76 2.90 Power -50 5.67 7.88 7.89 3.13 Power -30 5.98 9.14 8.13 3.78 Power -15 6.77 10.17 8.65 5.78 7.78 11.36 Power -10 7.31 11.30 9.06 Power - 5 8.56 13.25 9.15 - 2 8.22 18.47 Power 7.80 18.28 0 Power 7.80 29.63 4.22 32.90 Power 0.5 8.08 31.53 3.49 33.64 Power 1 7.22 31.68 1.63 34.94 Portfolio V2 5.47 4.24 8.12 3.99 7.07 6.99 Portfolio V4 8.62 6.87 Portfolio V6 8.56 10.24 9.01 10.08 Portfolio V8 9.94 13.66 9.26 13.42 Portfolio V10 11.21 17.16 9.38 16.84

Geometric Means and Standard Deviations of Annual Returns using 24 Value-Weighted Industry Indices, without Leverage (32 quarter estimating period, simple probability assessment)

*Standard Deviation is for the variable $\ln(1+r_t)$.

Geometric Means and Standard Deviations of Annual Returns using 12 Equal-Weighted Industry Indices, with Leverage, 1936-86: 32, 28 and 40 Quarter Estimating Periods (simple probability assessment)

| | 32 00 | arter | 28 0 | iarter | 40 Quarter | |
|-----------------------|--------|--------|--------|---------|------------|--------|
| Portfolio | G.Mean | S.Dev* | G.Mean | S.Dev" | G.Mean | S.Dev* |
| Petroleum | 14.38 | 23.87 | 14.38 | 23.87 | 14.38 | 23.87 |
| Finance & Real Estate | 13.83 | 25.07 | 13.83 | 25,07 | 13.83 | 25.07 |
| Consumer Durables | 14.06 | 27.88 | 14.06 | 27.88 | 14.06 | 27.88 |
| Basic Industries | 13.80 | 21.36 | 13.80 | 21,36 | 13.80 | 21.36 |
| Food & Tobacco | 14.13 | 20.06 | 14.13 | 20,06 | 14.13 | 20.06 |
| Construction | 13.14 | 26.86 | 13.14 | 26.86 | 13.14 | 26,86 |
| Capital Goods | 12.96 | 24.72 | 12.96 | 24.72 | 12.96 | 24.72 |
| Transportation | 14.04 | 29.70 | 14.04 | 29.70 | 14,04 | 29.70 |
| Utilities | 12.66 | 21.58 | 12.66 | 21.58 | 12.66 | 21.58 |
| Textiles & Trade | 13.92 | 28.53 | 13.92 | 28,53 | 13,92 | 28.53 |
| Services | 15.44 | 32.39 | 15.44 | 32.39 | 15.44 | 32.39 |
| Leisure | -14.66 | 30.67 | 14.66 | 30.67 | 14,66 | 30.67 |
| Risk-Free | 3.90 | 3.47 | 3.90 | 3.47 | 3.90 | 3.47 |
| Inflation | 4.16 | 3.86 | 4.16 | 3.86 | 4.16 | 3.86 |
| Power -75 | 4.86 | 4.18 | 5.12 | 4.30 | ن 4.70 | 3.77 |
| Power -50 | 5.32 | 5.08 | 5.70 | 5.38 | 5.08 | 4.31 |
| Power -30 | 6.22 | 7.05 | 6.69 | 7.58 | 5,80 | 5.79 |
| Power -15 | 7.62 | 9.84 | 8.41 | 10.79 | 7.25 | 8.39 |
| Power -10 | 8.92 | 11.66 | 9,61 | 12.85 | 8.43 | 9.67 |
| Power -5 | 11.09 | 15.31 | 12.10 | 17.36 | 10.69 | 13.51 |
| Power -2 | 14.13 | 21.70 | 15.35 | 24.91 | 13.74 | 20.52 |
| Power O | 18.07 | 33.95 | 15.85 | 36.33 | 16.01 | 33,83 |
| Power 0.5 | 16.99 | 45.29 | 14.93 | 47.74 | 14.89 | 44.74 |
| Power 1 | 15.07 | 55.11 | 17.48 | 55.84 | 14.63 | 52.64 |
| Portfolio E2 | 6.40 | ,5,45 | 6.40 | 5,45 | 6.40 | 5.45 |
| Portfolio E4 | 8.72 | 9.72 | 8.72 | 9.72 | 8.72 | 9.72 |
| Portfolio E6 | 10,84 | 14.34 | 10.84 | 14.34 | 10.84 | 14.34 |
| Portfolio E8 | 12.76 | 19.07 | 12.76 | 19.07 | 12.76 | 19.07 |
| Portfolio ElO | 14.48 | 23.88 | 14.48 | 23,88 | 14.48 | 23,88 |
| Portfolio El2 | 15.56 | 28.72 | 15.56 | 28.72 | 15.56 | 28.72 |
| Portfolio E14 | 16.63 | 33.34 | 16.63 | 33.34 | 16,63 | 33,34 |
| Portfolio El6 | 17.33 | 37.26 | 17.33 | 37.26 | 17.33 | 37.26 |
| Portfolio E18 | 17.88 | 40,98 | 17.88 | 40.98 ຶ | 17.88 | 40.98 |
| Portfolio E20 | 18.68 | 43.60 | 18.68 | 43,60 | 18.68 | 43.60 |
| Portfolio V2 | 5.49 | 4.26 | 5.49 | 4.26 | 5.49 | 4.26 |
| Portfolio V4 | 6.98 | 6.99 | 6,98 | 6.99 | 6.98 | 6.99 |
| Portfolio V6 | 8.36 | 10.21 | 8.36 | 10.21 | 8.36 | 10.21 |
| Portfolio V8 | 9.62 | 13.59 | 9.62 | 13.59 | 9.62 | 13.59 |
| Portfolio V10 | 10.78 | 17.08 | 10.78 | 17.08 | 10.78 | 17.08 |
| Portfolio V12 | 11.39 | 20.60 | 11.39 | 20.60 | 11.39 | 20.60 |
| Portfolio V14 | 12.00 | 24.05 | 12.00 | 24.05 | 12.00 | 24.05 |
| Portfolio V16 | 12.41 | 26.98 | 12.41 | 26.98 | 12.41 | 26,98 |
| Portfolio -V18 | 12.71 | 29.83 | 12.71 | 29.83 | 12.71 | 29.83 |
| Portfolio V20 | 13.08 | 32.13 | 13.08 | 32.13 | 13.08 | 32.13 |

*Standard Deviation is for the variable ln(1+r).

1

Geometric Means and Standard Deviations of Annual Returns using 12 Equal-Weighted Industry Indices, with Leverage, 1966-86: 32, 28 and 40 Quarter Estimating Periods (simple probability assessment)

| | 32 Qi | uarter 🔒 | 28 Q | uarter 🔔 | 40 Qu | 0 Quarter , | |
|-----------------------|--------|--------------------|--------|--------------------|--------|-------------|--|
| Portfolio | G.Mean | S.Dev [*] | G.Mean | S.Dev [*] | G.Mean | S.Dev | |
| Petroleum | 11.82 | 27.24 | 11.82 | 27.24 | 11.82 | 27.24 | |
| Finance & Real Estate | 12.45 | 26.25 | 12.45 | 26.25 | 12.45 | 26.25 | |
| Consumer Durables | 13.42 | 28.12 | 13.42 | 28.12 | 13.42 | 28.12 | |
| Basic Industries | 12.69 | 17.41 | 12.69 | 17.41 | 12.69 | 17.41 | |
| Food & Tobacco | 16.02 | 20.99 | 16.02 | 20.99 | 16.02 | 20.99 | |
| Construction | 12.97 | 24.84 | 12.97 | 24.84 | 12.97 | 24.84 | |
| Capital Goods | 11.88 | 25.02 | 11.88 | 25.02 | 11.88 | 25.02 | |
| Transportation | 10.97 | 24.40 | 10.97 | 24.40 | 10.97 | 24.40 | |
| Utilities | 12.13 | 17.33 | 12.13 | 17.33 | 12.13 | 17.33 | |
| Textiles & Trade | 14.18 | 31.28 , | 14.18 | 31.28 | 14.18 | 31.28 | |
| Services | 14.22 | 32.55 | 14.22 | 32.55 | 14.22 | 32.55 | |
| Leisure | 15.50 | 30.15 | 15.50 | 30.15 | 15.50 | 30.15 | |
| Risk-Free | 7.49 | 2.71 | 7.49 | 2.71 | 7.49 | 2.71 | |
| Inflation | 6.12 | 3,15 | 6,12 | 3.15 | 6.12 | 3.15 | |
| Power -75 | 7.93 | 2.88 | 7.86 | 2.99 | 7.81 | 2.75 | |
| Power -50 | 8.14 | 3.12 | 8.04 | 3.31 | 7.96 | 2.91 | |
| Power -30 | 8.55 | 3.76 | 8.39 | 4.16 | 8.26 | 3.43 | |
| Power -15 | 9.51 | 5.82 | 9.16 | 6.70 | 8,95 | 5.23 | |
| Power -10 | 10.37 | 7.98 | 9.83 | 9.32 | 9.55 | 7.19 | |
| Power -5 | 12.32 | 13.78 | 11.52 | 16.18 | 10.62 | 12.19 | |
| Power -2 | 15.27 | 22.27 | 13.83 | 25.07 | 12.62 | 20.90 | |
| 'ower O | 19.05 | 33.96 | 14.52 | 35.51 | 14.21 | 33.83 | |
| Power 0.5 | 15.44 | 39.57 | 14.57 | 38,40 | 11.87 | 41.06 | |
| 'ower l | 14.97 | 46.38 | 15.69 | 44.18 | 13.96 | 49.14 | |
| ortfolio E2 | 9.20 | 5.11 | 9.20 | 5.11 | 9.20 | 5.11 | |
| ortfolio E4 | 10.69 | 9.38 | 10.69 | 9,38 | 10.69 | 9.38 | |
| ortfolio E6 | 11.95 | 13.89 | 11.95 | 13.89 | 11.95 | 13.89 | |
| ortfolio E8 | 12.97 | 18.49 | 12.97 | 18.49 | 12.97 | 18.49 | |
| ortfolio ElO | 13.75 | 23.15 | 13.75 | 23.15 | 13.75 | 23.15 | |
| ortfolio El2 | 13.67 | 27.95 | 13.67 | 27.95 | 13.67 | 27.95 | |
| ortfolio El4 | 13.95 | 32.15 | 13.95 | 32.15 | 13.95 | 32.15 | |
| ortfolio El6 | 14.10 | 35.44 | 14.10 | 35.44 | 14.10 | 35.44 | |
| ortfolio El8 | 14.26 | 37.98 | 14.26 | 37.98 | 14.26 | 37.98 | |
| ortfolio E20 | 14.27 | 40.71 | 14.27 | 40.71 | 14.27 | 40.71 | |
| ortfolio V2 | 8.12 | 3.99 | 8.12 | 3,99 | 8.12 | 3,99 | |
| ortfolio V4 | 8.62 | 6.87 | 8.62 | 6.87 | 8.62 | 6.87 | |
| ortfolio V6 | 9.01 | 10.08 | 9.01 | 10,08 | 9.01 | 10.08 | |
| ortfolio V8 | 9.26 | 13.42 | 9.26 | 13.42 | 9.26 | 13.42 | |
| ortfolio V10 | 9.38 | 16.84 | 9.38 | 16.84 | 9.38 | 16.84 | |
| ortfolio V12 | 8.77 | 20.39 | 8.77 | 20.39 | 8.77 | 20.39 | |
| ortfolio V14 | 8.43 | 23.79 | 8.43 | 23.79 | 8.43 | 23.79 | |
| ortfolio V16 | 8.12 | 26.87 | 8.12 | 26.87 | 8,12 | 26 87 | |
| ortfolio V18 | 7.74 | 29.78 | 7 74 | 29.78 | 7 74 | 29 78 | |
| ortfollo V20 | 7 15 | 22.00 | 7 16 | 12 00 | 7 16 | 22.70 | |

* Standard Deviation is for the variable ln(1+r).

Table 7c

| | Geometric Means and Standard Deviations of Annual Returns using 12 | |
|-----|---|-----------|
| | Equal-Weighted Industry Indices, without Leverage, 1934-86: | |
| 32, | 28 and 40 Quarter Estimating Periods (simple probability assessment | :) |

| | 32 Quarter | | 28 Quarter | | 40 Quarter | |
|-----------------------|------------|---------------|------------|--------|------------|--------|
| Portfolio | G.Mean | S.Dev* | G.Mean | S.Dev* | G.Mean | S.Dev* |
| Petroleum | 14.38 | 23.87 | 14.38 | 23.87 | 14.38 | 23.87 |
| Finance & Real Estate | 13.83 | 25.07 | 13.83 | 25.07 | 13.83 | 25.07 |
| Consumer Durables | 14.06 | 27.88 | 14.06 | 27.88 | 14.06 | 27.88 |
| Basic Industries | 13.80 | 21.36 | 13.80 | 21.36 | 13.80 | 21.36 |
| Food & Tobacco | 14.13 | 20.06 | 14.13 | 20.06 | 14.13 | 20.06 |
| Construction | 13.14 | 26.86 | 13.14 | 26.86 | 13.14 | 26.86 |
| Capital Goods | 12.96 | 24.72 | 12.96 | 24.72 | 12.96 | 24.72 |
| Transportation | 14.04 | 29.70 | 14.04 | 29.70 | 14.04 | 29.70 |
| Utilities | 12.66 | 21.58 | 12.66 | 21.58 | 12.66 | 21.58 |
| Textiles & Trade | 13,92 | 28.53 | 13.92 | 28.53 | 13.92 | 28.53 |
| Services | 15.44 | 32,39 | 15.44 | 32.39 | 15.44 | 32.39 |
| Leisure | 14,66 | 30.67 | 14.66 | 30.67 | 14.66 | 30.67 |
| Risk-Free | 3.90 | 3.47 | 3,90 | 3.47 | 3.90 | 3.47 |
| Inflation | 4.16 | 3.86 | 4.16 | 3.86 | 4.16 | 3.86 |
| Power -75 | 4.86 | 4.18 | 5.12 | 4.30 | . 4.70 | 3.77 |
| Power -50 | 5.32 | 5.08 | 5.70 | 5.38 | 5,08 | 4.31 |
| Power -30 | 6.22 | 7.05 | 6.69 | 7.58 | 5,80 | 5.79 |
| Power -15 | 7.55 | 8.20 | 8.06 | 8.98 | 7.27 | 7.37 |
| Power -10 | 8.51 | 9.41 | 9.12 | 10.67 | 8.30 | 8.46 |
| Power -5 | 10.42 | 13.23 | 11.18 | 14.85 | 10.26 | 11.88 |
| Power -2 | 12.44 | 17.55 | 11.91 | 19.44 | 11.32 | 17.09 |
| Power O | 13.47 | 25.26 | 12.65 | 27.19 | 11.99 | 24.85 |
| Power 0.5 | 13.06 | 26.93 | 12.86 | 28.28 | 12.79 | 26.00 |
| Power 1 | 12.48 | 28.81 | 14.24 | 29.17 | 13.49 | 26.81 |
| Portfolio E2 | 6.40 | 5.45 | 6.40 | 5.45 | 6.40 | 5.45 |
| Portfolio E4 | 8.72 | 9.72 | 8.72 | 9.72 | 8,72 | 9.72 |
| Portfolio E6 | 10.84 | 14.34 | 10.84 | 14.34 | 10.84 | 14.34 |
| Portfolio E8 | 12.76 | 19.07 | 12.76 | 19.07 | 12.76 | 19.07 |
| Portfolio ElO | 14.48 | 23.88 | 14.48 | 23.88 | 14.48 | 23,88 |
| Portfolio V2 | 5.49 | 4.26 | 5.49 | 4.26 | 5.49 | 4,26 |
| Portfolio V4 | 6.98 | 6.99 | 6.98 | 6.99 | 6.98 | 6.99 |
| Portfolio V6 | 8.36 | 10.21 | 8.36 | 10.21 | 8.36 | 10.21 |
| Portfolio V8 🚽 | 9.62 | 13.5 9 | 9.62 | 13,59 | 9.62 | 13.59 |
| Portfolio V10 | 10.78 | 17.08 | 10.78 | 17.08 | 10.78 | 17.08 |

*Standard Deviation is for the variable $\ln(1+r_t)$.

ı,

Table 7d

Geometric Means and Standard Deviations of Annual Returns using 12 Equal-Weighted Industry Indices, without Leverage, 1966-86: 32, 28 and 40 Quarter Estimating Periods (simple probability assessment)

| <i>.</i> | 32 Qu | arter _ | 28 Quarter | | 40 Quarter | |
|-----------------------|--------------------|--------------------|------------|--------------------|------------|-------|
| Portfolio | G.Mean | S.Dev [*] | G.Mean | S.Dev [*] | G.Mean | S.Dev |
| Petroleum | 11.82 [°] | 27.24 | 11.82 | 27.24 | 11.82 | 27.24 |
| Finance & Real Estate | 12.45 | 26.25 | 12.45 | 26.25 | 12.45 | 26.25 |
| Consumer Durables | 13.42 | 28.12 | 13.42 | 28.12 | 13.42 | 28.12 |
| Basic Industries | 12.69 | 17.41 | 12.69 | 17.41 | 12.69 | 17.41 |
| Food & Tobacco | 16.02 | 20.99 | 16.02 | 20.99 | 16.02 | 20.99 |
| Construction | 12.97 | 24.84 | 12.97 | 24.84 | 12.97 | 24.84 |
| Capital Goods | 11.88 | 25.02 | 11.88 | 25.02 | 11.88 | 25.02 |
| Transportation | 10.97 | 24.40 | 10.97 | -24.40 | 10.97 | 24.40 |
| Utilities | 12.13 | 17.33 | 12.13 | 17.33 | 12.13 | 17.33 |
| Textiles & Trade | 14.18 | 31.28 | 14.18 | · 31.28 | 14.18 | 31.28 |
| Services | 14.22 | 32,55 | 14.22 | 32.55 | 14.22 | 32.55 |
| Leisure | 15.50 | 30.15 | 15.50 | 30.15 | 15.50 | 30.15 |
| Risk-Free | 7.49 | 2.71 | 7.49 | 2.71 | 7.49 | 2.71 |
| Inflation | 6.12 | 3,15 | 6.12 | 3.15 | 6.12 | 3.15 |
| Power -75 | 7.93 | 2.88 | 7.86 | 2.99 | 7.81 | 2.75 |
| Power -50 | 8.14 | 3.12 | 8.04 | 3.31 | 7.96 | 2.91 |
| Power -30 | 8.55 | 3.76 | 8.39 | 4.16 | 8.26 | 3.43 |
| Power -15 | 9.51 | 5.82 | 9.16 | 6.70 | - 8.95 | 5.23 |
| Power -10 | 10.37 | 7.98 | 9.83 | 9.32 | 9.55 | 7.19 |
| Power -5 | 12.34 | 13.79 | 11.68 | 15.87 | 10.60 | 12.18 |
| Power -2 | 14.46 | 19`05 | 12.38 | 21.19 | 11.35 | 18.35 |
| Power 0 | 15.14 | 26.05 | 13.27 | 25.59 | 13.23 | 26.04 |
| Power 0.5 | 14.88 | 26.19 | 13.32 | 26.27 | 14.57 | 25.64 |
| Power 1 | 14.42 | 26.17 | 13.72 | 27.18 | 15.67 | 26.72 |
| Portfolio E2 | 9.20 | 5.11 | 9.20 | 5.11 | 9.20 | 5.11 |
| Portfolio E4 | 10.69 | 9.38 | 10.69 | 9.38 | 10.69 | 9.38 |
| Portfolio E6 | 11.95 | 13.89 | 11.95 | 13.89 | 11.95 | 13.89 |
| Portfolio E8 | 12.97 | 18.49 | 12.97 | 18.49 | 12.97 | 18.49 |
| Portfolio ElO | 13.75 | 23.15 | 13.75 | 23.15 | 13.75 | 23.15 |
| Portfolio V2 | 8.12 | 3.99 | 8.12 | 3.99 | 8.12 | 3.99 |
| Portfolio V4 | 8.62 | 6.87 | 8.62 | 6.87 | 8.62 | 6.87 |
| Portfolio V6 | 9.01 | 10.08 | 9.01 | 10.08 | 9.01 | 10.08 |
| Portfolio V8 | 9.26 | 13.42 | 9.26 | 13.42 | 9.26 | 13.42 |
| Portfolio V10 | 9.38 | 16.84 | 9.38 | 16.84 | 9.38 | 16.84 |

*Standard Deviation is for the variable $ln(1+r_t)$.

Geometric Means and Standard Deviations of Annual Returns using 12 Value-Weighted Industry Indices, with Leverage, 1936-86: 32, 28 and 40 Quarter Estimating Periods (simple probability assessment)

| | 32 Qu | arter _ | 28 Quarter | | 40 Quarter | |
|-----------------------|---------|--------------------|------------|--------------------|------------|-------|
| Portfolio | G.Mean | S.Dev [*] | G.Mean | S.Dev [*] | G.Moan | S.Dev |
| Petroleum | 12.28 | 18.96 | 12.28 | 18,96 | 12.28 | 18.96 |
| Finance & Real Estate | 11.26 | 20.13 | 11,26 | 20.13 | 11.26 | 20.13 |
| Consumer Durables | . 11.10 | 23.54 | 11.10 | 23.54 | 11.10 | 23.54 |
| Basic Industries | 10.49 | 17.56 | 10,49 | 17.56 | 10.49 | 17,56 |
| Food & Tobacco | 11.19 | 16.45 | 11.19 | 16.45 | 11,19 | 16.45 |
| Construction | 8,94 | 21.63 | 8,94 | 21.63 | 8,94 | 21.63 |
| Capital Goods | 10.80 | 20.57 | 10.80 | 20.57 | 10.80 | 20.57 |
| Transportation | 9.44 | 23.00 | 9.44 | 23.00 | 9.44 | 23,00 |
| Utilities | 9.80 | 14.31 | 9.80 | 14.31 | 9.80 | 14.31 |
| Textiles & Trade | 10.71 | 23.27 | 10.71 | 23.27 | 10.71 | 23.27 |
| Services | 14.15 | 28.27 | 14.15 | 28.27 | 14.15 | 28.27 |
| Leisure | 11.80 | 28.85 | 11.80 | 28.85 | 11.80 | 28.85 |
| Risk-Free | 3.90 | 3.47 | 3.90 | 3.47 | 3.90 | 3.47 |
| Inflation | 4.16 | 3.86 | 4.16 | 3.86 | 4.16 | 3.86 |
| Power -75 | 5,05 | 4.66 | 5.26 | 4.70 | . 4.82 | 4.22 |
| Power -50 | 5.59 | 5.96 | 5.86 | 6.05 | . 5.25 | 5.19 |
| Power -30 | 6.34 | 7,95 | 6.71 | 8.01 | 5,93 | 7.03 |
| Power -15 | 7.51 | 11.12 | 8.08 | 11.64 | 7.14 | 9.70 |
| Power -10 | 8.08 | 13.11 | 8.70 | 14.01 | 7.95 | 11.31 |
| Power -5 | 9.56 | 16.56 | 10.08 | 18.17 | 9.57 | 14.52 |
| Power -2 | 11.89 | 22.13 | 12.09 | 23.64 | 11.74 | 19.88 |
| Power O | 13.61 | 33.36 | 12.93 | 32.79 | 12.85 | 28.47 |
| Power 0.5 | 13.75 | 44.09 | 12.82 | 43,90 | 13.06 | 35.79 |
| Power 1 | 10,42 | 55.89 | 12.41 | 53.02 | 11.45 | 49.30 |
| Portfolio V2 | 5.49 | 4.26 | 5.49 | 4.26 | 5.49 | 4.26 |
| Portfolio V4 | 6.98 | 6.99 | 6.98 | 6.99 | 6.98 | 6,99 |
| Portfolio V6 | 8.36 | 10.21 | 8.36 | 10.21 | 8.36 | 10.21 |
| Portfolio V8 | 9.62 | 13.59 | 9.62 | 13.59 | 9.62 | 13,59 |
| Portfolio V10 | 10.78 | 17.08 | 10.78 | 17.08 | 10.78 | 17.08 |
| Portfolio V12 | 11.39 | 20.60 | 11.39 | 20.60 | 11.39 | 20,60 |
| Portfolio V14 | 12.00 | 24.05 | 12.00 | 24.05 | 12.00 | 24.05 |
| Portfolio V16 | 12.41 | 26.98 | 12.41 | 26.98 | 12.41 | 26.98 |
| Portfolio V18 | 12.71 | 29.83 | 12.71 | 29.83 | 12.71 | 29.83 |
| Portfolio V20 | 13 08 | 32 13 | 13 08 | 12 12 - | 13 08 | 32 13 |

*Standard Deviation is for the variable $\ln(1+r_t)$.

Table 8b

| | 32 Qu | arter _ | 28 Qu | arter _ | * 40 Quarter, | | |
|-----------------------|--------------------------|--------------------|--------|---------------------------|---------------|-------|--|
| Portfolio | G.Mean | S.Dev [*] | G.Mean | S.Dev | G.Mean | S.Dev | |
| Petroleum | 10,93 | 21.21 | 10.93 | 21.21 | 10.93 | 21.21 | |
| Finance & Real Estate | 10.68 | 18.75 | 10,68 | 18.75 | 10.68 | 18.75 | |
| Consumer Durables | 8.17 | 23.83 | 8,17 | 23.83 | 8.17 | 23.83 | |
| Basic Industries | 8,58 | 15.15 | 8.58 | 15.15 | 8.58 | 15.15 | |
| Food & Tobacco | 12:96 | 17.39 | 12.96 | 17.39 | 12.96 | 17.39 | |
| Construction | 7.99 | 22.18 | 7.99 | 22.18 | 7.99 | 22.18 | |
| Capital Goods | 7.92 | 20.54 | 7.92 | 20.54 | 7.92 | 20.54 | |
| Transportation | 7.03 | 20.85 | 7.03 | 20.85 | X.03 | 20.85 | |
| Utilities | 9.68 | 13.69 | 9.68 | 13.69 | 9 68 | 13.69 | |
| Textiles & Trade | 9.60 | 26.96 | 9.60 | 26.96 | 9.60- | 26.96 | |
| Services | 10.67 | 28.10 | 10.67 | 28.10 ⁻ | 10.67 | 28.10 | |
| Leisure | 12.31 | 30.18 | 12.31 | 30.18 | 12.31 | 30.18 | |
| Risk-Free | 7.49 | 2.71 | 7.49 | 2.71 | 7.49 | 2.71 | |
| Inflation | 6.12 | 3.15 | 6.12 | 3.15 | 6.12 | 3.15 | |
| Power -75 | 7.81 [°] | 2.90 | 7.69 | 2.95 | 7.69 | 2.70 | |
| Power -50 | 7.96 | 3.12 | 7.79 | 3.22 | 7.79 | 2.79 | |
| Power -30 | 8.25 | · 3.73 | 7.97 | 3.91 | 7.97 | 3.09 | |
| Power -15 | 8.93 | 5.62 | 8.34 | 6.01 | 8.40 | 4.27 | |
| Power -10 | 9.52 | 7.59 | 8.63 | 8.17 | 8.78 | 5.63 | |
| Power -5 | 10.39 | 12.16 | 8.87 | 12.75 | 9.10 | 8.55 | |
| Power -2 | 11.31 | 19.52 | 9.23 | 19.65 | 9.06 | 14.34 | |
| Power 0 | 9.34 | 30.29 | 8.04 | 26.04 | 6.32 | 25.17 | |
| Power 0.5 | 8.87 | 35.97 | 7.55 | 32.73 | 3.20 | 32.98 | |
| Power 1 | 0.02 | 49.37 | 3.67 | 42.86 | 0.85 | 47.43 | |
| Portfolio V2 | 8.12 | 3.99 | 8.12 | 3,99 | 8.12 | -3,99 | |
| Portfolio V4 | 8.62 | 6.87 | 8.62 | 6.87 | 8.62 | 6.87 | |
| Portfolio V6 | 9.01 | 10.08 | 9.01 | 10.08 | 9.01 | 10.08 | |
| Portfolio V8 | 9.26 | 13.42 | 9.26 | 13.42 | 9.26 | 13.42 | |
| Portfolio V10 | 9.38 | 16.84 | 9.38 | 16.84 | 9.38 | 16.84 | |
| Portfolio V12 | 8.77 | 20.39 | 8.77 | 20.39 | 8.77 | 20.39 | |
| Portfolio V14 | 8.43 | 23.79 | 8.43 | 23.79 | 8.43 | 23.79 | |
| Portfolio V16 | 8.12 | 26.87 | 8.12 | 26.87 | 8.12 | 26.87 | |
| Portfolio V18 | 7.74 | 29.78 | 7.74 | 29.78 | * 7.74 | 29.78 | |
| Portfolio V20 🖌 | 7.15 | 32.90 | 7.15 | 32.90 | 7.15 | 32.90 | |

Geometric Means and Standard Deviations of Annual Returns using 12 Value-Weighted Industry Indices, with Leverage, 1966-86: 32, 28 and 40 Quarter Estimating Periods (simple probability assessment)

*Standard Deviation is for the variable $ln(1+r_t)$.

Table 8c

| | 32 Qu | arter " | 28 Qu | arter | 40 Qu | 40 Quarter | |
|-----------------------|--------|--------------------|--------|--------------------|--------|------------|--|
| Portfolio | G.Mean | S.Dev [*] | G.Mean | S.Dev [*] | G.Mean | S.Dev* | |
| Petroleum | 12.28 | 18,96 | 12.28 | 18.96 | 12.28 | 18,96 | |
| Finance & Real Estate | 11.26 | 20.13 | 11.26 | 20.13 | 11.26 | 20.13 | |
| Consumer Durables | 11.10 | 23.54 | 11.10 | 23.54 | 11.10 | 23.54 | |
| Basic Industries | 10.49 | 17,56 | 10.49 | 17.56 | 10.49 | 17.56 | |
| Food & Tobacco | 11.19 | 16,45 | 11.19 | 16.45 | 11.19 | 16,45 | |
| Construction | 8.94 | 21.63 | 8.94 | 21.63 | 8,94 | 21.63 | |
| Capital Goods 🔪 | 10.80 | 20.57 | 10.80 | 20.57 | 10,80 | 20.57 | |
| Transportation | 9.44 | 23.00 | 9.44 | 23.00 | 9.44 | 23.00 | |
| Utilities | 9.80 | 14.31 | 9.80 | 14.31 | 9.80 | 14.31 | |
| Textiles & Trade 👘 | 10.71 | 23.27 | 10.71 | 23.27 | 10,71 | 23.27 | |
| Services | 14.15 | 28.27 | 14.15 | 28.27 | 14.15 | 28.27 | |
| Leisure | 11.80 | 28.85 | 11.80 | 28.85 | 11.80 | 28.85 | |
| Risk-Free | 3.90 | 3.47 | 3.90 | 3.47 | 3,90 | 3.47 | |
| Inflation | 4.16 | 3.86 | 4.16 | <i>,</i> 3,86 | 4.16 | 3.86 | |
| Power -75 | 5.05 | 4.66 | 5.26 | 4.70 | 4.82 | 4.22 | |
| Power -50 | 5.59 | 5,96 | 5.86 | 6.05 | 5.25 | 5.19 | |
| Power -30 | 6.34 | 7.95 | 6.67 | 7.99 | 5.93 | 7.03 | |
| Power -15 | 7.03 | 9.18 | 7.46 | 9.80 | 6.92 | 8.25 | |
| Power -10 | 7.68 | 10.47 | 8.11 | 11.43 | 7.75 | 9.20 | |
| Power -5 | 9.12 | 13.63 | 9.11 | 14.33 | 9.35 | 11.99 | |
| Power -2 | 10.01 | 17.23 | 9.44 | 17.76 | 9.62 | 14.79 | |
| Power 0 | 10.46 | 25.56 | 11.51 | 24.22 | 9.54 | 21.36 | |
| Power 0.5 | 9.93 | 28.61 | 11.97 | 27.26 | 9.52 | 25.26 | |
| Power 1 | 9.45 | 31.22 | 11.37 | 28.53 | 10.31 | 28.08 | |
| Portfolio V2 | 5.49 | 4.26 | 5.49 | 4.26 | 5,49 | 4.26 | |
| Portfolio V4 | 6.98 | 6.99 | 6.98 | 6.99 | 6.98 | 6.99 | |
| Portfolio V6 | 8.36 | 10.21 | 8.36 | 10.21 | 8.36 | 10.21 | |
| Portfolio V8 | 9.62 | 13.59 | 9.62 | 13,59 | 9.62 | 13.59 | |
| Portfolio V10 | 10.78 | 17.08 | 10,78 | 17.08 | 10.78 | 17.08 | |

Geometric Means and Standard Deviations of Annual Returns using 12 Value-Weighted Industry Indices, without Leverage, 1936-86: 32, 28 and 40 Quarter Estimating Periods (simple probability assessment)

* Standard Deviation is for the variable $ln(1+r_t)$.

| | | | د | | | |
|--------------------------------|--------|---------------------|--------|--------------------|--------|---------|
| а | 32 Qu | arter _ | 28 Qu | arter _ | 40 Qu | arter . |
| Portfolio | G.Mean | S.Dev [*] | G.Mean | S.Dev [*] | G.Mean | S.Dev |
| Petroleum | 10.93 | 21.21 | 10.93 | 21.21 | 10.93 | 21.21 |
| Finance & Real Estate | 10.68 | 18.75 | 10.68 | 18.75 | 10.68 | 18.75 |
| Consumer ₀ Durables | 8.17 | 23.83 | 8.17 | 23.83 | 8.17 | 23.83 |
| Basic Industries | 8.58 | 15.15 | 8.58 | 15.15 | 8.58 | 15.15 |
| Food & Tobacco | 12.96 | 17.39 | 12.96 | 17.39 | 12.96 | 17.39 |
| Construction | 7.99 | 22.18 | 7.99 | 22.18 | • 7.99 | 22.18 |
| Capital Goods | 7.92 | 20 [°] .54 | 7.92 | 20.54 | 7.92 | 20.54 |
| Transportation | 7.03 | 20.85 | 7.03 | 20.85 | 7.03 | 20.85 |
| Utilities | 9.68 | 13.69 | 9,68 | 13.69 | 9.68 | 13.69 |
| Textiles & Trade | 9.60 | 26.96 | 9.60 | 26.96 | 9.60 | 26.96 |
| Services | 10.67 | 28.10 | 10.67 | 28.10 | 10.67 | 28.10 |
| Leisure | 12.31 | 30.18 | 12.31 | 30.18 | 12.31 | 30.18 |
| Risk-Free | 7.49 | 2.71 | 7.49 | 2.71 | 7.49 | 2.71 |
| Inflation | 6.12 | 3.15 | 6.12 | 3.15 | 6.12 | 3.15 |
| Power -75 | 7.81 | 2.90 | 7.69 | 2.95 | 7.69 | 2.70 |
| Power -50 | 7.96 | 3.12 | 7.79 | 3.22 | 7.79 | 2.79 |
| Power -30 | 8.25 | 3.73 | 7.97 | 3.91 | 7.97 | 3.09 |
| Power -15 | 8.93 | 562 | 8.34 | 6.01 | 8.40 | 4.27 |
| Power -10 | 9:52 | 7.59 | 8.63 | 8.17 | 8.78 | 5.63 |
| Power -5 | 10.29 | 12.01 | 8.77 | 12.47 | 9.12 | 8.57 |
| Power -2 | 10.61 | 16.92 | 8.33 | 16.13 | 8.36 | 13.51 |
| Power O | 8.23 | 23.87 | 8.77 | 20.80 | 6.00 | 20.61 |
| Power 0.5 | 6.64 | 28.39 | 8.36 | 23.97 | 5.42 | 26.01 |
| Power 1 | 3.85 | 33.66 | 7.00 | 27.29 | 5.24 | 31.02 |
| Portfolio V2 | 8.12 | 3.99 | . 8.12 | 3.99 | 8.12 | 3.99 |
| Portfolio V4 | 8.62 | 6.87 | 8.62 | 6.87 | 8.62 | 6.87 |
| Portfolio V6 | 9.01 | 10.08 | 9.01 | 10.08 | 9.01 | 10.08 |
| Portfolio V8 | 9.26 | 13,42 | 9.26 | 13,42 | 9.26 | 13.42 |
| Portfolio V10 | 9.38 | 16.84 | 9.38 | 16.84 | 9.38 | 16.84 |

Geometric Means and Standard Deviations of Annual Returns using 12 Value-Weighted Industry Indices, without Leverage, 1966-86: 32, 28 and 40 Quarter Estimating Periods (simple probability assessment)

* Standard Deviation is for the variable $ln(1+r_t)$.

Geometric Means and Standard Deviations of Annual Returns using 12 Equal-Weighted Industry Indices, with Leverage, 1934-86: Inflation Adapter vs. Sum-of-the-Digits vs. All-of-History (32 quarter estimating period)

| · • • • | Inflation | Adapter. | Sum-of- | Digits | ۸11-of- | llistory |
|-----------------------|-----------|--------------------|--------------|--------------------|---------|----------|
| Portfolio | G.Mean | S.Dev [*] | G.Mean | S.Dev | G.Mean | S.Dev |
| Petroleum | 14,98 | 24.21 | 14.98 | 24.21 | 14,98 | ·24.21 |
| Finance & Real Estate | 14.76 | 25.22 | 14.76 | 25.22 | 14,76 | 25.22 |
| Consumer Durables | 15.12 | 28.04 | 15.12 | 28.04 | 15,12 | 28,04 |
| Basic Industries | 14,60 | 21.38 | 14.60 | 21.38 | 14.60 | 21.38 |
| Food & Tobacco | 14.58 | 19.78 | 14.58 | 19.78 * | 14.58 | 19.78 |
| Construction | 13.70 | 26.80 | 13.70 | 26,80 | 13.70 | 26.80 |
| Capital Goods | 14.58 | 25.54 | 14.58 | 25.54 | 14.58 | 25.54 |
| Transportation | 14.22 | 29.43 | 14.22 | 29,43 | 14.22 | 29.43 |
| Utilities | 13.22 | 23.05 | 13.22 | 23.05 | 13.22 | 23,05 |
| Textiles & Trade | 14.81 | 28.26 | 14.81 | 28.26 | 14.81 | 28.26 |
| Services | 16.40 | 32.39 | 16.40 | 32.39 | 16.40 | 32.39 |
| Leisure | 15.64 | 30.40 | 15.64 | 30.40 | 15,64 | 30.40 |
| Risk-Free | 3.76 | 3.47 | 3.76 | 3.47 | 3.76 | 3.47 |
| Inflation | 4.09 | 3.80 | 4.09 | 3.80 | 4.09 | 3.80 |
| Power -75 | 4.71 | 4.08 | 4.91 | 4.25 | 3.96 | 3.42 |
| Power -50 | 5.17 | 4.89 | 5.46 | 5.22 | 4.06 | 3.41 |
| Power -30 | 5.99 | 6.80 | 6.49 | 7.40 | 4.25 | 3.42 |
| Power -15' | 7.44 | 9.35 | 8.10 | 10.37 | 4.70 | 3.58 |
| Power -10 | 8.73 | 11.30 | 9.55 | 12.52 | 5.12 | 3.88 |
| Power -5 | 11.00 | 14.12 | 11.83 | 17.01 | 6.25 | 5.22 |
| Power -2 | 14.24 | 19.67 | 15.12 | 24.99 | 8.66 | 9.30 |
| Power 0 | 17.51 | 30.13 | 15.74 | 39.00 | 14.33 | 23.89 |
| Power 0.5 | 17.90 | 42.09 | 15.41 | 48.69 | 17.39 | 36.72 |
| Power 1 | 14.63 | 53.87 | 16.84 | 50.90 | 15.37 | 45.69 |
| Portfolio E2 | 6.47 | 5.40 | 6.47 | 5.40 | 6.47 | 5.40 |
| Portfolio E4 | 8.99 | 9.72 | 8.99 | 9.72 | 8.99 | 9.72 |
| Portfolio E6 | 11.31 | 14.39 | 11.31 | 14.39 | 11.31 | 14.39 |
| Portfolio E8 | 13.42 | 19.14 | 13.42 | 19.14 | 13.42 | 19.14 |
| Portfolio ElO | 15.31 | 23.96 | 15.31 | 23.96 | 15.31 | 23.96 |
| Portfolio E12 | 16.56 | 28.80 | ្ញុំ 16.56 | 28.80 | 16.56 | 28,80 |
| Portfolio El4 | 17.77 | 33.41 | 17.77 | 33,41 | 17.77 | 33,41 |
| Portfolio El6 | 18.61 | 37.35 | 18.61 | 37.35 | 18.61 | 37.35 |
| Portfolio E18 | 19.29 | 41.08 📐 | 19.29 | 41.08 | 19.29 | 41.08 |
| Portfolio E20 | 20.18 | 43.74 | 20.18 | 43.74 | 20.18 | 43.74 |
| Portfolio V2 | 5.47 | 4.24 | 5.47 | 4.24 | 5.47 | 4.24 |
| Portfolio V4 | 7.07 | 6.99 | 7.07 | 6.99 | 7.07 | 6.99 |
| Portfolio V6 | 8.56 | 10.24 | 8.56 | 10.24 | 8.56 | 10.24 |
| Portfolio V8 | 9.94 | 13.66 | 9.94 | 13.66 | 9,94 | 13,66 |
| Portfolio V10 | 11.21 | 17.16 | 11.21 | 17.16 | 11.21 | 17.16 |
| Portfolio V12 | 11.93 | 20.69 | 11.93 | 20.69 | 11.93 | 20.69 |
| Portfolio V14 | 12.65 | 24.15 | 12.65 | 24.15 [°] | 12.65 | 24.15 |
| Portfolio V16 | 13.18 | 27.11/ | 13.18 | 27,11 | 13.18 | 27.11 |
| Portfolio V18 | 13.57 | 30,00 | 13.57 | 30,00 | 13.57 | 30.00 |
| Portfolio V20 | 14.04 | 32.34 | 14.04 | 32.34 | 14.04 | 32.34 |

*Standard Deviation is for the variable ln(1+r).

Table 9b

Geometric Means and Standard Deviations of Annual Returns using 12 Equal-Weighted Industry Indices, with Leverage, 1966-86: Inflation Adapter vs. Sum-of-the-Digits vs. All-of-History (32 quarter estimating period)

| Portfolio | Inflation G.Mean | Adapter S.Dev | Sum-of G.Mean | -Digits S.Dev | All-of G.Mean | -llistory S.Dev |
|-----------------------|---------------------|------------------|------------------|------------------|------------------|--------------------|
| Petroleum | 11.82 | 27.24 | 11.82 | 27.24 | 11.82 | 27.24 |
| Finance & Real Estate | 12.45 | 26.25 | • 12.45 | 26.25 | 12.45 | 26.25 |
| Consumer Durables | 13.42 | 28.12 | 13.42 | 28.12 | 13.42 | 28.12 |
| Basic Industries | 12.69 | * 17.41 | 12.69 | 17.41 | 12.69 | 17.41 |
| Food & Tobacco | 16.02 | 20.99 | 16.02 | 20.99 | 16.02 | 20.99 |
| Construction | 12.97 | 24,84 | 12.97 | 24.84 | 12.97 | 24.84 |
| Capital Goods | 11.88 | 25.02 | 11,88 | 25.02 | 11.88 | 25.02 |
| Transportation | [·] 10.97 | 24.40 | 10.97 * | 24.40 | 10.97 | 24.40 |
| Utilities | 12.13 | 17.33 | 12.13 | 17.33 | 12.13 | 17.33 |
| Textiles & Trade | 14.18 | 31.28 | 14.18 | 31.28 | 14.18 | 31.28 |
| Services | 14.22 | 32.55 | 14.22 | 32.55 | 14.22 | 32.55 |
| Leisure | 15.50 | 30.15 | 15.50 | 30.15 | 15.50 | 30.15 |
| Risk-Free | 7.49 | 2.71 | 7.49 | 2.71 | 7.49 | 2.71 |
| Inflation | 6.12 | 3.15 | 6.12 | 3.15 | 6.12 | 3.15 |
| Power -75 | 7.91 | 2:73 | 8.01 | 3.01 | 7.64 | 2.65 |
| Power -50 | 8.12 | 2.85 | 8.27 | 3.37 | 7.71 | 2.64 |
| Power -30 | 8.52 | 3.24 | 8.76 | 4.33 | 7.86 | 2.65 |
| Power -15 | 9.45 | 4.65 | 9.87 | 7.16 | 8.20 | 2.89 |
| Power -10 | 10.30 | 6.23 | 10.85 | 10.01 | 8.51 | 3.31 |
| Power -5 | 12.17 | 10.31 | 12.59 | 16.83 | 9.33 | 4.95 |
| Power -2 | 14.80 | 17.59 | 14.84 | 26.60 | 10.99 | 9.42 |
| Power 0 | 15.84 | 25.37 | 13.55 | 36.99 | 13.01 | 23.98 |
| Power 0.5 | 15.82 | 30.77 | 14.20 | 39.42 | 12.57 | 30.16 |
| Power 1 | 12.17 | 38.72 | 12.59 | 39.69 | 12.57 | 42.73 |
| Portfolio E2 | 9.20 | 5.11 | 9.20 | 5.11 | 9.20 | 5.11 |
| Portfolio E4 | 10.69 | 9.38 | 10.69 | 9.38 | 10.69 | 9.38 |
| Portfolio E6 | 11.95 | 13.89 | 11.95 | 13.89 | 11.95 | 13.89 |
| Portfolio E8 | 12.97 | 18.49 | 12.97 | 18.49 | 12.97 | 18.49 |
| Portfolio E10 | 13.75 | 23.15 | 13.75 | 23.15 | 13.75 | 23.15 |
| Portfolio E12 | · 13.67 | 27.95 | 13.67 | 27.95 | 13.67 | 27.95 |
| Portfolio E14 | 13.95 | 32.15 | 13.95 | 32.15 | 13.95 | 32.15 |
| Portfolio El6 | 14.10 | 35.44 | 14.10 | 35.44 | 14.10 | 35.44 |
| Portfolio El8 | 14.26 | 37.98 | 14.26 | 37.98 | 14.26 | 37.98 |
| Portfolio E20 | 14.27 | 40.71 | 14.27 | 40.71 | 14.27 | 40.71 |
| Portfolio V2 | 8.12 | 3.99 | 8.12 | 3.99 | 8.12 | ,3.99 |
| Portfolio V4 | 8.62 | 6.87 | 8.62 | 6.87 | 8.62 | 6.87 |
| Portfolio V6 | 9.01 | 10.08 | 9.01 | 10.08 | 9.01 | 10.08 |
| Portfolio V8 | 9.26 | 13.42 | 9.26 | 13.42 | 9.26 | 13.42 |
| Portfolio V10 | 9.38 | 16.84 | 9.38 | 16.84 | 9.38 | 16.84 |
| Portfolio V12 | 8.77 | 20.39 | 8.77 | 20.39 | 8.77 | 20.39 |
| Portfolio V14 | 8.43 | 23.79 | 8.43 | 23.79 | 8.43 | 23.79 |
| Portfolio V16 | 8.12 | 26.87 | 8.12 | 26.87 | 8.12 | 26.87 |
| Portfolio V18 | 7.74 | 29.78 | 7.74 | 29.78 | 7.74 | 29.78 |
| Portfolio V20 | 7.15 | 32.90 | 7.15 | 32.90 | 7.15 | 32.90 |

* Standard Deviation is for the variable ln(1+r).

ذ

Table 9c

Geometric Means and Standard Deviations of Annual Returns using 12 Equal-Weighted Industry Indices, without Leverage, 1934-86: Inflation Adapter vs. Sum-of-the-Digits vs. All-of-History (32 quarter estimating period)

| Portfolio | Inflation G.Mean | Adapter S.Dev | Sum-of-Digits G.Mean S.Dev | | All-of- G.Mean | History _* S.Dev |
|-----------------------|---------------------|------------------|-------------------------------|-------|-------------------|-------------------------------|
| Petroleum | 14.98 | 24.21 | 14.98 | 24.21 | 14.98 | 24.21 |
| Finance & Real Estate | 14.76 | 25.22 | 14.76 | 25.22 | 14.76 | 25.22 |
| Consumer Durables | 15.12 | 28.04 | 15.12 | 28.04 | 15.12 | 28.04 |
| Basic Industries | 14.60 | 21.38 | 14.60 | 21.38 | 14.60 | 21.38 |
| Food & Tobacco | 14.58 | 19.78 | 14.58 | 19.78 | 14.58 | 19.78 |
| Construction | 13.70 | 26.80 | 13.70 | 26.80 | 13.70 | 26.80 |
| Capital Goods | 14.58 | 25.54 | 14.58 | 25.54 | 14.58 | 25.54 |
| Transportation | 14.22 | 29.43 | 14.22 | 29.43 | 14.22 | 29.43 |
| Utilities | 13.22 | 23.05 | 13.22 | 23.05 | 13.22 | 23.05 |
| Textiles & Trade | 14.81 | 28.26 | 14.81 | 28.26 | 14.81 | 28.26 |
| Services | 16.40 | 32.39 | 16.40 | 32.39 | 16.40 | 32.39 |
| Leisure | 15.64 | 30.40 | 15.64 | 30.40 | 15.64 | 30.40 |
| Risk-Free - | 5.76 | 3.47 | 3.76 | 3.47 | 3.76 | 3.47 |
| Inflation | 4.09 | 3.80 | 4.09 | 3.80 | 4.09 | 3.80 |
| Power -75 | 4.71 | 4.08 | 4.91 - | 4.25 | 3.96 | 3.42 |
| Power -50 | 5.17 | 4.89 | 5.46 | 5.22 | 4.06 | 3.41 |
| Power -30 | 5.99 | 6.80 | 6.49 | 7.40 | 4.25 | 3.42 |
| Power -15 | 7.34 | 7.94 | 7.96 | 8.82 | 4.70 | 3,58 |
| Power -10 | 8.34 | 8.91 | 8.96 | 10.50 | 5.12 | 3.88 |
| Power -5 | 10.34 | 11.72 | 11.15 | 15.06 | 6,25 | 5.22 |
| Power - 2 | 12.11 | 15.07 | 12.01 | 20.15 | 8.66 | 9.30 |
| Power 0 | 13.42 | 22.16 | 12.86 | 25.67 | 13.28 | 21.90 |
| Power 0.5 | 13.04 | 24.65 | 13.98 | 27,36 | 13.49 | 22.99 |
| Power 1 | 12.58 | 27.02 | 13.73 | 26.76 | 11.96 | 25.66 |
| Portfolio E2 | 6.47 | 5.40 | 6.47 | 5.40 | 6.47 | 5.40 |
| Portfolio E4 | 8.99 | 9.72 | 8.99 | 9.72 | 8.99 | 9.72 |
| Portfolio E6 | 11.31 | 14.39 | 11.31 | 14.39 | 11.31 | 14.39 |
| Portfolio E8 | 13.42 | 19.14 | 13.42 | 19.14 | 13.42 | 19.14 |
| Portfolio ElO | 15.31 | 23.96 | 15.31 | 23,96 | 15,31 | 23.96 |
| Portfolio V2 | 5.47 | 4.24 | 5.47 | 4.24 | 5.47 | 4.24 |
| Portfolio V4 | 7.07 | 6.99 | 7.07 | 6.99 | 7.07 | 6.99 |
| Portfolio V6 | 8.56 | 10.24 | 8,56 | 10.24 | 8.56 | 10.24 |
| Portfolio V8 | 9.94 | 13.66 | 9.94 | 13.66 | 9.94 | 13.66 |
| Portfolio V10 | 11.21 | 17.16 | 11.21 | 17.16 | 11.21 | 17.16 |

* Standard Deviation is for the variable $ln(1+r_t)$.

Table 9d

Geometric Means and Standard Deviations of Annual Returns using 12 Equal-Weighted Industry Indices, without Leverage, 1966-86: Inflation Adapter vs. Sum-of-the-Digits vs. All-of-History (32 quarter estimating period)

| Portfolio | Inflation Adapter * G.Mean S.Dev | | Sum-of-Digits G.Mean S.Dev | | All-of-History G.Mean S.Dev | |
|-----------------------|-------------------------------------|-------|-------------------------------|-------|--------------------------------|---------------|
| Petroleum | 11.82 | 27.24 | 11.82 | 27.24 | 11.82 | 27.24 |
| Finance & Real Estate | 12.45 | 26.25 | 12.45 | 26.25 | 12.45 | 26.25 |
| Consumer Durables | 13.42 | 28.12 | 13.42 | 28.12 | 13.42 | 28.12 |
| Basic Industries | 12.69 | 17.41 | 12.69 | 17.41 | 12.69 | 17.41 |
| Food & Tobacco | 16.02 | 20.99 | 16.02 | 20.99 | 16.02 | 20.99 |
| Construction | 12.97 | 24.84 | 12.97 | 24.84 | 12.97 | 24.84 |
| Capital Goods | 11.88 | 25.02 | 11.88 | 25.02 | 11.88 | 25.02 |
| Transportation | 10.97 | 24.40 | 10.97 | 24.40 | 10.97 | 24.40 |
| Utilities | 12.13 | 17.33 | 12.13 | 17.33 | 12.13 | 17.33 |
| Textiles & Trade | 14.18 | 31,28 | 14.18 | 31.28 | 14.18 | 31.28 |
| Services | 14.22 | 32.55 | 14.22 | 32.55 | 14.22 | 32.55 |
| Leisure | 15.50 | 30.15 | 15.50 | 30.15 | 15.50 | 30.15 |
| Risk-Free | 7.49 | 2.71 | 7.49 | 2.71 | 7.49 | 2.71 |
| Inflation | . 6.12 | 3.15 | 6.12 | 3.15 | 6.12 | 3.15 |
| Power -75 | 7.91 | 2.73 | 8.01 | 3.01 | 7.64 | 2.65 |
| Power -50 | 8.12 | 2.85 | 8.27 | 3.37 | 7.71 | 2.64 |
| Power -30 | 8.52 | 3.24 | 8.76 | 4.33 | 7.86 | 2.65 |
| Power -15 | 9.45 | 4.65 | 9.87 | 7.16 | 8.20 | 2.89 |
| Power -10 | 10.30 | 6.23 | 10.85 | 10.01 | 8.51 | 3.31 |
| Power -5 | 12.09 | 10.09 | 12.67 | 16.63 | 9,33 | 4.95 |
| Power -2 | 13.01 | 13.54 | 12.35 | 21.84 | 10.99 | 9.42 |
| Power O | 13.61 | 19.15 | 12.50 | 24.24 | 12.49 | 22.39 |
| Power 0.5 | 13.35 | 19.69 | 12.86 | 24.39 | 10.94 | ~22.71 |
| Power 1 | 13.92 | 20.52 | 13.03 | 22,47 | 12.16 | 25.97 |
| Portfolio E2 | 9.20 | 5.11 | 9.20 | 5.11 | 9.20 | 5.11 |
| Portfolio E4 | 10.69 | 9.38 | 10.69 | 9.38 | 10.69 | 9.38 |
| Portfolio E6 | 11.95 | 13.89 | 11.95 | 13.89 | 11.95 | 13.89 |
| Portfolio E8 | 12.97 | 18.49 | 12.97 | 18.49 | 12.97 | 18.49, |
| Portfolio ElO | 13.75 | 23.15 | 13.75 | 23.15 | 13.75 | 23.15 |
| Portfolio V2 | 8.12 | 3.99 | 8.12 | 3.99 | 8.12 | 3.99 |
| Portfolio V4 | 8.62 | 6.87 | 8.62 | 6.87 | 8.62 | 6.87 |
| Portfolio V6 | 9.01 | 10.08 | 9.01 | 10.08 | 9.01 | 10.08 |
| Portfolio V8 | 9.26 | 13.42 | 9.26 | 13.42 | 9.26 | 13.42 |
| Portfolio V10 | 9.38 · | 16.84 | 9.38 | 16.84 | 9.38 | 16.84 |

* Standard Deviation is for the variable $\ln(1+r_t)$.

Table 10a

Geometric Means and Standard Deviations of Annual Returns using 12 Value-Weighted Industry Indices, with Leverage, 1934-86: Inflation Adapter vs. Sum-of-the-Digits vs. All-of-History (32 quarter estimating period)

| Portfolio | Inflation G.Mean | Adapter _* S.Dev | Sum-of- G.Mean | Digits S.Dev | All-of- G.Mean | History S.Dev |
|-----------------------|---------------------|-------------------------------|--------------------|-----------------|---------------------------|------------------|
| , Petroleum | 12.20 | 18.96 | 12.20 | 18.96 | 12.20 | 18,96 |
| Finance & Real Estate | 12.08 | 20.29 | 12.08 | 20.29 | 12.08 | 20.29 |
| Consumer Durables | 11.82 | 23.87 | 11.82 | 23.87 | 11.82 [#] | 23.87 |
| Basic Industries | 10.95 | 17.67 | 10.95 | 17.67 | 10,95 | 17.67 |
| Food & Tobacco | 11.44 | 16.17 | 11.44 | 16.17 | 11.44 | 16.17 |
| Construction | 9.61 | 21.91 | 9.61 | 21.91 | 9.61 | 21.91 |
| Capital Goods | 11.73 | 20.74 | 11.73 | 20.74 | 11.73 | 20.74 |
| Transportation | 9.31 | 22.73 | 9.31 | 22.73 | 9.31 | 22.73 |
| Utilities | 10.33 | 15.50 | 10.33 | 15.50 | 10.33 | 15.50 |
| Textiles & Trade | 11.28 | 22 <i>.</i> 98 | 11.28 | 22.98 | 11.28 | 22.98 |
| Services | 12.75 | 31.39 | . 12.75 | 31.39 | 12.75 | 31.39 |
| Leisure | 12.64 | 28.62 | 12.64 | 28.62 | 12.64 | 28.62 |
| Risk-Free | 3.76 | 3.47 | 3.76 | 3.47 | 3.76 | 3.47 |
| Inflation | 4.09 | 3.80 | 4.09 | 3.80 | 4.09 | 3.80 |
| Power -75 | 4.84 | 4.57 | 5.04 | 4.69 | 3.91 | 3.43 |
| Power -50 | 5.36 | 5.77 | 5.65 | 6.01 | 3.98 | 3.42 |
| Power -30 | 6.07 | 7.67 | 6.58 | 8.15 | 4.12 | 3.43 |
| Power -15 | 7.27 | 10,84 | 7.91 | 11.52 | 4.46 | 3.56 |
| Power -10 | 7.81 | 12,82 | 8.73 | 13.99 | 4.77 | 3.82 |
| Power -5 | 9.43 | 15,96 | [*] 10.27 | 17.96 | 5.58 | 4.96 |
| Power -2 | 12.18 | 20,95 | 11.94 | 24.87 | 7.28 | 8.47 |
| Power O | 14,46 | 30,76 | 12.19 | 36.49 | 10.16 | 19.95 |
| Power 0.5 | 14.26 | 41.04 | 11.58 | 48.73 | 13.13 | 29.84 |
| Power 1 | 15.82 | 50.21 | 11.51 | 57.65 | 14.96 * | 44.85 |
| Portfolio V2 | 5.47 | 4.24 | 5.47 | 4.24 | 5.47 | 4.24 |
| Portfolio V4 | 7.07 | 6.99 | 7.07 | 6.99 | 7.07 | 6.99 |
| Portfolio V6 | 8.56 | 10.24 | 8,56 | 10.24 | 8,56 | 10.24 |
| Portfolio V8 | 9.94 | 13,66 | 9.94 | 13.66 | 9,94 | 13.66 |
| Portfolio V10 | 11.21 | 17.16 | 11.21 | 17.16 | 11.21 | 17,16 |
| Portfolio V12 | 11.93 | 20.69 | Д 1.93 | 20.69 | 11.93 | 20.69 |
| Portfolio Vl4 | 12.65 | 24.15 | 12.65 | 24.15 | 12.65 | 24.Ì5 |
| Portfolio V16 | 13.18 | 27.11 | 3.18 | 27.11 | 13.18 | 27.11 |
| Portfolio V18 | 13.57 | 30.00 | 13.57 | 30.00 | 13.57 | 30.00 |
| Portfolio V20 | 14.04 | 32.34 | 14,04 | 32.34 | 14.04 | 32,34 |

* Standard Deviation is for the variable $ln(1+r_t)$.

Table 10b

Geometric Means and Standard Deviations of Annual Returns using 12 Value-Weighted Industry Indices, with Leverage, 1966-86: Inflation Adapter vs. Sum-of-the-Digits vs. All-of-History (32 quarter estimating period)

| Portfolio | Inflation G.Mean | Adapter* S.Dev | Sum-of- G.Mean | Digits S.Dev | All-of- C.Mean | History S.Dev |
|-----------------------|---------------------|-------------------|-------------------|-----------------|-------------------|------------------|
| Petroleum | 10.93 | 21.21 | 10.93 | 21.21 | 10.93 | 21.21 |
| Finance & Real Estate | 10.68 | 18.75 | 10.68 | 18.75 | 10.68 | 18.75 |
| Consumer Durables | 8.17 | 23.83 | 8.17 | 23.83 | 8.17 | 23.83 |
| Basic Industries | 8.58 | 15.15 | 8.58 | 15.15 | 8.58 | 15.15 |
| Food & Tobacco | 12.96 | 17.39 | 12.96 | 17.39 | 12.96 | 17.39 |
| Construction | 7.99 | 22.18 | 7.99 | 22.18 | 7.99 | 22.18 |
| Capital Goods | 7.92 | 20.54 | 7.92 | 20.54 | 7.92 | 20.54 |
| Transportation | 7.03 | 20.85 | 7.03 | 20.85 | 7.03 | 20.85 |
| Utilities | 9.68 | 13.69 | 9.68 | 13.69 | 9.68 | 13.69 |
| Textiles & Trade | 9.60 | 26.96 | 9.60 | 26.96 | 9.60 | 26.96 |
| Services | 10.67 | 28.10 | 10.67 | 28.10 | 10.67 | 28.10 |
| Leisure | 12.31 | 30.18 | 12.31 | 30.18 | 12.31 | 30.18 |
| Risk-Free | 7.49 | 2.71 | 7.49 | 2.71 | 7.49 | 2.71 |
| Inflation | 6.12 | 3.15 | 6.12 | 3.15 | 6.12 | 3.15 |
| Power -75 | 7.86 | 2.80 | 7.85 | 3.00 | 7.57 | 2.67 |
| Power -50 | 8.03 | 2.98 | 8.02 | 3.34 | 7.61 | 2.66 |
| Power -30 | 8.36 | 3.49 | 8.35 | 4.21 | 7.68 | 2.67 |
| Power -15 | 9.11 | 5.14 | 9.07 | 6.73 | 7.87 | 2.80 |
| Power -10 | 9.72 | 6.79 | 9.69 | 9.31 | 8.03 | 3.05 |
| Power -5 | 10,93 | 10.65 | c10.25 | 14.71 | 8.46 | 4.11 |
| Power -2 | 12.62 | 17.12 | 10.46 | 23.80 | 9.29 | 7.26 |
| Power O | 12.35 | / 25.39 | 9.07 | 33.92 | 9.75 | 18.82 |
| Power 0.5 | 11.26 | 27.84 | 9.09 | 36.96 | 7.82 | 24.66 |
| Power 1 | 9.60 | 32.07 | 7.81 | 39.29 | 7.28 | 43.60 |
| Portfolio V2 | 8.12 | 3.99 | 8.12 | 3.99 | 8.12 | 3.99 |
| Portfolio V4 | 8.62 | 6.87 | 8.62 | 6.87 | 8.62 | 6.87 |
| Portfolio V6 | 9.01 | 10.08 | 9.01 | 10.08 | 9.01 | 10.08 |
| Portfolio V8 | 9.26 | 13.42 | 9.26 | 13.42 | 9.26 | 13.42 |
| Portfolio V10 | 9.38 | 16.84 | 9.38 | 16.84 | 9.38 | 16.84 |
| Portfolio V12 | 8.77 | 20.39 | 8.77 | 20.39 | 8.77 | 20.39 |
| Portfolio V14 | 8.43 | 23.79 | 8.43 | 23.79 | 8.43 | 23.79 |
| Portfolio V16 | 8.12 | 26.87 | 8.12 | 26.87 | 8.12 | 26.87 |
| Portfolio V18 | 7.74 | 29.78 | 7.74 | 29.78 | 7.74 | 29.78 |
| Portfolio V20 | 7.15 | 32.90 | 7.15 | 32.90 | 7.15 | 32.90 |

* Standard Deviation is for the variable $ln(1+r_t)$.

Table 10c

| | | | σ. | • | | |
|-----------------------|---------------------|--------------------|-------------------|-----------------|-------------------|-------------------|
| Portfolio | Inflation G.Mean | Adapter * S.Dev | Sum-of- G.Mean | Digits S.Dev | All-of- G.Mean | llistory S.Dev |
| | · · · · · · | | | | | |
| Petroleum | 12.20 | 18.96 | 12.20 | 18.96 | 12.20 | 18,96 |
| Finance & Real Estate | 12.08 | 20.29 | 12.08 | 20.29 | 12.08 | 20.29 |
| Consumer Durables | 11.82 | 23.87 | 11.82 | 23.87 | 11.82 | 23.87 |
| Basic Industries | 10.95 | 17.67 | 10,95 | 17.67 | 10.95 | 17.67 |
| Food & Tobacco | 11.44 | 16.17 | 11.44 | 16.17 | 11.44 | 16.17 |
| Construction | 9.61 | 21.91 | 9.61 | 21.91 | 9.61 | 21.91 |
| Capital Goods | 11.73 | 20.74 | 11.73 | 20.74 | 11.73 | 20.74 |
| Transportation | 9.31 | 22.73 | 9.31 | 22.73 | 9.31 | 22.73 |
| Utilities | 10.33 | 15.50 | 10.33 | 15.50 | 10.33 | 15.50 |
| Textiles & Trade | 11.28 | 22.98 | 11.28 | 22.98 | 11.28 | 22.98 |
| Services | 12.75 | 31,39 | 12.75 | 31.39 | 12.75 | 31.39 |
| Leisure | 12.64 | 28.62 | 12.64 | 28.62 | 12.64 | 28.62 |
| Risk-Free | 3.76 | 3.47 | 3.76 | 3.47 | 3.76 | 3.47 |
| Inflation | 4.09 | 3.80 | 4.09 | 3.80 | 4.09 | 3.80 |
| Power -75 | 4.84 | 4.57 | 5.04 | 4.69 | 3.91 | 3.43 |
| Power -50 | 5.36 | 5.77 | 5.65 | 6.01 | 3.98 | 3.42 |
| Power -30 | 6.07 | 7.67 | 6.58 | 8.15 | 4.12 | 3,43 |
| Power -15 | 6.81 | 9.02 | . 7.40 | 9.80 | 4.46 | 3.56 |
| Power -10 | 7.51 | 10.27 | 8.20 | 11.38 | 4.77 | 3.82 |
| Power -5 | 9.06 | 12.86 | 9.34 | 14.78 | 5.58 | 4.96 |
| Power -2 | 9,95 | 15.76 | 9.56 | 19.13 | 7.28 | 8.47 |
| Power 0 | 11.13 | 23.25 | 10.56 | 25.53 | 9.94 . | 19.04 |
| Power 0.5 | ° 11.26 | 25.21 | 11.96 | 26.87 | 11.38 | 22.31 |
| Power 1 | 12.86 | 25.57 | 11.90 | 27.94 | 12.08 | 23.72 |
| Portfolio V2 | 5.47 | 4.24 | 5.47 | 4.24 | 5.47 | 4.24 |
| Portfolio V4 | 7.07 | 6.99 | 7.07 | 6,99 | 7.07 | 6.99 |
| Portfolio V6 | 8,56 | 10.24 | 8.56 | 10.24 | 8.56 | 10.24 |
| Portfolio V8 | 9.94 | 13.66 | 9,94 | 13.66 | 9.94 | 13,66 |
| Portfolio V10 | 11.21 | 17.16 | 11.21 | 17.16 | 11.21 | 17.16 |

Geometric Means and Standard Deviations of Annual Returns using 12 Value-Weighted Industry Indices, without Leverage, 1934-86: Inflation Adapter vs. Sum-of-the-Digits vs. All-of-History (32 quarter estimating period)

* Standard Deviation is for the variable $ln(l+r_t)$.

Table 10d

Geometric Means and Standard Deviations of Annual Returns using 12 Value-Weighted Industry Indices, without Leverage, 1966-86: Inflation Adapter vs. Sum-of-the-Digits vs. All-of-History (32 quarter estimating period)

| | Inflation Adapter | | Sum-of | -Digits | All-of-History. | |
|-----------------------|-------------------|--------------------|--------|--------------------|-----------------|--------|
| Portfolio | G.Mean | S.Dev [*] | G.Mean | S.Dev [*] | G.Mean | S.Dev* |
| Petroleum | 10.93 | 21.21 | 10.93 | 21.21 | 10.93 | 21.21 |
| Finance & Real Estate | 10.68 | 18.75 | 10.68 | 18.75 | 10.68 | 18.75 |
| Consumer Durables | 8.17 | 23.83 | 8.17 | 23.83 | 8.17 | 23.83 |
| Basic Industries | 8.58 | 15.15 | 8,58 | 15.15 | 8.58 | 15.15 |
| Food & Tobacco | 12.96 | 17.39 | 12.96 | 17.39 | 12.96 | 17.39 |
| Construction | 7.99 | 22.18 | 7.99 | 22.18 | 7.99 | 22.18 |
| Capital Goods | 7.92 | 20.54 | 7.92 | 20.54 | 7.92 | 20.54 |
| Transportation | 7.03 | 20.85 | 7.03 | 20.85 | 7.03 | 20.85 |
| Utilities | 9.68 | 13.69 | 9.68 | 13.69 | 9.68 | 13.69 |
| Textiles & Trade | 9.60 | 26.96 | 9.60 | 26.96 | 9.60 | 26.96 |
| Services | 10.67 | 28.10 | 10.67 | 28.10 | 10.67 | 28.10 |
| Leisure | 12.31 | 30.18 | 12.31 | 30.18 | 12.31 | 30.18 |
| Risk-Free | 7.49 | 2.71 | 7.49 | 2.71 | 7.49 | 2.71 |
| Inflation | 6.12 | 3.15 | 6.12 | 3.15 | 6.12 | 3.15 |
| Power -75 | 7.86 | 2.80 | 7.85 | 3.00 | 7.57 | 2.67 |
| Power -50 | 8.03 | 2.98 | 8.02 | 3.34 | 7.61 | 2.66 |
| Power -30 | 8.36 | 3.49 | 8.35 | 4.21 | 7.68 | 2.67 |
| Power -15 | 9.11 . | 5.14 | 9.07 | 6.73 | 7.87 | 2.80 |
| Power -10 | 9.85 | 7.06 | 9.71 | 9.35 | 8.03 | 3.05 |
| Power -5 | 10.81 | 10.32 | 9.99 | 14.09 | 8.46 | *4.11 |
| Power -2 | 11.10 | 13.79 | 9.38 | 19.89 | 9.29 | 7.26 |
| Power O | 10.41 | 18.67 | 9.85 | 22.92 | 9.64 | 18.35 |
| Power 0.5 | 9.43 | 19.41 | 10.38 | 23.09 | 7.97 | 20.55 |
| Power 1 | 10.25 | 19.16 | 10.18 | 22.36 | 8.80 | 23.52 |
| Portfolio V2 | 8.12 | 3.99 | 8.12 | 3.99 | 8.12 | 3.99 |
| Portfolio V4 | 8.62 | 6.87 | 8.62 | 6.87 | 8.62 | 6.87 |
| Portfolio V6 | 9.01 | 10.08 | 9.01 | 10.08 | 9.01 | 10.08 |
| Portfolio V8 | 9.26 | 13.42 | 9.26 | 13.42 | 9.26 | 13.42 |
| Portfolio V10 | 9.38 | 16.84 | 9.38 | 16.84 | 9.38 | 16.84 |

* Standard Deviation is for the variable $\ln(1+r_t)$.

Table lla

Geometric Means and Standard Deviations of Annual Returns using 12 Equal-Weighted Industry Indices, with Leverage, 1934-86: Disaster States Scenario for J = 1, 6, and 12 (32 quarter estimating period, simple probability assessment)

| | D. State 1 | | D. State.6 | | D. State 12 | |
|-----------------------|--------------|--------------------|------------|--------------------|-------------|--------|
| Portfolio | G.Mean | S.Dev [*] | G.Mean | S.Dev [*] | G.Mean | S.Dev* |
| Petroleum | 14.98 | 24.21 | 14.98 | 24.21 | 14.98 | 24.21 |
| Finance & Real Estate | 14.76 | 25.22 | 14.76 | 25.22 | 14.76 | 25.22 |
| Consumer Durables | 15.12 | 28.04 | 15.12 | 28.04 | 15.12 | 28.04 |
| Basic Industries | 14.60 | 21.38 | 14.60 | 21.38 | 14.60 | 21.38 |
| Food & Tobacco | 14.58 | 19,78 | 14.58 | 19.78 | 14.58 | 19.78 |
| Construction | 13.70 | 26,80 | 13.70 | 26.80 | 13.70 | 26.80 |
| Capital Goods | 14.58 | 25.54 | 14.58 | 25.54 | 14,58 | 25.54 |
| Transportation | 14.22 | 29.43 | 14.22 | 29.43 | 14.22 | 29.43 |
| Utilities | 13.22 | 23.05 | 13.22 | 23.05 | 13.22 | 23.05 |
| Textiles & Trade | 14.81 | 28.26 | 14.81 | 28.26 | 14.81 | 28.26 |
| Services | 16.40 | 32.39 | 16.40 | 32.39 | 16.40 | 32.39 |
| Leisure | 15.64 | 30,40 | 15.64 | 30.40 | 15.64 | 30.40 |
| Risk-Free | 3 .76 | 3.47 | 3.76 | 3.47 | 3.76 | 3.47 |
| Inflation | 4.09 | 3.80 | 4.09 | 3.80 | 4.09 | 3.80 |
| Power -75 | 4.57 | 4.12 | 4.24 | 3.99 | 4.13 | 3.90 |
| Power -50 | 4.96 | 4.93 | 4.47 | 4.63 | 4.30 | 4.45 |
| Power -30 | 5.70 | 6.83 | 4.90 | 6.28 | 4.63 | 5.92 |
| Power -15 | 6.82 | 9.27 | 5.41 | 8.12 | 4.99 | 7.53 |
| Power -10 | 7.96 | 10.91 | 5,83 | 9.80 | 5.17 | 9.28 |
| Power -5 | 9.72 | 13.37 | 6.45 | 11.09 | 5.50 | 10.61 |
| Power -2 | 12.33 | 18.93 | 7.19 | 12.18 | 5.70 | 10.96 |
| Power O | 17.02 | 29.46 | 9.17 | 15.72 | 5.92 | 11.63 |
| Power 0.5 | 18.64 | 39.62 | 9.97 | 20.32 | 6.11. | 12.47 |
| Power 1 | 16.59 | 55.48 | 14.35 | 48.67 | 3.58 | 22.95 |
| Portfolio E2 | 6.47 | 5.40 | 6.47 | 5.40 | 6.47 | 5,40 |
| Portfolio E4 | 8.99 | 9.72 | 8.99 | 9.72 | 8.99 | 9.72 |
| Portfolio E6 | 11.31 | 14.39 | 11.31 | 14.39 | 11.31 | 14.39 |
| Portfolio E8 | 13.42 | 19.14 | 13.42 | 19.14 | 13.42 | 19.14 |
| Portfolio ElO | 15.31 | 23.96 | 15,31 | 23,96 | 15.31 | 23.96 |
| Portfolio El2 | 16.56 | 28.80 | 16.56 | 28.80 | 16.56 | 28.80 |
| Portfolio El4 | 17.77 | 33.41 | 17.77 | 33.41 | 17.77 | 33.41 |
| Portfolio El6 | 18.61 | 37.35 | 18.61 | 37.35 | 18.61 | 37.35 |
| Portfolio E18 | 19.29 | 41.08 | 19.29 | 41.08 | 19.29 | 41.08 |
| Portfolio E20 | 20.18 | 43.74 | 20.18 | 43.74 | 20,18 | 43.74 |
| Portfolio V2 | 5.47 | 4.24 | 5.47 | 4.24 | 5.47 | 4.24 |
| Portfolio V4 | 7.07 | 6.99 | 7.07 | 6.99 | 7.07 | 6.99 |
| Portfolio V6 | 8,56 | 10.24 | 8.56 | 10.24 | 8.56 | 10.24 |
| Portfolio V8 | 9.94 | 13.66 | 9.94 | 13.66 | 9.94 | 13,66 |
| Portfolio V10 | 11.21 | 17.16 | 11.21 | 17.16 | 11.21 | 17.16 |
| Portfolio Vl2 | 11.93 | 20.69 | 11.93 | 20.69 | 11,93 | 20.69 |
| Portfolio V14 | 12.65 | 24.15 | 12.65 | 24.15 | 12.65 | 24.15 |
| Portfolio V16 | 13.18 | 27.11 | 13.18 | 27.11 | 13.18 | 27.11 |
| Portfolio Vl8 | 13.57 | 30.00 | 13.57 | 30.00 | 13.57 | 30.00 |
| Portfolio V20 | 14.04 | 32.34 | 14.04 | 32.34 | 14.04 | 32.34 |

Standard Deviation is for the variable ln(l+r).

ą

ø

Geometric Means and Standard Deviations of Annual Returns using 12 Equal-Weighted Industry Indices, with Leverage, 1966-86: Disaster States Scenario for J - 1, 6, and 12 (32 quarter estimating period, simple probability assessment)

| | D. State 1 | | D. State 6 | | D. State 12 | |
|-----------------------|------------|--------------------|------------|--------------------|-------------|--------------------|
| Portfolio | G.Mean | S,Dev [*] | G.Mean | S.Dev [*] | G.Mean | S.Dev [*] |
| Petroleum | 11.82 | 27,24 | 11.82 | 27.24 | 11.82 | 27.24 |
| Finance & Real Estate | 12,45 | 26.25 | 12.45 | 26.25 | 12.45 | 26.25 |
| Consumer Durables | 13.42 | 28.12 | 13.42 | 28.12 | 13.42 | 28.12 |
| Basic Industries | 12.69 | 17.41 | 12.69 | 17.41 | 12.69 | 17.41 |
| Food & Tobacco | 16.02 | 20.99 | 16.02 | 20.99 | 16.02 | 20.99 |
| Construction | 12.97 | 24.84 | 12.97 | 24.84 | 12.97 | 24.84 |
| Capital Goods | 11.88 | 25.02 | 11.88 | 25.02 | 11.88 | 25.02 |
| Transportation | 10.97 | 24.40 | 10.97 | 24.40 | 10.97 | 24.40 |
| Utilities | 12.13 | 17.33 | 12.13 | 17,33 | 12.13 | 17.33 |
| Textifes & Trade | 14.18 | 31.28 | 14.18 | 31.28 | 14.18 | 31.28 |
| Services | 14.22 | 32.55 | 14.22 | 32.55 | 14.22 | 32.55 |
| Leisure | 15.50 | 30.15 | 15.50 | 30.15 | 15.50 | 30.15 |
| Risk-Free | 7.49 | 2.71 | 7.49 | 2.71 | 7.49 | 2.71 |
| Inflation | 6.12 | 3.15 | 6.12 | 3.15 | 6.12 | 3.15 |
| Power -75 | 7.86 | 2.80 | 7.61 | 2.75 | 7.51 | 2.71 |
| Power -50 | 8,04 | 2.94 | 7.67 | 2.78 | 7.52 | 2.72 |
| Power -30 | 8,39 | 3.34 | 7.79 | 2.84 | 7.54 | 2.73 |
| Power -15 | 9.20 | 4.74 | 8.07 | 3.07 | 7.59 | 2.75 |
| Power -10 | 9.95 | 6.29 | 8.34 | 3.35 | 7.63 | 2.79 |
| Power -5 | 11.82 | 10.68 | 9.04 | 4.29 | 7.75 | 2.89 |
| Power -2 | 14.53 | 18.56 | 10.36 | 6.53 | 8.01 | 3.25 |
| Power 0 | 18.73 | 30.36 | 13.46 | 12.75 | 8.87 | 5.11 |
| Power 0.5 | 18.94 | 36.29 | 14,79 | 18.77 | 9.54 | 7.18 |
| Power 1 | 12.71 | 44.70 | 15.48 | 23.36 | 11.56 | 10.53 |
| Portfollo E2 | 9.20 | 5.11 | 9,20 | 5.11 | 9.20 | 5.11 |
| Portfollo E4 | 10.69 | 9.38 . | 10.69 | 9.38 | 10.69 | 9.38 |
| Portfollo E6 | 11,95 | 13,89 | 11.95 | 13.89 | 11.95 | 13,89 |
| Portfolio E8 | 12.97 | 18.49 | 12.97 | 18.49 | 12.97 | 18.49 |
| Portfolio E10 | 13.75 | 23.15 | 13.75 | 23.15 | 13.75 | 23.15 |
| Portfolio E12 | 13.67 | 27.95 | 13.67 | 27,95 | 13.67 | 27.95 |
| Portfolio El4 | 13.95 | 32.15 | 13.95 | 32.15 | 13.95 | 32.15 |
| Portfolio El6 | 14.10 | 35.44 | 14.10 | 35.44 | 14.10 | 35.44 |
| Portfolio E18 | 14.26 | 37,98 | 14.26 | 37.98 | 14.26 | 37.98 |
| Portfolio E20 | 14.27 | 40.71 | 14.27 | 40.71 | 14.27 | 40.71 |
| Portfolio V2 | 8.12 | 3,99 | 8.12 | 3.99 | 8.12 | 3.99 |
| Portfolio V4 | 8.62 | 6.87 | 8.62 | 6.87 | 8.62 | 6.87 |
| Portfolio V6 | 9.01 | 10.08 | 9.01 | 10.08 | 9.01 | 10.08 |
| Portfolio V8 | 9.26 | 13.42 | 9.26 | 13.42 | 9.26 | 13.42 |
| Portfollo V10 | 9.38 | 16.84 | 9.38 | 16.84 | 9.38 | 16.84 |
| Portfolio V12 | 8.77 | 20.39 | 8.77 | 20.39 | 8.77 | 20.39 |
| Portfollo Vi4 | 8.43 | 23.79 | 8.43 | 23,79 | 8.43 | 23.79 |
| Portfolio V16 | 8.12 | 26.87 | 8.12 | 26.87 | 8.12 | 26.87 |
| Portfolio V18 | 7.74 | 29.78 | 7.74 | 29.78 | 7.74 | 29,78 |
| Portfolio V20 | 7.15 | 32.90 | 7.15 | 32.90 | 7.15 | 32.90 |
| | | · · · · • | | | | |

*Standard Deviation is for the variable ln(l+r).

Table 11c

Geometric Means and Standard Deviations of Annual Returns using 12 Equal-Weighted Industry Indices, without Leverage, 1934-86; Disaster States Scenario for J - 1, 6, and 12 (32 quarter estimating period, simple probability assessment)

| | D. State 1 | | D. State 6 | | D. State 12 | |
|-----------------------|------------|--------------------|------------|--------|-------------|-------|
| Portfolio | G.Mean | S.Dev [*] | G.Mean | S.Dev* | G.Mean | S.Dev |
| Petroleum | 14.98 | 24.21 | 14.98 | 24.21 | 14.98 | 24.21 |
| Finance & Real Estate | 14.76 | 25.22 | 14,76 | 25.22 | 14.76 | 25.22 |
| Consumer Durables | 15.12 | 28.04 | 15.12 | 28.04 | 15.12 | 28.04 |
| Basic Industries | 14.60 | 21.38 | 14.60 | 21.38 | 14.60 | 21.38 |
| Food & Tobacco | 14.58 | 19.78 | 14,58 | 19.78 | 14.58 | 19.78 |
| Construction | 13.70 | 26.80 | 13,70 | 26.80 | 13,70 | 26.80 |
| Capital Goods | 14.58 | 25.54 | 14,58 | 25.54 | 14.58 | 25.54 |
| Transportation | 14.22 | 29.43 | 14.22 | 29.43 | 14.22 | 29.43 |
| Utilities | 13.22 | 23.05 | 13.22 | 23.05 | 13.22 | 23.05 |
| Textiles & Trade | 14.81 | 28.26 | 14.81 | 28.26 | 14.81 | 28.26 |
| Services | 16.40 | 32.39 | 16.40 | 32.39 | 16.40 | 32.39 |
| Leisure | 15.64 | 30.40 | 15.64 | 30,40 | 15.64 | 30.40 |
| Risk-Free | 3.76 | 3.47 | 3.76 | 3.47 | 3.76 | 3.47 |
| Inflation | 4.09 | 3.80 | 4.09 | 3.80 | 4.09 | 3.80 |
| Power -75 | 4.57 | 4.12 | 4.24 | 3,99 | 4.13 | 3.90 |
| Power -50 | 4.96 | 4,93 | 4.47 | 4.63 | 4.30 | 4.45 |
| Power -30 | 5.70 | 6.83 | 4,90 | 6.28 | 4.63 | 5.92 |
| Power -15 | 6.80 | 7.78 | 5.41 | 7.46 | 5.00 | 7.39 |
| Power -10 | 7.57 | 8.58 | 5.57 | 7.63 | 5.02 | 7.46 |
| Power -5 | 9.12 | 11.14 | 6.09 | 8.22 | 5.12 | 7.56 |
| Power -2 | 11.49 | 15.62 | 6.69 | 9.33 | 5.35 | 7.94 |
| Power 0 | 13.51 | 23.00 | 8.26 | 12.68 | 5.52 | 8.46 |
| Power 0.5 | `14.32 | 26.39 | 8.59 | 15.96 | 5,50 | 9.60 |
| Power 1 | 13.83 | 28.84 | 11.70 | 24.85 | 5,03 | 12.52 |
| Portfolio E2 | 6.47 | 5.40 | 6.47 | 5.40 | 6.47 | 5.40 |
| Portfolio E4 | 8.99 | 9.72 | 8.99 | 9.72 | 8.99 | 9.72 |
| Portfolio E6 | 11.31 | 14.39 | 11.31 | 14.39 | 11.31 | 14.39 |
| Portfolio E8 | 13.42 | 19.14 | 13.42 | 19.14 | 13.42 | 19.14 |
| Portfolio E10 | 15.31 | 23,96 | 15.31 | 23,96 | 15,31 | 23.96 |
| Portfolio V2 | 5.47 | 4.24 | 5.47 | 4.24 | 5.47 | 4.24 |
| Portfolio V4 | 7.07 | 6.99 | 7.07 | 6.99 | 7.07 | 6,99 |
| Portfolio V6 | 8.56 | 10.24 | 8.56 | 10.24 | 8,56 | 10.24 |
| Portfolio V8 | 9.94 | 13.66 | 9.94 | 13.66 | 9.94 | 13.66 |
| Portfolio V10 | 11.21 | 17.16 | 11.21 | 17.16 | 11.21 | 17.16 |

*Standard Deviation is for the variable $\ln(1+r_t)$.

Table 11d

Ľ

Geometric Means and Standard Deviations of Annual Returns using 12 Equal-Weighted Industry Indices, without Leverage, 1966-86: Disaster States Scenario for J - 1, 6, and 12 (32 quarter estimating period, simple probability assessment)

| | D. State 1 🖕 | | D. State 6 🛨 | | D. State 12 | |
|-----------------------|--------------|-------|--------------|-------|-------------|-------|
| Portfolio | G.Mean | S.Dev | G.Mean | S.Dev | G.Mean | S.Dev |
| Petroleum | 11.82 | 27.24 | 11.82 | 27.24 | 11.82 | 27.24 |
| Finance & Real Estate | 12.45 | 26.25 | 12.45 | 26.25 | 12.45 | 26.25 |
| Consumer Durables | 13.42 | 28.12 | 13.42 | 28.12 | 13.42 | 28.12 |
| Basic Industries | 12.69 | 17.41 | 12.69 | 17.41 | 12.69 | 17.41 |
| Food & Tobacco | 16.02 | 20.99 | 16.02 | 20.99 | 16.02 | 20.99 |
| Construction | 12.97 | 24.84 | 12.97 | 24.84 | 12.97 | 24.84 |
| Capital Goods | 11.88 | 25.02 | 11.88 | 25.02 | 11.88 | 25.02 |
| Transportation | 10.97 | 24.40 | 10.97 | 24.40 | 10.97 | 24.40 |
| Utilities | 12.13 | 17.33 | 12.13 | 17.33 | 12.13 | 17.33 |
| Textiles & Trade | 14.18 | 31.28 | 14.18 | 31.28 | 14.18 | 31.28 |
| Services | 14.22 | 32.55 | 14.22 | 32.55 | 14.22 | 32.55 |
| Leisure | 15.50 | 30.15 | 15.50 | 30.15 | 15.50 | 30.15 |
| Risk-Free | 7.49 | 2.71 | 7.49 | 2.71 | 7.49 | 2.71 |
| Inflation | 6.12 | 3.15 | 6.12 | 3.15 | 6.12 | 3.15 |
| Power -75 | 7.86 | 2.80 | 7.61 | 2.75 | 7.51 | 2.71 |
| Power -50 | 8.04 | 2.94 | 7.67 | 2.78 | 7.52 | 2.72 |
| Power -30 | 8.39 | 3.34 | 7.79 | 2.84 | 7.54 | 2.73 |
| Power -15 | 9.20 | 4.74 | 8.07 | 3.07 | 7.59 | 2.75 |
| Power -10 | 9.95 | 6.29 | 8.34 | 3.35 | 7.63 | 2.79 |
| Power -5 | 11.82 | 10.68 | 9.04 | 4.29 | 7.75 | 2.89 |
| Power -2 | 13.88 | 17.46 | 10.36 | 6.53 | 8.01 | 3.25 |
| Power 0 | 15.31 | 23.68 | 12.85 | 11.82 | 8.87 | 5.11 |
| Power 0.5 | 15.70 | 26.33 | 12.92 | 13.55 | 9.55 | 7.19 |
| Power 1 | 14.47 | 26.20 | 12.29 | 14.95 | 10.98 | 9.78 |
| Portfolio EŹ | 9,20 | 5,11 | 9.20 | 5.11 | 9.20 | 5.11 |
| Portfolio E4 | 10.69 | 9.38 | 10.69 | 9.38 | 10.69 | 9.38 |
| Portfolio E6 | 11.95 | 13.89 | 11.95 | 13.89 | 11.95 | 13.89 |
| Portfolio E8 | 12.97 | 18.49 | 12.97 | 18.49 | 12.97 | 18.49 |
| Portfolio E10 | 13.75 | 23.15 | 13.75 | 23.15 | 13.75 | 23.15 |
| Portfolio V2 | 8.12 | 3.99 | 8.12 | 3.99 | 8.12 | 3.99 |
| Portfolio V4 | 8.62 | 6,87 | 8.62 | 6.87 | 8.62 | 6.87 |
| Portfolio V6 | 9.01 | 10,08 | 9.01 | 10.08 | 9.01 | 10.08 |
| Portfolio V8 | 9.26 | 13.42 | 9.26 | 13.42 | 9.26 | 13.42 |
| Portfolio V10 | 9.38 | 16.84 | 9.38 | 16.84 | 9.38 | 16.84 |

 $\langle \boldsymbol{\varphi} \rangle$

* Standard Deviation is for the variable $\ln(1+r_t)$.

Table 12a

Geometric Means and Standard Deviations of Annual Returns using 12 Value-Weighted Industry Indices, with Leverage, 1934-86: Disaster States Scenario for J - 1, 6, and 12 (32 quarter estimating period, simple probability assessment)

| | D. State 1 | | D. State 6 | | D. State 12 | |
|-----------------------|------------|--------------------|------------|--------------------|-------------|--------|
| Portfolio | G.Mean | S.Dev [*] | G.Mean | S.Dev [*] | G. Mean | S.Dev* |
| Petroleum | 12.20 | 18.96 | 12.20 | 18.96 | 12.20 | 18.96 |
| Finance & Real Estate | 12.08 | 20.29 | 12.08 | 20.29 | 12.08 | 20.29 |
| Consumer Durables | 11.82 | 23.87 | 11.82 | 23.87 | 11.82 | 23.87 |
| Basic Industries | 10.95 | 17.67 | 10.95 | 17.67 | 10.95 | 17.67 |
| Food & Tobacco | 11.44 | 16.17 | 11.44 | 16.17 | 11.44 | 16.17 |
| Construction | 9.61 | 21.91 | 9.61 | 21.91 | 9.61 | 21.91 |
| Capital Goods | 11.73 | 20.74 | 11,73 | 20.74 | 11.73 | 20.74 |
| Transportation | 9.31 | 22.73 | 9.31 | 22.73 | 9.31 | 22.73 |
| Utilities | 10.33 | 15.50 | 10,33 | 15,50 | 10.33 | 15.50 |
| Textiles & Trade | 11.28 | 22.98 | 11.28 | 22,98 | 11,28 | 22.98 |
| Services | 12.75 | 31.39 | 12.75 | 31,39 | 12,75 | 31.39 |
| Leisure | 12.64 | 28.62 | 12.64 | 28,62 | 12.64 | 28.62 |
| Risk-Free | 3.76 | 3.47 | 3.76 | 3.47 | 3.76 | 3.47 |
| Inflation | 4.09 | 3,80 | 4.09 | 3.80 | 4.09 | 3.80 |
| Power -75 | 4.75 | 4,54 | 4.41 | 4.21 | 4.25 | 4.03 |
| Power -50 | 5.22 | 5.71 | 4.72 | 5.08 | 4.49 | 4.72 |
| Power -30 | 5.92 | 7.72 | 5,29 | 7.10 | 4.93 | 6.44 |
| Power -15 | 6.90 | 10.50 | 5.59 | 9.11 | 5.02 | 8.20 |
| Power -10 | 7.37 | 12.39 | 6.05 | 10.76 | 5.40 | 9.87 |
| Power -5 | 8.66 | 14.88 | 6.43 | 12.35 | 5.35 | 11.79 |
| Power -2 | 10.81 | 19.51 | 7.73 | 14.47 | 5.46 | 12.65 |
| Power 0 | 13.01 | 28.85 | 10.45 | 19.41 | 6.04 | 13.80 |
| Power 0.5 | 13.69 | 38,70 | 12.17 | 24.11 | 6.43 | 14.20 |
| Power 1 . | 13.56 | 51.43 | 14.06 | 40.83 | 11.39 | 24.00 |
| Portfolio V2 | 5.47 | 4.24 | 5.47 | 4.24 | 5.47 | 4.24 |
| Portfolio V4 | 7.07 | 6,99 | 7.07 | 6,99 | 7.07 | 6.99 |
| Portfolio V6 | 8.56 | 10.24 | 8.56 | 10.24 | 8.56 | 10.24 |
| Portfolio V8 | 9.94 | 13.66 | 9.94 | 13.66 | 9.94 | 13.66 |
| Portfolio V10 | 11.21 | 17.16 | 11.21 | 17.16 | 11.21 | 17.16 |
| Portfolio V12 | 11.93 | 20.69 | 11.93 | 20.69 | 11.93 | 20.69 |
| Portfolio V14 | 12.65 | 24.15 | 12.65 | 24.15 | 12.65 | 24,15 |
| Portfolio V16 | 13.18 | 27.11 | 13.18 | 27.11 | 13.18 | 27.11 |
| Portfolio V18 | 13.57 | 30.00 | 13.57 | 30.00 | 13.57 | 30.00 |
| Portfolio V20 | 14.04 | 32.34 | 14.04 | 32.34 | 14.04 | 32.34 |

*Standard Deviation is for the variable $ln(1+r_t)$.

Table 12b

Geometric Means and Standard Deviations of Annual Returns using 12 Value-Weighted Industry Indices, with Leverage, 1966-86: Disaster States Scenario for J = 1, 6, and 12 (32 quarter estimating period, simple probability assessment)

| Portfolio G. Mean S. Dev G. Mean S. Dev G. Mean S. Dev Petroleum 10.93 21.21 10.93 21.21 10.93 21.21 Finance 6 Real Estate 10.68 18.75 10.68 18.75 10.68 18.75 Consumer Durables 8.17 23.83 8.17 23.83 8.17 23.83 Basic Industries 8.58 15.15 8.58 15.15 8.58 15.15 Food & Tobacco 12.96 17.39 12.96 17.39 12.96 17.39 Construction 7.99 22.18 7.99 22.18 7.99 22.18 Transportation 7.03 20.85 7.03 20.85 7.03 20.85 Transportation 7.03 20.85 7.03 20.85 7.03 20.85 Transportation 7.03 20.85 7.03 20.85 7.03 20.81 13.69 Services 10.67 28.10 10.67 28.10 1 | | D. State 1 | | D. State 6 | | D. State 12 | |
|---|-----------------------|---------------|--------------------|------------|--------------------|-------------|-------|
| Petroleum 10.93 21.21 10.93 21.21 10.93 21.21 Finance & Real Estate 10.68 18.75 10.68 18.75 10.68 18.75 Consumer Durables 8.17 23.83 8.17 23.83 8.17 23.83 8.17 23.83 Basic Industries 8.58 15.15 8.58 15.15 8.58 15.15 Food & Tobacco 12.96 17.39 12.96 17.39 12.96 17.39 Construction 7.99 22.18 7.99 22.18 7.99 22.054 7.92 20.54 7.92 20.54 7.92 20.54 7.92 20.54 7.92 20.54 7.92 20.54 7.92 20.54 7.92 20.54 7.92 20.54 7.92 20.54 7.92 20.54 7.92 20.54 7.92 20.54 7.92 20.54 7.92 20.54 7.92 20.54 7.92 20.54 7.92 20.54 7.92 20.54 7.92 | Portfolio | G.Mean | S.Dev [*] | G.Mean | S.Dev [*] | G.Mean | S.Dev |
| Finance & Real Estate 10.68 18.75 10.68 18.75 10.68 18.75 Consumer Durables 8.17 23.83 8.17 23.83 8.17 23.83 Basic Industries 8.58 15.15 8.58 15.15 8.58 15.15 Food & Tobacco 12.96 17.39 12.96 17.39 12.96 17.39 Canstruction 7.99 22.18 7.99 22.18 7.99 22.18 Capital Goods 7.92 20.54 7.92 20.54 7.92 20.54 Transportation 7.03 20.85 7.03 20.85 7.03 20.85 Utilities 9.68 13.69 9.68 13.69 9.68 13.69 9.68 13.69 Services 10.67 28.10 10.67 28.10 10.67 28.10 10.67 28.10 10.67 28.10 10.10.67 28.10 3.15 10.18 12.31 30.18 12.31 30.18 12.31 30.18 12.31 30.18 12.31 30.18 12.31 30.18 12.31 | Petroleum | 10.93 | 21.21 | 10.93 | 21.21 | 10.93 | 21.21 |
| Consumer Durables 8.17 23.83 8.17 23.83 8.17 23.83 8.17 23.83 Basic Industries 8.58 15.15 8.58 15.15 8.58 15.15 Food & Tobacco 12.96 17.39 12.96 17.39 12.96 17.39 Capital Goods 7.92 20.54 7.92 20.54 7.92 20.54 Transportation 7.03 20.85 7.03 20.85 7.03 20.85 Transportation 7.03 20.85 7.03 20.85 7.03 20.85 Services 10.67 28.10 10.67 28.10 10.67 28.10 10.67 28.10 30.18 12.31 30.18 12.31 30.18 12.31 30.18 12.31 30.18 12.31 30.18 12.31 30.18 12.31 30.18 12.31 30.18 12.31 30.18 12.31 30.18 12.31 30.18 12.31 30.18 12.31 30.18 12.31 1 | Finance & Real Estate | 10.68 | 18.75 | 10.68 | 18.75 | 10.68 | 18.75 |
| Basic Industries 8.58 15.15 8.58 15.15 8.58 15.15 Food & Tobacco 12.96 17.39 12.96 17.39 12.96 17.39 Construction 7.99 22.18 7.99 22.18 7.99 22.18 Capital Goods 7.92 20.54 7.92 20.54 7.92 20.54 Transportation 7.03 20.85 7.03 20.85 7.03 20.85 Utilities 9.68 13.69 9.68 13.69 9.68 13.69 Textiles & Trade 9.60 26.96 9.60 26.96 9.60 26.96 Services 10.67 28.10 10.67 28.10 10.67 28.10 Leisure 12.31 30.18 12.31 30.18 12.31 30.18 12.31 30.18 Risk-Free 7.49 2.71 7.49 2.70 7.50 2.70 Power -50 7.87 2.97 7.62 2.74 7.51 | Consumer Durables | 8.17 | 23.83 | 8.17 | 23.83 | 8.17 | 23.83 |
| Food & Tobacco 12.96 17.39 12.96 17.39 12.96 17.39 Construction 7.99 22.18 7.99 22.18 7.99 22.18 Capital Goods 7.92 20.54 7.92 20.54 7.92 20.54 Transportation 7.03 20.85 7.03 20.85 7.03 20.85 Utilities 9.68 13.69 9.68 13.69 9.68 13.69 Services 10.67 28.10 10.67 28.10 10.67 28.10 Leisure 12.31 30.18 12.31 30.18 12.31 30.18 12.31 30.18 Risk-Free 7.49 2.71 7.49 2.71 7.49 2.71 Inflation 6.12 3.15 6.12 3.15 6.12 3.15 Power -75 7.74 2.83 7.58 2.72 7.50 2.70 Power -10 9.15 5.87 8.08 3.20 7.57 2.70 Power -2 10.11 9.17 8.54 4.01 7.64 | Basic Industries | 8.58 | 15.15 | 8.58 | 15.15 | 8.58 | 15.15 |
| Construction 7.99 22.18 7.99 22.18 7.99 22.18 Capital Goods 7.92 20.54 7.92 20.54 7.92 20.54 Transportation 7.03 20.85 7.03 20.85 7.03 20.85 Utilities 9.68 13.69 9.68 13.69 9.68 13.69 9.60 26.96 Services 10.67 28.10 10.67 28.10 10.67 28.10 10.67 28.10 10.67 28.10 10.67 28.10 10.67 28.10 10.67 28.10 10.67 28.10 10.67 28.10 10.67 28.10 10.67 28.10 10.67 28.10 10.67 28.10 10.85 17.49 2.71 7.49 2.71 7.49 2.71 7.49 2.71 7.49 2.71 7.49 2.71 7.49 2.70 7.52 2.70 7.00 2.06 7.55 2.70 7.00 2.06 7.55 2.70 7.00 <t< td=""><td>Food & Tobacco</td><td>12.96</td><td>17.39</td><td>12.96</td><td>17.39</td><td>12.96</td><td>17.39</td></t<> | Food & Tobacco | 12.96 | 17.39 | 12.96 | 17.39 | 12.96 | 17.39 |
| Gapital Goods 7.92 20.54 7.92 20.54 7.92 20.54 Transportation 7.03 20.85 7.03 20.85 7.03 20.85 Utilities 9.68 13.69 9.68 13.69 9.68 13.69 Textiles & Trade 9.60 26.96 9.60 26.96 9.60 26.96 Services 10.67 28.10 10.67 28.10 10.67 28.10 Leisure 12.31 30.18 12.31 30.18 12.31 30.18 Risk-Free 7.49 2.71 7.49 2.71 7.49 2.71 Inflation 6.12 3.15 6.12 3.15 6.12 3.15 Power -50 7.87 2.97 7.62 2.74 7.51 2.70 Power -10 9.15 5.87 8.08 3.20 7.57 2.70 Power -5 10.11 9.17 8.54 4.01 7.64 2.76 Power -5 10.11 9.17 8.54 4.01 7.64 2.76 Power -5 </td <td>Construction</td> <td>7.99</td> <td>22.18</td> <td>7.99</td> <td>22.18</td> <td>7.99</td> <td>22.18</td> | Construction | 7.99 | 22.18 | 7.99 | 22.18 | 7.99 | 22.18 |
| Transportation 7.03 20.85 7.03 20.85 7.03 20.85 7.03 20.85 7.03 20.85 13.69 9.68 13.69 9.68 13.69 9.68 13.69 9.68 13.69 9.68 13.69 9.68 13.69 9.68 13.69 9.68 13.69 9.68 13.69 9.68 13.69 9.68 13.69 9.60 26.96 9.60 26.96 9.60 26.96 9.60 26.96 9.60 26.96 9.60 26.96 9.60 26.96 9.60 26.96 9.60 26.96 9.60 26.96 9.60 26.96 9.60 26.96 9.61 30.18 12.31 30.18 12.31 30.18 12.31 30.18 12.31 30.18 12.31 30.18 12.31 30.18 12.31 30.18 12.31 30.18 12.31 30.18 12.31 30.15 7.07 10.70 10.70 10.70 10.70 10.70 10.70 10.70 10.70 < | Capital Goods | 7.92 | 20.54 | 7.92 | 20.54 | 7.92 | 20.54 |
| Utilities 9.68 13.69 9.68 13.69 9.68 13.69 9.68 13.69 9.68 13.69 9.68 13.69 9.68 13.69 9.68 13.69 9.68 13.69 9.68 13.69 9.68 13.69 9.68 13.69 9.68 13.69 9.68 13.69 9.68 13.69 9.68 13.69 9.68 13.69 9.68 13.69 9.60 26.96 9.60 26.96 9.60 26.96 9.60 26.96 9.60 26.96 9.68 13.69 9.68 13.69 9.68 13.69 9.68 13.69 9.68 13.69 9.68 13.69 9.68 13.69 9.68 13.69 9.68 13.69 9.68 10.67 28.10 10.67 28.10 10.67 28.10 10.67 28.10 10.67 28.10 10.67 28.10 10.67 28.10 10.67 28.17 7.50 2.70 11.50 13.15 6.12 3.15 7.70 2.77 7.50 2.70 10 10.91 5.87 8.08 3.20 7.57 2 | Transportation | 7.03 | 20.85 | 7.03 | 20.85 | 7.03 | 20.85 |
| Textiles & Trade9.6026.969.6026.969.6026.96Services10.6728.1010.6728.1010.6728.10Leisure12.3130.1812.3130.1812.3130.18Risk-Free7.492.717.492.717.492.71Inflation6.123.156.123.156.123.15Power -757.742.837.582.727.502.70Power -507.872.977.622.747.512.70Power -158.654.547.902.967.552.70Power -109.155.878.083.207.572.70Power -510.119.178.544.017.642.76Power -211.2515.139.335.717.772.97Power 010.2923.2911.089.737.933.72Power 16.0435.0412.4114.777.845.01Portfolio V28.123.998.123.998.123.99Portfolio V48.626.878.626.878.626.87Portfolio V48.626.878.626.878.626.87Portfolio V48.626.878.626.878.626.87Portfolio V109.3816.849.3816.849.3816.84Portfolio V109.3816.849.3816.849.3816.84 | Utilities | 9.68 | 13.69 | 9.68 | 13.69 | 9.68 | 13.69 |
| Services 10.67 28.10 10.18 12.31 30.18 12.31 30.18 12.31 30.18 12.31 30.18 12.31 30.18 12.31 30.18 12.31 30.18 12.31 30.15 6.12 3.15 6.12 3.15 6.12 3.15 6.12 3.15 6.12 3.15 7.17 7.49 2.71 7.49 2.70 7.62 2.74 7.51 2.70 7.00 7.93 3.72 7.00 7.90 2.95 <t< td=""><td>Textiles & Trade</td><td>9.60</td><td>26.96</td><td>9.60</td><td>26.96</td><td>9.60</td><td>26.96</td></t<> | Textiles & Trade | 9.60 | 26.96 | 9.60 | 26.96 | 9.60 | 26.96 |
| Leisure12.3130.1812.3130.1812.3130.18Risk-Free7.492.717.492.717.492.71Inflation6.123.156.123.156.123.15Power -757.742.837.582.727.502.70Power -507.872.977.622.747.512.70Power -158.654.547.902.967.552.70Power -109.155.878.083.207.572.70Power -510.119.178.544.017.642.76Power -211.2515.139.335.717.772.97Power 010.2923.2911.089.737.933.72Power 16.0435.0412.4114.777.845.01Portfolio V28.123.998.123.998.123.99Portfolio V48.626.878.626.878.626.87Portfolio V48.626.878.626.878.626.87Portfolio V109.3816.849.3816.849.3816.84Portfolio V128.7720.398.7720.398.7720.39Portfolio V148.4323.798.4323.798.4323.79Portfolio V168.1226.878.1226.878.1226.87Portfolio V187.7429.787.7429.787.7429.78 </td <td>Services</td> <td>10.67</td> <td>28.10</td> <td>10.67</td> <td>28.10</td> <td>10.67</td> <td>28.10</td> | Services | 10.67 | 28.10 | 10.67 | 28.10 | 10.67 | 28.10 |
| Risk-Free 7.49 2.71 7.49 2.71 7.49 2.71 Inflation 6.12 3.15 6.12 3.15 6.12 3.15 Power -75 7.74 2.83 7.58 2.72 7.50 2.70 Power -50 7.87 2.97 7.62 2.74 7.51 2.70 Power -30 8.10 3.33 7.70 2.79 7.52 2.70 Power -15 8.65 4.54 7.90 2.96 7.55 2.70 Power -10 9.15 5.87 8.08 3.20 7.57 2.70 Power -2 11.25 15.13 9.33 5.71 7.77 2.97 Power 0 10.29 23.29 11.08 9.73 7.93 3.72 Power 0.5 8.20 29.33 11.29 11.09 7.40 3.29 Power 1 6.04 35.04 12.41 14.77 7.84 5.01 Portfolio V2 8.12 3.99 8.12 3.99 8.12 3.99 Portfolio V4 8.62 | Leisure | 12.31 | 30.18 | 12.31 | 30.18 | 12.31 | 30.18 |
| Inflation6.123.156.123.156.123.15Power -757.742.837.582.727.502.70Power -507.872.977.622.747.512.70Power -308.103.337.702.797.522.70Power -158.654.547.902.967.552.70Power -109.155.878.083.207.572.70Power -510.119.178.544.017.642.76Power -211.2515.139.335.717.772.97Power 010.2923.2911.089.737.933.72Power 0.58.2029.3311.2911.097.403.29Power 16.0435.0412.4114.777.845.01Portfolio V28.123.998.123.998.123.99Portfolio V48.626.878.626.878.626.87Portfolio V48.626.878.626.878.626.87Portfolio V109.3816.849.3816.849.3816.84Portfolio V128.7720.398.7720.398.7720.39Portfolio V148.4323.798.4323.798.4323.79Portfolio V168.1226.878.1226.878.1226.87Portfolio V187.7429.787.7429.787.7429.78 <td>Risk-Free</td> <td>7.49</td> <td>2.71</td> <td>7.49</td> <td>2.71</td> <td>7.49</td> <td>2.71</td> | Risk-Free | 7.49 | 2.71 | 7.49 | 2.71 | 7.49 | 2.71 |
| Power -75 7.74 2.83 7.58 2.72 7.50 2.70 Power -50 7.87 2.97 7.62 2.74 7.51 2.70 Power -30 8.10 3.33 7.70 2.79 7.52 2.70 Power -15 8.65 4.54 7.90 2.96 7.55 2.70 Power -15 8.65 4.54 7.90 2.96 7.55 2.70 Power -10 9.15 5.87 8.08 3.20 7.57 2.70 Power -5 10.11 9.17 8.54 4.01 7.64 2.76 Power -2 11.25 15.13 9.33 5.71 7.77 2.97 Power 0 10.29 23.29 11.08 9.73 7.93 3.72 Power 0 10.29 23.29 11.08 9.73 7.93 3.72 Power 1 6.04 35.04< | Inflation | 6.12 | 3.15 | 6.12 | 3.15 | 6.12 | 3.15 |
| Power - 50 7.87 2.97 7.62 2.74 7.51 2.70 Power - 30 8.10 3.33 7.70 2.79 7.52 2.70 Power - 15 8.65 4.54 7.90 2.96 7.55 2.70 Power -10 9.15 5.87 8.08 3.20 7.57 2.70 Power -5 10.11 9.17 8.54 4.01 7.64 2.76 Power -5 10.11 9.17 8.54 4.01 7.64 2.76 Power -2 11.25 15.13 9.33 5.71 7.77 2.97 Power 0 10.29 23.29 11.08 9.73 7.93 3.72 Power 0.5 8.20 29.33 11.29 11.09 7.40 3.29 Power 1 6.04 35.04 12.41 14.77 7.84 5.01 Portfolio V2 8.12 3.99 8.12 3.99 8.12 3.99 Portfolio V4 8.62 6.87 8.62 6.87 8.62 6.87 Portfolio V4 8.62 | Power 75 | , , , | 2 03 | 7 50 | 2 72 | 7 50 | 0 70 |
| Power-307.872.977.622.747.512.70Power-308.103.337.702.797.522.70Power-158.654.547.902.967.552.70Power-109.155.878.083.207.572.70Power-210.119.178.544.017.642.76Power-211.2515.139.335.717.772.97Power010.2923.2911.089.737.933.72Power0.58.2029.3311.2911.097.403.29Power16.0435.0412.4114.777.845.01PortfolioV28.123.998.123.998.123.99Portfolio V28.123.998.123.998.123.99Portfolio V48.626.878.626.878.626.87Portfolio V48.626.878.626.878.626.87Portfolio V48.6213.429.2613.429.2613.42Portfolio V109.3816.849.3816.849.3816.84Portfolio V128.7720.398.7720.398.7720.39Portfolio V148.4323.798.4323.798.4323.79Portfolio V168.1226.878.1226.878.1226.87Portfoli | Power - 50 | 7.74 | 2.03 | 7.50 | 2.12 | 7.50 | 2.70 |
| Power-158.103.337.702.797.322.70Power-109.155.878.083.207.572.70Power-510.119.178.544.017.642.76Power010.2923.2911.089.737.933.72Power010.2923.2911.089.737.933.72Power010.2923.2911.089.737.845.01Power0.58.2029.3311.2911.097.403.29Power16.0435.0412.4114.777.845.01PortfolioV28.123.998.123.998.123.99PortfolioV48.626.878.626.878.626.87PortfolioV48.626.878.626.878.626.87PortfolioV48.626.878.626.878.626.87PortfolioV39.2613.429.2613.429.2613.42PortfolioV109.3816.849.3816.849.3816.84PortfolioV128.7720.398.7720.398.7720.39PortfolioV148.4323.798.4323.798.4323.79PortfolioV168.1226.878.1226.878.1226.87PortfolioV168.1226.87 <th< td=""><td>Power - 30</td><td>9 10</td><td>2.37</td><td>7.02</td><td>2.74</td><td>7.51</td><td>2.70</td></th<> | Power - 30 | 9 10 | 2.37 | 7.02 | 2.74 | 7.51 | 2.70 |
| Power-109.155.878.083.207.532.70Power-510.119.178.544.017.642.76Power-211.2515.139.335.717.772.97Power010.2923.2911.089.737.933.72Power010.2923.2911.089.737.933.72Power010.2923.2911.089.737.933.72Power0.58.2029.3311.2911.097.403.29Power16.0435.0412.4114.777.845.01PortfolioV28.123.998.123.998.123.99PortfolioV48.626.878.626.878.626.87PortfolioV48.626.878.626.878.626.87PortfolioV48.626.878.626.878.626.87PortfolioV109.3816.849.3816.849.3816.84PortfolioV109.3816.849.3816.849.3816.84PortfolioV128.7720.398.7720.398.7720.39PortfolioV128.7720.398.7220.398.4323.79PortfolioV148.4323.798.4323.798.4323.79PortfolioV168.1226.87< | Power -15 | 0.10 | 5.55 | 7.70 | 2.79 | 7.52 | 2.70 |
| Power-510.119.178.544.017.642.76Power-211.2515.139.335.717.772.97Power010.2923.2911.089.737.933.72Power010.2923.2911.089.737.933.72Power0.58.2029.3311.2911.097.403.29Power16.0435.0412.4114.777.845.01PortfolioV28.123.998.123.998.123.99PortfolioV48.626.878.626.878.626.87PortfolioV48.626.878.626.878.626.87PortfolioV48.626.878.626.878.626.87PortfolioV48.626.878.626.878.626.87PortfolioV48.626.878.626.878.626.87PortfolioV109.3816.849.3816.849.3816.84PortfolioV109.3816.849.3816.849.3816.84PortfolioV128.7720.398.7720.398.7720.39PortfolioV148.4323.798.4323.798.4323.79PortfolioV168.1226.878.1226.878.1226.87PortfolioV168.1226.87< | Power -10 | 0.05 | 4.54 | 7.90 | 2.96 | 7.55 | 2.70 |
| Nover10.119.178.344.017.642.76Power010.2923.2911.089.737.933.72Power010.2923.2911.089.737.933.72Power0.58.2029.3311.2911.097.403.29Power16.0435.0412.4114.777.845.01PortfolioV28.123.998.123.998.123.99PortfolioV48.626.878.626.878.626.87PortfolioV69.0110.089.0110.089.0110.08PortfolioV89.2613.429.2613.429.2613.42PortfolioV109.3816.849.3816.849.3816.84PortfolioV128.7720.398.7720.398.7720.39PortfolioV148.4323.798.4323.798.4323.79PortfolioV168.1226.878.1226.878.1226.87PortfolioV187.7429.787.7429.787.7429.78PortfolioV207.1532.907.1532.907.1532.90 | Power -5 | 9.1J 10.11 | J.07 0 17 | 0.00 | 5.20 | 7.57 | 2.70 |
| Power010.2923.2911.089.335.717.772.97Power010.2923.2911.089.737.933.72Power0.58.2029.3311.2911.097.403.29Power16.0435.0412.4114.777.845.01PortfolioV28.123.998.123.998.123.99PortfolioV48.626.878.626.878.626.87PortfolioV69.0110.089.0110.089.0110.08PortfolioV89.2613.429.2613.429.2613.42PortfolioV109.3816.849.3816.849.3816.84PortfolioV128.7720.398.7720.398.7720.39PortfolioV148.4323.798.4323.798.4323.79PortfolioV168.1226.878.1226.878.1226.87PortfolioV187.7429.787.7429.787.7429.78PortfolioV187.1532.907.1532.907.1532.90 | Power -2 | 11 25 | 9.17 | 0.34 | 4.01 | 7.64 | 2.76 |
| Power10.2923.2911.089.737.933.72Power0.58.2029.3311.2911.097.403.29Power16.0435.0412.4114.777.845.01PortfolioV28.123.998.123.998.123.99PortfolioV48.626.878.626.878.626.87PortfolioV69.0110.089.0110.089.0110.08PortfolioV89.2613.429.2613.429.2613.42PortfolioV109.3816.849.3816.849.3816.84PortfolioV128.7720.398.7720.398.7720.39PortfolioV148.4323.798.4323.798.4323.79PortfolioV168.1226.878.1226.878.1226.87PortfolioV187.7429.787.7429.787.7429.78PortfolioV207.1532.907.1532.907.1532.90 | Power O | 10.20 | 13.13 | 9.33 | 5.71 | 7.77 | 2.97 |
| Nover13.2029.3311.2911.097.403.29Power16.0435.0412.4114.777.845.01PortfolioV28.123.998.123.998.123.99PortfolioV48.626.878.626.878.626.87PortfolioV69.0110.089.0110.089.0110.08PortfolioV89.2613.429.2613.429.2613.42PortfolioV109.3816.849.3816.849.3816.84PortfolioV128.7720.398.7720.398.7720.39PortfolioV148.4323.798.4323.798.4323.79PortfolioV168.1226.878.1226.878.1226.87PortfolioV187.7429.787.7429.787.7429.78PortfolioV207.1532.907.1532.907.1532.90 | Power 0 5 | 9 20 | 23.23 | 11.00 | 9.75 | 7.93 | 3.72 |
| PortfolioV28.123.998.123.998.123.99PortfolioV48.626.878.626.878.626.87PortfolioV69.0110.089.0110.089.0110.08PortfolioV89.2613.429.2613.429.2613.42PortfolioV109.3816.849.3816.849.3816.84PortfolioV128.7720.398.7720.398.7720.39PortfolioV148.4323.798.4323.798.4323.79PortfolioV168.1226.878.1226.878.1226.87PortfolioV187.7429.787.7429.787.7429.78PortfolioV207.1532.907.1532.907.1532.90 | Power 1 | 6.04 | 29.33 | 11.29 | 11.09 | 7.40 | 5.29 |
| PortfolioV28.123.998.123.998.123.99PortfolioV48.626.878.626.878.626.87PortfolioV69.0110.089.0110.089.0110.08PortfolioV89.2613.429.2613.429.2613.42PortfolioV109.3816.849.3816.849.3816.84PortfolioV128.7720.398.7720.398.7720.39PortfolioV148.4323.798.4323.798.4323.79PortfolioV168.1226.878.1226.878.1226.87PortfolioV187.7429.787.7429.787.7429.78PortfolioV207.1532.907.1532.907.1532.90 | I UWBI | 0.04 | 33.04 | 12.41 | 14.// | 7.84 | 5.01 |
| PortfolioV48.626.878.626.878.626.87PortfolioV69.0110.089.0110.089.0110.08PortfolioV89.2613.429.2613.429.2613.42PortfolioV109.3816.849.3816.849.3816.84PortfolioV128.7720.398.7720.398.7720.39PortfolioV148.4323.798.4323.798.4323.79PortfolioV168.1226.878.1226.878.1226.87PortfolioV187.7429.787.7429.787.7429.78PortfolioV207.1532.907.1532.907.1532.90 | Portfolio V2 | 8.12 | 3.99 | 8.12 | 3.99 | 8.12 | 3.99 |
| Portfolio V69.0110.089.0110.089.0110.08Portfolio V89.2613.429.2613.429.2613.42Portfolio V109.3816.849.3816.849.3816.84Portfolio V128.7720.398.7720.398.7720.39Portfolio V148.4323.798.4323.798.4323.79Portfolio V168.1226.878.1226.878.1226.87Portfolio V187.7429.787.7429.787.7429.78Portfolio V207.1532.907.1532.907.1532.90 | Portfolio V4 | 8.62 | 6.87 | 8.62 | 6.87 | 8.62 | 6.87 |
| Portfolio V89.2613.429.2613.429.2613.42Portfolio V109.3816.849.3816.849.3816.84Portfolio V128.7720.398.7720.398.7720.39Portfolio V148.4323.798.4323.798.4323.79Portfolio V168.1226.878.1226.878.1226.87Portfolio V187.7429.787.7429.787.7429.78Portfolio V207.1532.907.1532.907.1532.90 | Portfolio V6 | 9.01 | 10.08 | 9.01 | 10.08 | 9.01 | 10.08 |
| Portfolio V109.3816.849.3816.849.3816.84Portfolio V128.7720.398.7720.398.7720.39Portfolio V148.4323.798.4323.798.4323.79Portfolio V168.1226.878.1226.878.1226.87Portfolio V187.7429.787.7429.787.7429.78Portfolio V207.1532.907.1532.907.1532.90 | Portfolio V8 | 9.26 | 13.42 | 9.26 | 13.42 | 9.26 | 13.42 |
| Portfolio V128.7720.398.7720.398.7720.39Portfolio V148.4323.798.4323.798.4323.79Portfolio V168.1226.878.1226.878.1226.87Portfolio V187.7429.787.7429.787.7429.78Portfolio V207.1532.907.1532.907.1532.90 | Portfolio V10 | 9.38 | 16.84 | 9.38 | 16.84 | 9.38 | 16.84 |
| Portfolio V148.4323.798.4323.798.4323.79Portfolio V168.1226.878.1226.878.1226.87Portfolio V187.7429.787.7429.787.7429.78Portfolio V207.1532.907.1532.907.1532.90 | Portfolio V12 | 8.77 | 20.39 | 8.77 | 20.39 | 8.77 | 20.39 |
| Portfolio V168.1226.878.1226.878.1226.87Portfolio V187.7429.787.7429.787.7429.78Portfolio V207.1532.907.1532.907.1532.90 | Portfolio V14 | 8.43 | 23.79 | 8.43 | 23.79 | 8.43 | 23.79 |
| Portfolio V187.7429.787.7429.787.7429.78Portfolio V207.1532.907.1532.907.1532.90 | Portfolio V16 | 8.12 | 26.87 | 8.12 | 26.87 | 8.12 | 26.87 |
| Portfolio V20 7.15 32.90 7.15 32.90 7.15 32.90 | Portfolio V18 | 7.74 | 29.78 | 7.74 | 29.78 | 7.74 | 29.78 |
| | Portfolio V20 | 7.15 | 32.90 | 7.15 | 32.90 | 7.15 | 32.90 |

* Standard Deviation is for the variable $\ln(1+r_t)$.

Table 12c

Geometric Means and Standard Deviations of Annual Returns using 12 Value-Weighted Industry Indices, without Leverage, 1934-86: Disaster States Scenario for J = 1, 6, and 12 (32 quarter estimating period, simple probability assessment)

| | D. State 1 | | - D. State 6 | | D. State 12 | |
|-----------------------|------------|--------------------|--------------|--------------------|-------------|--------------------|
| Portfolio | G.Mean | S.Dev [*] | G.Mean | S.Dev [*] | G.Mean | S.Dev [*] |
| Petroleum | 12.20 | 18,96 | 12.20 | 18.96 | 12.20 | 18.96 |
| Finance & Real Estate | 12.08 | 20.29 | 12.08 | 20.29 | 12.08 | 20.29 |
| Consumer Durables | 11.82 | 23.87 | 11.82 | 23.87 | 11.82 | 23.87 |
| Basic Industries | 10.95 | 17.67 | 10.95 | 17.67 | 10.95 | 17.67 |
| Food & Tobacco | 11.44 | 16.17 | 11.44 | 16.17 | 11.44 | 16.17 |
| Construction | 9.61 | 21.91 | 9.61 | 21,91 | 9.61 | 21.91 |
| Capital Goods | 11.73 | 20.74 | 11,73 | 20.74 • | 11,73 | 20.74 |
| Transportation | 9.31 | 22.73 | 9.31 | 22.73 | 9.31 | 22.73 |
| Utilities | 10.33 | 15.50 | 10.33 | 15,50 | 10.33 | 15,50 |
| Textiles & Trade | 11.28 | 22.98 | 11.28 | 22.98 | 11.28 | 22.98 |
| Services | 12.75 | 31.39 | 12.75 | 31.39 | 12.75 | 31.39 |
| Leisure | 12.64 | 28.62 | 12.64 | 28,62 | 12.64 | 28.62 |
| Risk-Free | 3.76 | 3.47 | 3.76 | 3.47 | 3.76 | 3,47 |
| Inflation | 4.09 | 3.80 | 4.09 | 3.80 | 4.09 | 3.80 |
| Power -75 | 4.75 | 4.54 | 4.41 | 4.21 | 4.25 | 4.03 |
| Power -50 | 5.22 | 5.71 | 4.72 | 5.08 | 4,49 | 4.72 |
| Power -30 | 5.92 | 7.72 | 5.29 | 7.10 | 4.93 | 6.44 |
| Power -15 | 6.46 | 8.75 | 5.42 | 8.28 | 5,03 | 8.07 |
| Power -10 | 7.00 | 9.60 | 5.63 | 8.47 | 4,99 | 8.26 |
| Power -5 | 8,26 | 12.03 | 6.14 | 9.36 | 5.05 | 8.60 |
| Power -2 | 9.71 | 15.34 | 7.02 | 10.95 | 5.33 | 9.25 |
| Power O | 10.37 | 22.87 | 9.96 | 15.95 | 5.62 | 9.82 |
| Power 0.5 | 10.19 | 25.75 | 10.80 | 19.54 | 6.10 | 10.35 |
| Power 1 | 10.77 | 28.24 | 11,29 | 21.25 | 8.34 | 15.07 |
| Portfolio V2 | 5.47 | 4.24 | 5.47 | 4.24 | 5.47 | 4.24 |
| Portfolio V4 | 7.07 | 6.99 | 7.07 | 6.99 | 7.07 | 6.99 |
| Portfolio V6 | 8.56 | 10.24 | 8,56 | 10.24 | 8.56 | 10.24 |
| Portfolio V8 | 9.94 | 13.66 | 9.94 | 13.66 | 9.94 | 13.66 |
| Portfolio V10 | 11.21 | 17.16 | 11.21 | 17.16 | 11.21 | 17.16 |

*Standard Deviation is for the variable $ln(1+r_t)$.
Table 12d

Geometric Means and Standard Deviations of Annual Returns using 12 Value-Weighted Industry Indices, without Leverage, 1966-86: Disaster States Scenarion for J = 1, 6, and 12 (32 quarter estimating period, simple probability assessment)

| | D. S | tate 1 🔔 | D. St | tate 6 | D. Sta | ate 12 |
|-----------------------|--------|---------------------|---------------------|--------------------|--------|--------|
| Portfolio 🔮 | G.Mean | S.Dev [*] | .G.Mean | S.Dev [*] | G.Mean | S.Dev |
| Petroleum | 10.93 | 21.21 | 10.93 | 21.21 | 10.93 | 21.21 |
| Finance & Real Estate | 10.68 | 18.75 | 10.68 | 18.75 | 10.68 | 18.75 |
| Consumer Durables | 8.17 | 23.83 | 8.17 | 23.83 | 8.17 | 23.83 |
| Basic Industries | 8.58 | 15.15 | 8.58 | 15.15 | 8.58 | 15.15 |
| Food & Tobacco | 12.96 | 17.39 | 12.96 | 17.39 | 12.96 | 17.39 |
| Construction | 7.99 | 22.18 | 7.99 | 22.18 | 7.99 | 22.18 |
| Capital Goods | 7.92 | 20.54 ^{,4} | 7.92 | 20.54 | 7.92 | 20.54 |
| Transportation | 7.03 | 20.85 | 7.03 | 20.85 | 7.03 | 20.85 |
| Utilities | 9.68 | 13.69 | 9.68 | 13.69 | 9.68 | 13.69 |
| Textiles & Trade | 9.60 | -26.96 | .9.60 | 26.96 | | 26.96 |
| Services | 10.67 | 28.10 | 10.67 | 28.10 | 10.67 | 28.10 |
| Leisure | 12.31 | 30.18 | 12.31 | 30.18 | 12.31 | 30.18 |
| Risk-Free | 7.49 | 2.71 | 7.49 | 2.71 | 7.49 | 2.71 |
| Inflation | 6.12 | 3.15 | 6.12 | 3.15 | 6.12 | 3.15 |
| Power -75 | 7.74 | 2.83 | 7.58 | 2.72 | 7.50 | 2.70 |
| Power -50 | 7.87 | 2.97 | 7.62 | 2.74 | 7.51 | 2.70 |
| Power -30 | 8.10 | 3.33 | 7.70 | 2.79 | 7.52 | 2.70 |
| Power -15 | 8.65 | 4.54 | 7.90 | 2.96 | 7.55 | 2.70 |
| Power -10 | 9.15 | 5.87 | ₹8.08 | 3.20 | 7.57 | 2.70 |
| Power -5 | 10.11 | 9.17 | 8.54 | 4.01 | 7.64 | 2.76 |
| Power -2 | 10.45 | 14.09 | 9.31 | 5.66 | 7.77 | 2.97 |
| Power 0 . | 9.48 | 18.28 | 10.78 | 9.15 | 7.84 | 3.49 |
| Power 0.5 | 7.64 | 23.29 | 10.88 | 9.20 | 7.79 | 4.16 |
| Power 1 | 6.36 | 25.76 | 11.26 | 8.83 | 7.99 | 5.47 |
| Portfolio V2 | 8.12 | 3.99 | 8.12 | 3.99 | 8.12 | 3.99 |
| Portfolio V4 | 8.62 | 6.87 | 8,62 | 6.87 | 8,62 | 6.87 |
| Portfolio V6 | 9.01 | 10.08 | 9.01 | 10.08 | 9.01 | 10.08 |
| Portfolio V8 | 9.26 | 13.42 | 9.26 | 13.42 | 9.26 | 13.42 |
| Portfolio V10 | 9.38 | 16.84 | 9.38 | 16.84 | 9.38 | 16.84 |

* Standard Deviation is for the variable $ln(1+r_t)$.

Table 13a

.

| Period | 1 Rp | RL | B | PETR | FINA | COUR | BASI | FTOB | CONS | CAPG | TRAN | UTIL | TEXT | SERV | LEIS |
|--|---|--------------------------------------|----------------------------------|---------------------------------------|------------------------------|------|--|------------------------------|----------------------|------------------------------|------------------------------|---|------------------------------|--------------------------------------|------------------------------|
| 1934-1 1934-3 1934-3 1934-4 1935-1 1935-2 1935-3 | 1.08 -0.16 -0.28 0.45 -0.44 0.62 0.76 | 0.9 0.9 0.9 0.9 0.9 | 17 16 16 16 17 16 | | | | 0,03 0,03 0.02 0.02 0.03 0.03 0.01 | 0,01 0,02 0,01 0,03 | | | | | | | |
| 1935-4 1936-1 1936-2 1936-3 | 0.89 0.66 -0.30 0.66 | 0.9 0.9 0.9 0.9 | 7 6 6 7 | | | | 0.03 0.04 0.02 0.03 | 0.01 | | 0.02 | | | | | |
| 1936-4 1937-1 1937-2 1937-3 1937-3 | 1.03 0.69 -0.63 -0.62 -0.84 | 0.9 0.9 0.9 | 6 6 6 6 | | | | 0.03 0.04 0.04 0.04 | | | 0.01 | | | | | |
| 1938-1 1938-2 1938-3 1938-4 | -0.70 1.42 0.21 0.60 | 0.9 0.9 0.9 | 7 7 6 | | | | 0.03 0.02 0.03 0.02 | | | | | , | | 0.01 0.01 0.03 | |
| 1939-1 1939-2 1939-3 1939-4 1940-1 | -1.10 -0.08 2.47 -0.64 0.05 | 0.99 0.96 0.96 0.95 | 5 | 0.01 | | | 0.04 0.04 0.04 0.05 | 0.01 | | | • | | | 0,01 | |
| 1940-2 1940-3 1940-4 1941-1 | -1.68 0.87 0.46 -0.33 | 0.93 | | 0.01 | | | 0,06 0,04 0,07 0,02 0,02 | 0.04 | | 0.04 0.07 | | | | | |
| 1941-2 1941-3 1941-4 1942-1 1942-2 | 0.20 0.42 -1.42 0.46 0.10 | 0.91 0.92 0.91 0.92 0.93 | | | | | | | | 0.09 0.08 0.09 0.06 | | | | 0.02 | |
| 1942-3 1942-4 1943-1 1943-2 1943-2 | 0.80 1.61 4.85 2.85 | 0.92 0.91 0.90 0.89 | | 0.05 0.01 | | | | | | 0.08 | 0.04 | | | 0.02 0.05 0.06 | |
| 1943-4 1944-1 1944-2 1944-3 | 0,14 2,58 2,44 0,83 | 0.89 0.90 0.90 0.89 | | | | | | 0.01 | | | 0.03 0.02 0.02 0.01 | | | 0.08 0.08 0.08 0.09 0.09 | |
| 1944-4 1945-1 1945-2 1945-3 1945-4 | 0.63 0.74 1.67 1.40 4.31 | 0.88 0.91 0.88 0.85 0.80 | | | | | | 0.06 0.02 0.07 0.09 | | | 0.03 0.01 0.02 | | | 0.06 0.05 0.04 0.03 | |
| 1946-1 1946-2 1946-3 1946-4 | 2.64 2.13 -6.75 0.76 | 0.73 0.68 0.67 0.76 | | | | | | 0.17 0.09 0.11 0.17 | | | | | 0.09 0.23 0.22 0.04 | 0.02 | 0.04 |
| 1947-2 1947-3 1947-4 1940-1 | +1.84 0.76 0.69 0.62 | 0.72 0.74 0.79 0.74 | | 0.07 0.05 0.22 | | | | 0.25 0.22 0.11 0.02 | | | | | | 0.03 | 0.02 0.07 0.07 0.14 |
| 1948-2 1948-3 1948-4 1949-1 1949-2 | 6.68 -6.71 -0.63 0.33 -1.74 | 0.74 0.62 0.69 0.71 6.70 | | 0.25 0.33 0.23 0.23 0.21 | 0.01 | | | | | | 0.01 | | | | 0.01 0.05 0.08 0.05 |
| 1949-3 1949-4 1950-1 1950-2 1950-3 | 6.11 2.32 1.77 0.17 | 0.73 0.73 0.62 0.61 | | 0,07 0,10 0,12 0,14 | 0.17 0.10 | | | | 0, 10 | | | 0.07 | | 0.03 0.07 0.09 0.04 | 0.00 |
| 1950-4 1951-1 1951-2 1951-3 | 2.30 2.09 0.83 6.76 | 0.63 0.64 0.64 0.65 | | 0, 17 0, 21 0, 21 0, 26 | | | | 0.02 | 0.01 | | | 0,14 0,10 0,08 0,08 0,01 | | 0,07 0,10 0.07 0.07 0.03 | |
| 1951-4 1962-1 1952-2 1952-3 1952-4 | 0.69 4.14 0.21 -1.65 3.19 | 0.63 0.62 0.62 0.64 0.63 | | 0.26 0.27 0.28 0.29 0.25 | | | | | 0.11 0.10 0.09 | | | 0.07 | | 0.00 | |
| 1953-1 1953-2 1953-3 1953-4 | 1.39 -0.96 -1.80 2.63 | 0.63 0.64 0.69 0.69 | | 0.24 0.24 0.27 0.23 | | | | | | | | 0, 11 0, 13 0, 12 0, 04 0, 08 | | | |
| 1954-1 1954-2 1954-3 1954-4 1955-1 | 1.47 2.97 12.30 2.97 | 0.69 0.69 0.68 0.26 0.29 | | 0,22 0,24 0,22 0,10, 0,04 | 0.44 | | | | | | | 0.10 0.68 0.11 0.21 | | | |
| 1955-2 1965-3 1956-4 1956-1 | 3.86 -1.08 3.10 6.08 | 0.23 0.18 0.22 0.14 | | 0.07 | 0.45 0.68 0.48 0.29 | | | | | | | 0.24 0.24 0.29 0.66 | | | |
| 1956-3 1956-4 1967-1 1957-2 | -1.46 4.85 4.67 1.23 | 0. 15 | -0.08 -0,28 -0,30 | 0.05 0.29 0.30 0.28 | 0.20 | | | | 0.33 | | | 0.48 0.46 0.69 0.77 0.75 | | 0.10 0.22 | |
| 1957-3 1957-4 1958-1 1958-2 1958-3 | -9.92 -0.92 11.80 9.84 8.81 | | -0.41 -0.07 -0.07 -0.16 | 0.48 0.23 0.16 0.17 | | | | | | | | 0,82 0.67 0.65 0.66 | | 0.12 0.26 0.26 0.31 | |
| 1958-4 1959-1 1959-2 1959-3 | 16.02 9.77 -2.66 -1.62 | | -0.36 -0.11 -0.11 -0.11 | 0.09 | | | | | 0. 12 | | | 0.96 0.94 0.80 0.78 0.56 | | 0.32 0.32 0.31 0.33 0.43 | |

Portfolio Composition and Realized Returns for Power -15 Managing 12 Equal-Weighted Industries, with leverage, 1934-86 (32 quarter estimating period, simple probability assessment approach)



164

Table 13b

| Period | Ro | RI | A | PEIR | FINA | CDUR | BASI | FTOB | CONS | CAPG | TRAN | UTIL | TEXT | SERV | LEIS |
|--|--|--|---|--|------------------------------|------|--|--------------|--------------|----------------------|--|---------|------|--|------------------------------|
| 1934-1 1934-2 1934-3 1934-3 1935-1 1935-2 1935-3 1935-3 1936-3 1936-3 1936-3 1936-3 1936-3 1936-3 1936-3 1936-3 | 18.62 -5.68 -7.42 8.46 -9.95 10.89 15.08 17.28 17.28 12.39 -7.67 12.61 21.65 13.31 | 0.37 0.20 0.24 0.33 0.29 0.31 0.29 0.31 0.29 0.27 0.26 0.20 0.20 0.20 | | | | | 0.63 0.80 0.70 0.67 0.71 0.60 0.67 0.71 0.73 0.75 0.70 0.71 0.71 | 0.06 0.19 | | | | · · · · | | | |
| 1937-2 1937-4 1938-1 1938-3 1938-3 1938-3 1938-3 1939-1 1939-2 1939-4 1939-4 1940-1 1940-2 1940-3 1940-4 1941-1 1941-2 | -13.36 -11.90 -17.81 -14.63 26.73 2.76 10.61 -23.69 -1.64 52.13 -12.37 -26.29 17.16 6.79 -8.73 3.34 | 0.23 0.28 0.35 0.33 0.44 0.21 0.04 0.19 0.12 | -0.24 -0.27 -0.28 -0.44 -0.23 -0.64 | 0.03 0.02 0.02 | | 0.04 | 0.77 0.65 0.67 0.667 0.70 0.961 0.85 1.21 1.28 1.28 1.28 1.28 1.24 1.28 | | | 0.19 0.63 1.20 | | | | 0. 12 0.09 0.42 | |
| 1941-3 1941-4 1942-1 | 6.76 -20.50 6.98 0.66 | | -0,14 -0,33 -0,20 | | | | | | | 1.33 0.90 1.00 | | | | 0.30 | |
| 1942-2 1942-4 1942-4 1943-1 1943-2 1943-3 1943-3 1943-4 1944-1 1944-2 1944-2 1944-3 1944-3 1944-3 1945-1 1945-4 1945-3 1945-3 1946-2 1946-3 | 0.66 20.04 40.42 16.42 34.76 34.76 34.76 34.76 34.76 8.99 24.84 4.62 26.67 4.67 3.40 26.39 | | -0.09 -0.31 -0.64 -0.64 -0.62 -0.68 -0.42 -0.42 -0.42 -0.42 -0.24 -0.24 -0.23 -0.33 -0.33 | 0.75 0.02 | | | | 0.13 | | 1.09 | 0.68 0.65 0.38 0.37 0.28 0.42 0.42 0.42 0.75 0.65 | | | 0.18 0.64 0.84 1.07 1.20 1.29 1.29 1.30 1.11 0.67 0.61 0.66 0.46 0.34 | 0. 13 0.21 0.04 |
| 1946-4 1947-1 1947-2 1947-3 | -2.69 -1.99 -13.38 2.68 | | -0.33 | | | | · | | | | 0.61 0.67 0.63 | | | 0.03 | 0.45 0.46 0.77 0.70 |
| 1947-4 1948-1 1948-2 | -6.19 -2.74 23.33 | | -0.33 -0.33 -0.33 | 0.19 0.48 | | | | | | | 0.38 | | | 0.77 | 1,33 0,26 |
| 1948-3 1948-4 1949-1 1949-2 | -18.46 -6.04 0.81 -12.13 | | -0.33 -0.33 -0.33 -1.00 | 0.64 0.41 0.90 | 0.63 | | • | | | | 0.61 0.92 | | | | 0.18 0.57 |
| 1949-3 1949-4 1950-1 | 36.17 13.39 -1.61 | | -1.00 -1.00 -1.00 | | 1.68 0.77 0.57 | | | | | | | 0.02 | | 0.32 | |
| 1950-2 1950-3 1950-4 1951-1 1951-2 1951-3 1951-3 1952-1 1952-2 1952-3 1952-3 1952-3 1952-3 1953-2 | -1,13 31,13 -10,41 8,99 -2,60 24,24 0,87 16,28 -1,43 -10,36 11,02 3,28 -7,21 | , | -1.00 -1.00 -1.00 -0.33 -0.33 -0.33 -0.33 -0.33 -0.33 -0.33 -0.33 -0.33 -0.33 -0.33 | 0.33 0.13 0.76 1.33 1.33 1.33 1.33 1.33 1.33 1.33 2.00 | 0.00 | | | | i | | , | | | 1.67 1.87 1.24 0.85 | |
| 1953-3 1953-4 1954-1 | -18.12 13.00 36.79 6.76 | | -1.00 -1.00 -1.00 | 2.00 2.00 2.00 2.00 | | | | | | | | | | | |
| 1954-3 1954-4 1955-1 | 18, 10 32, 70 4, 73 | | -1.00 -1.00 -0.67 | 2.00 2.00 1.67 | | | | | | | | | | | |
| 1955-2 1955-3 1955-4 1956-1 1956-2 1956-3 1956-4 1957-1 | 9.95 -3.19 10.92 -2.23 -5.17 11.96 -5.83 | | -0.43 -0.43 -0.43 -0.43 -0.43 -0.43 -0.43 -0.43 | 0.48 0.71 0.68 0.35 1.43 1.43 | 0.95 0.72 1.43 0.75 | | | | 1.08 | | | | | | |
| 1957-2 1957-3 1957-4 1958-1 | 14.88 -18.99 -19.82 | | -0.43 -0.43 -0.43 -0.43 | 1,43 1,43 1,43 | | | | | | | | | | | |
| 1958-2 1958-3 1958-4 | 22.60 17.41 11.68 | | -1.00 -1.00 -0.43 | 1.26 0,49 | 0.76 1.61 1.43 | | | | 1 11 | | | | | | |
| 1959-1 1959-2 1959-3 | 5.42 2.10 -6.71 | | -0.11 -0.11 -0.11 | | | | | | 0.29 0.63 | | | | | 0.82 0.58 | |

. •

.

Portfolio Composition and Realized Returns for Power O Managing 12 Equal-Weighted Industries, with leverage, 1934-86 (32 quarter estimating period, simple probability assessment approach)

165

.

I

•

Table 13b (continued)

.

| Period | Rp | RL | 8 | PETR | FINA | COUR | BASI | FTOB | CONS | CAPG | TRAN | UTIL | TEXT | SERV | LEIS |
|--|--|----------------------|--|--------------------------------------|------|------|----------------------|------------------------------|------|------|--------------------------------------|--|----------------------|--|--|
| 1959-4 1960-1 1960-2 1960-3 1960-4 1961-1 | 3.80 -4,74 -6.02 -0.08 23.87 13.63 | | -0.11 -0.11 -0.11 -0.11 -0.43 -0.43 | | | | | | | | | | | 1.11 1.11 1.11 0.76 1.38 | 1.11 0.67 0.05 |
| 1961-2 1961-3 1961-4 1962-1 1962-2 1962-3 | 1.02 -4.40 9.36 -1.82 -33.88 3.68 | | +0,43 -0,43 -0,43 -0,43 -0,43 -0,43 | | | | | | × | | | •. | | 0.81 0.54 0.76 1.13 1.02 1.43 | 0.61 0.89 0.67 0.30 0.41 |
| 1962-4 1963-1 1963-2 1963-3 1963-4 | 16.62 9.07 4.01 -3.48 1.61 | | -1.00 -1.00 -1.00 -1.00 -1.00 | | | | | | | | | 0,20 0,74 1,04 1,02 | | 1.80 1.26 0.96 2.00 0.98 | |
| 1964-1 1964-2 1964-3 1964-4 1965-1 1965-2 | -0,17 5,12 8,79 4,19 6,62 -5,00 | | -0,43 -0,43 -0,43 -0,43 -0,43 -0,43 | | | | | | | | | 0.77 1.43 1.43 1.43 1.43 0.62 | 0.81 | 0.66 | |
| 1965-3 1965-4 1966-1 1966-2 1966-3 | 9,15 9,78 12,64 -7,70 -25,08 | | -0.43 -0.43 -0.43 -0.43 -0.43 | | | | | | | | 1.21 1.43 1.43 | 0.75 0.32 | 0,68 1,10 0,22 | | |
| 1966-4 1967-1 1967-2 1967-3 1967-4 1968-1 | 10.71 33.13 13.69 11.39 7.17 ~6.05 | | -0,43 -0,43 -0,43 -0,43 -0,43 -0,43 | 0.21 1.30 1.37 | | | | | | | 0.91 1.11 0.70 0.07 | | | | 0.62 0.32 0.73 1.15 0.13 0.05 |
| 1968-3 1968-4 1969-1 1969-2 1969-3 | 9.23 10.10 -9.28 -12.12 -13.39 | | -0.25 -0.25 -0.26 -0.25 -0.25 | 1.26 1.25 0.17 0.57 1.00 | | | | | | | 1.08 0.68 0.26 | | | | |
| 1969-4 1970-1 1970-2 1970-3 1970-4 1971-1 | -13.19 -2.64 -39.02 42.27 13.26 20.27 | | -0.25 -0.25 -0.23 -0.64 -0.64 | 0.83 0.83 1.47 1.64 | | | | | | | 0.26 0.94 1.25 0.40 0.07 | | | | |
| 1971-2 1971-3 1971-4 | -0.19 -6.18 4.27 | | -0.64 -0.64 -0.64 | 1.27 1.54 1.24 | | | | | | | 0.09 | | 0.21 | | 0.27 |
| 1972-2 1972-3 1972-4 | -2.27 4.49 15.48 | | -0.82 -0.82 -0.82 | 1.29 0.09 1.82 | | 0.82 | | | | | | | 0.17 | | 0.63 |
| 1973-1 1973-2 1973-3 1973-4 | -12,84 -12,23 24,25 9,61 | | -0.64 -0.64 -0.33 -0.09 | 1.64 1.64 1.33 1.09 | | | | | | | | | | | |
| 1974-1 1974-2 1974-3 | ~12.44 -18.11 -12.03 | 0.24 | -0.07 | 1.07 1.00 0.76 | | | | | | | | | | | |
| 1976-1 1976-2 1976-3 | 7.35 28.34 -10.80 | 0.09 | | 0.91 0.90 1.00 | | | | | | | | | | | |
| 1976-4 1976-1 1976-2 1976-3 | -2.65 13.42 5.78 3.06 | 0.33 0.29 0.04 | | 0.67 0.65 0.36 0.62 | | | 0.64 | 0. 16 | | | | | | | |
| 1976-4 1977-1 1977-2 | 7.07 -1.12 14.40 | 0,06 | | 0.73 0.85 0.83 | • | | 0.21 0.06 0.10 | | | | | 0.09 0.07 | | | |
| 1977-4 1977-4 1978-1 1978-2 | 3.27 -6.04 9.29 | | -0.17 -0.15 | 0.90 1.17 1.15 | | | | | : | | | 0.10 | | | |
| 1978-3 1978-4 1979-1 1979-2 | 22.12 -13.30 19.65 17.02 | | -0.83 -0.30 | 1.32 0.83 0.70 0.92 | | | | | | | 0.27 0.27 | | | | 0.24 0.20 0.30 0.08 |
| 1979-3 1979-4 1980-1 | 18.12 17.39 -2.93 | | -0.04 -0.20 | 1.01 1.20 1.00 | | | | | | | | | | | ŏ: ŏ3 |
| 1980-3 1980-4 1981-1 | 42.88 28.75 -7.43 | , | -0.88 | 1.69 1.00 | | | | | | | | | | | |
| 1981-2 1981-3 1981-4 1982-1 | -13.44 -13.78 13.90 -17.25 | | -0.48 -0.01 -0.52 | 1.35 0.37 0.48 | | | | | | | | | | 0,13 0.63 0.52 1.62 | |
| 1982-2 1982-3 1982-4 | 1.77 28.25 61.22 | | -0.02 -0.54 -1.00 | | | | | | | | | | | 1.02 1.54 2.00 | |
| 1983-2 1983-3 1983-4 | 20.21 -13.92 -10.74 | | -1.00 -1.00 -1.00 | | | | | | | | | | | 2.00 2.00 2.00 2.00 | |
| 984-1 984-2 984-3 | -21.92 1.86 15.25 | | -1.00 -1.00 -1.00 | | | | | | · . | | | | | 2.00 2.00 2.00 | 0.16 |
| 985-1 986-2 985-3 | 26.38 14.49 -12.90 | | -1.00 -1.00 -1.00 | | | | | | | | | | | 1.70 0.91 1.11 | 0.30 1.09 0.89 |
| 985-4 986-1 986-2 986-3 | 30,43 24,42 15,19 -16,18 | | -1.00 -1.00 -1.00 -1.00 | | | | | U.59 1.67 2.00 2.00 | | | | | | 1.41 0.33 | |
| 986-4 | 7.12 | | -1.00 | | | | | 2.00 | | | | | | | |

.

Table 14a

| Period | Rp | RL | B PETR | FINA | CDUR | BASI | FTOB | CONS | CAPG | TRAN | UTIL | TEXT | SERV | LEIS |
|--|---|--|--|--------------------------------------|------|--|--|---|--------------------------------------|--|--|----------------------|--|--|
| 1934-1 1934-2 1934-3 1934-4 1935-1 1935-3 1935-3 1935-3 1936-1 1936-2 1936-2 1936-2 1936-3 1936-4 1937-2 1937-2 | 1.08 -0.16 -0.28 -0.46 -0.44 0.62 0.76 0.66 -0.30 0.65 1.03 0.69 -0.63 -0.63 | 0.97 0.96 0.96 0.97 0.96 0.97 0.96 0.97 0.96 0.96 0.96 0.96 0.96 0.96 0.96 0.96 | | | | 0.03 0.03 0.02 0.03 0.01 0.03 0.03 0.03 0.04 0.03 0.04 0.04 0.04 | 0.01 0.02 0.01 0.03 0.01 | | 0.02 0.01 | | | | | |
| 1937-4 1938-2 1938-3 1938-4 1939-4 1939-2 1939-3 1939-4 1940-2 1940-3 1940-3 | -0,84 -0,70 1,42 0,21 0,60 -1,10 -0,08 2,47 -0,64 0,05 -1,68 0,45 | 0.97 0.97 0.96 0.95 0.95 0.95 0.96 0.96 0.93 0.93 0.93 0.93 0.93 | 0.01 0.01 0.01 | | | 0.03 0.03 0.02 0.03 0.02 0.04 0.04 0.04 0.05 0.04 0.05 0.04 0.07 0.07 0.02 | 0.01 0.04 | | 0.04 | | | | 0.01 0.01 0.03 0.01 | |
| 1941-1 1941-2 1941-3 1941-4 1942-1 | -0.33 0.20 0.42 -1.42 0.46 | 0.91 0.92 0.91 0.92 | | | | 0.01 | | | 0.09 0.08 0.09 0.06 0.07 | | | | 0.02 | |
| 1942-2 1942-4 1943-2 1943-2 1943-3 1943-3 1944-2 1944-2 1944-2 1944-2 1944-2 1946-3 1946-3 1946-4 | 0.80 1.61 4.85 2.85 1.29 0.14 2.58 2.44 0.83 0.74 1.40 4.31 2.64 | 0.93 0.91 0.99 0.89 0.89 0.90 0.90 0.90 0.90 0.91 0.88 0.91 0.85 0.85 0.80 | 0.05 0.01 | | | | 0.01 0.06 0.02 0.07 0.09 0.17 0.17 | • | 0.08 0.08 | 0.04 0.03 0.02 0.01 0.01 0.01 0.03 0.01 0.02 | | 0.09 | 0.02 0.05 0.06 0.08 0.09 0.09 0.09 0.05 0.05 0.05 0.04 0.03 0.03 0.02 | |
| 1946-2 1946-3 1946-4 1947-1 1947-2 1947-2 1947-3 1947-3 1948-1 1948-2 1948-3 1948-4 1948-4 | 2.13 -6.75 0.75 -0.69 -1.84 0.69 0.65 6.68 -6.71 -0.63 0.33 | 0.68 0.67 0.75 0.72 0.72 0.74 0.74 0.74 0.69 0.71 | 0.07 0.05 0.22 0.23 0.33 0.23 0.23 | 0.01 | | ħ | 0.09 0.11 0.17 0.25 0.22 0.11 0.02 | | | 0.01 | | 0.23 0.22 0.04 | 0.03 | 0.04 0.02 0.07 0.07 0.14 0.05 0.08 0.05 |
| 1949-2 1949-3 1949-4 1950-1 1950-2 1950-3 1950-4 1951-1 1951-2 1951-3 1951-4 1952-1 | -1,74 6,11 2,32 1,77 0,17 4,61 2,30 2,09 0,83 6,76 0,69 4,14 | 0.70 0.73 0.62 0.61 0.62 0.63 0.64 0.64 0.64 0.65 0.63 0.62 | 0.21 0.07 0.10 0.14 0.17 0.21 0.21 0.26 0.26 0.26 0.26 | 0. 17 0. 10 | | | 0.02 | 0. 10 0. 01 0. 02 0. 11 0. 10 | | | 0.07 0.21 0.14 0.10 0.08 0.08 0.01 | | 0.03 0.07 0.09 0.04 0.07 0.10 0.07 0.07 0.03 | 0.08 |
| 1952-2 1952-3 1952-4 1953-1 1953-3 1953-3 1953-4 1954-1 1954-3 1954-4 1955-1 | 0.21 -1.55 3.19 1.39 -0.96 -1.80 2.63 6.17 1.47 2.97 12.30 2.97 | 0.62 0.63 0.63 0.63 0.69 0.69 0.69 0.69 0.69 0.69 0.69 0.69 | 0.28 0.25 0.24 0.24 0.27 0.27 0.22 0.22 0.22 0.22 0.24 0.22 0.24 | 0.44 0.60 | | | | 0.09 | | | 0.07 0.11 0.13 0.12 0.04 0.08 0.10 0.08 0.11 0.21 0.07 | | | |
| 1955-2 1956-3 1955-4 1956-1 1956-2 1956-3 1956-3 1956-4 1957-1 1957-2 1957-3 1957-4 1957-4 | 3.86 -1.08 3.10 6.08 -0.77 -1.46 4.69 3.24 1.77 -7.86 -0.95 | 0.23 0.18 0.22 0.14 0.16 0.15 | 0.07 0.01 0.05 0.28 0.28 0.25 0.42 0.23 0.23 | 0,45 0,68 0,48 0,29 0,28 | | | | 0.06 0.33 | | | 0.24 0.29 0.66 0.48 0.46 0.60 0.66 0.66 0.66 0.65 0.61 | | 0.08 0.12 0.18 0.02 0.24 0.24 | |
| 1958-2 1958-3 1958-4 1959-1 1959-2 1959-3 | 8.84 6.42 12.09 8.89 -2.17 -1.44 | | 0, 16 0, 09 0, 06 | 0.03 | | | | 0. 12 | | | 0.68 0.70 0.69 0.73 0.70 0.49 | | 0.27 0.21 0.21 0.27 0.30 0.39 | |

Portfolio Composition and Realized Returns for Power -15 Managing 12 Equal-Weighted Industries, without leverage, 1934-86 (32 quarter estimating period, simple probability assessment approach)

| Per iod | Rp | RL | В | PETR | FINA | CDUR | BASI | FTOB | CONS | CAPG | TRAN | UTIL | TEXT | SERV | LEIS |
|---|---|---|---|--|------|------|------------------------------|--|------|------|--|--|----------------------|--|--|
| $\begin{array}{c} 1959-4\\ 1950-1\\ 1960-2\\ 1960-3\\ 1960-3\\ 1960-3\\ 1961-3\\ 1961-3\\ 1961-3\\ 1961-3\\ 1962-2\\ 1962-2\\ 1962-4\\ 1962-4\\ 1963-3\\ 1963-3\\ 1963-3\\ 1964-4\\ 1964-3\\ 1964-4\\ 1964-4\\ 1964-4\\ 1964-4\\ 1964-4\\ 1964-4\\ 1965-1\\$ | 4.03 -0.74 3.60 1.61 13.36 1.70 3.63 5.19.60 2.74 4.32 3.19.60 2.101 1.62 1.01 1.26 2.13 3.23 3.23 1.95 2.68 | 0.69 0.68 0.68 0.69 0.69 0.69 0.69 0.69 0.67 | | | | | | | | | • | $\begin{array}{c} 0.61\\ 0.60\\ 0.60\\ 0.61\\ 0.61\\ 0.61\\ 0.60\\ 0.60\\ 0.60\\ 0.50\\ 0.26\\ 0.35\\ 0.25\\ 0.25\\ 0.24\\ 0.35\\ 0.41\\ 0.43\\$ | | 0.37 0.35 0.29 0.39 0.30 0.34 0.34 0.34 0.34 0.34 0.08 0.15 0.08 0.06 | 0,02 0,10 0,05 0,14 0,13 0,14 0,13 0,15 0,13 0,16 0,14 0,11 |
| 1965-2 1965-3 1965-4 1966-1 1966-2 1966-3 1966-4 1966-4 1967-4 1967-3 1967-3 1967-3 1967-3 1968-2 1968-3 1968-4 1968-4 | -1,49 2,66 1,21 0,64 -0,84 -3,04 3,64 -0,84 -3,04 2,61 -0,96 3,86 4,16 4,16 4,16 4,16 4,16 4,16 4,16 4,1 | 0.67 0.666 0.62 0.72 0.72 0.75 0.72 0.75 0.67 0.65 0.65 0.65 0.63 | | 0.04 0.13 0.33 0.33 0.35 0.39 0.40 0.37 | | | | | | | 0.09 0.16 0.14 0.08 0.06 0.05 | 0,43 0,40 0,44 0,21 0,10 0,05 0,13 0,14 0,05 | 0.08 0.08 0.09 | | 0.01 0.01 0.03 |
| 1969-2 1969-3 1969-4 1970-1 1970-2 1970-3 1970-4 1971-2 1971-3 1971-4 1971-3 1971-4 1972-1 1972-2 | -0.69 -1.88 -1.04 0.93 -1.92 6.88 2.87 3.65 1.08 0.49 1.46 0.66 | 0.67 0.72 0.85 0.86 0.86 0.82 0.80 0.78 0.82 0.82 0.82 0.82 0.81 | | 0.33 0.28 0.25 0.13 0.09 0.11 0.18 0.20 0.22 0.18 0.18 0.18 0.18 0.18 | | | 0.03 | | | | 0.02 0.05 | | | | 0.01 |
| 1972-3 1972-4 1973-1 1973-2 1973-3 1973-4 1974-1 1974-2 1974-3 1974-4 1976-1 1976-2 | 1.98 2.68 -0.26 0.39 3.74 2.68 0.44 0.63 1.98 3.09 1.98 3.69 | 0.83 0.83 0.82 0.87 0.90 0.887 0.90 0.98 0.95 0.95 0.94 0.94 | | 0.13 0.17 0.18 0.13 0.10 0.12 0.11 0.08 0.05 0.06 0.06 0.06 | | 0.04 | | | | | | | | | |
| 1975-3 1975-4 1976-1 1976-2 1976-3 1976-3 1977-1 1977-1 1977-3 1977-4 1978-2 1978-2 1978-3 | 0.67 1.27 1.69 1.68 1.42 1.69 0.94 1.60 0.58 1.60 0.93 2.23 3.46 | 0.95 0.96 0.96 0.92 0.94 0.94 0.94 0.93 0.91 0.91 0.91 0.91 | | 0,04 0,03 0.03 0.04 0.04 0.05 0.05 0.05 0.05 0.07 0.06 0.09 0.09 0.09 | | | 0.05 0.02 0.01 0.01 | 0.01 | : | | 0.04 | 0.01 0.01 0.02 0.03 | | | |
| 1978-4 1979-1 1979-2 1979-3 1979-4 1980-1 1980-2 1980-3 1980-4 | 0.69 3.62 3.43 3.72 3.86 2.33 6.02 6.02 6.48 | 0.88 0.94 0.93 0.90 0.89 0.89 0.94 0.84 0.84 | | 0.08 0.04 0.07 0.10 0.11 0.11 0.06 0.16 0.13 | | | | | | | 0,04 | | | | 0.02 0.01 0.01 0.01 |
| 1981-1 1981-2 1981-3 1981-4 1982-1 1982-3 1982-3 1982-4 1983-1 | 2.61 1.90 1.40 4.61 1.10 3.32 6.28 8.49 6.62 | 0.89 0.87 0.93 0.88 0.91 0.88 0.91 0.88 0.69 0.69 0.74 | | 0.11 0.12 0.08 0.04 0.01 | | | | 0.06 | | | | 0.09 0.02 | | 0,02 0,03 0,11 0,09 0,12 0,16 0,22 | 0.01 0.03 |
| 1983-2 1983-3 1983-4 1984-1 1984-2 1984-3 1984-4 1986-1 1985-2 1985-3 1985-3 | 4.39 0.64 0.45 -0.84 2.33 4.08 6.46 4.69 1.02 | 0.76 0.75 0.65 0.72 0.76 0.76 0.76 0.65 0.65 0.60 | | 0.01 | | | | 0.03 0.01 0.06 0.04 0.04 0.24 0.24 0.27 0.39 | | | | 0.11 0.02 | | 0.24 0.21 0.24 0.26 0.17 0.17 0.12 0.09 0.08 0.01 | 0.03 0.08 |
| 1986-1 1986-2 1986-3 1986-4 | 6.80 4.73 -1.77 2.18 | 0.68 0.66 0.63 0.48 | | | | | | 0.42 0.44 0.39 0.20 | | | | 0,08 0.32 | | | |

Table 14b

•

| Period | Rp | RL | в В | PETR | FINA | CDUR | BASI | FTOB | CONS | CAPG | TRAN | UTIL | TEXT | SERV | LEIS |
|---|---|--|--------|--|------------------------------|--------------|--|----------------|----------------------|--------------------------------------|--|------|------|---|--|
| 1934 - 1 1934 - 2 1934 - 3 1934 - 3 1935 - 1 1935 - 2 1935 - 2 1935 - 3 1935 - 3 1935 - 4 1936 - 2 1936 - 3 1936 - 4 1937 - 1 1937 - 2 1937 - 3 | 18.62 -6.68 -7.42 -9.95 10.89 16.08 17.28 17.28 17.28 17.26 17.67 12.61 12.61 13.31 -13.36 -11.90 | 0.37 0.20 0.24 0.33 0.33 0.33 0.29 0.27 0.26 0.27 0.26 0.29 0.22 0.22 0.22 0.22 0.23 | | | | | 0.63 0.80 0.70 0.67 0.67 0.67 0.67 0.71 0.73 0.76 0.71 0.73 0.76 0.71 0.71 | 0. 06 0. 19 | | | | | | | |
| 1937-4 1638-1 1938-2 1938-3 1938-4 1939-1 1939-2 1939-3 1939-4 1940-1 1940-2 1940-3 | - 17.81 -14.63 26.73 2.76 10.61 -23.69 - 1.64 62.13 -9.86 -20.72 12.06 12.02 | 0.36 0.33 0.44 0.21 0.04 0.19 0.12 | | 0.03 | | 0.04 0.10 | 0.65 0.67 0.44 0.70 0.66 0.96 0.81 0.81 1.00 1.00 1.00 | | | | | , | | 0, 12 0,09 0,42 | |
| 1941-1 1941-2 1941-3 1941-4 1942-1 1942-2 | -6.66 1.63 6.11 -16.26 6.05 0.65 | | | | | 0.31 0.73 | 0.69 0.06 | | | 0.21 1.00 1.00 0.75 1.00 | | | | 0.25 | |
| $\begin{array}{c} 1942-3\\ 1942-4\\ 1943-1\\ 1943-3\\ 1943-4\\ 1944-1\\ 1944-1\\ 1944-1\\ 1944-3\\ 1944-3\\ 1944-3\\ 1946-1\\ 1946-3\\ 1946-3\\ 1946-3\\ 1946-3\\ 1946-3\\ 1946-3\\ 1946-3\\ 1946-4\\ 1946-3\\ 1946-1\\$ | 9.67 16.03 68.64 22.912 0.02 7.12 6.39 24.23 7.60 6.39 22.16 1.022 2.44 2.44 2.44 2.44 2.44 2.44 2.44 | | | 0.28 | | | | · | | 0.93 | 0.18 0.63 0.30 0.10 0.13 0.64 0.71 0.64 0.71 0.79 0.61 0.55 | | · | 0.07 0.66 0.37 0.63 0.70 0.90 0.87 0.46 0.66 0.22 0.29 0.28 0.228 0.34 0.03 | 0.21 0.04 0.49 0.46 |
| 1947-2 1947-3 1947-4 1948-1 1948-2 1948-3 1948-4 1949-3 1949-3 1949-3 1949-4 1960-1 1960-2 1950-3 | -10.08 3.18 -4.49 -2.77 16.04 -12.84 -6.28 1.67 -6.83 20.42 3.02 -1.91 0.94 16.83 | | | 0. 12 0. 19 0. 26 0. 06 0. 11 | 0.40 0.74 | | | | | | 0.65 0.63 0.61 0.61 0.68 0.94 0.28 | | | 0.68 0.21 0.10 0.26 1.00 1.00 1.00 | 0.44 0.37 1.00 0.27 0.06 0.11 |
| $\begin{array}{c} 1960-4\\ 1961-1\\ 1961-2\\ 1951-3\\ 1951-3\\ 1952-1\\ 1952-3\\ 1952-3\\ 1952-3\\ 1953-2\\ 1953-3\\ 1953-4\\ 1954-1\\ 1954-3\\ 1954-4\\ 1954-4\\ 1954-4\\ 19564-5\\ 19566-5\\ 19566-5\\ 1956656-5\\ 1956656-5\\ 1956656-5\\ 195665655555555555555555555555555555555$ | -6.20 4.11 18.38 12.42 -0.86 12.42 -0.86 12.42 -7.65 8.65 -7.65 -8.63 -8.63 -8.63 -9.65 -9.65 -9.65 -9.65 -9.65 -9.65 -9.65 -9.65 -9.65 -9.65 -9.65 -0.55 -0 | | | 0.24 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00 | | | | | | | 0.16 | | | 1.00 0.84 0.76 | |
| 1956-2 1956-3 1956-4 1956-1 1956-2 1956-3 1966-4 1957-1 1967-2 1957-3 1957-4 | 6.62 -1.70 9.98 7.62 0.76 -3.29 8.78 -3.67 10.83 -12.88 -13.46 | | | 1.00 0.67 0.80 0.78 0.31 1.00 1.00 1.00 | 0.43 0.20 1.00 0.22 | | | | 0.69 | | | | | | |
| 1968-2 1968-3 1968-4 1969-1 1969-2 1969-3 | 8.24 13.80 7.72 8.63 6.00 1.94 -6.03 | | | 1.00 1.00 0.79 | 0.21 | | | | 1.00 0.24 0.49 | | | | | 0.76 0.61 | |

Portfolio Composition and Realized Returns for Power O Managing 12 Equal-Weighted Industries, without leverage, 1934-86 (32 quarter estimating period, simple probability assessment approach)

169

•

| Period | Rp | RL | 8 | PETR | FINA | CDUR | BASI | F108 | CONS | CAPG | TRAN | UTIL | TEXT | SERV | LEIS |
|---|--|--|---|--|------|------|------------------------------|------------------------------|------|------|--|------------------------------|----------------------|--|--|
| 1959-4 1960-1 1960-3 1960-3 1960-4 1961-1 1961-2 1961-3 1961-4 1962-1 1962-3 1962-3 1963-3 1963-3 1963-4 1963-4 1963-4 | 3.66 -4.10 -4.36 0.07 16.94 9.27 1.10 -2.95 6.89 -0.72 -22.78 2.99 8.93 4.01 2.66 -1.05 -2.62 -0.42 4.00 | | | | | | | | | | | 0.15 0.12 0.39 1.00 | | 1.00 1.00 1.00 0.51 1.00 0.56 0.30 0.88 0.76 1.00 1.00 0.85 1.00 0.85 1.00 | 1.00 0.49 0.44 0.70 0.60 0.12 0.24 |
| 1964-3 1966-1 1965-2 1965-2 1965-3 1966-1 1966-2 1966-2 1966-3 1966-4 1967-3 1967-3 1967-3 1967-3 | 6.67 3.35 -3.21 7.80 9.00 -4.90 -4.90 -17.90 22.07 10.67 5.93 5.95 | | | 0.77 | | | | | | | 1.00 1.00 0.67 0.89 0.42 | 1.00 1.00 0.13 0.25 | 0.87 0.75 1.00 | | 0.33 0.11 0.68 1.00 0.23 0,15 |
| 1968-2 1968-3 1968-3 1969-1 1969-2 1969-2 1969-3 1969-4 1970-2 1970-2 1970-3 1970-3 1970-3 1970-3 | -3,043 20,76 8,40 -7,663 -11,04 -9,65 -10,40 -0,26 -30,77 34,39 10,333 10,333 13,78 -1,92 | | | 0.46 0.46 0.46 | | | | | | | 0.04 1.00 0.76 0.34 0.37 1.00 1.00 0.64 0.39 | | | | 0.65 |
| 1971 - 3 1971 - 4 1972 - 12 1972 - 3 1972 - 4 1972 - 4 1973 - 2 1973 - 4 1973 - 4 1973 - 4 1974 - 3 1974 - 4 1974 - 3 1974 - 4 1976 - 2 1976 - 4 1976 - 2 | -3.32 -3.329 10.268 -6.086 -7.702 -7.233 9.066 -11.41 -12.03 -28.340 -28.340 -2.655 13.42 | 0.24 0.39 0.09 0.10 0.33 0.29 | | 0.82 0.81 1.00 1.00 1.00 1.00 1.00 1.00 0.76 0.61 0.90 0.65 0.36 | | 0.74 | Q. 64 | 0. 16 | | | 0.33 | | 0.20 | | 0.06 0.64 1.00 0.26 0.19 |
| 1976-3 1976-4 1977-4 1977-2 1977-3 1977-4 1978-3 1978-3 1978-3 1978-4 1978-4 1979-1 1979-3 1979-3 1979-4 | 3.06 7.07 -1.12 14.40 -7.92 3.27 -8.37 12.62 -9.65 17.02 17.02 17.02 | 0,04 0,06 | | 0.62 0.73 0.85 0.83 1.00 0.90 1.00 1.00 1.00 0.67 0.61 0.70 0.92 0.98 1.00 | | | 0.34 0.21 0.06 0.10 | | | | 0, 11 | 0.09 0.07 0.10 | | | 0.33 0.28 0.30 0.08 0.02 |
| 1980 - t 1980 - 2 1980 - 3 1980 - 4 1981 - 1 1981 - 2 1981 - 4 1981 - 4 1982 - 1 1982 - 4 1982 - 4 1983 - 2 1983 - 4 1983 - 4 1984 - 4 1984 - 4 | -2.93 24.16 18.63 -7.66 13.90 -10.81 19.06 21.60 21.60 21.60 21.60 -9.62 -9.62 -9.62 -9.62 -9.62 -9.64 | | | 1.00 1.00 1.00 1.00 1.00 0.92 0.37 0.48 | | | | | | | | | | 0.083 0.622 1.000 1.000 1.000 1.000 1.000 1.000 1.000 | |
| 1986-1 1985-2 1986-3 1986-3 1986-4 1986-1 1986-2 1986-3 1986-4 | 13,83 8,33 -4,47 15,06 12,23 7,67 -7,05 4,64 | | | | | | | 0.26 0.77 1.00 1.00 | | | | | | 1.00 0.60 0.82 1.00 0.74 0.23 | 0, 40 0, 18 |

, at L

Table 14b (continued)

Table 15a

Portfolio Composition and Realized Returns for Power -15 Managing 12 Value-Weighted Industries, with leverage, 1934-86 (32 quarter estimating period, simple probability assessment approach)

171

·

Table 15a (continued)

| Period | Rp | AL | в | PETR | FINA | CDUR | BASI | FTOB | CONS | CAPG | IRAN | UTIL | TEXT | SERV | LEIS |
|----------------------------|-----------------------|----------------------|----------------|----------------------|------|--------------|------|-------|------|-------------------------|-------|----------------------|----------------------|-------------------------|--------------|
| 1959-4 1960-1 1960-2 | 6.67 0.64 6.29 | | -0.11 -0.11 | | | 0,08 0.09 | | 0.06 | | 0. 15 0. 16 | | 0.48 0.30 | | 0.34 0.33 | |
| 1960-3 1960-1 | -0.23 | | -0.11 -0.43 | | | | | 0.26 | | 0.21 | | 0.43 0.35 0.32 | | 0.32 0.30 0.43 | |
| 1961-1 1961-2 1961-3 | 0.41 | | -0.43 | | | | | 0.49 | | 0.17 | | 0.31 0.36 | | 0.46 0.43 | |
| 1961-4 1962-1 | 10.56 | | -0,43 | | | | | 0.68 | | 0.17 | | 0.44 | | 0.32 | |
| 1962-2 1962-3 | -24.35 | 0.61 | -0, 15 | 0.04 | | | | 0.22 | | õ. i3 | | 0.67 | | 0, 12 | |
| 1963-1 1963-2 | 2.43 | 0.62 | | 0.01 | | | | | | | | 0.37 | | | |
| 1963-3 1963-4 | 2.07 | 0.62 | | 0.01 | | | | | | | | 0.36 | | | |
| 1964-2 | 2.31 | 0.63 | | | | | | | | | | 0, 38 0, 37 | | | |
| 1964-4 1965-1 | 1.36 | 0.61 | | 6 | | | | | | | | 0.39 | 0.04 | | |
| 1965-3 1965-4 | 1.34 | 0.65 | | | | | | | | | | 0, 19 0, 14 | 0.17 | | |
| 1966-1 1966-2 | -0.23 | 0.65 | | | | | | | | | 0, 10 | 0.19 | 0.18 0.26 0.16 | | |
| 1966-4 1967-1 | 2.11 | 0.82 | | 0.02 | | | | | | 0.10 | 0,18 | | 0.13 | | |
| 1967-2 1967-3 1967-4 | 1.80 | 0.76 | | 0.09 | | | | | | 0.16 | 0.04 | | | | |
| 1968-1 1968-2 | -0.72 | 0.73 | | 0.14 | | | | | | 0, 13 0, 14 0, 04 | | | | | |
| 1968-3 1968-4 1969-1 | 4.49 2.92 | 0.62 | | 0.35 | | | | | | 0.03 | 0, 02 | | | | |
| 1969-2 1969-3 | -0.97 -1.31 | 0.67 | | 0.31 | | | | | | 0.02 | 0.05 | | | | |
| 1969-4 1970-1 1970-2 | 0,80 0,95 -1,61 | 0.81 0.90 0.88 | | 0.12 | | | | | | 0.07 | | | | | |
| 1970-3 1970-4 | 3.67 | 0.87 | | 0.02 | | | | | | 0.12 0.13 0.16 | | | | | |
| 1971-2 1971-3 | 0.73 1.33 | 0.80 | | 0.03 | | | | | | 0.13 | | | 0.04 0.09 | | |
| 1971-4 1972-1 | 2.07 | 0.81 | | | | | | | | 0,10 | | | 0.07 0.09 0.11 | | 0.04 |
| 1972-3 1972-4 | 0.24 | 0.82 | | | | | | | | 0.02 | | | 0.08 | | 0.08 |
| 1973-1 1973-2 | 0.37 | 0.84 | | | | | | | | 0.14 | | | | | 0.03 |
| 1973-4 1974-1 | 0.62 | 0.90 | | 0.03 | | | | | | 0.10 | | | | | 0. 02 |
| 1974-2 1974-3 | 1.68 | 0.99 | | | 0.01 | | | 0.01 | | 0.01 | | | | | |
| 1976-1 1975-2 | 1.65 | 0.99 | | 0.01 | | | | 0.02 | | | | | | | |
| 1976-3 1976-4 | 1.05 | 0.94 | | 0.05 0.03 | | | | 0.01 | | | | | | | |
| 1976-2 | 1.35 | 0.95 | | 0.02 | | | 0.06 | 0.05 | | | | | | | |
| 1976-4 1977-1 | 1.38 | 0.94 | | 0.05 | | | 0.01 | 0.01 | | | | | | | |
| 1977-3 | 0.76 | 0.92 | | 0.03 0.06 0.06 | | | | | | | | 0.03 | | | |
| 1978-1 1978-2 | 0.83 | 0.91 | | 0.09 | | | | | | | | 0.03 | | | |
| 1978-4 | 1.42 | 0.89 0.91 0.98 | | 0.09 0.06 0.01 | | | | | | | | | | | 0.03 |
| 1979-2 1979-3 | 2.63 | 0.97 | | 0.03 | | | | | | | | | | | 0.01 |
| 1980-1 | 3.23 | 0.94 | | 0.09 | | | | | | | | | | | |
| 1980-3 1980-4 | 3.96 | 0.83 | | 0.17 0.12 | | | | | | | | | | | |
| 1981-2 | 2.65 | 0.93 | | 0.07 0.05 | | | | | | | | | | 0.02 | |
| 1981-4 1982-1 | 3,80 | 0.98 | | | | | | | | | | | | 0.06 | |
| 1982-3 1982-4 | 5.04 8.62 | 0.92 | | | | | | | | | | | | 0.04 | |
| 1983-1 1983-2 | 5.49 3.71 | 0.77 0.78 | | | | | | | | | | 0.03 | | 0.22 | |
| 1983-4 1984-1 | -0.29 | 0.78 0.69 0.77 | | | | | | | | | | 0.07 | | 0.22 | |
| 1984-2 | 2.16 | 0.80 | | 0.04 | | | | | | | | | č |). 23). 16 | |
| 1986-1 1986-2 | 6.64 | 0.82 0.73 0.62 | | 0.01 | | | | 0. 12 | | | | | |), 18), 18), 16 | |
| 1986-3 1986-4 | -0.36 | 0.62 | | 0.07 | | | | 0.36 | | | | | 0 | . 12 | |
| 1986-2 1986-3 | 10.31 | 0.37 | | 0.10 | | | | 0.44 | | | | | | | |
| 1986-4 | 3.47 | 0.38 | | 0. 16 | | | | 0.35 | | | c | . 13 | | | |

...

Table 15b

| Period | Rp | RL | B | PETR | FINA | CDUR | BASI | FTOB | CONS | CAPG | TRAN | UTIL | TEXT | SERV | LEIS |
|--|---|--|--|---|-------|--|--|---|--|--|------|--------------|--|--|--|
| 1934-1 1934-2 1934-3 1934-3 1934-4 1935-1 1935-3 1935-3 1935-4 1936-1 1936-3 1936-3 1936-3 1936-3 1936-3 1936-3 1937-1 1937-2 | 6.26 -8.87 -1.25 7.29 -4.95 12.26 8.90 8.60 7.31 -1.98 4.24 4.24 4.07 0.63 -6.19 -4.84 | 0.49 0.32 0.09 0.10 0.30 0.37 0.38 0.37 0.38 0.37 0.42 0.37 0.45 0.65 | | | | 0.61 0.63 0.11 0.11 0.31 0.09 0.26 0.09 | | 0, 15 0, 80 0, 90 0, 75 0, 40 | 0.46 | 0.30 0.63 0.26 0.26 0.64 0.28 0.28 0.49 | | | | | |
| 1938-1 1938-2 1938-3 1938-4 1939-1 1939-2 1939-3 1939-4 1940-1 | - 10.88 16.87 6.77 6.29 - 19.01 0.70 6.89 - 0.80 - 1.49 | 0.66 0.73 0.48 0.39 0.21 0.38 0.31 | -0. 12 | 9. 17 | | 0.20 | | | 0.44 0.27 0.62 0.61 0.79 0.62 0.69 0.62 | | | , , | | | |
| 1940-3 1940-4 1941-1 1941-2 1941-3 1941-3 1942-1 1942-2 1942-3 | -20.50 19.07 -0.69 -14.67 6.86 4.91 -21.97 -14.70 1.77 9.10 | | -0.65 -0.33 -0.91 -0.84 -0.35 -0.76 -0.26 | | | 0.61 0.28 0.12 | 1.12 0.62 1.91 1.42 1.30 0.64 1.23 0.82 | | 0.69 | | | | 0.03 0.12 0.16 0.04 1.21 0.03 | 0.06 | |
| 1942-3 1942-3 1943-2 1943-3 1943-4 1944-3 1944-3 1944-3 1944-3 1944-3 1944-3 1944-3 1944-3 1944-1 1946-3 1946-3 1946-3 | 5.70 65.31 40.97 16.64 16.64 32.13 38.35 -1.72 20.16 22.76 22.76 22.76 22.76 22.76 22.76 22.76 22.76 22.76 22.76 22.76 22.76 22.72 20.16 | | -0.16 -0.64 -0.39 -0.43 -0.41 -0.41 -0.27 -0.26 -0.33 -0.33 | 0.06 0.21 0.21 | | 0.96 0.02 0.05 0.19 | | | | | 0.01 | 0.63 0.63 | | 0.19 0.82 1.00 1.27 1.18 1.34 1.34 1.31 1.27 1.26 1.33 1.27 1.26 1.33 1.27 0.84 | 0.04 0.23 0.16 |
| 1946-4 1947-2 1947-3 1947-3 1948-1 1948-3 1948-3 1948-3 1948-3 1948-4 1949-2 1949-2 1949-3 1949-4 1950-1 1950-3 | -4,96 -3,47 -10,23 -1,40 -6,98 -8,90 -13,47 -1,19 -13,47 -1,77 37,84 12,98 7,94 20,98 | | -0.33 -0.33 -0.33 -0.33 -0.33 -0.33 -0.33 -0.33 -1.00 -1.00 -1.00 -1.00 -1.00 | 0.05 | | | | | | | | | | 0.66 0.62 1.27 0.46 0.66 1.20 0.17 0.87 0.92 1.13 1.42 2.00 2.00 2.00 | 0.44 0.38 0.67 0.68 0.13 1.16 0.47 0.41 0.83 0.68 |
| $1950-1\\1951-2\\1951-3\\1951-3\\1952-1\\1952-2\\1952-2\\1952-2\\1952-4\\1953-2\\1953-2\\1953-2\\1953-2\\1953-4\\1955-4\\1955-4\\1955-2\\1955-2\\1955-2\\1955-3\\$ | $\begin{array}{c} -6, 06\\ 11, 67\\ -4, 82\\ 6, 43\\ 13, 80\\ -10, 885\\ -9, 28\\ -4, 63\\ -9, 28\\ -4, 63\\ -9, 28\\ -4, 63\\ -9, 28\\ -4, 63\\ -10, 08\\ 32, 69\\ 29, 46\\ 29, 46\\ 29, 46\\ 29, 46\\ 4, 72\\ 9, 46\\ 4, 72\\ 9, 46\\ 4, 72\\ 16, 60\\ 14, 14, 14\\ 14, 14, 14, 14\\ 14, 14, 14, 14\\ 14, 14, 14, 14\\ 14, 14, 14, 14\\ 14, 14, 14, 14\\ 14, 14, 14, 14, 14\\ 14, 14, 14, 14, 14, 14, 14, 14, 14, 14,$ | | -1.00 -0.33 -0.33 -0.33 -0.33 -0.33 -0.33 -0.33 -0.33 -0.33 -1.00 -1.00 -1.00 -1.00 -1.00 -1.00 -1.00 -1.00 -0.67 -0.63 | 0.13 0.84 1.02 1.33 1.33 1.33 1.33 1.68 2.00 2.00 2.00 2.00 2.00 2.00 2.00 2.0 | D. 42 | 0.88 1.09 0.76 1.43 | | | | | | | | 2.00 2.00 1.33 1.21 0.49 0.32 | |
| 1956-1 1956-2 1956-2 1956-3 1956-4 1957-2 1957-3 1957-3 1957-3 1957-4 1958-1 1958-3 1958-3 1958-3 1958-4 1959-1 1959-3 | 8.28 6.22 -4.70 1.32 1.52 -7.32 -7.32 -16.81 -13.85 20.37 11.53 20.38 20.38 20.38 20.38 -12.02 -6.39 | | -0.43 -0.43 -0.43 -0.43 -0.43 -0.43 -0.43 -0.43 -0.43 -0.43 -0.43 -0.43 -1.00 -0.43 -1.00 -0.43 -1.00 -0.11 -0.11 | 0.22 0.40 0.85 0.49 1.43 2.00 1.83 0.40 | | 1.43 1.43 1.21 1.43 1.43 1.03 0.68 0.48 0.94 | | | | 0. 17 1. 03 1. 11 0. 89 1. 11 | | | | 0.22 | |

Portfolio Composition and Realized Returns for Power O Managing 12 Value-Weighted Industries, with leverage, 1934-86 (32 quarter estimating period, simple probability assessment approach)

Table 15b (continued)



174

Table 16a

| Period | Rp | RL | 8 | PETR | FINA | CDUR | BASI | FT08 | CONS | CAPG | TRAN | UTIL | TEXT | SERV | LEIS |
|-------------------------------------|-------------------------|----------------------|---|----------------------|----------------------|----------------------------------|----------------------|----------------------|----------------------|----------------------|------|----------------------|----------------------|----------------------|-------------------------|
| 1934-1 1934-2 1934-3 | 0.62 -0.31 0.00 | 0.97 0.96 0.94 | • | | | 0.03 0.02 | | 0.02 | | | | | | | |
| 1934-4 1935-1 1935-2 | -0.21 -0.21 0.76 | 0.95 0.94 0.96 | | | | | | 0.05 0.05 0.03 | | 0.02 | | | | | |
| 1935-4 1936-1 1936-2 | 0.64 | 0.96 | | | | 0.01 | | | | 0.04 | | | | | |
| 1936-3 1936-4 1937-1 | 0.26 0.40 0.05 | 0.96 0.97 0.96 | | | | 0.01 | | | | 0.04 | | | | | |
| 1937-2 1937-3 1937-4 | -0.21 -0.20 -0.60 | 0.98 0.98 0.98 | | | | | | | 0.02 | 0,01 | | | | | |
| 1938-1 1938-2 1938-3 | -0.67 0.90 0.37 | 0.98 0.98 0.97 | | | | | | | 0.02 0.02 0.03 | | | 1 | | | |
| 1938-4 1939-1 1939-2 | 0,34 -0,99 0,04 | 0.97 0.96 0.97 | | | | | | | 0.03 0.04 0.03 | | | | | | |
| 1939-4 1940-1 1940-2 | -0.43 | 0.93 | | 0.04 0.01 | | 0.02 | | | 0.01 | | | | | | |
| 1940-3 1940-4 1941-1 | 1.04 0.19 -0.86 | 0.88 0.90 0.88 | | | | | 0.08 0.07 0.11 | 0.03 0.01 | 0.00 | | | | 0.01 0.03 | | |
| 1941-2 1941-3 1941-4 | 0.64 0.60 -1.78 | 0.88 0.89 0.86 | | | | | 0.09 0.09 0.04 | | | | | | 0.03 0.02 0.10 | | |
| 1942-1 1942-2 1942-3 | -1.07 0.18 0.68 | 0.90 0.93 0.93 | | | | 0.07 | 0.08 0.07 | | | | | | 0.02 | | |
| 1943-1 1943-2 1943-3 | 4.63 | 0.88 | | 0.01 | | 0.01 | | | | | | 0.06 0.04 | | 0.01 | |
| 1943-4 1944-1 1944-2 | 0.10 2.18 2.73 | 0.89 0.92 0.91 | | 0.03 0.01 | | 0.01 | | | | | | | | 0.08 | |
| 1944-3 1944-4 1945-1 | 0.98 -0.03 0.69 | 0.90 0.91 0.91 | | | | | | | | | | | | 0.10 0.09 0.09 | |
| 1946-2 1945-3 1945-4 | 1.48 1.79 2.60 | 0.91 0.89 0.85 | | | | | | 0.06 | | | 0.01 | 0. OZ | | 0,09 0,09 0,08 | |
| 1946-2 1946-3 1946-4 | 0.72 -6.28 -0.67 | 0.65 | | | | | | | | | 0.01 | 0.02 0.04 | 0.23 0.20 0.19 | 0.03 | 0.05 |
| 1947-1 1947-2 1947-3 | -0,73 -1,11 -0.68 | 0.77 0.79 0.77 | | 0.01 0.11 | | | | | | | 0.02 | 0.04 | 0.11 | 0.03 | 0.07 |
| 1947-4 1948-1 1948-2 | -0.90 -0.16 2.80 | 0.82 0.77 0.76 | | 0.13 0.13 | | | | | | | | | | 0.04 0.06 | 0.14 0.04 0.11 |
| 1948-3 1948-4 1949-1 | 0.26 | 0.64 | | 0.19 | | | | | | | | | | 0.03 0.02 0.03 | 0, 15 0, 16 0, 13 |
| 1949-3 1949-4 1950-1 | 4,31 2,06 | 0.70 | | 0.10 | 0.15 | | | | | | | | | 0.04 0.08 0.12 | 0,13 0,13 0,06 |
| 1950-2 1950-3 1950-4 | 0.83 5.74 2.63 | 0,65 0.60 0,60 | | 0.13 0.18 0.17 | 0.10 | | | 0.05 | | | | | 0.16 | 0.13 | |
| 195 1 - 1 195 1 - 2 195 1 - 3 | 2.34 1.17 5.19 | 0.69 0.68 0.60 | | 0.19 0.19 0.23 | | | 0.01 0.05 | | | | | | 0.08 | 0.16 0.14 0.12 | |
| 1951-4 1952-1 1952-2 | 1.68 3.39 0.62 | 0.65 | | 0,22 0.26 0.32 | 0.05 | | 0.16 0.14 0.09 | | | | | | | 0.05 | |
| 1952-4 1953-1 1953-2 | 3.42 | 0.68 | | 0.29 | 0.10 | | 0.02 | | | | | | | | |
| 1953-3 1953-4 1954-1 | -1.03 3.27 5.85 | 0.64 0.61 0.60 | | 0.26 0.29 0.29 | 0.05 | | 0.04 | | | | | 0.09 | | | |
| 1954-2 1954-3 1964-4 | 1.48 6.10 11.41 | 0.69 0.69 0.05 | | 0.30 0.26 0.12 | 0.31 | 0.01 0.17 | 0.03 | | | | | 0.11 0.11 0.34 | | 0.01 | |
| 955-2 955-3 | 6.89 0.41 | 0.14 | | 0.07 | 0.39 0.32 0.23 | 0.22 | 0.14 0.33 | | | | | 0,13 0,40 0,36 | | 0.06 0.05 0.08 | |
| 956-1 956-2 956-3 | 6.59 -2.32 -1.89 | | | 5.02 | 0.07 | 0, 10 0, 17 0, 14 0, 07 | 0.24 | | | | | 0.40 0.60 0.37 | | 0.08 0.09 0.21 | |
| 956-4 957-1 957-2 | 4.49 0.82 6.39 | | | 0.15 0.18 0.21 | | 0,22 0,26 0,19 | 0.06 | | 0, 12 | 0.07 0.18 0.13 | | 0.08 | | 0.14 0.30 0.32 | |
| 957-3 957-4 968-1 | -7.23 -4.43 6.41 | | | 0.22 0.18 0.18 | | 0.28 0.23 0.08 | | | | 0.25 | | 0.01 0.09 0.39 | | 0.23 0.43 0.36 | |
| 958-3 958-4 959-1 | 7.13 17.64 10.15 | | | 0.07 | | 0.03 | | | | 0.06 0.17 0.21 | | 0.38 0.41 0.46 | | 0.37 0.34 0.32 | |
| 969-2 969-3 | 1.43 | | | | | 0.03 | | | | 0.20 0.30 | | 0.45 0.43 0.24 | | 0.35 0.37 0.43 | |

.

Portfolio Composition and Realized Returns for Power -15 Managing 12 Value-Weighted Industries, without leverage, 1934-86 (32 quarter estimating period, simple probability assessment approach)

175

معہ ب

•

| Period | Rp | RL | 8 PEIR | FINA | CDUR | BASI | FTOB | CONS | CAPG | TRAN | UTIL | TEXT | SERV | LEIS |
|----------------------------|----------------|----------------------|--------------|------|--------------|------|--------------|------|----------------|--------------|----------------|--------------|--------------|------|
| 1959-4 1960-1 | 6.24 0.61 | | | | 0,08 0,10 | | 0, 17 | | 0, 16 0, 16 | | 0.43 0.25 | | 0.33 0.32 | |
| 1960-2 1960-3 | 4.71 | | | | | | 0.10 | | 0.21 | | 0.38 0.29 | | 0.31 | |
| 1960-4 1961-1 | 17.17 | | | | | | 0,30 | | 0.19 | | 0.13 | | 0.38 | |
| 1961-2 | 0,70 3,48 | | | | | | 0.28 | | 0.20 | | 0.15 | | 0.37 | |
| 1962-1 | -6.96 | | | | | | 0.53 | | 0.20 | | 0.53 | | 0.27 | |
| 1962-3 1962-4 | 2.49 | 0.61 | 0.04 0.01 | | | | 0.10 | | | | 0.34 | | | |
| 1963-1 1963-2 | 2,43 | 0,62 | 0.02 | | | | | | | | 0.38 | | | |
| 1963-3 | 2.07 | 0.62 | 0.01 | | | | | | | | 0.30 | | | |
| 1964-2 | 2.31 | 0.62 | | | | | | | | | 0.37 | | | |
| 1964-4 1965-1 | 1.36 | 0.61 | | | | | | | | | 0.35 | 0.04 0.11 | | |
| 1965-2 1965-3 | 0.49 | 0.64 | | | | | | | | | 0, 19 0, 14 | 0.17 | | |
| 1965-1 | -0.23 | 0.65 | | | | | | | | 0,10 | 0.19 | 0.26 | | |
| 1966-3 1966-4 | -3.60 | 0.69 | | | | | | | 0, 10 | 0.18 0.02 | | 0.13 | | |
| 1967 - 1 1967 - 2 | 4,67 | 0.82 | 0.02 | | | | | | 0.13 | 0.04 | | | | |
| 1967-3 | 3.13 | 0.76 | 0.11 0.16 | | | | | | 0.13 | | | | | |
| 1968-2 | 4.20 | 0.68 | 0.24 | | | | | | 0.08 | | | | | |
| 1968-4 1969-1 | 2,92 | 0.60 0.63 | 0.36 0.32 | | | | | | 0,02 | 0,02 | | | | |
| 1969-2 1969-3 | -0.97 -1.31 | 0.67 0.76 | 0.31 0.20 | | | | | | 0.02 0.05 | | | | | |
| 1969-4 1970-1 | 0.80 | 0.81 | 0, 12 | | | | | | 0.07 0,10 | | | | | |
| 1970-3 | 3.67 | 0.87 | 0.02 | | | | | | 0.13 | | | | | |
| 1971-1 1971-2 | 3.72 0.73 | 0.80 0.78 | 0.03 | | | | | | 0, 13 0, 12 | | | 0.04 0.09 | | |
| 1971-3 1971-4 | 1.33 | 0.82 | | | | | | | 0,10 | | | 0.07 | | |
| 1972-1 1972-2 | 2.65 | 0.80 | | | | | | | 0.04 0.02 | | | 0.11 | | 0.04 |
| 1972-4 1973-1 | 1.33 0.37 | 0.83 | | | | | | | 0.17 | | | | | 0.03 |
| 1973-2 1973-3 | 0.62 | 0.86 0.90 | | | | | | | 0.14 0.10 | | | | | |
| 1973-4 1974-1 | 0.62 | 0.93 | 0.03 | | | | | | 0.05 0.01 | | | | | 0.02 |
| 1974-3 | 1.67 | 0,99 | | 0.01 | | | 0.01 | | | | | | | |
| 1976 - 1 1975 - 2 | 1.65 | 0.99 0.97 | 0.01 | | | • | 0.03 | | | | | | | |
| 1976-3 1976-4 | 1.05 | 0.94 | 0.05 0.03 | | | | 0.01 | | | | | | | |
| 1976-2 | 1.35 | 0,92 | 0.02 | | | 0.06 | 0.05 | | | | | | | |
| 1976-4 1977-1 | 1,38 0,61 | 0.94 0.93 | 0.05 | | | 0.01 | 0.01 | | | | | | | |
| 1977-2 1977-3 | 1.65 0.76 | 0.94 0.92 | 0.03 0.06 | | | | | | | | 0.03 | | | |
| 1978-1 | 0.83 | 0.92 | 0.05 | | | | | | | | 0.03 | | | |
| 1978-3 | 3.11 | 0.89 | 0,09 | | | | | | | | | | | 0.03 |
| 1979-1 1979-2 | 2.69 | 0.98 0.97 | 0.01 0.03 | | | | | | | | | | | 0.01 |
| 1979-3 1979-4 | 3.13 2.98 | 0.93 | 0.07 0.09 | | | | | | | | | | | |
| 1980-2 | 4,12 | 0.94 | 0.06 | | | | | | | | | | | |
| 1980-4 1981-1 | 4.83 | 0.88 | 0.12 | | | | | | | | | | | |
| 1981-2 1981-3 | 2.65 | 0.93 0.94 | 0.05 | | | | | | | | | | 0.02 | |
| 1981-4 1982-1 1982-2 | 3.80 | 0.98 | | | | | | | | | | | 0,02 | |
| 1982-3 | 5.04 8.62 | 0,92 | | | | | | | | | 0.03 | | 0.04 | |
| 1983-1 1983-2 | 5.49 3.71 | 0.11 0.18 | | | | | | | | | 0,03 | | 0.22 | |
| 1983-3 1983-4 | -0.24 | 0.78 | | | | | | | | | 0.07 | | 0.22 0.25 | • |
| 1984-1 1984-2 1984-1 | 2.16 | 0.80 | 0.04 | | | | | | | | | | 0.23 0.16 | |
| 1984-4 1986-1 | 2.12 | 0.82 0.73 | 0.01 | | | | 0 12 | | | | | | 0.18 | |
| 1986-2 1986-3 | 4,28 | 0.62 | 0.04 0.07 | | | | 0.23 0.36 | | | | | | 0.12 | |
| 1985-4 1986-1 | 8.72 9.20 | 0.61 | 0.10 0.10 | | | | 0.38 0.44 | | | | | | 0.00 | |
| 986-3 | -4.27 | 0.37 0.36 0.38 | 0.12 0.13 | | | | 0.50 | | | | | | | |
| | | | V. 10 | | | | 0.33 | | | | U. 13 | | | |

176

.

.

Table 16b

| Period | Rp | RL | B | PETR | FINA | CDUR | BASI | FTOB | CONS | CAPG | TRAN | UTIL | TEXT | SERV | LEIS |
|--|---|--|------|--|------|--|--------------------------------------|---|--|--|------|----------------|------|--|--|
| 1934-1 1934-2 1934-3 1934-4 1935-2 1935-3 1935-4 1935-4 1936-3 1936-3 1936-3 1936-3 1936-3 1937-1 1937-2 1937-3 | 6.26 -8.82 -1.29 -4.95 12:26 8.90 8.60 7.31 -1.98 4.24 6.07 0.63 -6.19 -4.84 | 0.49 0.32 0.09 0.10 0.31 0.37 0.38 0.42 0.37 0.46 0.66 0.66 0.66 | . s. | | | 0.61 0.53 0.11 0.11 0.31 0.09 0.26 0.09 | | 0, 15 0, 80 0, 90 0, 75 0, 40 | 0, 45 0, 41 | 0.30 0.63 0.62 0.26 0.68 0.64 0.28 0.49 | | | | | |
| 1938-1 1938-2 1938-3 1938-4 1939-1 1939-2 1939-3 1939-4 1940-1 1940-2 1940-3 | - 10. 88 16. 87 6. 29 - 19. 01 0. 70 6. 89 - 0. 80 - 1. 28 - 24. 47 16. 20 | 0.66 0.73 0.48 0.39 0.21 0.38 0.31 | | 0. 17 | | 0.20 | | | 0.34 0.27 0.62 0.61 0.79 0.62 0.69 0.62 1.00 | | | • | | | |
| 1940-4 1941-1 1941-2 1941-3 1941-4 1942-1 1942-2 1942-3 1942-4 1943-1 1943-2 | -3,46 -9,03 -2,63 -16,28 -11,69 -11,69 -11,77 9,10 13,74 64,19 30,61 | | | • | | 0.44 1.00 0.60 0.12 1.00 0.83 0.21 0.04 | 0.05 0.03 1.00 1.00 0.82 | | 0.95 0.63 | | | 0.01 | 0.50 | 0.06 0.17 0.77 0.96 | |
| 1943-3 1943-4 1944-1 1944-2 1944-2 1944-3 1946-1 1946-2 1946-3 1945-4 1946-1 | 13.85 1.26 27.37 28.81 9.21 -1.19 6.91 16.08 17.19 20.21 6.18 | | | | | | | | | | | х ^с | · | 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00 | |
| 1946-3 1946-3 1946-4 1947-1 1947-2 1947-3 1947-4 1948-1 1948-2 1948-3 1948-4 | -23.12 -4.96 -3.47 -7.61 -4.68 -7.98 -0.03 -10.16 -1.44 | | | | | | | | | | | | | 0.77 0.84 0.66 0.62 1.00 0.47 0.66 1.00 0.18 0.18 0.49 | 0.23 0.16 0.44 0.38 0.53 0.34 0.82 0.82 0.51 |
| 1949-1 1949-2 1949-3 1949-4 1960-1 1950-2 1950-3 1950-4 1951-1 1961-2 1951-3 | 1.49 7.49 26.82 4.30 0.49 10.82 -2.20 6.21 -3.73 | | | • | | | | | | | | | | 0.87 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.0 | 0.13 |
| 1951-4 1952-1 1952-3 1952-3 1952-4 1953-1 1953-3 1963-3 1953-4 1954-1 1954-2 | 4, 18 10, 63 0, 81 -7, 92 -3, 17 -3, 40 -4, 61 8, 22 16, 82 2, 85 | | | 0.69 0.76 1.00 1.00 1.00 1.00 1.00 1.00 1.00 | | | | | | | | | | 0.24 | |
| 1954-3 1954-4 1955-1 1955-2 1955-3 1956-4 1956-1 1956-3 1956-3 1956-4 1956-4 | 15.23 12.89 3.22 10.34 10.64 6.14 4.00 -3.11 1.31 1.47 -4.84 | | | 1.00 0.77 0.35 0.67 0.09 | | 0.23 0.64 0.33 1.00 1.00 0.91 1.00 0.91 | | | | | | | | | |
| 957-2 957-3 958-1 958-1 958-2 958-3 958-3 958-3 958-4 959-1 959-2 959-3 | 11.89 -11.66 -9.26 4.15 10.76 6.72 13.73 3.06 11.61 -6.62 | | | 0.65 0.76 0.33 1.00 1.00 1.00 0.34 | | 0.35 0.24 0.67 | | | | 0.66 1.00 0.85 1.00 | | | | 0, 15 | |

Portfolio Composition and Realized Returns for Power O Managing 12 Value-Weighted Industries, without leverage, 1934-86 (32 quarter estimating period, simple probability assessment approach)

SERV 0.02 TEXT LEIS FTOB TRAN UTIL FINA COUR HASI CONS CAPG 0.98 1.00 1.00 0.99 0.18 0.80 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00 0.897 1.00 0.897 1.00 0.64 $\begin{array}{c} R_{0} \\ \hline R_{0} \\$ PETR RL B 0.01 0.82 0.20 0.11 0.03 0.46 1.00 1.00 1.00 1.00 1.00 0.80 0.09 $\begin{array}{c} 0.20\\ 0.91\\ 1.00\\ 1.00\\ 1.00\\ 1.00\\ 1.00\\ 1.00\\ 1.00\\ 1.00\\ 1.00\\ 1.00\\ 1.00\\ 1.00\\ 1.00\\ 1.00\\ 1.00\\ 1.00\\ 1.00\\ 1.00\\ 0.03\\ \end{array}$ 0.16 0.22 0.29 0.76 0.36 0.11 0,60 0.30 0.97 1.00 1.00 1.00 1.00 1.00 1.00 1.00 0.62 0.18 0.38 0.54 0.28 0.76 0.87 1.00 0.87 0.60 0.69 0.23 0.24 0. 13 0.13 0.60 0.12 $\begin{array}{c} 0.83\\ 0.41\\ 0.07\\ 0.89\\ 0.83\\ 0.83\\ 0.83\\ 0.83\\ 0.83\\ 0.80\\ 0.85\\ 0.65\\ 1.009\\ 0.65\\ 1.009\\ 0.65\\ 1.009\\ 0.65\\ 1.009\\ 0.65\\ 1.000\\ 0.65\\ 1.000\\ 0.73\\ 1.000\\$ 0.71 0.79 0.11 0.16 0.16 0.01 0.01 0.64 0.17 0.20 0.41 0.35 0.20 0,62 0.46 0.01 $1980-1 \\ 1980-2 \\ 1980-3 \\ 1980-4 \\ 1980-3 \\ 1980-4 \\ 1980-3 \\ 1980-4 \\ 1980-3 \\ 1980-4 \\ 1980-3 \\ 1980-3 \\ 1981-3 \\ 1982-2 \\ 1982-3 \\ 1$ $\begin{array}{c} 0.27\\ 0.89\\ 0.25\\ 1.00\\ 1.00\\ 1.00\\ 1.00\\ 1.00\\ 1.00\\ 1.00\\ 1.00\\ 1.00\\ 1.00\\ 1.00\\ 1.00\\ 1.00\\ 1.00\\ 1.00\\ 0.81 \end{array}$ 0.11 0.76 0.29 0.19 1.00 1.00



178

Table 17a

| Number of | | | | 2 | Powe | ers | | | 1 | |
|-------------------|------------|-------|--------|------------|----------|----------------|-------|---------|--------|-------|
| Times in: | - 75 | - 50 | - 30 | -15 | -10 | - 5 | - 2 | . 0 | . 5 | 1 |
| Petroleum | 103 | 104 | 112 | 111 | 109 | 109 | 108 | 100 | 84 | 60 |
| Finance & R.E. | 9 | 9 | 9 | - 10 | 12 | 15 | 18 | 13 | 6 | 1 |
| Consumer Durables | _ 1 | 1 | 1 | 1 | 1. | ₆ 1 | 1 | 2 | 5 | 7 |
| Basic Industries | 23 | 29 | 33 | 34 | 34 | 35 | 35 | 35 | 32 | 5 |
| Food & Tobacco | 27 | 31 | 37 | 38 | 38 | 37 | 27 | 9 | 5 | 1 |
| Construction | 7 | 8 | 8 | 9 | 10 | 12 | 11 | 4 | 6 | 23 |
| Capital Goods | 10 | 10 | 10 | 11 | 11 | 11 | 10 | 9 | 6 | 3 |
| Transportation | 13 | 16 | 18 | 20 | 20 | 23 | 31 | 42 | 43 | 42 |
| Utilities | 72 | 76 | 77 | 78 | 78 | 75 | 40 | 16 | 8 | 4 |
| Textiles & Trade | 7 | 7 | 7 | 7 | 7 | 8 | 11 | 6 | 5 | 10 |
| Services | 68 | 72 | 74 | 76 | 76 | 75 | 76 | 69 | 66 | 44 |
| Leisure | 23 | 25 | 28 | 33 | 33 | 34 | 36 | 37 | 30 | 19 |
| Lending | 212 | 212 | 207 | 189 | 181 | 156 | 94 | 30 | | |
| Borrowing | • • • | | | 23 | 26 | 35 | 104 | 163 | 198 | 206 |
| Neither | ••• | ••• | 5 | ••• | 5 | 21 | 14 | 19 | 14 | 6 |
| Max. Lent (%) | 100 | 99 | 98 | .97 | 96 | 92 | 83 | 44 | | • • • |
| Min. Lent(%) | 52 | 28 | 6 | 14 | 24 | 1 | 1 | 4 | • • •, | ••• |
| Loss Quarters | 21 | 28 | 36 | 44 | 51 | 63 | 71 | 81 | 83 🐔 | 83 |
| In O Industries | 3 | | | | م ••• | | | • • • | | |
| In 1 Industry | 101 | 92 | 74 | 65 | 65 | 65 | 74 | 101 | 136 | 205 |
| In 2 Industries | 63 | 67 | 78 | 84 | 83 | 81 | 93 | 92 | 71 | 7 |
| In 3 Industries | 44 | 51 | -57 | ~58 | 59 | 57 | 37 | 19 | 3 | |
| In 4 Industries | 1 | 1 | 2 | 4 | 4 | 8 | 7 | • • • | 1 | |
| In 5 Industries | | 1 | 1 | 1 | 1 | 1 | 1 | • • • | 1 | |
| In 6 Industries | | | | | | | • • • | • • • | | |
| In 7 Industries | • • • | • • • | • •/ • | | | | | · • • • | | • • • |
| In 8 Industries | | | • • • | | | | | • • • | • • • | • • • |
| In 9 Industries | | | | • • • | · · · . | • • • | ••• | | • • • | |
| In 10 Industries | ••• | | | • • • | | • • • | • • • | | • • • | |
| In 11 Industries | • • • | • • • | | | | | | • • • | • • • | |
| In 12 Industries | | | | | | | | | | |

Summary of Portfolio Compositions of Power Policies Managing 12 Equal-Weighted Industries, With Leverage, 1934-86 Quarter estimating period simple probability assessment approach)

| Number of | * | | | - | Powe | ers | | | | |
|-------------------|-------|-------|------------------|-------|-------|-----------------|-------|-------|------------|-------|
| Times in: | -75 | - 50 | - 30 | -15 | -10 | - 5 | - 2 | 0 | . 5 | 1 |
| Petroleum | 103 | 104 | 112 | 111 | 109 | 109 | 109 | 88 | 71 | 60 |
| Finance & R.E. | 9 | 9 | 9 | 11 | _ 14 | 18 | 17 | 8 | 3 | 1 |
| Consumer Durables | 1 | 1 | 1 | 1 | 1 | 1 | 1 | 5 | 15 | 7 |
| Basic Industries | 23 | 29 | 33 | 34 | 34 | 35 | 35 | - 35 | 23 | 4 |
| Food & Tobacco | 27 | 31 | 37 | 38 | 38 | 37 | 25 | 7 | 4 | 1 |
| Construction | 7 | 8 | 8 | 9 | 11 | 14 | 7 | 4 | 15 | 24 |
| Capital Goods | 10 | 10 | 10 | 11 | 11 | 11 | 10 | 7 | 6 | 3 |
| Transportation | 13 | 16 | 18 | 20 | 20 | 23 | 35 | 43 | 45 | 42 |
| Utilities | 72 | 76 | 77 | 78 | 78 | 66 | 25 | 12 | 4 | 4 |
| Textiles & Trade | 7 | 7 | 7 | 7 | 7 | 8 | 11 | 5 | 4 | 4 |
| Services | 68 | 72 | 74 | 76 | 74 | 75 | 73 | 71 | 59 | 43 |
| Leisure | 23 | 25 | 28 | 33 | 33 | 34 | 38 | 37 | 23 | 19 |
| Lending | 212 | 212 | 207 | 189 | 181 | 156 | 94 | 30 | | |
| Borrowing | • • • | | | | | • • • | • • • | | | |
| Neither | • • • | ••• | <mark>، 5</mark> | 23 | 31 | 56 | 118 | 182 | 212 | 212 |
| Max. Lent (%) | 100 | 99 | 98 | 97 | 96 | 92 | 83 | 44 | , • • • | |
| Min. Lent(%) | - 52 | 28 | 6 | 14 | 24 | 1 | 1 | 4 | • • • | |
| Loss Quarters | 21 | 28 | 36 | 44 | 50 | [`] 65 | 71 | 79 | 81 | 84 |
| In 0 Industries | 3 | · • • | • • • | | | • • • | | | ••• | ••• |
| In 1 Industry | 101 | 92 | 74 | 65 | ÷ 65 | 68 | 82 | 118 | 154 | 212 |
| In 2 Industries | 63 | 67 | 78 | 84 | 84 | 81 | 91 | 81 | 56 | |
| In 3 Industries | 44 | 51 | 57 | 57 | 56 | 52 | 34 | 11 | 2 | |
| In 4 Industries | 1 | 1 | 2 | 5 | 6 | 10 | 5 | 1 | • • • • | |
| In 5 Industries | • • • | 1 | 1 | 1 | 1 | 1 | | 1 | • • • | |
| In 6 Industries | • • • | | | | • • • | | • • • | | | |
| In 7 Industries | • • • | | | • • • | • • • | . \ . | • • • | • • • | | |
| In 8 Industries | • • • | | | | • • • | ••• | | • • • | | |
| In 9 Industries | | | | | | | | • • • | | |
| In 10 Industries | • • • | | | | | | | • • • | | • • • |
| In 11 Industries | • • • | | | | | • • • | | | | |
| In 12 Industries | | | | | | | | | | |

Summary of Portfolio Compositions of Power Policies Managing 12 Found-Weighted Industries Without Leverage 1934-86

.

| | Summary | of Portfo | lio Compos | sitions (| of Power | Policies | Managi | ing 12 |
|-----|---------|------------|------------|-----------|-----------|-----------|---------|-----------|
| | ĪV | Value-Weig | nted Indu | stries, N | With Leve | erage, 19 | 34 - 86 | , i |
| (32 | quarter | estimatin | g period, | simple | probabil | ity asses | sment | approach) |

| Number of | t) | | | | Pow | ers | • | | | |
|-------------------|-------|-------|-----------|-------------|-------|--------------|-------|-------|-------|----------|
| Times in: | -75 · | - 50 | - 30 | -15 | -10 | - 5 | - 2 | 0 | . 5 | 1 |
| Petroleum. | 83 | 88 | 97 | 100 | 102 | 105 | 100 | 63 | 51 | 40 |
| Finance & R.E. | 11 | 11 | 12 | 13 | 13 | 15 | 14 | 2 | 1 | 1 |
| Consumer Durables | . 15 | 17 | 19 | 26 | 29 | 32 | 32 | 30 | 34 | 28 |
| Basic Industries | 25 ่ | 28 | 29 | 30 | 28 | 23 | 14 | 10 | 8 | |
| Food & Tobacco | 29 | 30 | 33 | 34 | 32 | 28 | . 22 | 15 | 8 | 3 |
| Construction | 7 | 11 | 12 | 14 | 15 | 15 | · 13 | 14 | 18 | 16 |
| Capital Goods | 56 | 58 | 60 | 59 * | 59 | 56 | 58 | 62 | 54 | 38 |
| Transportation | 8 | 11 | 12 | 13 | 13 | 13 | 13 | 5 | 2 | 1 |
| Utilities | 58 | 60 | 61 | 61 | 56 | 36 | 23 | 14 | 8 | 9 |
| Textiles & Trade | 29 | 31 | 32 | 32 | 32 | 32 | 27 | 17 | 8 | . 7 |
| Services | 76 | 78 | 80 | 83 | 83 | 80 | 76 | 65 | 61 | 56 |
| Leisure | 20 | 22 | 23 | 23 | 23 | 24 | 24 | 28 | 22 | 21 |
| Lending | 212 | 212 | 195 | 183 | 181 | 150 | 103 | 41 | 12 | 1 |
| Borrowing | | | | 23 | 28 | 41 | 95 | 149 | 171 | 195 |
| Neither | ••• | ••• | 17 | 6 | 3 | 21 | 14 | 22 | 29 | 16 |
| Max. Lent (%) | 100 | 100 | 100 | 100 | 100 | 100 | 100 | 100 | 100 | 100 |
| Min. Lent(%) | 42 | 13 | - 1 | 5 | 6 | 1 | 1 | 1 | 3 | 100 |
| Loss Quarters | 24 | 30 | 37 | 46 | 53 | 63 | 72 | 78 | 78 | 72 |
| In 0 Industries | 11 - | 7 | 3 | 1 | 1 | - 1 | 1 | 1 | 1 | 1 |
| In l Industry | 84 | 73 | 68 | 65 | 61 | 60 | 66 | 111 | 156 | 202 |
| In 2 Industries | 57 | 63 | 67 | 69 | 71 | 77 | 96 | 88 | 46 | 9 |
| In 3 Industries | 34 | 43 | 41 | 40 | 47 | 55 | 38 | 10 | 9 | • • • |
| In 4 Industries | 16 | 16 | 23 | 25 | 22 | 15 | 11 | 2 | | • • • |
| In 5 Industries | 7 | 7 | 7 | 8 | 8 | 4 | • • • | • • • | • • • | |
| In 6 Industries | 3 | 3 | 3 | 3 | 1 | | • • • | | • • • | |
| In 7 Industries | • • • | • • • | • • • | 1 | 1 | ••• | • • • | | • • • | • • • |
| In 8 Industries | • • • | | • • • | | • • • | r | | • • • | • • • | • • • |
| In 9 Industries | • • • | | | • • • | • • • | | | • • • | | • • • |
| In 10 Industries | • • • | • • • | • • • | • • • | ••• | | | | | |
| In 11 Industries | • • • | | • • • . | • • • | | | | ••• | | |
| In 12 Industries | | | • • • | | | | | | | |

181

,

| Times in: -75 -50 -30 -15 -10 -5 -2 0 ,5 1 Petroleum 83 88 -97 100 102 106 92 57 48 40 Finance & R.E. 11 11 12 13 15 14 9 1 1 1 Consumer Durables 15 17 19 26 30 32 31 30 22 Food & Tobacco 29 30 33 33 32 24 22 13 8 3 Construction 7 11 12 14 15 14 13 15 17 16 Capital Goods 56 58 60 59 57 54 63 58 52 38 Textles Tarasportation 8 11 12 13 13 12 2 12 14 15 14 12 14 15 14 15 16 16 16 16 16 16 | Number of | | | | | Powe | ers | | 1 | | |
|--|-------------------|-------|-------|-------|----------|-------------|-------|-------|---------------|------------|-------|
| Petroleum 83 88 97 100 102 106 92 57 48 400 Finance & R.E. 11 11 12 13 15 14 9 1 1 1 Consumer Durables 15 17 19 26 30 32 31 30 26 Basic Industries 25 28 29 26 23 11 7 3 Food & Tobacco 29 30 33 33 32 24 22 13 8 3 Construction 7 11 12 14 15 14 13 15 17 16 Capital Goods 56 58 60 61 59 46 30 20 7 1 Transportation 8 11 12 13 13 12 32 32 32 32 24 28 24 21 21 Leisure 20 22 23 23 23 <th>Times in:</th> <th>-75</th> <th>- 50</th> <th>- 30</th> <th>-15</th> <th>-10</th> <th>- 5</th> <th>-2</th> <th>0</th> <th><u>,</u> 5</th> <th>1</th> | Times in: | -75 | - 50 | - 30 | -15 | -10 | - 5 | -2 | 0 | <u>,</u> 5 | 1 |
| Finance & R.E. 11 11 12 13 15 14 9 1 1 1 Consumer Durables 15 17 19 26 30 32 31 30 28 Basic Industries 25 28 29 26 23 11 7 3 Food & Tobacco 29 30 33 33 32 24 22 13 8 3 Construction 7 11 12 14 15 14 13 15 17 16 Capital Goods 56 58 60 59 57 54 63 58 52 36 Transportation 8 11 12 13 13 12 3 2 1 17 16 14 13 15 <td>Petroleum</td> <td>83</td> <td>88</td> <td> 97</td> <td>100</td> <td>102 。</td> <td>106</td> <td>92</td> <td>57</td> <td>48</td> <td>40</td> | Petroleum | 83 | 88 | 97 | 100 | 102 。 | 106 | 92 | 57 | 48 | 40 |
| Consumer Durables 15 17 19 26 30 32 32 31 30 28 Basic Industries 25 28 29 29 26 23 11 7 3 Food & Tobacco 29 30 33 33 32 24 22 13 8 3 Construction 7 11 12 14 15 14 13 15 17 16 Capital Goods 56 58 60 59 57 54 63 58 52 38 Transportation 8 11 12 13 13 12 3 21 2 1 Textiles & Trade 29 31 32 32 32 32 24 28 24 21 21 Leisure 20 22 23 23 23 23 24 28 24 21 21 Borrowing <td< td=""><td>Finance & R.E.</td><td>11</td><td>11</td><td>12</td><td>13</td><td>15</td><td>14</td><td>9</td><td>1</td><td>1</td><td>÷ 1</td></td<> | Finance & R.E. | 11 | 11 | 12 | 13 | 15 | 14 | 9 | 1 | 1 | ÷ 1 |
| Basic Industries 25 28 29 29 26 23 11 7 73 Food & Tobacco 29 30 33 33 32 24 22 13 8 3 Construction 7 11 12 14 15 14 13 15 17 16 Capital Goods 56 58 60 59 57 54 63 58 52 38 Transportation 8 11 12 13 13 12 3 2 1 Textiles & Trade 29 31 32 32 32 32 26 7 6 4 60 56 Services 76 78 80 83 83 77 76 64 60 56 Leisure 20 22 23 23 24 28 24 21 21 Lending 212 195 183 181 150 103 41 12 1 </td <td>Consumer Durables</td> <td>15</td> <td>17</td> <td>19</td> <td>26</td> <td>30</td> <td>32</td> <td>32</td> <td>31</td> <td>- 3Q</td> <td>28</td> | Consumer Durables | 15 | 17 | 19 | 26 | 30 | 32 | 32 | 31 | - 3Q | 28 |
| Food & Tobacco 29 30 33 33 32 24 22 13 8 33 Construction 7 11 12 14 15 14 13 15 17 16 Capital Goods 56 58 60 59 57 54 63 58 52 38 Transportation 8 11 12 13 13 13 12 3 2 1 Utilities 58 60 61 59 46 30 20 7 1 Textles & Trade 29 31 32 32 32 32 26 7 6 7 Services 76 78 80 83 83 77 76 64 60 56 Lesure 20 22 23 23 24 28 24 21 21 Borrowing <t< td=""><td>Basic Industries</td><td>25</td><td>. 28</td><td>29</td><td>29</td><td>26</td><td>23</td><td>11</td><td>7</td><td>3</td><td></td></t<> | Basic Industries | 25 | . 28 | 29 | 29 | 26 | 23 | 11 | 7 | 3 | |
| Construction 7 11 12 14 15 14 13 15 17 16 Capital Goods 56 58 60 59 57 54 63 58 52 38 Transportation 8 11 12 13 13 14 3 2 1 Textiles 58 60 61 59 46 30 20 7 1 Textiles 6 Trade 29 31 32 32 32 32 26 7 6 76 Services 76 78 80 83 83 77 76 64 60 56 Leisure 20 22 23 23 24 28 24 21 21 Borrowing 17 29 31 62 109 <td>Food & Tobacco</td> <td>29</td> <td>30</td> <td>33</td> <td>33</td> <td>ີ 32</td> <td>24</td> <td>22</td> <td>13</td> <td>8</td> <td>3</td> | Food & Tobacco | 29 | 30 | 33 | 33 | ີ 32 | 24 | 22 | 13 | 8 | 3 |
| Capital Goods 56 58 60 59 57 54 63 58 52 38 Transportation 8 11 12 13 13 12 3 2 1 Utilities 58 60 61 59 46 30 20 7 1 Textiles & Trade 29 31 32 32 32 32 26 7 6 7 Services 76 78 80 83 83 77 76 64 60 56 Leisure 20 22 23 23 23 24 28 24 21 21 Borrowing | Construction | . 7 | 11 | 12 | 14 | 15 | 14 | 13 | 15 | 17 | 16 |
| Transportation 8 11 12 13 13 14 3 2 1 Utilities 58 60 61 59 46 30 20 7 1 Textiles & Trade 29 31 32 33 33 77 76 64 60 64 30 31 11 12 13 13 14 13 16 16 17 200 211< | Capital Goods | 56 | 58 | 60 | 59 | 57 | 54 | 63 | 58 | 52 | 38 |
| Utilities 58 60 61 59 46 30 20 7 1 Textiles & Trade 29 31 32 32 32 32 26 7 6 76 Services 76 78 80 83 83 77 76 64 60 56 Leisure 20 22 23 23 23 24 28 24 21 21 Lending 212 212 195 183 181 150 103 41 12 17 Borrowing | Transportation | 8 | 11 | 12 | 13 | 13 | 13 | 12 | 3 | 2 | 1 |
| Textiles & Trade 29 31 32 32 32 32 26 7 6 7 Services 76 78 80 83 83 77 76 64 60 56 Leisure 20 22 23 23 23 24 28 24 21 21 Lending 212 212 195 183 181 150 103 41 12 17 Borrowing </td <td>Utilities</td> <td>58</td> <td>60</td> <td>61</td> <td>59</td> <td>46</td> <td>30</td> <td>20</td> <td>7.</td> <td>ī</td> <td></td> | Utilities | 58 | 60 | 61 | 59 | 46 | 30 | 20 | 7. | ī | |
| Services 76 78 80 83 83 77 76 64 60 56 Leisure 20 22 23 23 23 24 28 24 21 21 Lending 212 212 195 183 181 150 103 41 12 17 Borrowing </td <td>Textiles & Trade</td> <td>29</td> <td>31</td> <td>32</td> <td>32</td> <td>32</td> <td>32</td> <td>26</td> <td>7</td> <td>6</td> <td>7</td> | Textiles & Trade | 29 | 31 | 32 | 32 | 32 | 32 | 26 | 7 | 6 | 7 |
| Leisure 20 22 23 23 23 24 28 24 21 21 Lending 212 212 195 183 181 150 103 41 12 17 Borrowing | Services | 76 | 78 | 80 | 83 | 83 | 77 | 76 | 64 | 60 | 56 |
| Lending Borrowing 212 212 195 183 181 150 103 41 12 13 Neither 17 29 31 62 109 171 200 211 Max. Lent (%) 100 <td< td=""><td>Leisure</td><td>20</td><td>22</td><td>23</td><td>23</td><td>23</td><td>24</td><td>28</td><td>24</td><td>21</td><td>21</td></td<> | Leisure | 20 | 22 | 23 | 23 | 23 | 24 | 28 | 24 | 21 | 21 |
| Lending 212 212 195 183 181 150 103 41 12 1 Borrowing 17 29 31 62 109 171 200 211 Max. Lent (%) 100 | | | | | | • | | 7 | | | |
| Borrowing 17 29 31 62 109 .171 200 211 Max. Lent (%) 100 <td>Lending</td> <td>212</td> <td>212</td> <td>195</td> <td>183</td> <td>181</td> <td>150</td> <td>103</td> <td>~ 4<u>1</u></td> <td>12</td> <td>1</td> | Lending | 212 | 212 | 195 | 183 | 181 | 150 | 103 | ~ 4 <u>1</u> | 12 | 1 |
| Neither 17 29 31 62 109 171 200 211 Max. Lent (%) 100 | Borrowing | • • • | ••• | • • • | | • • • | • • • | | • • • | | |
| Max. Lent (%)100100100100100100100100100100Min. Lent(%)42131561113100Loss Quarters24303746546270767671In 0 Industries11731111111In 1 Industry84736865616068140175211In 2 Industries5763676971831016634In 3 Industries3443414251553452In 4 Industries1616232422138In 5 Industries77786In 6 Industries3332In 8 IndustriesIn 10 IndustriesIn 11 Industries | Neither | • • • | • • • | 17 | 29 | <u></u> 31 | 62 | 109 | : ~171 | 200 | 211 |
| Min. Lent(%) 42 13 1 5 6 1 1 1 3 100 Loss Quarters 24 30 37 46 54 62 70 76 76 71 In 0 Industries 11 7 3 1 <td>Max. Lent (%)</td> <td>100</td> | Max. Lent (%) | 100 | 100 | 100 | 100 | 100 | 100 | 100 | 100 | 100 | 100 |
| Loss Quarters 24 30 37 46 54 62 70 76 76 71 In 0 Industries 11 7 3 1 <td>Min. Lent(%)</td> <td>42</td> <td>13</td> <td>1</td> <td>5</td> <td>6</td> <td>1</td> <td>1</td> <td>1</td> <td>3</td> <td>100</td> | Min. Lent(%) | 42 | 13 | 1 | 5 | 6 | 1 | 1 | 1 | 3 | 100 |
| In 0 Industries 11 7 3 1 | Loss Quarters | 24 | 30 | 37 | 46 | 54 | 62 | 70 | 76 | 76 | 71 |
| In 1 Industry 84 73 68 65 61 60 68 140 175 211 In 2 Industries 57 63 67 69 71 83 101 66 34 In 3 Industries 34 43 41 42 51 55 34 5 2 In 4 Industries 16 16 23 24 22 13 8 | In O Industries | 11 | 7 | 3 | 1 | 1 | 1 | 1 | 1 | 1 | 1 |
| In 2 Industries 57 63 67 69 71 83 101 66 34 In 3 Industries 34 43 41 42 51 55 34 5 2 In 4 Industries 16 16 23 24 22 13 8 < | In l Industry | 84 | 73 | 68 | 65 | 61 | 60 | 68 | 140 | 175 | 211 |
| In 3 Industries 34 43 41 42 51 55 34 5 2 In 4 Industries 16 16 23 24 22 13 8 < | In 2 Industries | 57 | 63 | 67 | ,69 | 71 | - 83 | 101 | 66 | 34 | |
| In 4 Industries 16 16 23 24 22 13 8 In 5 Industries 7 7 7 8 6 | In 3 Industries | 34 | 43 | 41 | / 42 | 51 | 55 | 34 | 5 | 2 | |
| In 5 Industries 7 7 8 6 </td <td>In 4 Industries</td> <td>16</td> <td>16</td> <td>23</td> <td>24</td> <td>X 22</td> <td>13</td> <td>8</td> <td></td> <td></td> <td></td> | In 4 Industries | 16 | 16 | 23 | 24 | X 22 | 13 | 8 | | | |
| In 6 Industries 3 3 3 2 < | In 5 Industries | 7 | 7 | 7 | 8 | 6 | | | | | |
| In 7 Industries 1 | In 6 Industries | 3 | 3 | 3 | <u> </u> | . \ . | | | | | |
| In8 Industries <td>In 7 Industries</td> <td>• • •</td> <td></td> <td>• • •</td> <td>1</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> | In 7 Industries | • • • | | • • • | 1 | | | | | | |
| In 9 Industries | In 8 Industries | • • • | | • • • | • • • | | | | | | |
| In 10 Industries | In 9 Industries | | | | | | | | | ••• | ••• |
| In 11 Industries | In 10 Industries | | | | | | ••• | | • • • | • • • | |
| In 12 Industries \dots | In 11 Industries | | | - • • | ••• | ••• | • • • | • • • | ••• | ••• | • • • |
| | In 12 Industries | ••• | ••• | ••• | • • • | | ••• | ••• | • • • | • • • | • • • |

Summary of Portfolio Compositions of Power Policies Managing 12

۰, I

2

In

∄n

ĺn

6 Industries

7 Industries

8 Industries

| Number of | | - <u></u> | | | Pow | ers | | | | |
|------------------|-----------------|-----------|-------|-------------|--------------|------------|-----------|------------------|-----------|----------|
| Times in: | -75 | - 50 | - 30 | -15 | , -10 | - 5 | -2 | 0 | .5 | 1 |
| Basic Industries | 27 | 30 | 33 | 37 | 38 | 38 | 39 | 40 | 36 | 16 |
| Capital Goods | 15 | 16 | 19 | 20 | 20 | 26 | 25 | 18 | 13 | 6 |
| Construction | \$ | | 1 | 1 | 1 | 3 | 10 | 14 | 13 | · 33 |
| Consumer Goods | 47 [.] | 47 | 49 | 48 | 45 | 43 | 39 | . 39 | 30 | 18 |
| Energy | 113 | 116 | 121 | 124 | 123 | 119 | 119 | ⁻ 105 | -102 | 71 |
| Finance | 14 | 15 | 15 | 16 | 19 | 22 | 28 | 24 | 19 | 13 |
| Transportation | <u> </u> | - 19 | 22 | 25 | 26 | 26 | 29 | 41 | 44 | 48 |
| Utilities | 87 | 87 | 88 | 88 | 89 | 8 7 | 63 | 24 | 16 | × 10 |
| | × | | | | | | | | | |
| Lending | 212 | 212 | 212 | 190 | 181 | 154 | 95 | 29 | ••• | ·· • • • |
| Borrowing ` | ••• | • • • | • • • | 11 | 27 | 34 | 103 | 160 | 199 | 206 |
| Neither | . • • • | • • • | •••4 | 11 | 4 | 24 | 14 | 23 | 13 | 6 |
| Max. Lent (%) | 100 | 99 | 99 | 97) | 96 | 93 | 85 | 48 | | |
| Min. Lent(%) | 67 | 51 | 19 | 1 | 21 | 1 | 1 | 1 | • • • | |
| Loss Quarters | 23 | 29 | 35 | 43 | 52 | 67 | 74 | 80 | 82 | 83 |
| In O Industries | 2 | • • • | | | | | | ć ` | • • • | · • • • |
| In 1 Industry | 116 | 113 | 101 | 92 | 89 | 89 | 95 | 12 ⁹ | 157 | 209 |
| In 2 Industries | 77 | 80 | 86 | 93 | 98 | 94 | 94 | 73 | 49 | 3 |
| In 3 Industries | 17 | 19 | 25 | 27 | 24 | 29 | 23 | 10 | 6 | |
| In 4 Industries | | | • • • | • • • | 1 | • • • | • • • | ••• | • • • | |
| In 5 Industries | • • • | | | | • • | | | | | |

• • •

. . .

• • •

. .

. . .

. . .

. . .

. . .

. . .

. . .

. . .

. . .

. . .

. . .

ه م

. . .

. . .

. . .

. . .

. . .

. . .

. . .

. . .

. ...

. . .

• • •

. . .

. . .

. . .

Summary of Portfolio Compositions of Power Policies Managing 8 4

| Number of | | | | | Powe | ers | | · · · | | |
|------------------|-------|-----------|-------|-------|-------|-----------|-----|-------|------|-------|
| Times in: | -75 | -50 | - 30 | -15 | -10 | - 5 | -2 | 0 | . 5 | . 1 |
| Basic Industries | 34 | 41 | 42 | 46 | 46 | 47 | 40 | 33 | 28 | 11 |
| Capital Goods | 53 | 55 | 59 | 64 | 65 | 66 | 64 | 60 | 58 | 47 |
| Construction | 3 | 4 | 5 | 5 | 5 | 5 | 7 | 7 | 8 - | 20 |
| Consumer Goods | 51 | 53 | 55 | 54 | 51 | 44 | 38 | 24 | 17 | 7 |
| Energy | 101 | 111 | 120 | 129 | 129 | 128 | 122 | 107 | ~ 92 | 64 |
| Finance | 18 | 18 | 19 | 21 | 22 | 28 | 31 | 21 | 16 | 13 |
| Transportation | 31 | 32 | 34 | 38 | 38 | 38 | 40 | 42 | · 41 | 40 |
| Utilities | 84 | 85 | 90 | 91 | 90 | 78 | 60 | 23 | 15 | 8 |
| Lending | 212 | 212 | 212 | 183 | 181 | 152 | 100 | 42 | 13 | 2 |
| Borrowing | • • • | ••• | | 19 | 29 | 40 | 94 | 147 | 173 | 190 |
| Neither | ••• | ••• | • | 10- | 2 | 20 | 18 | 23 | 26 | 20 |
| Max. Lent (%) | 100 | 100 | 100 | 100 | 100 | 100 | 100 | 100 | 100 | 100 |
| Min. Lent(%) | 62 | 43 · | 6 | 1 | 14 | 1 | 2 | 1 | 11 | 100 |
| Loss Quarters | 22 | 27 | 33 | 47 | 52 | 61 | 72 | 79 | 79 | 77 |
| In O Industries | 14 | 8 | 5 | 3 | 3 | 3 | 3 | 2 | 2 | 2 |
| In 1 Industry | 79 | 76 | 67 | 62 | 61 | 60 | 69 | 120 | 153 | 210 |
| In 2 Industries | 71 | 73 | 77 | 70 | 72 | 81 | 90 | 76 | 49 | |
| In 3 Industries | 38 | 43 | 49 | 63 | 64 | 61 | 47 | 12 | 8 | |
| In 4/Industries | 10 | 12 | 14 | 13 | 11 | 6 | 3 | 1 | | |
| In 5 Industries | • • • | • • • | • • • | 1 | 1 | 1 | | 1 | | |
| In 6 Industries | • • • | • • • | • • • | • • • | • • • | | | • • • | | • • • |
| In 7 Industries | • • • | • • • | • • • | • • • | • • • | | | | | |
| In 8 Industries | • • • | • • • | ••• | | • • • | | | | | |

Summary of Portfolio Compositions of Power Policies Managing & Value-Weighted Industries, With Leverage, 1934-86 32 guarter estimating period, simple probability assessment approach)

184 Å

| Number of | | | | | Pow | ers | | | | |
|--------------------|-------------|-------|----------|---------|---------|-------|-------|-----------|-------------|------------|
| Times in: | -75 | - 50 | - 30 | -15 | -10 | - 5 | - 2 | 0 | . 5 | 1 |
| Banks & Fin. Serv. | • • • | ••• | ••• | | | 1 | · 3 | •••• | • • • | • • • |
| Chemicals, R. & P. | | | • • • | | • • • | • • • | • • • | • • • | • • • | • • • |
| Construction | 16 | 18 | 20 | 22 | 22 | 22 | 27 | 27 | 29 | 3 1 |
| Fabricated M.P. | 0 | ••• | | • • • | • • • | • • • | • • • | | • • • | • • • |
| Food | 3 | 3 | <u> </u> | 3 | 3 | 3 | 2 | ••• | • • • | • • • |
| Insurance | 48 | 50 | 54 | 55 | 55 | 54 | 26 | 15 | 13 | 2 |
| Machinery | 3 | 3 | 3 | 3 | 4 | 4 | 2 | - 1 | 3 | • • • |
| Metals | 1 | 2 | 2 | 2 | 2 | 2 | 5 | 1 | 1 | •••• |
| Mining | 42 / | 55 | 66 | 73 | 74 | 77 | 75 | 69 | 58 | 12 |
| Oil & Gas Expl. | 61 | 63 | 66 | 71 | 72 | 74 | 78 | 75 | 64 | 41 |
| Paper & Wood | 4 | 7 · | 7 | 7 | 7 | 8 | 8 | 11 | 10 | • • • |
| Petroleum & Coal | 47 | 59 | 60 | 58 | 58 | 50 | 34 | 9 | 3 | • • • |
| Railroads & Truck. | 1 | 2 | 6 | 7 | 8 | 9 | 10 | 9 | 10 P | 9 |
| Real Estate | 22 | 23 | 24 | 26 | 28 | 34 | 35 | 18 | ຸ 5 | 3 |
| Retail Tr., Food | 50 | 50 | 51 | 53 | 49 | 40 | 31 | 3 | · • • · | • • • |
| Retail Tr., Others | • • • | 1 | 1 | 1 | 2 | 2 | 3 | 2 | 2 | • • • |
| Services, Leisure | 18 | 18 | 20 | 24 | 27 | 31 | 39 | 52 | 55 | 50 |
| Services, Others | 36 | 38 | 39 (| 39 | 38 | 38 | 34 | 34 | 19 | 3 |
| Textile | 2 | 2 | 2 | 2 | 2 | 2 | 2 | | | ••• |
| Tobacco | 82 | 85 | 87 | 91 | 90 | 88 | 84、 | 64 | 41 | 16 |
| Transportation | 25 | 27 | 32 | 33 | 33 | 33 | 35 | 32 | 29 | 22 |
| Transportation Eq. | 1 | 1 | 2 | 2 | 2 | 2 | 2 | 1 | 1 | 1 |
| Utilities | 47 | 48 | 48 | 48 | 47 | 34 | 14 | | | • • • |
| Wholesale Tr. | 13 | 15 | 22 | 25 | 28 | 30 | 31 | 39 | 45 | 33 |
| Lending | 212 | 212 | 209 | 183 | 180 | 137 | 87 | 21 | | • • • |
| Borrowing | • • • | • • • | • • • | 23 | 31 | 54 | 114 | 172 | 202 | 208 |
| Neither | ••• | • • • | 3 | 6 | 1 | 21 | 11 | 19 | 10 | 4 |
| Max. Lent (%) | 100 | 99 | 99 | 97 | 95 | 92 | 82 | 39 | | • • • |
| Min. Lent(%) | 26 | 4 | 19 | 1 | 4 | 1 | 1 | 2 | ••• | • • • |
| Loss Quarters | 23 | 33 | 35 | 51 | 54 | 68 | 74 | 80 | 87 | 93 |
| In O Industries | 4 | | | | | | • • • | | ••• | • • • |
| In 1 Industry | 47 | 28 | 17 | 9 | 9 | 10 | 16 | 56 | 93 | 201 |
| In 2 Industries | 79 | 87 | 74 | 70 | 66 | 67 | 75 | 93 | 84 | 11 |
| In 3 Industries | 28 | 40 | 56 | 62 | 61 | 64 | 81 | 37 | 19 | • • • |
| In 4 Industries | 38 | 38 | 45 | · 46 | 54 | 56 | 29 | 22 | 11 | • • • |
| In 5 Industries | 15 | 18 | 18 | 24 | 21 | 12 | 11 | 3 | 4 | • • • |
| In 6 Industries | 1 | 1 | 2 | 1 | 1 | 3 | • • • | 1 | 1 | • • • |
| In 7 Industries | • • • | • • • | • • • | ••• | • • • • | ••• | • • • | • • • | • • • | • • • |
| | • • • | • • • | ••• | ••• | • • • | ••• | • • • | ••• | • • • | • • • |
| | ••• | • • • | ••• | • • • | • • • | ••• | ••• | • • • | • • • | • • • |
| In 24 Industries | • • • | • • • | • • • | • • • • | • • • | • • • | • • • | • • • | ••• | • • • |

Summary of Portfolio Compositions of Power Policies-Managing 24 Equal-Weighted Industries, With Leverage, 1934-86 (32 guarter estimating period, simple probability assessment approach)

ş

Summary of Portfolio Compositions of Power Policies Managing 24 Value-Weighted Industries, With Leverage, 1934-86 (32 quarter estimating period, simple probability assessment approach)

| Number of | | | | | Pow | ers | • | | | |
|--------------------|-------|------------------|-----------|------------|----------|------------|-----------|----------|--------|-------------|
| Times in: | - 75 | - 50 | - 30 | -15 | -10 | - 5 | - 2 | 0 | . 5 | 1 |
| Banks & Fin. Serv. | •••• | | | | | | | | ····· | · |
| Chemicals, R. & P. | 19 | 24 | 26 | 27 | 26 | 23 | 21 | 21 | 15 | 1 |
| Construction | 3 | 4 | 6 | 7 | 7 | 8 | 10 | 14 | 13 | - 9 |
| Fabricated M.P. | 2 | 2 | · 2 | 2 | 2 | 2 | 1 | | | |
| Food | 5 | 5 | 5 | 5 | 5 | 5 | 5 | 1 | | |
| Insurance | 41 | 41 | 43 | 45 | 44 | 40 | 27 | 11 | 13 | 4 |
| Machinery | 9 | 12 | 13 | 15 - | 15 | 16 | 14 | 13 | 12 | 10 |
| Metals | | | | í • • • | 1.17 | | | 1 | 1 | |
| Mining | 5 | 9 | 11 | 12 | 13 | 13 | 12 | 2 | | |
| Oil & Gas Expl. | 87 | 90 | 93 | 97 | 98 | 99 | 99 | 81 | 62 | 38 |
| Paper & Wood | 5 | 6 | 6 | 8 | 10 | 19 | 21 | 14 | 11 | 2 |
| Petroleum & Coal | 40 | 47 | 50 | 54 | 53 | 48 | 27 | 5 | | L |
| Railroads & Truck | 2 | 2 | 2 | 2 | 2 | 2 | 1 | - | • • • | • • • |
| Real Estate | 44 | 49 | 51 | 52 | - 54 | 56 | 55 | 50 | 32 | · · · 27 |
| Retail Tr. Food | 67 | 68 | 68 | 68 | 67 | . 62 | 51 | 12 | 7 | ر ے ۱ |
| Retail Tr Othere | 13 | 16 | 17 | 17 | 1.8 | 17 | 10 | · · | , 2 | 1 |
| Services laisuro | 20 | 28 | 17 | 17 /17 | 10 | 17 /.7 | 51 | 50 50 | 50 | 1 |
| Services, Leisure | 51 | 50 | 50 | 4J 5/. | 4J 60 | 47 51 | 50 | 51 | 01 | 47 |
| Toytilo | 21 | 52 | JZ 6~~ | 10 | 10 | 10 | JZ 0 |) I 7 | 40 | 40 |
| Tobacco | 0 | 9 | 101 | 10 | 10 | 10 | , o 00 | / | ••• | |
| Trangportation | 32 | <u>7</u> ∦ 07 | 101 | 101 | 97 | 90 | 90 | 74 | 49 | 21 |
| Transportation Fa | 20. | 27 | 20 | 20 | 20 - | 28 | 29 | 20 | 21 | 14 |
| litilities | 29 | 20 | 30 | 27 21 | 20 | 20 | 20 | 21 | 20 | 19 |
| | | - DO | رد ح | 21 | 1/ | 10 | 14 | 2 | 1 | - 1 |
| wholesale Ir. | 4 | . 0 | 0 | 8 | 8 | , , | 0 | 4 | Z | ช |
| Lending | 204 | 200 | 188 | 181 | 176 | 140 | 87 | . 8 | | |
| Borrowing | | 1 | 15 | 31 | 31 | 51 | 109 | 173 | 201 | 206 |
| Neither | 8 | 11 | 9 | ر ••• | 5 | 21 | . 16 | 31 | 11 | 6 |
| Max. Lent (%) | 100 | 99 | 98 | 97 | 95 | 91 | 82 | 42 | | |
| Min. Lent(%) | 4 | 1 | 3 | 23 | 6 | 1. | 5 | 2 | | |
| / | - | - | - | | - | - | - | - | | |
| Loss Quarters | 32 | 36 | 41 | 52 | 58 | 69 | 74 | 86 | 92 | 87 |
| In O Industries | · 1 | - ••• | | | | | | | • • • | ••• |
| In l Industry | 45 | 34 | 19 | 13 | 12 | 9 . | 10 | 45 | 95 | 184 |
| In 2 Industries | 61 | 57 | 64 | 54 | 53 | 59 | 72 | 96 | 86 | 25 |
| In 3 Industries | 42 | 45 | 48 | 58 | 57 | 57 | 67 | 59 | 27 | 2 |
| In 4 Industries | 21 | 31 | 37 | 44 | 53 | 51 | 47 | 7 | 4 | ·] |
| In 5 Industries | 22 | 20 | 21 | 28 | -27 | 28 | 10 | 5 | | •• • • • |
| In 6 Industries | 16 | 21 | 18 | 13 | 7 | 6 | 5 | | | |
| In 7 Industries | -3 | 3 | 4 | 2 | , 3 | 2 | 1 | ••• | ••• | |
| In 8 Industries | 1 | 1 | i | | | | • | • • • | • • • | ••• |
| n 9 Industries | • | - | - | ••• | ••• | ••• | ••• | • • • | ••• | |
| | ••• | ••• | • • • | ••• | • • • | | • • • | • • • | ••• | • • • |
| | ••• | • • • | • • • | ••• | • • • | | • • • | ••• | • • • | • • • |
| n 24 Industries | ••• | ••• | ••• | ••• | ••• | | • • • | ••• | ••• | • • • |
| a Ly Industries | • • • | ••• | • • • | • • • | ••• | ••• | • • • | ••• | ••• | ••• |

| Number of | | 2.0 | 2 | | Pow | ers | | | | |
|-------------------|------------|-------|-------|---------|-----------|-------|------------|-----|-------|-----|
| Times in: | -75 | - 50 | - 30 | -15 | -10 | -5 | - 2 | 0 | .5 | 1 |
| Petroleum | 95 | 100 | 103 | 105 | 106 | 109 | 105 | 93 | 72 | 50 |
| Finance & R.E. | 9 | 9 | 9 | 10 | 10 | 14 | 17 | 13 | 8 | 7 |
| Consumer Durables | < 1 | 1 | 3 | 5 | 5 | 7 | 7 | 7 | 8 | 6 |
| Basic Industries | 17 | 19 | 22 | 25 | 25 | 25 | 26 | 24 | 22 | 1 |
| Food & Tobacco | 27 | 28 | 29 | 31 | 31 | 31 | 25 | 15 | 9 | 1 |
| Construction | 6 | 6 | 6 | 8 | 9 | 12 | 12 | 6 | 8 | 18 |
| Capital Goods | 8 | 10 | 12 | 13 | 13 | 13 | 10 | 7 | 6 | 1 |
| Transportation | 13 | 19 | 22 | 24 | 26 | 26 | 33 | 43 | 44 | 45 |
| Utilities | 66 | 67 | 70 | 73 | 73 | 69 | 35 | 9 | 7 | 3 |
| Textiles & Trade | 6 | 6 | 6 | , 7 | 8 | 8 | 11 | 8 | 7 | 6 |
| Services | 71 | 75 | 79 | 81 | 78 | 76 | 74 | 82 | 73 | 47 |
| Leisure | 35 | 37 | 39 | 37 | 38 | 39 | 40 | 40 | 31 | 21 |
| Lending | 204 | 204 | 193 | 177 | 171 | 150 | 82 | 16 | 5 | 1 |
| Borrowing | ••• | • • • | • • • | 26 | 30 | 42 | 111 | 168 | 187 | 195 |
| Neither | • • • | • • • | 11 | 1 | 3 | 12 | 11 | 20 | 12 | 8 |
| Max. Lent (%) | 100 | 100 | 100 | 100 | 100 | 100 | 100 | 100 | 100 | 100 |
| Mín. Lent(%) | 49 | 23 | 1 | 14 | 7 | 2 | , 1 | 3 | 38 | 100 |
| Loss Quarters | 22 | 25 | 32 | 44 | 52 | 59 | 65 | 76 | 77 | 79 |
| In O Industries | 8 | 5 | 3 | 1 | 1 | 1 | 1 | 1 | 1 | 1 |
| In 1 Industry | 84 | 75 | 59 | 57 | 55 | 56 | 67 | 85 | 124 | 200 |
| In 2 Industries | 69 | 74 | 89 | 81 | 82 | 74 | 84 | 95 | 66 | 3 |
| In 3 Industries | 40 | 46 | 49 | 60 | 61 | 67 | 48 | 20 | 13 | |
| In 4 Industries | 3 | 4 | 4 | 5 | 5 | 6 | 4 | 3 | | |
| In 🔄 Industries | • • • | • • • | | ••• | • • • | • • • | • • • | | | |
| In 6 Industries | • • • | • • • | • • • | | • • • | | • • • | | | |
| In 7 Industries | • • • | | • • • | | • • • | | | | • • • | |
| In 8 Industries | • • • | * | • • • | · • • • | • • • | | | | | |
| In 9 Industries | • • • | | | | • • • | | | | • • • | |
| In 10 Industries | | | • • • | · • • • | | | | | | |
| In 11 Industries | ••• | ••• | | | • • • | | | | • • • | |
| In 12 Industries | | | | | | | | | | |

Summary of Portfolio Compositions of Power Tolicies Managing 12

187

. . .

. . .

. . .

. . .

| | Summary of Portfoli | o Compositions | of Power Pol | icies Manag | ing 12 |
|-----|---------------------|----------------|--------------|-------------|-----------|
| | Value-Weight | ed Industries, | With Leverag | e, 1934-86 | • |
| (28 | quarter estimating | period, simple | probability | assessment | approach) |

| Number of | | | | | Powe | ərs | | | | , | |
|-------------------|---------|--------------------|---------|-------|---------|-------|---------|-------|-------|-----|----------|
| Times in: | -75 | - 50 | - 30 | -15 | -10 | - 5 | - 2 | 0 | . 5 | 1 | |
| Petroleum | 83 | 90 | 94 | · 97 | 99 | 98 | 95 | 56 | 47 \ | 36 | |
| Finance & R.E. | 11 | 12 | 12 | 16 | 17 | 17 | 12 | 4 | 2 | 2 | |
| Consumer Durables | 6 | 12 | 14 | 21 | 25 | 30 | 29 | 23 | 25 | 10 | |
| Basic Industries | 28 | 32 | 33 | 33 | 32 | 29 | 22 | 15 | 10 | 1 | b |
| Food & Tobacco | 33 | 36 , | 36 | 36 | 32 | 25 | 23 | 13 | 8 | 6 | E. |
| Construction | 10 | 10 | 10 | 10 | 10 | 11 | 11 | 13 | 16 | 14 | |
| Capital Goods | 44 | 45 | 46 | 47 | 47 | 44 | 41 | 42 | 37 | 34 | V |
| Transportation | 17 | 19 | 20 | 20 | 21 | 21 | 21 | 15 | 13 | 7 | |
| Utilities | 62 | 62 | 63 | 61 | 53 | 37 | 24 | 13 | 13 | 9 | |
| Textiles & Trade | 33 | 34 | 34 | 35 | 35 | 34 | 28 | 15 | 12 | 9 | |
| Services | 77 | €, <mark>81</mark> | 83 | 84 | 86 | 84 | 79 | 76 | 67 | 64 | * |
| Leisure | 16 | 17 | 19 | 23 | 23 | 26 | 25 | 25 | 19 | 15 | |
| Lending | 204 | 202 [.] | 183 | 171 | 169 | 139 | 82 | 26 | 12 | . 3 | • |
| Borrowing | | • • • | 3 | 29 | 31 | 49 | 106 | 155 | 175 | 190 | |
| Neither | ••• | 2 | 18 | 4 | 4 | 16 | 16 | 23 | 17 | 41 | |
| Max. Lent (%) | 100 | 100 | 100 | 100 | 100 | 100 | 100 | 100 | 100 | 100 | |
| Min. Lent(%) | . 8 | 3 | 1. | 21 | 2 | 3 、 | 3 | 2 | 3 | 100 | |
| Loss Quarters | 24 | 25 | 32 | 46 | 51 | 59 | 67 | 74 | 74 | 73 | |
| In 0 Industries | 12 | 7 | 6 | 4 | 4 | 4 | 3 | 3 | 3 | 3 | |
| In 1 Industry | 69 | 61 | 56 | 55 | 53 | 51 👳 | 60 | 112 | 145 | 186 | • • |
| In 2 Industries | 57 | 63 | 65 | 62 | 63 | 74 | 84 | 71 | 46 | 15 | |
| In 3 Industries | 32 | 36 | 39 | 45 | 48 | 51 | 48 | 16 | 8 | | 0 |
| In 4 Industries | 29 | 31 | 30 | 24 | 24 | 17 | 7 | ່ 2 | 2 | | |
| In 5 Industries | ُرِ 5_، | - ົ 5 | 7 | 11 | 11 | 6 | 2 | • • • | • • • | | |
| In 6 Industries | · · · · | 1 | 1 | 3 | 1 | 1 | • • • | | • • • | | |
| In 7 Industries | | | | | | | <i></i> | | | | |
| In 8 Industries | | | | | | | | 4 | | | |
| In 9 Industries | • • • | | · · · · | | | | | | • • • | | |
| In 10 Industries | | | | | • • • | | • • • | | | | |
| In 11 Industries | · • • | | • • • | | | | | | • • • | | |
| In 12 Industries | , ··· | • • • | • • • | • • • | • • • • | • • • | • • • | • • • | • • • | A | |

| Number of | | | | | Pow | ers | | | - | ~ |
|-------------------|-------|----------|------------------------|------|-------|-------|-------|-------|-------|-------|
| Times in: | - 75 | - 50 | - 30 | -15 | -10 | -5 | · -2 | 0 | .5 | 1 |
| Petroleum | 109 | 111 | 114 | 114 | 114 | 112 | 109 | 97 | 74 | 59 |
| Finance & R.E. | 20 | 20 | 20 | 21 | 21 | 24 | 28 | 19 | 15 | 6 |
| Consumer Durables | 1 | 1 | 1 | 1. | 1 | 3 | 6 | .6 | 9. | 16 |
| Basic Industries | 23 | 27 | 30 | 31 | 33 | 32 | 32 | 34 | 32 | · 6 |
| Food & Tobacco | 42 | 45 | 47 | 51 | 51 | 51 | 46 | 12 | 2 | • • • |
| Construction | 3 | 3 | 4 | 5 | 5 | 8 | 6 | 5 | 6 | 17 |
| Capital Goods | 7 | 10 | 10 | .11 | 11 | 11 | 11 | 8 | 4 | • • • |
| Transportation | 7 | 8 | 10 | 10 | 10 | 11 | 15 | 39 | 42 | -40 |
| Utilities | 73 | 73 | 73 | 73 | 73 | 73 | 59 | 15 | 6 | 3 |
| Textiles & Trade | 12 | - 12 | 12 | 13 | 13 | 13 | 11 | 5 | 1 | 7 |
| Services | 60 | 65 | 68 | 74 | 77 | 77 | 79 | 70 | 62 | 36 |
| Leisure | 21 | 21 | 21 ¹ | 22 | 22 | 24 | 29 | 38 | 30 | 24 |
| Lending , | 204 | 204 | 201 | 189 | 183 | 144 | 98 | 25 | 1 | ••• |
| Borrowing | | • • • | • • • | 13 | 17 | 43 | 94 | 152 | 188 | 197 |
| Neither | ••• | • • • | 3 | 2 | 4 | 17 | 12 | 27 | 15 | 7 |
| Max, Lent (%) | 100 | 100 | 100 | . 99 | 99 | 97 | 95 | 84 | 69 | |
| Min. Lent(%) | 57 | 36 | 2 | 4 | 3 | 2 | 1 | 1 | 69 | ••• |
| Loss Quarters | 17 | 22 | 32 | 48 | 52 | 57 | 70 | 79 | 77 | 81 |
| In 0 Industries | 3 | , 1 | 1 | | | ••• | ••• | | ••• | |
| In 1 Industry | 84 | 74 | 67 | 62 | 57 | 55 | 57 | 100 | 136 | 194 |
| In 2 Industries | 67 | 76 | 76 | 74 | 80 | 80 | 79 | 70 | 57 | 10 |
| In 3 Industries | 42 | 44 | 51 | 58 | 57 | 57 | 57 | 28 | 11 | |
| In 4 Industries | 6 | <u>7</u> | 7 | 8 | 7 | 7 | 10 | 6 | • • • | |
| In 5 Industries | 2 | 2 | 2 | 2 | 3 | 5 | 1 | • • • | • • • | |
| In 6 Industries | • • • | • • • | • • • | | | • • • | • • • | • • • | | |
| In 7 Industries | ••• | • • • | • • • | ••• | | | | • • • | · · · | |
| In 8 Industries | | | • • • | | | | | | | |
| In 9 Industries | ••• * | ••• | • • • . | | | | • • • | • • • | | |
| In 10 Industries | | | ••• | | | | • • • | | | |
| In 11 Industries | | | | | • • • | | | | | |
| | | | | | | | | | | |

Summary of Portfolio Compositions of Power Policies Managing 12

Summary of Portfolio Compositions of Power Policies Managing 12 Value-Weighted Industries, With Leverage, 1936-86 (40 quarter estimating period, simple probability assessment approach)

| · · · · · · · · · · · · · · · · · · · | | | | | | | | | | |
|---------------------------------------|-------|---------------|-------|-------|-------|------------|-----------|------|-------|-------|
| Number of | | | | | Powe | ers | | | | |
| Times in: | - 75 | - 50 / | - 30 | -15 | - 10 | - 5 | - 2 | 0 | . 5 | 1 |
| Petroleum | 108 | 111 | 117 | 121 | 124 | 123 | 117 | ×81 | 54 | 34 |
| Finance & R.E. | 4 | 4 | 5 | 5 | 9 | 19 | 13 | 6 | | • • • |
| Consumer Durables | 18 | 25 | 31 | 41 | 42 | 46 | 45 | 43 | 37 | 25 |
| Basic Industries | 26 | 29 | 32 | 33 | 29 | 24 | 20 | 10 | •7 | |
| Food & Tobacco | 23 | 25 | 29 | 26 | 28 | 2 8 | 24 | 11 | 3 | • • • |
| Construction | 1 | 5 | 6 | 7 | 8 | 8 | 9 | 14 | 16 | 22 |
| Capital Goods | 53 | 57 | 58 | 60 | 62 | 69 | 70 | 64 | 55 | 43 |
| Transportation | 1 | 2 | 3 | 3 | 3 | 3 | 3 | 4 | 2 | |
| Utilities | 66 | 68 | 70 | 70 | 70 | 46 | 32 | 6 | 3 | 4 |
| Textiles & Trade | 34 | 38 | 41 | 42 | 42 | 43 | 43 | 16 | 11 | 5 |
| Services | 71 | 74 | 77 | 79 | 80 | 85 | 83 | 72 | 63 | 60 |
| Leisure | 11 | 15 | 16 | 19 | 19 | 19 | 23 | 25 | 26 | 24 |
| Lending | 204 | 204 | 195 | 185 | 181 | 139 | 101 | 46 | 18 | 2 |
| Borrowing | | | | 16 | 18. | 40 | 83 | 138 | 159 | 185 |
| Neither | • • • | • • • | 9 | 3 | 5 | 25 | 20 | 20 | 27 | 17 |
| Max. Lent (%) | 100 | 100 | 100 | 100 | 100 | 100 | 100 | 100 | 100 | 100 |
| Min. Lent(%) | 45 | 19 | 1 | 2 | 14 | 1 | 1 | 7 | 1 | 100 |
| Loss Quarters | 17 | 23 | 32 | 39 | 45 | 58 | 66 | 71 | 71 | 74 |
| In O Industries | 22 | 10 | 7 | 5 | 3 | 3 | 3 | . 3 | 3 | 2 |
| In 1 Industry | 53 | 53 | 43 | 38 | 37 | 33 | 31 | 92 | 139 | 187 |
| In 2 Industries | 61 | 63 | ÷65 | 63 | 64 | 64 | 86 | 76 | 51 | 1.5 |
| In 3 Industries | 39 | 47 | 53 | 62 | 64 | 71 | 60 | 25 | 8 | |
| In 4 Industries | 21 | 23 | 28 | 27 | 27 | 27 | 21 | 7 | 3 | |
| In 5 Industries | 8 | 7 | 7 | 6 | 5 | 5 | 3 | 1 | | • • • |
| In 6 Industries | | 1 | 1 | 3 | 2 | 1 | | | • • • | |
| In 7 Industries | • • • | • • • | • • • | • • • | 2 | | · · · · · | | | |
| In 8 Industries | | | ••• | | | • • • | | | | |
| In 9 Industries | | | • • • | | | | | •••• | | |
| In 10 Industries | | • • • | , | • • • | | • • • | | | | |
| In 11 Industries | | | | • • • | | | | | | |
| In 12 Industries | • • • | • • • | ••• | ••• | · · · | • • • | ••• | | | • • • |

Ľ

Table 22a

| Number of | | - | | | Powe | ers | | | | |
|-------------------|-------|-------|-------|-------|---------|-------|-------|-------|------------|-------|
| Times in: | - 75 | - 50 | - 30 | -15 | -10 | - 5 | - 2 | 0 | .5 | 1 |
| Petroleum | 89 | 91 | . 97 | 97 | | 96 | 96 | 82 | 68 | 58 |
| Finance & R.E. | 9 | 9 | 9 | 9 | 1 | 13 | 20 | 11 | 4 | 1 |
| Consumer Durables | 2 | 2 | 2 | 2 | 2 ` | 2 | 2 | 2 | 4 | 5 |
| Basic Industries | 19 | 25 | 26 | 27 | 27 | 22 | 27 | 27 | 31 | 5 |
| Food & Tobacco | 36 | 40 | 42 | 43 | 43 | 43 | 35 | 13 | . 8 | • • • |
| Construction | 6 | 6 | 6 | 7 · | ~ .7 | 8 | 7 | 3 | 10 | 24 |
| Capital Goods | 9 | 9 | 9 | 9 | 9 | 10 | 9 | 9 | 6 | -3 |
| Transportation | 7 | 10 | 12 | 12 | 13 | 14 | 20 | 28 | 30 | 30 |
| Utilities | 71 | 72 | 73 | 73 | 73 | 67 | 42 | 25 | 14 | 10 |
| Textiles & Trade | 4 | 4 | 4 | 4 | 5 | 5 | 8 | 7 | 6 | 14 |
| Services | 66 | 70 | 74 | 78 | 77 | 74 | 74 | 63 | 56 | 38 |
| Leisure | 22 | 26 | 27 | 30 | 31 | 30 | 35 | 34 | 26 | 20 |
| Lending | 212 | 212 | 209 | 191 | 181 | 166 | 99 | 44 | 19 | 14 |
| Borrowing | | | | 16 | 20 | 37 | 103 | 149 | 183 | 189 |
| Neither | • • • | ••• | 3 | 5 | 11 | . 9 | 10 | 19 | · 10 | 9 |
| Max. Lent (%) | 100 | 100 | 100 | 100 | 100 | 100 | 100 | 100 | 100 | 100 |
| Min. Lent(%) | 52 | 28 | 2 | • 20 | 12 | 3 | 7 | 1 | 42 | 100 |
| Loss Quarters | 21 | 26 | 31 | 37 | 42 | 55 | 65 | 74 | 75 | 79 |
| In O Industries | 22 | 18 | 15 | 14 | 14 | 14 | 14 | 14 | 14 | 14 |
| In 1 Industry | 79 | 68 | 64 | 59 | 58 | 64 | 72 | 105 | 137 | 188 |
| In 2 Industries | 73 | 83 | 83 | 88 | 87 | 82 | 82 | 81 | 58 | 10 |
| In 3 Industries | 37 | 42 | 49 | 48 | 48 | 47 | 37 | 11 | 2 | • • • |
| In 4 Industries | 1 | 1 | 1 | 3 | 5 | 5 | 7 | 1 | 1 | • • • |
| In 5 Industries | • • • | | ••• | ••• | ••• | • • • | • • • | • • • | • • • | |
| In 6 Industries | • • • | • • • | • • • | | • • • | • • • | • • • | | • • • | • • • |
| In 7 Industries | • • • | • • • | ••• | • • • | • • • | • • • | •••• | • • • | • • • | |
| In 8 Industries | • • • | ••• | • • • | • • • | | • • • | • • • | • • • | • • • | |
| In 9 Industries | • • • | • • • | ••• | ••• | • • • | • • • | • • • | • • • | • • • | |
| In 10 Industries | | • • • | • • • | • • • | • • • | • • • | • • • | • • • | • • • | |
| In 11 Industries | | • • • | | • • • | • • • | • • • | • • • | | • • • | |
| In 12 Industries | | | | • • • | | • • • | • • • | | | |

Summary of Portfolio Compositions of Power Policies Managing 12 Equal-Weighted Industries, With Leverage, 1934-86 (32 guarter estimating period, inflation adapter approach)

Table 22b

. .

In 12 Industries

• • •

. . .

. . .

| Number of | | | | | Powe | ers | | | | • |
|-------------------|-----------|-------|------|-----|------|-------|------|-------|------------|-----|
| Times in: | -75 | - 50 | - 30 | -15 | -10 | - 5 | - 2 | 0 | . 5 | 1 |
| Petroleum | 73 | 80 | 83 | 83 | 87 | 87 | 86 | 59 | 48 | 39 |
| Finance & R.E. | 9 | 10 | 10 | 10 | 11 | 13 | 11 | 2 | 1 | 1 |
| Consumer Durables | 15 | 19 | 23 | 26 | 28 | 31 | 31 | 32 | 33 | 23 |
| Basic Industries | 28 | 28 | 30 | 30 | 30 | 27 | 18 | 10 | 6 | |
| Food & Tobacco | 35 | 36 | 38 | 38 | 38 | 34 | 34 | 23 | 19 | 5 |
| Construction | 8 | 11 | 13 | 14 | 15 | 15 | 14 | 16 | 20 | 18 |
| Capital Goods | 34 | 36 | 38 | 41 | 41 | 38 | 37 | 40 | 31 | 27 |
| Transportation | 4 | 4 | 6 | 7 | 7 | 7 | 7 | 1 | | |
| Utilities | 56 | 58 | 59 | 61 | 58 | 40 | 21 | 10 | 6 | 9 |
| Textiles & Trade | 21 | 22 | 24 | 24 | 24 | 23 | 20 | 14 | 11 | 8 |
| Services. | 72 | 75 | 78 | 79 | 80 | 74 | 72 | 61 | 57 | 53 |
| Leisure | 20 | 21 | 22 | 23 | 23 | 25 | 27 | 30 | 22 | 17 |
| Lending | 212 | 212 | 201 | 185 | 178 | 157 | 103 | 56 | 34 | 21 |
| Borrowing | • • • | | | 18 | 27 | 41 | 93 | 144 | 163 | 180 |
| Neither | • • • | • • • | 11 | 9 | 7 | 14 | 16 | 12 | 15 | 11 |
| Max. Lent (%) | 100 | 100 | 100 | 100 | 100 | 100 | 100 | 100 | 100 | 100 |
| Min. Lent(%) | 36 | 4 | 10 | 4 | 9 | 1 | 1 | 1 | 1 | 100 |
| Loss Quarters | 24 | 28 | 36 | 47 | 54 | 59 | 63 | 69 | 68 | 67 |
| In O Industries | 33 | 25 | 22 | 22 | 22 | 21 | 21 | 21 | 21 | 21 |
| In l Industry | 72 | 72 | 67 | 65 | 62 | 63 | 66 | 104 | 139 | 182 |
| In 2 Industries | 51 | 54 | 53 | 52 | 54 | 56 | 75 | 68 | 42 | 9 |
| In 3 Industries | 32 | 34 | 39 | 37 | 40 | 52 | 38 | 18 | 9 | |
| In 4 Industries | 16 | 18 | 22 | 27 | 23 | 17 | 12 | 1 | 1 | |
| In 5 Industries | 7 | 8 | 8 | 7 | 7 | 3 | | | | |
| In 6 Industries | 1 | 1 | 1 | 1 | 3 | | | | | |
| In 7 Industries | | | | 1 | 1 | | | | | |
| In 8 Industries | . | | | - | | | | | | |
| In 9 Industries | | | | | | | | | ••• | ••• |
| In 10 Industries | | | | ••• | ••• | ••• | •••• | • • • | • • • | • • |
| In 11 Industries | - • • | | | | | • • • | ••• | • • • | ••• | |

Summary of Portfolio Compositions of Power Policies Managing 12 Value-Weighted Industries, With Leverage, 1934-86 (32 quarter estimating period. inflation adapter approach)

. . .

. . .

. . .

. . .

. . .

. . .

•••

| probability assessment approach) | | | | | | | | | | | | |
|----------------------------------|-------|-------|-------|-------|--------------|------------|-------|------------|-------|-------|--|--|
| Number of | | | | | Pow | ers | | | | | | |
| Times in: | -75 | - 50 | - 30 | - 1.5 | ` -10 | - 5 | - 2 | 0 | .5 | 1 | | |
| Petroleum | 94 | 99 | 104 | 105 | 106 | 105 | 102 | 89 | 73 | 49 | | |
| Finance & R.E. | 16 | 16 | 17 | 19 | 21 | 22 | 19 | ~20 | 12 | 3. | | |
| Consumer Durables | 1 | 1 | 1 | 2 | 2 | 2 | 2 | 7 | 12 | 17 | | |
| Basic Industries | 15 | 21 | 24 | 29 | 30 | 31 | 31 | 38 | 35 | 7 | | |
| Food & Tobacco | 43 | 45 | 47 | 46 | 46 | 44 | 38 | 23 | 14 | 4 | | |
| Construction | 4 | 7 | 7 | 15 | 19 | 21 | 14 | 6 | 9 | 20 | | |
| Capital Goods | 7 | 9 | 10 | 10 | 13 | 13 | 13 | 6 | 2 | | | |
| Transportation | 26 | 28 | 29 | 29 | 29 | 30 | 33 | 34 | 34 | 34. | | |
| Utilities | 72 | 72 | 72 | 73 | 73 | 63 | 41 | 15 | 7 | 6 | | |
| Textiles & Trade | 3 | 4 | 4 | 5 | 6 | <u>7</u> ~ | 5 | 8 | 10 | 7 | | |
| Services | 60 | 68 | 72 | 72 | 72 | 68 | 68 | 67 | 60 | 44 | | |
| Leisure | 39 | 40 | 43 | 39 | 37 | 37 | 37 | 40 | 35 | 28 · | | |
| Lending | 212 | 212 | 205 | 181 | 177 | 152 | 86 | 23 | 4 | 3 | | |
| Borrowing | | | • • • | 26 | 32 | 51 | 111 | 173 | 199 | 204 | | |
| Neither | | • • • | 7 | 5 | 3 | 9 | 15 | 16 | 9 | 5 | | |
| Max. Lent (%) | 100 | 100 | 100 | 100 | 100 | 100 | 100 | 100 | 100 | 100 | | |
| Min. Lent(%) | 55 | 33 | 1 | 17 | 8 | 1 | 6 | 1 | 84 | 100 | | |
| Loss Quarters | 26 | 31 | 36 | 51 | 57 | 68 | 77 | 81 | 82 | 86 | | |
| In O Industries | 6 | 4 | 4 | 4 | 3 | 3 | 3 | 3 | 3 | 3 | | |
| In l Industry | 84 | 68 | 61 | 56 | 56 | 56 | 72 | 97 | 132 | 199 | | |
| In 2 Industries | 75 | 85 | 84 | 86 | 83 | 92 | 89 | 82 | 62 | 10 | | |
| In 3 Industries | 42 | 48 | 51 | 50 | 52 | 44 | 39 | 28 | 13 | | | |
| In 4 Industries | 5 | 7 | 12 | 14 | 14 | 14 | 9 | 2 | 2 | | | |
| In 5 Industries | • • • | • • • | | 2 ^ | 4 | 3 | • • • | • • • | • • • | • • • | | |
| In '6 Industries | • • • | | • • • | • • • | ••• | ••• | | • • • | | • • • | | |
| In 7 Industries | | • • • | | • • • | | | ••• | | • • • | • • • | | |
| In 8 Industries | | • • • | | • • • | • • • | | | | • • • | • • • | | |
| In 9 Industries | • • • | • • • | | | • • • | • • • | | | • • • | • • • | | |
| In 10 Industries | ••• | • • • | • • • | • • • | • • • | | • • • | | | • • • | | |
| In 11 Industries | • • • | | | | • • • | | | | | • • • | | |
| In 12 Industries | • • • | • • • | ••• | • • • | • • • | • • • | ••• | ••• | ••• | ••• | | |

Summary of Portfolio Compositions of Power Policies Managing 12 Equal-Weighted Industries, With Leverage, 1934-86 (32 quarter estimating period, sum-of-the-digits probability assessment approach)

٤

193

Table 23b

| (32 | e-weig quarte pi | er esti cobabil | mating ity as | perio sessme | d, sum nt app | roach) | e-digi | ts | | , |
|-------------------|------------------------|--------------------|------------------|-----------------|------------------|--------|--------|-----------------|-------|-------|
| Number of | | | | | Powe | ers | | | | a |
| Times in: | -75 | - 50 | - 30 | -15 | -10 | - 5 | - 2 | 0 | . 5 | 1 |
| Petroleum | 97 | 101 | 105 | 110 | 110 | 110 | 101 | 65 | 53 | 36 + |
| Finance & R.E. | 11 | 11 | 12 | 12 | 14 | 15 | 16 | 6 | 4 | |
| Consumer Durables | 10 | 13 | 19 | 27 | 30 | 34 | 32 | 34 | 31 | 20 |
| Basic Industries | 21 | 24 | 27 | 27 | 25 | 22 | 18 | 13 | 8 | 2 |
| Food & Tobacco | 46 | 48 | 48 🌶 | - 48 | 50 | 45 | 35 | 26 | 15 | 6 |
| Construction | 7 | 11 | 14 | 14 | 16 | 16 | 17 | 17 | 18 | 23 |
| Capital Goods | 40 | 41 | · 42 | 44 | 45 | 48 | 45 | 38 | 30 | 19 |
| Transportation | 18 | 19 | 19 | 20 | 21 | 21 | 20 | ⁻ 13 | 11 | 8 |
| Utilities | 62 | 65 | 65 | 60 | 51 | 36 | 30 | 22 | 23 | 13 |
| Textiles & Trade | 23 | 28 | -29 | 30 | 30 | 29 | 24 | 17 | 15 | 9 |
| Services | 82 | 88 | 94 | 97 | 97 | 94 | 90 | 81 | 75 | 67 |
| Leisure | 22 | 23 | 26 | 27 | 29 | 31 | 32 | 30 | 29 | 29 |
| Lending | 212 | 212 | 191 | 180 | 172 | 147 | 85 | 29 | 11 | 4 |
| Borrowing | • • • | | | 31 | 34 | 55 | 108 | 160 | 182 | 200 |
| Neither | • • • | ••• | 21 | 1 | 6 | 10 | 19 | 23 | 19 | 8 |
| Max. Lent (%) | 100 | 100 | 100 | 100 | 100 | 100 | 100 | 100 | 100 | 100 |
| Min. Lent(%) | 46 | 19 | 3 | 4 | 7 | 2 | 2 | 2 | l | 100 |
| Loss Quarters | 29 | 32 | 40 | . 49 | 58 | 65 | 72 | 73 | 72 | 77 |
| In O Industries | 12 | 8 | 5 | 5 | 5 | 4 | 4 | 4 | 4 | 4 |
| In l Industry | 57 | 46 | 38 | 31 | 29 | 31 | 39 | 88 | 130 | 184 |
| In 2 Industries | 77 | 84 | 88 | 89 | 90 | 93 | 103 | 91 | 55 | 24 |
| In 3 Industries | 38 | 41 | 42 | 50 | 52 | 55 | 51 | 24 | 20 | |
| In 4 Industries | 26 | 30 | 36 | 29 | 29 | 26 | 13 | 5 | 3 | |
| In 5 Industries | 2 | 3 | 2 | 7 | 5 | 3 | 2 | | | |
| In 6 Industries | • • • | | 1 | 1 | 2 | | | | | |
| In 7 Industries | • • • | | | | | | | | • • • | |
| In 8 Industries | | | | | | | | | | |
| In 9 Industries | • • • | • • • | | | | | | • • • | | |
| In 10 Industries | | | | | | | | • • • | | |
| In 11 Industries | | | | | | | | | | |
| In 12 Industries | | | | | | | | | | |
| 12 1 | • • • | • • • | • • • | • • • | • • • | • • • | • • • | • • • | • • • | • • • |

Summary of Portfolio Compositions of Power Policies Managing 12 Value-Weighted Industries With Leverage 1934-86

| • | | - | | | - | | | | | |
|-------------------|-------|-------|-------|-------|-------|-------|-------|-------|-------|------------|
| Number of | | • | | | Pow | ers | | | | |
| Times in: | -75 | - 50 | - 30 | -15 | -10 | - 5 | -2 | 0 | .5 | 1 |
| Petroleum | 124 | 143 | 150 | 153 | 155 | 156 | 156 | 150 | 89 | 9 |
| Finance & R.E. | | | • • • | • • • | • • • | • • • | ••• | • • • | • • • | • • • |
| Consumer Durables | • • • | • • • | • • • | • • • | ••• | 1 | 2 | 4 | 9 | 208 |
| Basic Industries | 34 | 46 | 63 | 80 | 87 | 105 | 115 | 202 | 201 | |
| Food & Tobacco | 87 | 96 | 102 | 111 | 112 | 113 | 111 | 37 | 7 | • • • |
| Construction | • • • | • • • | • • • | • • • | | • • • | • • • | • • • | • • • | 5 |
| Capital Goods | | 2 | 2 | 3 | 3 | 3 | 2 | • • • | | • • • |
| Transportation | • • • | • • • | • • • | • • • | | • • • | • • • | • • • | 1 | ••• |
| Utilities | • • • | • • • | ••• | • • • | • • • | • • • | • • • | • • • | • • • | • • • |
| Textiles & Trade | • • • | • • • | • • • | | • • • | • • • | • • • | | | • • • |
| Services | 101 | 138 | 157 | 172 | 175 | 180 | 181 | 173 | 148 | • • • |
| Leisure | • • • | | | • • • | ••• | • • • | ••• | • • • | • • • | • • • |
| Lending | 212 | 212 | 212 | 212 | 212 | 212 | 212 | 51 | 5 | |
| Borrowing | | | | | | | | 118 | 196 | 205 |
| Neither | • • • | ••• | • • • | • • • | | ••• | ••• | 43 | 11 | 7 |
| Max. Lent (%) | 100 | 99 | 99 | 98 | 97 | 94 | 89 | 65 | 28 | |
| Min. Lent(%) | 97 | 96 | 94 | 88 | 82 | 68 | 35 | 2 | 5 | ••• |
| Loss Quarters | ` 15 | 17 | 18 | 25 | 31 | 50 | 64 | 80 | 82 | 83 |
| In O Industries | 6 | | | | | | | | | |
| In 1 Industry | 93 | 65 | 41 | 25 | 23 | 18 | 17 | 20 | 41 | 200 |
| In 2 Industries | 86 | 82 | 82 | 71 | 65 | 57 | 53 | 45 | 105 | 10 |
| In 3 Industries | 27 | 64 | 87 | 112 | 117 | 122 | 124 | , 132 | 60 | 2 |
| In 4 Industries | • • • | 1 | 2 | 4 | 7 | 15 | 18 | 15 | 6 | |
| In 5 Industries | | | • • • | | | •••• | | | | |
| In 6 Industries | • • • | | • • • | • • • | | • • • | | • • • | | • • • |
| In 7 Industries | ••• | | • • • | • • • | | • • • | | • • • | | |
| In 8 Industries | ••• | | • • • | • • • | • • • | • • • | • • • | • • • | | ••• |
| In 9 Industries | ••• | •_• • | | • • • | • • • | • • • | | • • • | | : * • • |
| In 10 Industries | | | | • • • | | • • • | | • • • | | |
| In 11 Industries | | | | | | | | | • • • | |
| In 12 Industries | ••• | | • • • | • • • | • • • | • • • | ••• | • • • | • • • | ••• |
| | | | | | | | | | | |

Summary of Portfolio Compositions of Power Policies Managing 12 Equal-Weighted Industries, With Leverage, 1934-86 (all-of-history simpleprobability assessment approach)

627

195

Ę

| Number of | | | | 6-1 | Pow | are | 29 | | | |
|-------------------|---------|-------|-------|---------|-------|-------|-------|---------|-------|------------|
| Times in: | -75 | - 50 | - 30 | -15 | -10 | - 5 | - 2 | 0 | .5 | 5 1 |
| Petroleum | 80 | 94 | 122 | 131 | 131 | 133 | 136 | 102 | 59 | 9 |
| Finance & R.E. | | | | ••• | | | | | | • • • |
| Consumer Durables | 18 | 52 | 79 | 100 | 108 | 120 | 133 | 165 | 191 | 208 |
| Basic Industries | 18 | 23 | 29 | 35 | 38 | 39 | 38 | 32 | 11 | |
| Food & Tobacco | 74 | 82 | 93 | 97 | 98 | 98 | 98 | 47 | 13 | |
| Construction | | | • • • | | • • • | • • • | • • • | • • • | • • • | 5 |
| Capital Goods | 69 | 78 | 86 | 92 | 92 | 93 | 91 | · 92 | 65 | • • • |
| Transportation | | | • • • | • • • - | • • • | • • • | • • • | • • • • | | • • • |
| Utilities | 、 • • • | 1 | 4 | . 5 | 7 | 8 | 7 | 5 | | |
| Textiles & Trade | • • • | • • • | • • • | | • • • | | • • • | • • • | • • • | |
| Services | • • • | 4 | 55 | 128 | 137 | 141 | 142 | 136 | 77 | • • • |
| Leisure | • • • | ••• | ••• | ••• | • • • | • • • | • • • | • • • | 1 | • • • |
| Lending | 212 | 212 | 212 | 212 | 212 | 212 | 212 | - 70 | 13 | 4 |
| Borrowing | | | | | | | | 87 | 174 | 199 |
| Neither | ••• | ••• | ••• | ••• | • • • | ••• | ••• | 55 | 25 | 9 |
| Max. Lent (%) | 100 | 100 | 100 | 100 | 100 | 100 | 100 | 100 | 100 | 100 |
| Min. Lent(%) | 97 | 95 | 92 | 85 | 78 | 59 | 19 | 1 | 4 | 100 |
| Loss Quarters | 15 | 16 | 19 | 27 | 33 | 47 | 58 | 67 | 74 | 77 |
| In O Industries | 22 | 9 | 7 | 6 | 5 | 5 | 4 | 4 | 4 | 4 |
| In l Industry | 125 | 93 | 47 | 26 | 23 | 22 | 18 | 18 | 67 | 194 |
| In 2 Industries | 61 | 91 | 86 | 54 | 52 | 45 | 46 | 61 | 82 | 14 |
| In 3 Industries | 4 | 17 | 43 | 63 | 60 | 61 | 61 | 81 | 51 | |
| In 4 Industries | • • • | . 2 | 25 | 51 | 58 | 60 | 64 | - 44 | 7 | |
| In 5 Industries | | | 4 | 11 | 12 | 17 | 18 | 4 | 1 | |
| In 6 Industries | | | | 1 | 2 | 2 | 1 | | | |
| In 7 Industries | | | •••• | | | | | | | |
| In 8 Industries | | | | | | | | | | |
| In 9 Industries | • • • | | | | | | | | | |
| In 10 Industries | • • • | • • • | | | | | • • • | | | • • • |
| In 11 Industries | | | | | | • • • | • • • | | • • • | |
| In 12 Industries | | | | | | | | | | |

Summary of Portfolio Compositions of Power Policies Managing 12 Value-Weighted Industries, With Leverage, 1934-86 (all-of-history simpleprobability assessment approach)

Table 25a

١.,

| (32 quarte | er esti simpl | lmating leprob <i>a</i> | g perio ability | od, dis 7 asses | aster ssment | state approa | scenar ich) | ioJ - | 1, | - |
|-------------------|------------------|----------------------------|--------------------|--------------------|-----------------|-----------------|----------------|-------|-------|-------|
| Number of | | | | | Pow | ers | | | | |
| Times in: | - 75 | - Š0 | - 30 | -15 | -10 | -5 | - 2 | 0 | .5 | 1 |
| Petroleum | 96 | 101 | 102 | 103 | 104 | 103 | 105 | 101 | 90 | 70 |
| Finance & R.E. | 9 | 9 | 9 | 9 | 11 | 12 | 14 | 13 | 9 | 2 |
| Consumer Durables | | | • • • | | | 1 | 2 | 3 | 5 | 6 |
| Basic Industries | 26 | 30 | 30 | 31 | 32 | 32 | .33 | 33 | 33 | 5 |
| Food & Tobacco | 17 | 18 | 20 | 20 | 20 | 21 | 18 | 8 | 4 | 1 |
| Construction | 6 | 7 | 8 | 8 | 8 | 8 | 9 | 3 | 9 | 23 |
| Capital Goods | 8 | 8 | 9 | 9 | 9 | 9 | 9 | 7 | 6 | 3 |
| Transportation | 11 | 15 | 18 | 21 | 22 | 22 | 26 | 35 | 34 | 38. |
| Utilities | 61 | 62 | 63 | 64 | 64 | 62 | 38 | 14 | 11 | 4 |
| Textiles & Trade | 6 | 8 | 8 | 8 | 8 | 9 | 9 | 5 | 4 | 10 |
| Services | 62 | 65 | 68 | 74 | 74* | 72 | ູ71 | 72 | 63 | 50 |
| Leisure | 20 - | ∽ 22 | 22 | 23 | 22 | 24 | 24 | 21 | 15 | 7 |
| Lending | 212 | 212 | 208 | 189 | 185 | 178 | 111 | 48 | 8 | • • • |
| Borrowing | • • • | • • • | | 21 | 23 | ʻ 32 | 80 | 141 | 186 | 200 |
| Neither | • • • | ••• | 4 | 2 | 4 | 2 | 21 | 23 | 18 | 12 |
| Max. Lent (%) | 100 | 100 | 99 | 99 | 98 | 97 | 93 | 80 | 60 | • • • |
| Min. Lent(%) | 54 | 31 | 2 | 28 | 5 | 8 | 2 | 1 | 1 | • • • |
| Loss Quarters | 21 | 26 | 30 | 39 | 46 | 59 | 69 | 79 | 80 | 83 |
| In O Industries | 6 | 1 | | ÷•• | • • • | ••• | | •••• | | • • • |
| In l Industry | 123 | 115 ° | 109 | 101 | 100 | 99 | 108 | 129 | 151 | 205 |
| In 2 Industries | 51 | 59 | 62 | 66 | 66 | 67 | 66 | 63 | 51 | · 7 |
| In 3 Industries | 31 | 36 | 40 | 43 | 42 | 42 | 34 - | 20 | 10 | • • • |
| In 4 Industries | 1 | 1 | 1 | 2 | 4 | 4 | 4 | | • • • | |
| In 5 Industries | • • • | • • • | | • • • | • • • | • • • | | ••• | | |
| In 6 Industries | ••• | • • • | • • • | • • • | • • • | • • • | | • • • | • • • | |
| In 7 Industries | • • • | • • • | | | | | | • • • | • • • | |
| In 8 Industries | • • • | • • • | | | | | | | | |
| In 9 Industries | ••• | • • • | | | • • • | · • • | | • • • | | • • • |
| In 10 Industries | • • • | • • • | | • • • | • • • | | | • • • | • • • | |
| In 11 Industries | • • • | | | | • • • | | | • • • | • • • | |
| In 12 Industries | | | | | • • • | | | • • • | • • • | • • • |

Summary of Portfolio Compositions of Power Policies Managing 12 Equal-Weighted Industries, With Leverage, 1934-86 (32 quarter estimating period, disaster state scenarioJ - 1, simpleprobability assessment approach)
Table 25b

| Number of Times in: -75 -50 -30 -15 -10 -5 -2 0 .5 Petroleum 75 81 86 90 89 90 90 64 53 4 Finance & R.E. 5 5 5 6 6 8 5 1 20 22 23 17 12 9 9 Food & Tobacco 21 24 23 22 18 13 7 6 Construction 6 10 15 16 17 18 17 18 19 2 Capital Goods 37 40 43 46 46 42 41 41 41 33 13 10 9 9 Capital Goods 37 40 43 46 46 42 41 41 41 33 12 10 Services 68 71 73 75 | (Sr quaree | simpl | eproba | bility | a, ars | sment | approa | ch) | 100 | - , | |
|--|-------------------|-------|--------|--------|------------|-------|--------|-------|-------------|------------|--------|
| Times in: -75 -50 -30 -15 -10 -5 -2 0 $.5$ Petroleum758186908990906453 $//$ Finance & R.E.55566851Consumer Durables1318202731312930-30 $//$ Basic Industries20202223171299Food & Tobacco2124232322181376Construction610151617181718192Capital Goods3740434646424141413Transportation1011121313131099Services6871737574747274645Leisure1519202122222222171Lending21221219518518117511763391Borrowing174432416261Max. Lent (%)10010010010010010010010010010010Min. Lent(%)45186218814 | Number of | | | | | Pow | ers | | - | | ****** |
| Petroleum75818690899090645364Finance & R.E.55566851Consumer Durables1318202731312930-302Basic Industries20202223171299Food & Tobacco2124232322181376Construction610151617181718192Capital Goods3740434646424141412Transportation1011121313131099Utilities52535555523123 42 10Services6871737574747274645Leisure1519202122222222171Lending21221219518518117511763391Neither174432416261Max. Lent (%)100100100100100100100100100100100Min. Lent(%)45186218814510 | Times in: | -75 | - 50 | - 30 | -15 | -10 | - 5 | - 2 | 0 | , 5 | 1 |
| Finance & R.E. 5 5 5 6 6 8 5 1 Consumer Durables 13 18 20 27 31 31 29 30- 30 30 Basic Industries 20 20 22 23 17 12 9 9 Food & Tobacco 21 24 23 23 22 18 13 7 6 Construction 6 10 15 16 17 18 17 18 19 2 Capital Goods 37 40 43 46 46 42 41 41 41 3 Transportation 10 11 12 13 13 13 10 9 Utilities 52 53 55 52 31 23 ¥3 12 10 Services 68 71 73 75 74 74 72 74 64 5 Leisure 15 19 20 21 22 | Petroleum | 75 | 81 | 86 | 90 | 89 | 90 | 90 | 64 | 53 | 40 |
| Consumer Durables 13 18 20 27 31 31 29 30 30 2 Basic Industries 20 20 22 23 17 12 9 9 Food & Tobacco 21 24 23 23 22 18 13 7 6 Construction 6 10 15 16 17 18 17 18 19 2 Capital Goods 37 40 43 46 46 42 41 41 41 21 Transportation 10 11 12 13 13 13 13 10 9 Utilities 52 53 55 52 31 23 \$23 12 10 Services 68 71 73 75 74 74 72 74 64 5 Leisure 15 19 20 21 22 22 22 17 1 Max. Lent (%) 100 100 100 1 | Finance & R.E. | 5 | 5 | 5 | ' 6 | 6 | 8 | 5 | 1 | | |
| Basic Industries 20 20 20 22 23 17 12 9 9 Food & Tobacco 21 24 23 23 22 18 13 7 6 Construction 6 10 15 16 17 18 17 18 19 2 Capital Goods 37 40 43 46 46 42 41 41 41 2 Transportation 10 11 12 13 13 13 10 9 Utilities 52 53 55 52 31 23 ¥3 12 10 Services 68 71 73 75 74 74 74 64 5 Leisure 15 19 20 21 22 22 22 17 1 Lending 212 212 195 185 181 175 117 63 39 1 Borrowing 17 4 3 | Consumer Durables | 13 | 18 | 20 | 27 | 31 | 31 | 29 | 30- | 30 | 24 |
| Food & Tobacco 21 24 23 23 22 18 13 7 6 Construction 6 10 15 16 17 18 17 18 19 2 Capital Goods 37 40 43 46 46 42 41 41 41 3 Transportation 10 11 12 13 13 13 13 10 9 Utilities 52 53 55 52 31 23 ¥3 12 10 Services 68 71 73 75 74 74 72 74 64 52 Leisure 15 19 20 21 22 22 22 17 1 Lending 212 212 195 185 181 175 117 63 39 1 Borrowing 17 4 3 24 16 26 1 Max. Lent (%) 100 100 100 | Basic Industries | 20 | 20 | 20 | 22 | 23 | 17 | 12 | 9 | 9 | 1 |
| Construction 6 10 15 16 17 18 17 18 19 2 Capital Goods 37 40 43 46 46 42 41 < | Food & Tobacco | 21 | 24 | 23 | 23 | 22 | 18 | 13 | 7 | 6 | 3 |
| Capital Goods 37 40 43 46 46 42 41 <td>Construction</td> <td>6</td> <td>10</td> <td>15</td> <td>16</td> <td>17</td> <td>18</td> <td>17</td> <td>18</td> <td>19</td> <td>20</td> | Construction | 6 | 10 | 15 | 16 | 17 | 18 | 17 | 18 | 19 | 20 |
| Transportation 10 11 12 13 13 13 13 13 13 13 10 9 Utilities 52 53 55 55 52 31 23 53 12 10 Services 68 71 73 75 74 74 72 74 64 52 Leisure 15 19 20 21 22 22 22 17 1 Lending 212 212 195 185 181 175 117 63 39 1 Borrowing 23 27 34 71 133 147 16 Neither 17 4 4 3 24 16 26 1 Max. Lent (%) 100 100 100 100 100 100 100 100 100 100 100 100 100 100 100 100 100 100 100 10 10 <t< td=""><td>Capital Goods</td><td>37</td><td>40</td><td>43</td><td>46</td><td>46</td><td>42</td><td>41</td><td>41</td><td>41</td><td>35</td></t<> | Capital Goods | 37 | 40 | 43 | 46 | 46 | 42 | 41 | 41 | 41 | 35 |
| Utilities 52 53 55 55 52 31 23 53 12 1 Textiles & Trade 21 21 23 23 24 25 23 12 10 Services 68 71 73 75 74 74 72 74 64 5 Leisure 15 19 20 21 22 22 22 17 1 Lending 212 212 195 185 181 175 117 63 39 1 Borrowing 23 27 34 71 133 147 16 Neither 17 4 4 3 24 16 26 1 Max. Lent (%) 100 100 100 100 100 100 100 100 100 100 100 100 100 100 100 100 100 100 100 10 10 10 10 10 < | Transportation | 10 | 11 | 12 | 13 | 13 | 13 | 13 | 10 | . 9 | 6 |
| Textiles & Trade 21 21 23 23 24 25 23 12 10 Services 68 71 73 75 74 74 72 74 64 5 Leisure 15 19 20 21 22 22 22 22 17 1 Lending 212 212 195 185 181 175 117 63 39 1 Borrowing 23 27 34 71 133 147 18 Neither 17 4 4 3 24 16 26 1 Max. Lent (%) 100 <td>Utilities</td> <td>52</td> <td>53</td> <td>55</td> <td>55</td> <td>52</td> <td>31</td> <td>23</td> <td>}₽3</td> <td>12</td> <td>10</td> | Utilities | 52 | 53 | 55 | 55 | 52 | 31 | 23 | } ₽3 | 12 | 10 |
| Services 68 71 73 75 74 74 72 74 64 5 Leisure 15 19 20 21 22 22 22 22 22 22 17 1 Lending 212 212 212 195 185 181 175 117 63 39 1 Borrowing 23 27 34 71 133 147 18 Neither 17 4 4 3 24 16 26 16 Max. Lent (%) 100 10 10 10 10 10 10 10 10 10 10 10< | Textiles & Trade | 21 | 21 | 23 | 23 | 24 | 25 | 23 | 912 | 10 | 5 |
| Leisure 15 19 20 21 22 22 22 22 17 1 Lending 212 212 195 185 181 175 117 63 39 1 Borrowing 23 27 34 71 133 147 18 Neither 17 4 4 3 24 16 26 1 Max. Lent (%) 100 </td <td>Services</td> <td>68</td> <td>71</td> <td>73</td> <td>75</td> <td>74</td> <td>74</td> <td>72</td> <td>74</td> <td>64</td> <td>57</td> | Services | 68 | 71 | 73 | 75 | 74 | 74 | 72 | 74 | 64 | 57 |
| Lending 212 212 195 185 181 175 117 63 39 1 Borrowing 23 27 34 71 133 147 18 Neither 17 4 4 3 24 16 26 1 Max. Lent (%) 100 <t< td=""><td>Leisure</td><td>15</td><td>19</td><td>20</td><td>21</td><td>22</td><td>. 22</td><td>22</td><td>22</td><td>17</td><td>13</td></t<> | Leisure | 15 | 19 | 20 | 21 | 22 | . 22 | 22 | 22 | 17 | 13 |
| Borrowing 23 27 34 71 133 147 18 Neither 17 4 4 3 24 16 26 1 Max. Lent (%) 100 10 | Lending | 212 | 212 | 195 | 185 | 181 | 175 | 117 | 63 | 39 | 10 |
| Neither 17 4 4 3 24 16 26 1 Max. Lent (%) 100 <td< td=""><td>Borrowing</td><td></td><td></td><td>• • •</td><td>23</td><td>27</td><td>34</td><td>71</td><td>133</td><td>147</td><td>183</td></td<> | Borrowing | | | • • • | 23 | 27 | 34 | 71 | 133 | 147 | 183 |
| Max. Lent (%) 100 | Neither | ••• | ••• | 17 | 4 | 4 | 3 | 24 | 16 | 26 | 19 |
| Min. Lent(%) 45 18 6 2 18 8 1 4 5 10 Loss Quarters 22 29 36 40 48 53 62 75 74 7 In 0 Industries 34 20 11 11 10 10 10 10 1 In 1 Industries 34 20 11 11 10 10 10 1 1 1 10 10 10 1 | Max. Lent (%) | 100 | 100 | 100 | 100 | 100 | 100 | 100 | 100 | 100 | 100 |
| Loss Quarters 22 29 36 40 48 53 62 75 74 74 In 0 Industries 34 20 11 11 10 11 10 10 10 3 10 3 10 3 10 3 11 10 10 3 11 11 11 11 11 11 11 11 | Min. Lent(%) | 45 | 18 | 6 | 2 | 18 | 8 | 1 | 4 | 5 | 100 |
| In 0 Industries 34 20 11 11 10 | Loss Quarters | 22 | 29 | 36 | 40 | 48 | 53 | . 62 | 75 | 74 | 73 |
| In 1 Industry 91 97 97 86 82 80 84 115 137 19 In 2 Industries 41 46 52 59 64 73 83 76 62 1 In 3 Industries 23 21 23 24 26 34 30 10 3 In 4 Industries 17 22 23 23 23 14 5 1 <td>In 0 Industries</td> <td>34</td> <td>20</td> <td>11</td> <td>11</td> <td>10</td> <td>10</td> <td>10</td> <td>10</td> <td>10</td> <td>10</td> | In 0 Industries | 34 | 20 | 11 | 11 | 10 | 10 | 10 | 10 | 10 | 10 |
| In 2 Industries 41 46 52 59 64 73 83 76 62 1 In 3 Industries 23 21 23 24 26 34 30 10 3 In 4 Industries 17 22 23 23 23 14 5 1 In 5 Industries 3 3 6 4 1 | In l Industry | 91 | 97 | 97 | 86 | 82 | 80 | 84 | 115 | 137 | 190 |
| In 3 Industries 23 21 23 24 26 34 30 10 3 In 4 Industries 17 22 23 23 23 14 5 1 In 5 Industries 3 3 6 4 1 <td>In 2 Industries</td> <td>41</td> <td>46</td> <td>52</td> <td>59</td> <td>64</td> <td>73</td> <td>83</td> <td>76</td> <td>62 ັ</td> <td>12</td> | In 2 Industries | 41 | 46 | 52 | 59 | 64 | 73 | 83 | 76 | 62 ັ | 12 |
| In 4 Industries 17 22 23 23 23 14 5 1 In 5 Industries 3 3 6 4 1 </td <td>In 3 Industries</td> <td>23</td> <td>21</td> <td>23</td> <td>24</td> <td>26</td> <td>34</td> <td>30</td> <td>10</td> <td>3</td> <td></td> | In 3 Industries | 23 | 21 | 23 | 24 | 26 | 34 | 30 | 10 | 3 | |
| In 5 Industries 3 3 6 4 1 <td< td=""><td>In 4 Industries</td><td>17</td><td>22</td><td>23</td><td>23</td><td>23</td><td>14</td><td>5</td><td>1</td><td></td><td></td></td<> | In 4 Industries | 17 | 22 | 23 | 23 | 23 | 14 | 5 | 1 | | |
| In 6 Industries 3 3 2 2 < | In 5 Industries | 3 | 3 | 3 | 6 | 4 | 1 | • • • | | | |
| In 7 Industries | In 6 Industries | 3 | 3 | 3 | 2 | 2 | | • • • | • • • | • • • | |
| In 8 Industries | In 7 Industries | • • • | | | 1 | 1 | • • • | • • • | | | |
| | In 8 Industries | • • • | | • • • | • • • | | | • • • | • • • | • • • | |
| In 9 Industries | In 9 Industries | ••• | • • • | • • • | | • • • | | | | • • • | |
| In 10 Industries | In 10 Industries | • • • | • • • | • • • | • • • | ••• | | | | | |
| In 11 Industries | In 11 Industries | ••• | | • • • | | • • • | | | | | |
| In 12 Industries | In 12 Industries | ••• | • • • | • • • | ••• | • • • | | • • • | | • • • | • • • |

Summary of Portfolio Compositions of Power Policies Managing 12 Value-Weighted Industries, With Leverage, 1934-86 (32 quarter estimating period, disaster state scenarioJ = 1, simpleprobability assessment approach)

198

Table 26a

| (1 | simpl | eproba | bility | asses | sment | approa | ich) | | - | |
|------------------------|-------|--------|---------|-------|-------|------------|------------|-------|-------|---------|
| Number of | | | | | Pow | vers | | | i i | |
| Times in: | - 75 | - 50 | - 30 | -15 | -10 | - 5 | - 2 | 0 | . 5 | 1 |
| Petroleum | 42 | 47 | 54 | 57 | 56 | 55 | 55 | 54 | 53 | 42 |
| Finance & R.E. | 6 | 6 | 6 | 6 | . 6 | 6 | 8 | 7 | 7 | 7 |
| Consumer Durables | | | • • • | 1 | 1 | ° 1 | 1 | 1 | 2 | • • • |
| Basic Industries | 3 | 4 | 4 | 9 | 12 | 14 | 14 | 14 | 14 | 9 |
| Food & Tobacco | 6 | 6 | 7 | 8 | 8 | 9 | 9 | 8 | 6 | 4 |
| Construction | 4 | 5 | 5 | 16 | 16 | 17 | 17 | 15 | 16 | 23 |
| Capital Goods | • • • | | • • • • | • • • | • • • | | • • • | • • • | • • • | • • • |
| Transportation | | 4 | 4 | 9 | 11 | 12 | 12 | 15 | 17 | 20 |
| Utilities | 33 | 34 | 34 | 35 | 35 | 35 | 33 | 13 | 8 | • • • |
| Textiles & Trade | 4 | 4 | 6 | 6 | 7 | 7 | 8 | 8 | 6 | 2 |
| Services | 37 | 39 | 43 | 46 | 47 | 46 | 47 | 44 | 46 | 40 |
| Leisure | 1 | 2 | 3 | 3 | 3 | 4 | 4 | 4 | 4 | 1 |
| Lending | 212 | 212 | 212 | 191 | 189 | 187 | 178 | 146 | 116 | 66 |
| Borrowing | • • • | | | 13 | 21 | 23 | 32 | 51 | 77 | 127 |
| Neither | | ••• | • • • | 8 | 2 | . 2 | 2 | 15 | 19 | 19 |
| Max. Lent (%) | 100 | 100 | 100 | 100 | 100 | 100 | 100 | 100 | 100 | 100 |
| Min. Lent(%) | 61 | 42 | 4 | 1 | 44 | 3 | 17 | 3 | 3 | 100 |
| Loss Quarters | 12 | 12 | 19 | 28 | 33 | 38 | 46 | 51 | 55 | 61 |
| In O Industries | 117 | 107 | 98 | 75 | 69 | 67 | 66 | 66 | 66 | 66 |
| In l Industry | 62 | 68 | 73 | 89 | 94 | 93 | 9 5 | 112 | 114 | 144 |
| In 2 Industries | 26 | 29 | 31 | 38 | 40 | 43 | 41 | 31 | 31 | 2 |
| In 3 Industries | 6 | 7 | 9 | 9 | 8 | 9 | 9 | 3 | 1 | |
| In 4 Industries | 1 | 1 | 1 | 1 | 1 | ••• | 1 | • • • | • • • | |
| In 5 Industries | • • • | • • • | • • • | • • • | • • • | | • • • | • • • | • • • | |
| In <u>6</u> Industries | • • • | | • • • | | • • • | • • • | • • • | • • • | • • • | • • • |
| In 7 Industries | • • • | | • • • | | • • • | • • • | (| | • • • | • • • |
| In 8 Industries | • • • | | • • • | • • • | • • • | | ••• | ••• | ••• | • • • |
| In 9 Industries | • • • | • • • | • • • | • • • | • • • | | • • • | • • • | | |
| In 10 Industries | • • • | | • • • | • • • | • • • | • • • | • • • | • • • | • • • | • • • |
| In 11 Industries | ••• | • • • | • • • | • • • | • • • | • • • | • • • | • • • | | • • • • |
| In 12 Industries | • • • | • • • | • • • | • • • | • • • | ••• | • • • | • • • | ••• | ••• |

Summary of Portfolio Compositions of Power Policies Managing 12 Equal-Weighted Industries, With Leverage, 1934-86 (32 quarter estimating period, disaster state scenarioJ = 6, simpleprobability assessment approach)

Table 26b

| | | | | | ~ | | | | | |
|------------------------|------------|--------|------|-------------|-----|-------|-----|-----|--------|-------|
| Number of Fimes in: | - 75 | - 50 | - 30 | -15 | -10 | -5 | -2 | 0 | 5 | 1 |
| | | | | | | | | | | |
| Petroleum | 30 | 32 | 34 | 39 | 40 | 43 | 41 | 33 | 31 | 16 |
| Finance & R.E. | 3 | 3 | 3 | 3 | 3 | 4 | 4 | 2 | • • • | • • • |
| Consumer Durables | <u>∼</u> 9 | 9 | 10 | 12 | 12 | 13 | 13 | 13 | 14 | 13 |
| Basic Industries | 12 | 12 | 12 | 13 | 13 | 9 | 8 | 3 | 2 | • • • |
| Food & Tobacco | 22 | 22 | 22 | 20 | 20 | 17 | 9 | 6 | 5 | 4 |
| Construction | • • • | 1 | 1 | 3 | 4 | 4 | 7 | 6 | 7 | 10 |
| Capital Goods | 6 | 9 | 10 | 13 | 14 | 15 | 8 | | | |
| Transportation | • • • | 2 | 2 | 2 | 2 | 2 | 2 | 2. | 3 | 2 |
| Utilities | 38 | 38 | 38 | 36 | 32 | 22 | 8 | 6 | 5 | 3 |
| Textiles & Trade | 1 | 1 | 1 | 1 | 1 | 1 | 1 | · 2 | 3 | 1 |
| Services ° | 50 | 55 | 55 | 56 | 57 | 58 | 59 | 58 | 56 | 56 |
| Leisure | 4 | 5 | 7 | 9 | 9 | 9 | 9 | 9 | 8 | 7 |
| Lending | 212 | 212 | 207 | 189 | 188 | 180 | 176 | 146 | 121 | 101 |
| Borrowing | | | | ` 16 | 23 | 23 | 33 | 46 | 75 | 100 |
| Neither | ••• | • • 42 | 5 | 7 | 1 | 9 | 3 | 20 | 16 | 11 |
| Max. Lent (%) | 100 | 100 | 100 | 100 | 100 | 100 | 100 | 100 | 100 | 100 |
| Min. Lent(%) | 54 | 32 | 1 | 31 | 5 | 27 | 5 | 1 | 9 | 100 |
| Loss Quarters | 11 | 13 | 16 | 20 | 22 | 27 | 33 | 35 | 36 | 40 |
| In O Industries | 119 | 111 | 108 | 106 | 106 | 104 | 102 | 101 | 101 | 101 |
| In 1 Industry | 52 | 58 | 60 | 56 | 55 | 55 | 65 | 83 | 89 | 110 |
| In 2 Industries | 11 | 12 | 13 | 19 | 21 | 22 | 33 | 27 | 21 | 1 |
| In 3 Industries | 22 | 20 | 18 | 18 | 16 | 26 | 10 | 1 | 1 | |
| In 4 Industries | 5 | 8 | 9 | 8 | 10 | 5 | 2 | | | |
| In 5 Industries | 3 | 3 | 4 | 3 | 2 | • • • | | | | |
| In 6 Industrieș | • • • | • • • | | 2 | 2 | • • • | | | | |
| In 7 Industries | • • • | | | | | • • • | | | | |
| In 8 Industries | | | | | | • • • | | | | • • • |
| In 9 Industries | | • • • | | | | | | | · . ·. | ÷ |
| In 10 Industries | | | | | | • • • | | | | |
| In 11 Industries | | | • | • • • | | • • • | | | | |
| | | | | | | | | | | |

Summary of Portfolio Compositions of Power Policies Managing 12 Value-Weighted Industries, With Leverage, 1934-86 (32 quarter estimating period, disaster state scenarioJ - 6, simpleprobability assessment approach)

ì

Table 27a

| | | | · · · · · · · · · · · · · · · · · · · | ······· | | | | | | |
|-------------------|---------|------------|---------------------------------------|---------|-------|----------|-------------|-------|-------|-------|
| Number of | 76 | 50 | 20 | 15 | Powe | ers 5 | 2 | 0 | 5 | 1 |
| limes in: | -75 | - 50 | · - 20 | -15 | -10 | | - 2 | | | ۲ |
| Petroleum | 4 | 4 | 7 | 8 | 8 | 10 | 10 | 9 | 9 | 7 |
| Finance & R.E. | 6 | 6 9 | 6 | 6 | 6 | 6 | 6 | 5 | 5 | 5 |
| Consumer Durables | | • • • | | | • • • | • • •. • | • • • | • • • | • • • | • • • |
| Basic Industries | · • • • | • • • | 1 | 1 | 1 | 1 | 1 | · 2 | 1 | |
| Food & Tobacco | • • • | | | | • • • | • • • | • • • | • • • | • • • | 1 |
| Construction | 5 | 5 | 5 | 6 | 8 | 8 | 8 | 7 | 7 | 9 |
| Capital Goods | | • • • | | • • • | ••• | ••• | • • • | • • • | • • • | |
| Transportation | • • • | • • • | | • • • | • • • | • • • | 1 | 1 | 1 | 2 |
| Utilities | 29 | 31 | 31 | 31 | 31 | 31 | 31 | 18 | 16 | 7 |
| Textiles & Trade | • • • | | | • • • | • • • | | 1 | 1 | 1 | • • • |
| Services | 24 | 25 | 27 | 28 | €29 | 28 | 28 | 26 | 25 | 25 |
| Leisure | • • • | ••• | • • • | ••• | • • • | • • • | | ••• | ••• | • • • |
| Lending | 212 | 212 | 212 | 194 | 190 | 189 | Ì 86 | 178 | 173 | 156 |
| Borrowing | ••• | • • • | | 7 | 17 | 23 | .23 | 31 | 33 ີ | - 44 |
| Neither | ••• | • • • | • • • | 11 | 5 | ••• | 3 | 3 | 6 | 12 |
| Max. Lent (%) | 100 | 100 | 100 | 100 | 100 | 100 | 100 | 100 | 100 | 100 |
| Min. Lent(%) | 64 | 46 | 11 | 13 | 14 | 48 | 5 | 12 | 27 | 100 |
| Loss Quarters | 7 | 8 | 10 | 12 | 13 | 14 | 15 | 19 | 20 | 23 |
| In O Industries | 173 | 170 | 164 | 162 | 159 | 158 | 157 | 156 | 156 | 156 |
| In l Industry | 14 | 17 | 23 | 24 | 27 | 28 | 28 | 44 | 48 | 56 |
| In 2 Industries | 21 | 21 | 21 | 22 | 22 | 22 | 23 | 11 | 7 | |
| In 3 Industries | 4 | 4 | 4 | 4 | 4 | 4 | 4 | 1 | 1 | |
| In 4 Industries | ,••• | | • • • | | ••• | | • • • | | | |
| In 5 Industries | • • • | • • • | • • • | • • • | • • • | • • • | | • • • | | |
| In 6 Industries | • • • | • • • | • • • | • • • | • • • | • • • | | ••• | • • • | |
| In 7 Industries | • • • | ••• | • • • | | • • • | | • • • | ••• | • • • | |
| In 8 Industries | • • • | • • • | • • • | | • • • | • • • | • • • | | • • • | |
| In 9 Industries | • • • | • • • | • • • | | • • • | • • • | • • • | | • • • | |
| In 10 Industries | • • • | • • • | • • • | • • • | • • • | • • • | • • • | | • • • | |
| In 11 Industries | • • • | • • • | • • • | • • • | • • • | ••• | • • • | | ••• | |
| In 12 Industries | | • • • | | ••• | • • • | • • • | • • • | • • • | | |

Summary of Portfolio Compositions of Power Policies Managing 12 Equal-Weighted Industries, With Leverage, 1934-86 (32 quarter estimating period, disaster state scenarioJ = 12, simpleprobability assessment approach)

Value-Weighted Industries, With Leverage, 1934-86 (32 quarter estimating period, disaster state scenarioJ - 12. simpleprobability assessment approach) Number of Powers Times in: -75 - 50 - 30 -15 -10 - 5 - 2 0 . 5 1 2 3 2 2 Petroleum 1 2 3 2 2 2 Finance & R.E. 2 3 3 3 3 3 3 2 9 11 12 12 12 13 **t**2 Consumer Durables 12 10 9 8 9 9 2 **Basic Industries** 8 8 8 5 1 . . . 20 20 20 19 19 18 7 3 3 Food & Tobacco 3 1 Construction . 5 4 Capital Goods 4 4 4 2 Transportation . Utilities 34 35 35 28 24 22 9 5 5 4 Textiles & Trade . Services 23 25 29 34 34 35 35 34 34 34 Leisure . 212 191 189 185 180 175 160 Lending 212 212 176 21 Borrowing 8 23 24 32 33 46 • • • Neither 2 8 13 4 4 4 6 , 100 100 100 100 100 100 Max. Lent (%) 100 100 100 100 25 60 40 39 12 24 27 Min. Lent(%) 1 8 100 5 7 7 9 Loss Quarters 9 13 15 16 16 16 In 0 Industries 175 172 168 161 161 160 160 160 160 160 In 1 Industry 4 7 11 18 18 18 33 44 49 52 In 2 Industries 8 6 6 14 19 17 16 8 3 . . . 20 22 In 3 Industries 22 13 12 12 2 ه هر ه In 4 Industries 4 2 1 2 3 3 In 5 Industries 3 1 4 4 2 In 6 Industries . In 7 Industries . In 8 Industries . · · · · In 9 Industries . In 10 Industries . In 11 Industries :• *• • . . . ÷. In 12 Industries

Summary of Portfolio Compositions of Power Policies Managing 12

. . .

. . .

. . .

. . .

. . .

. . .

• • •

. . .

. . .

. . .

Paired t-tests : Selected Active Strategies Managing 12 Equal-Weighted Industry Indices vs. Passive and Semi-Passive Strategies (32 quarter estimating period, simple probability assessment approach)

| Comparison | | c | d | $\sigma(d)$ | t |
|---------------------------------------|-------------|-------------------|--------------|-------------|--------|
| · · · · · · · · · · · · · · · · · · · | Panel A : 1 | .934-86, Wi | Ith Leverage | | ······ |
| Power -75 vs. Risk-fr | cee asset | | .0022 | .0107 | 3.04** |
| Power -50 vs. Risk-fr | cee asset | | .0033 | .0161 | 2.97** |
| Power -30 vs. Risk-fr | cee asset | ~ | .0054 | .0256 | 3.05** |
| Power -5 vs. Portfol | Lio V8 | $\langle \rangle$ | ,0019 | .0519 | .54 |
| Power -2 vs. Portfol | lio V12 | | .0043 | .0720 | .86 |
| Power O vs. Portfol | io V2O | | .0091 | . 1022 | 1.30 |
| Power 0 vs. Service | s | | . 0040 | .1063 | .55 |
| | Panel B : 1 | 966-86, Wi | th Leverage | | |
| Power -75 vs. Risk-fr | ee asset | | .0010 | . 0047 | 1.99* |
| Power -50 vs. Risk-fr | ee asset | | .0015 | .0070 | 1.98* |
| Power -30 vs. Risk-fr | ee asset | · · · | .0025 | .0115 | 1.97* |
| Power -30 vs. Portfol | io V2 | | .0010 | .0115 | .80 |
| Power -15 vs. Portfol | io V2 | | .0032 | .0147 | 1.99* |
| Power -10 vs. Portfol | io V4 | | .0040 | .0238 | 1.54 |
| Power -5 vs. Portfol | io V8 | • | .0069 | .0466 | 1.36 |
| Power -2 vs. Portfol | ío V12 | | .0145 | .0714 | 1.86* |
| Power -2 vs. Portfol | .io V14 | | .0153 | .0809 | 1.73* |
| Power -2 vs. Portfol | .io E10 | | .0033 | .0836 | .36 |
| Power 0 vs. Portfol | io V20 | | .0263 | .1151 | 2.10* |
| Power 0 vs. Portfol | io El4 | • | .0109 | .1101 | .91 |
| Power O vs. Service | S · | | .0104 | . 1146 | .83 |
| Power .5 vs. Portfol | io E18 | | .0026 | 1351 | 17 |

Table 28 (continued)

| Compan | riso | n | | ā | σ(d) | t |
|--------|------|-----|-----------------------|------------------|----------------|-------|
| | • | | Panel C : 1966-86, | Without Leverage | | |
| Power | -75 | vs. | Risk-free asset | .0010 | .0047 | 1.99* |
| Power | - 50 | vs. | Risk-free asset | .0015 | .0070 | 1.98* |
| Power | - 30 | vs. | Risk-free asset | .0025 | .0115 | 1.97* |
| Power | - 30 | vs. | Portfolio V2 | .0010 | .0115 | . 80 |
| Power | -15 | vs. | Portfolio V2 | .0032 | .0147 | 1.99* |
| Power | -10 | vs. | Portfolio V4 | ,0040 | .0238 | 1.54 |
| Power | - 5 | ٧s. | Portfolio V8 | .0069 | .0465 | 1.37 |
| Power | - 2 | vs. | Portfolio V10 | .0114 | .0576 | 1.81* |
| Power | - 2 | vs. | Portfolio E8 | .0033 | .0649 | .46 |
| Power | 0 | vs. | Petroleum | .0073 | .0729 | . 92 |
| Power | 0 | vs. | Finance & Real Estate | .0059 | .0889 | . 61 |
| Power | 0 | vs. | Construction | .0048 | .0898 | .49 |
| Power | 0 | vs. | Capital Goods | <i>:</i> °0072 | .0804 | . 82 |
| Power | 0 | vs. | Transportation | .0092 | .0929 | . 91 |
| Power | . 5 | vs. | Petroleum | .0068 | .0749 | .83 |
| Power | . 5 | vs. | Finance & Real Estate | .0053 | , 091 0 | .54 |
| Power | . 5 | vs. | Construction | ,0042 | .0912 | . 42 |
| Power | . 5 | vs. | Capital Goods | .0066 🛝 | .0802 | .76 |
| Power | . 5 | vs. | Transportation | .0087 | .0939 | . 84 |
| Power | 1 | vs. | Petroleum | .0058 | .0779 | .68 |
| Power | 1 | vs. | Finance & Real Estate | .0043 | .0923 | .43 |
| Power | 1 | vs. | Construction | .0032 | .0924 | . 32 |
| Power | 1 | vs. | Capital Goods | .0056 | .0805 | . 64 |
| Power | 1 | vs. | Transportation | .0077 | .0943 | .74 |

* Significant at the 5% level. * ** Significant at the 1% level.

Ø

Table 29

۰.

Paired t-tests : Selected Active Strategies Managing 12 Value-Weighted Industry Indices vs. Passive Strategies

(32 quarter estimating period, simple probability assessment approach)

| Compa | rison | | | d | σ(d) | t |
|---|--|--|--|---|---|---|
| <u> </u> | | | Panel A : 1966-86 | , With Leverage | | ···· · |
| ₽ower | -75 v | s. Risk-fr | ee asset | .0007 | .0048 | 1.42 |
| Power | -50 v | s. Risk-fr | ee asset | .0011 | .0071 | 1.41 |
| Power | -30 v | s. Risk-fr | ee asset | .0018 | .`0116 | 1.40 |
| Power | -15 v | s. Portfol | io V2 | .0019 | .0178 | . 96 |
| Power | -10 v | s. Portfol | io V4 | ,0020 | .0290 | .65 |
| Power | -5 v | s. Portfol | io V6 | .0032 | .0449 | .65 |
| Power | -5 v | s. Utiliti | es | .0016 | .0636 | . 23 |
| Power | -2 v | s. Portfol: | io V12 | .0058 | .0808 | .65 |
| Power | -2 v | s. Finance | & Real Estate | .0014 | .0886 | .15 |
| Power | -2 v | s. Capital | Goods | .0077 | .0879 | . 80 |
| Power | -2 v | s. Transpo | rtation | .0098 | .0995 | . 90 |
| | | | Panel B : 1966-86, | Without Leverage | | |
| Power | -75 v | s. Risk-fr | ee asset | .0007 | .0048 | 1.42 |
| TOMOT | | | | | | |
| Power | -50 v | s. Risk-fr | ee asset | .0011 | .0071 | 1.41 |
| Power Power | -50 v -30 v | s. Risk-fro s. Risk-fro | ee asset ee asset | .0011 .0018 | .0071 .0116 | 1.41 1.40 |
| Power Power Power | -50 v -30 v -15 v | s. Risk-fro s. Risk-fro s. Portfol | ee asset ee asset io V2 | .0011 .0018 .0019 | .0071 .0116 .0178 | 1.41 1.40 .96 |
| Power Power Power Power | -50 v -30 v -15 v -10 v | s. Risk-fr s. Risk-fr s. Portfol s. Portfol | ee asset ee asset io V2 io V4 | .0011 .0018 .0019 .0020 | .0071 .0116 .0178 .0290 | 1.41 1.40 .96 .65 |
| Power Power Power Power Power | -50 v -30 v -15 v -10 v -5 v | s. Risk-fro s. Risk-fro s. Portfol s. Portfol s. Portfol | ee asset ee asset io V2 io V4 io V6 | .0011 .0018 .0019 .0020 .0029 | .0071 .0116 .0178 .0290 .0424 | 1.41 1.40 .96 .65 .63 |
| Power Power Power Power Power Power | -50 v -30 v -15 v -10 v -5 v -5 v | s. Risk-fro s. Risk-fro s. Portfol s. Portfol s. Portfol s. Portfol | ee asset ee asset io V2 io V4 io V6 io V8 | .0011 .0018 .0019 .0020 .0029 .0023 | .0071 .0116 .0178 .0290 .0424 .0522 | 1.41 1.40 .96 .65 .63 .41 |
| Power Power Power Power Power Power Power | -50 v -30 v -15 v -10 v -5 v -5 v -2 v | s. Risk-fro s. Risk-fro s. Portfol s. Portfol s. Portfol s. Portfol | ee asset ee asset io V2 io V4 io V6 io V8 io V10 | .0011 .0018 .0019 .0020 .0029 .0023 .0028 | .0071 .0116 .0178 .0290 .0424 .0522 .0625 | 1.41 1.40 .96 .65 .63 .41 .41 |

* Significant at the 5% level.

** Significant at the 1% level.

| | | | | | | | <u></u> | | |
|--------|------|----------|------------|---------------------|--------|-----------|---------|---------------------|------|
| Strate | egy | | ā | 1934 - 1986 σ(d) | 5 t | | d | 1966 - 1986 σ(d) | t |
| | | , | . <u>.</u> | Panel A | : With | Leverage | <u></u> | | |
| Power | -75 | | .0003 | .0042 | 1.15 | | . 0001 | .0012 | . 57 |
| Power | - 50 | | .0005 | .0063 | 1.09 | | .0001 | .0018 | . 56 |
| Power | - 30 | | .0008 | .0094 | 1.23 | | .0002 | .0030 | . 55 |
| Power | -15 | | .0004 | .0120 | . 52 | | .0003 | .0057 | .53 |
| Power | -10 | | .0006 | .0123 | . 70 | | .0005 | .0083 | . 51 |
| Power | - 5 | | .0002 | .0170 | . 20 | | .0006 | .0144 | . 36 |
| Power | - 2 | | .0006 | .0279 | . 30 | | .0017 | .0272 | . 58 |
| Power | 0 | | .0029 | .0519 | .81 | | .0015 | .0460 | . 30 |
| Power | . 5 | | .0046 | .0681 | . 98 | | .0050 | .0536 | .85 |
| Power | 1 | | .0062 | .0793 | 1.13 | | .0035 | .0654 | .49 |
| | | | ~ | Panel B : | Withou | t Leverag | ,e | | |
| Power | - 75 | | .0003 | .0042 | 1.15 | | .0001 | .0012 | . 57 |
| Power | - 50 | | .0005 | .0063 | 1.09 | | .0001 | .0018 | . 56 |
| Power | - 30 | | .'0008 | . 0094 | 1.23 | | .0002 | .0030 | .55 |
| Power | -15 | | .0001 | .0084 | . 20 | | .0003 | .0057 | . 53 |
| Power | -10 | | . 0002 | . 0102 | . 23 | | .0005 | .0083 | . 51 |
| Power | - 5 | | .0002 | .0160 | .16 | | .0006 | .0141 | .41 |
| Power | - 2 | 3. | .0010 | . 0228 | . 64 | | .0010 | .0194 | .49 |
| Power | 0 | X | .0017 | .0368 | . 66 | | .0018 | .0314 | .60 |
| Power | . 5 | | .0005 | .0374 | . 21 | | .0025 | .0323 | .70 |
| Power | 1 | | .0003 | .0405 | .10 | | .0023 | .0363 | . 59 |

Paired t-tests : 12 vs. 8 Industry Universe, Equal-Weighted (32 quarter estimating period, simple probability assessment approach)

206

, .

Table 30b

| | , _, _, _, _, _, _, _, _, _, _, _, | 1934 - 1986 | 5 | | 1966 - 1980 | 6 |
|-----------|---------------------------------------|-------------|-------------|--------|-------------|------|
| Strategy | d | σ(d) | t | đ | $\sigma(d)$ | t |
| | | Panel A | : With Leve | erage | | |
| Power -75 | .0008 | .0049 | 2.33* | . 0004 | .0023 | 1.52 |
| Power -50 | .0012 | .0073 | 2.32* | .0006 | .0034 | 1.51 |
| Power -30 | .0012 | .0091 | 1.89* | .0009 | .0056 | 1.50 |
| Power -15 | .0012 | .0141 | 1.24 | .0017 | .0107 | 1.46 |
| Power -10 | .0005 | .0170 | .41 | .0024 | .0154 | 1.43 |
| Power -5 | .0001 | .0257 | .03 | .0032 | .0232 | 1.26 |
| Power -2 | .0015 | .0396 | . 57 | .0058 | .0402 | 1.32 |
| Power O | .0047 | .0710 | .96 | .0058 | .0686 | .78 |
| Power .5 | .0068 | .0980 | 1.00 | .0030 | .0879 | . 31 |
| Power 1 | .0068 | .1373 | .72 | .0128 | .1236 | .95 |
| | · · · · · · · · · · · · · · · · · · · | Panel B : | Without Le | verage | | |
| Power -75 | .0008 | .0049 | 2.33* | 0004 | .0023 | 1.52 |
| Power -50 | .0012 | .0073 | 2.32* | .0006 | .0034 | 1.51 |
| Power -30 | .0012 | .0091 | 1.89* | . 0009 | .0056 | 1.50 |
| Power -15 | .0002 | .0119 | . 24 | .0017 | .0107 | 1.46 |
| Power -10 | .0002 | .0162 | .21 | .0024 | .0154 | 1.43 |
| Power -5 | .0009 | .0217 | . 58 | .0030 | .0193 | 1.43 |
| Power -2 | .0013 | .0334 | . 57 | .0043 | .0292 | 1.34 |
| Power O | .0045 | .0566 | 1.16 | .0047 | .0495 | .87 |
| Power 5 | . 0044 | .0648 | . 99 | .0004 | .0594 | .68 |
| Power 1 | .0030 | .0729 | . 60 | 0062 | .0743 | 77 |

Paired t-tests : 12 vs. 8 Industry Universe, Value-Weighted (32 quarter estimating period, simple probability assessment approach)

* Significant at the 5% level.

Table 31a

| | | | | - B | | • |
|-----------|--------|--------------------|--------------|--------|---------------------|--------|
| Strategy | 1 ā | .934 - 198 σ(d) | 6 t | ā | 1966 - 1980 σ(d) | 6 t |
| | | Panel A | : With Leve | erage | | |
| Power -75 | 0007 | .0060 | -1.75* | 0007 | .0040 | -1.51 |
| Power -50 | 0010 | .0086 | -1.66* | 0010 | .0059 | -1.50 |
| Power -30 | 0011 | .0093 | -1.69 | 0016 | .0097 | -1.48 |
| Power -15 | 0014 | .0162 | -1.30 | 0029 | .0186 | -1.44 |
| Power -10 | 0013 | .0221 | 84 | 0041 | . 0269 | -1.40 |
| Power -5 | 0017 | .0350 | 71 | 0047 | .0454 | 96 |
| Power -2 | 0031 | .0546 | 82 | 0046 | .0716 | 59 |
| Power O | .0047 | .0846 | . 80 | .0194 | .0894 | 1.99* |
| Power .5 | .0110 | .1312 | 1.22 | .0293 | .1206 | 2.23* |
| Power 1 | .0165 | .1800 | 1.33 | .0527 | .1709 | 2.82** |
| | | Panel B | : Without Le | verage | | · · |
| Power -75 | 0007 | .0060 | -1.75* | 0007 | .0040 | -1.51 |
| Power -50 | 0010 | .0086 | -1.66* | 0010 | .0059 | -1.50 |
| Power -30 | 0010 | .0090 | -1.66* | 0016 | .0097 | -1.48 |
| Power -15 | 0010 | .0153 | 91 | 0029 | .0186 | -1.44 |
| Power -10 | 0012 | .0209 | 81 | 0041 | .0269 | -1.40 |
| Power -5 | 0013 | .0296 | 64 | 0027 | .0381 | 64 |
| Power -2 | . 0004 | .0432 | .13 | .0034 | .0533 | . 58 |
| Power 0 | .0027 | .0623 | . 64 | .0131 | .0630 | 1.90* |
| Power .5 | .0052 | .0823 | .93 | .0215 | .0718 | 2.75** |
| Power 1 | . 0046 | .1001 | . 67 | .0273 | . 0930 | 2.69** |
| | | | | | | |

Paired t-tests : 12 vs. 24 Industry Universe, Equal-Weighted (32 quarter estimating period, simple probability assessment approach)

* Significant at the 5% level.

** Significant at the 1% level.

| | | | 1934 - 198 | 6 | · · · | 1966 - 198 | 6 |
|----------|------------|--------|-------------|-------------|--------|------------|------|
| Strateg | у | ā | $\sigma(d)$ | t | d | σ(d) | t |
| | | | Panel A | : With Leve | erage | | ~ |
| Power - | 75 | 0011 | .0116 | -1.44 | .0001 | .0035 | .26 |
| Power - | 50 | 0006 | .0113 | 82 | .0002 | .0052 | .27 |
| Power -3 | 30 | 0001 | .0152 | 11 | .0003 | .0086 | . 29 |
| Power - | 15 | 0000 | .0230 | 01 | .0006 | .0167 | .34 |
| Power -1 | 10 | 0003 | .0267 | 19 | .0011 | .0243 | .40 |
| Power · | - 5 | .0005 | .0373 | .21 | .0025 | .0420 | . 54 |
| Power - | - 2 | .0038 | .0568 | .97 | .0080 | .0708 | 1.04 |
| Power | 0 | .0138 | .0978 | 2.06* | .0170 | .1179 | 1.33 |
| Power . | . 5 | .0164 | .1282 | 1.86* | .0236 | .1547 | 1.40 |
| Power | 1 | .0204 | .1613 | 1.84* | .0298 | .1904 | 1.43 |
| | | | Panel B : | Without Lev | verage | | |
| Power -7 | 75 | 0011 | .0116 | -1.44 | .0001 | .0035 | .26 |
| Power -5 | 50 | 0003 | .0113 | 83 | .0002 | .0052 | .27 |
| Power -3 | 30 | .0003 | .0145 | . 34 | .0003 | .0086 | . 29 |
| Power -1 | L 5 | .0001 | .0187 | .07 | .0006 | .0167 | . 34 |
| Power -1 | L O | .0003 | .0238 | .19 | .0011 | .0243 | .40 |
| Power - | - 5 | . 0007 | .0327 | . 32 | .0026 | .0392 | .61 |
| Power - | · 2 | .0037 | .0483 | 1.11 | .0064 | .0604 | .97 |
| Power | 0 | . 0064 | .0723 | 1.28 | .0094 | .0926 | .93 |
| Power . | 5 | .0051 | .0723 | 1.02 | .0075 | .0890 | .77 |
| Power | 1 | . 0069 | .0781 | 1.29 | .0054 | .0905 | .55 |

Paired t-tests : 12 vs. 24 Industry Universe, Value-Weighted (32 quarter estimating period, simple probability assessment approach)

* Significant at the 5% level.

| | | (simpl | e probabil | lity assessme | nt approach) | | |
|----------|---------|--|------------|---------------|--------------|-------------|------|
| | |] | 1936 - 198 | 6 | | 1966 - 1986 | 5 |
| Strate | egy | ā | σ(d) | t | d | σ(d) | ° t |
| <u> </u> | · · · · | ان من من | Panel A | : With Lever | cage | | |
| Power | -75 | 0006 | .0030 | -2.93** | .0001 | .0019 | .72 |
| Power | - 50 | 0009 | .0044 | -2.92** | . 0002 | .0028 | . 73 |
| Power | - 30 | 0011 | .0064 | -2.48** | . 0004 | .0046 | .75 |
| Power | -15 | 0018 | .0111 | -2.33* | .0008 | .0090 | . 80 |
| Power | -10 | 0016 | .0140 | -1.62 | .0012 | .0132 | .85 |
| Power | - 5 | 0023 | .0224 | -1.44 | .0018 | .0232 | .71 |
| Power | - 2 | 0026 | .0343 | -1.10 | .0031 | .0376 | . 76 |
| Power | 0 | .0047 | .0638 | 1.06 | .0097 | .0560 | 1.59 |
| Power | . 5 | .0044 | .0739 | . 86 | .0019 | .0590 | . 29 |
| Power | 1 | 0052 | .0850 | 87 | 0016 | .0748 | 19 |
| | | | Panel B | Without Lev | erage | | |
| Power | - 75 | 0006 | . 0030 | -2.93** | . 0001 | .0019 | .72 |
| Power | - 50 | 0009 | .0044 | -2.92** | .0002 | .0028 | . 73 |
| Power | - 30 | 0011 | . 0064 | -2.48** | ,0004 | .0046 | .75 |
| Power | -15 | 0012 | . 0098 | -1.74* | . 0008 | .0090 | . 80 |
| Power | -10 | 0014 | .0135 | -1.49 | .0012 | .0132 | . 85 |
| Power | - 5 | 0017 | .0204 | -1.20 | .0015 | .0225 | , 60 |
| Power | - 2 | .0012 | .0300 | . 56 | .0046 | .0347 | 1.21 |
| Power | 0 | .0018 | .0410 | .63 | .0041 | .0399 | . 94 |
| Power | . 5 | . 0004 | .0451 | . 14 | .0034 | .0451 | . 70 |
| Power | 1 | 0039 | .0437 | -1.27 | ,0015 | .0429 | . 33 |

Paired t-tests : 32 vs. 28 Quarter Estimating Period, - 12 Equal-Weighted Industry Indices (simple probability assessment approach)

Table 32a

đ

* Significant at the 5% level.

** Significant at the 1% level.

 $\overset{\bullet}{\overset{\bullet}}$

₽

Table 32b

۰,

| <u></u> | | (51mp1 | e probabi | | | | |
|---------|------|---------|------------|--------------|--------|-----------|------|
| - | | | 1936 - 198 | 6 | _ 1 | 966 - 198 | 6 |
| Strate | egy | đ | σ(d) | t | d | σ(d) | t |
| <u></u> | | <u></u> | Panel A | : With Leve | erage | | |
| Power | -75 | 0005 | .0054 | -1.35 | . 0003 | .0018 | 1.34 |
| Power | - 50 | 0006 | .0061 | -1.51 | . 0004 | .0027 | 1.35 |
| Power | - 30 | 0009 | .0083 | -1.52 | . 0007 | .0045 | 1.37 |
| Power | -15 | 0013 | .0113 | -1.68* | .0013 | .0087 | 1.41 |
| Power | -10 | 0014 | .0139 | -1.48 | .0020 | .0128 | 1.45 |
| Power | - 5 | 0012 | .0224 | 76 | .0035 | .0240 | 1.33 |
| Power | - 2 | 0004 | .0326 | 19 | .0047 | .0375 | 1.15 |
| Power | 0 | .0015 | .0495 | .43 | .0030 | .0489 | :56 |
| Power | . 5 | .0021 | .0653 | .45 | .0031 | .0609 | .46 |
| Power | 1 | 0044 | .0687 | 92 | 0089 | .0931 | 88 |
| | | | Panel B | : Without Le | verage | | |
| Power | - 75 | 0005 | .0054 | -1.35 | . 0003 | .0018 | 1.34 |
| Power | - 50 | 0006 | .0061 | -1.51 | . 0004 | .0027 | 1.35 |
| Power | - 30 | 0008 | .0066 | -1.70* | .0007 | .0045 | 1.37 |
| Power | -15 | 0010 | .0096 | -1.46 | .0013 | .0087 | 1.41 |
| Power | -10 | 0010 | .0129 | -1.09 | .0020 | .0128 | 1.45 |
| Power | - 5 | . 0000 | .0203 | .01 | .0035 | .0237 | 1.34 |
| Power | - 2 | .0013 | .0280 | .66 | .0052 | .0329 | 1,45 |
| Power | 0 | 0023 | .0314 | -1.07 | 0012 | .0333 | 34 |
| Power | .5 | 0046 | .0386 | -1.70* | 0040 | .0481 | 76 |
| Power | 1 | 0044 | .0463 | -1.34 | 0075 | .0657 | 104 |

Paired t-tests : 32 vs. 28 Quarter Estimating Period, 12 Value-Weighted Industry Indices (simple probability assessment approach)

* Significant at the 5% level.

| | | _ 1 | .936 - 1986 | 6 | _ 1 | 966 - 1986 | 5 | | | | |
|---------|------------|--------|-------------|--------------|--------|------------|-------|--|--|--|--|
| Strateg | у | d | σ(d) | t | d | σ(d) | t | | | | |
| | | | Panel A | : With Lever | age | | | | | | |
| Power - | 75 | .0004 | .0033 | 1.67* | .0003 | .0020 | 1.28 | | | | |
| Power - | 50 | .0006 | .0050 | 1.62 | .0004 | .0029 | 1.28 | | | | |
| Power - | 30 | .0010 | .0077 | 1.85* | .0007 | .0048 | 1.28 | | | | |
| Power - | 15 | .0009 | .0117 | 1.07 | .0013 | .0092 | 1.28 | | | | |
| Power - | 10 | .0011 | .0158 | 1.01 | .0019 | .0133 | 1.28 | | | | |
| Power | - 5 | .0009 | .0254 | .51 | .0038 | .0238 | 1.47 | | | | |
| Power | - 2 | .0009 | .0395 | .31 | .0058 | .0409 | 1.30 | | | | |
| Power | , 0 | . 0044 | .0663 | .95 | .0104 | .0717 | 1.33 | | | | |
| Power | . 5 | . 0045 | .0953 | .68 | .0079 | .0982 | . 73 | | | | |
| Power | 1 | .0010 | .0935 | .15 | .0022 | .0908 | . 22 | | | | |
| | | | Panel B : | Without Lev | erage | | | | | | |
| Power - | 75 | .0004 | .0033 | 1.67* | . 0003 | .0020 | 1.28 | | | | |
| Power - | 50 | .0006 | .0050 | 1.62 | . 0004 | .0029 | 1.28 | | | | |
| Power - | 30 | .0010 | .0077 | 1.85* | . 0007 | .0048 | 1.28 | | | | |
| Power - | 15 | .0006 | .0103 | . 90 | .0013 | .0092 | 1.28 | | | | |
| Power - | 10 | .0005 | .0142 | .49 | .0019 | .0133 | 1.28 | | | | |
| Power | - 5 | . 0004 | .0238 | .22 | .0039 | .0237 | 1.50 | | | | |
| Power | -2 | .0025 | .0332 | 1.07 | .0069 | .0379 | 1.67* | | | | |
| Power | 0 | .0033 | .0483 | . 97 | . 0042 | .0474 | .81 | | | | |
| Power | .5 | .0006 | .0500 | . 17 | .0007 | .0487 | .13 | | | | |
| Power | 1 | 0022 | .0518 | 62 | 0027 | .0500 | 50 | | | | |

Paired t-tests : 32 vs. 40 Quarter Estimating Period, 12 Equal-Weighted Industry Indices (simple probability assessment approach)

1

Table 33a

* Significant at the 5% level.

÷.,

.

| (simple probability assessment approach) | | | | | | | | | | |
|--|---|---|--|--|--|--|--|--|--|--|
| | · · · · · · · · · · · · · · · · · · · | 1936 - 1980 | 6 | 1 | L966 - 1980 | 5 | | | | |
| egy | d | σ(d) | t | ā | $\sigma(d)$ | t | | | | |
| , | | Panel A | : With Leve | rage . | | | | | | |
| -75 | .0006 | .0033 | 2.37** | .0003 | .0022 | 1.11 | | | | |
| - 50 | . 0008 | .0050 | 2.36** | .0004 | .0033 | 1.10 | | | | |
| - 30 | .0010 | .0069 | 1.96* | .0006 | .0054 | 1.09 | | | | |
| -15 | . 0009 | .0114 | 1.07 | .0012 | .0105 | 1.05 | | | | |
| -10 | .0003 | .0157 | .26 | .0017 | .0152 | 1.02 | | | | |
| - 5 | 0000 | .0261 | 01 | .0029 | .0272 | . 99 | | | | |
| - 2 | .0003 | .0394 | . 12 | .0051 | .0415 | 1.13 | | | | |
| 0 | .0017 | .0703 | . 34 | .0070 | .0700 | .91 | | | | |
| . 5 | .0015 | .0965 | . 22 | .0134 | .0789 | 1.55 | | | | |
| 1 | 0023 | ,0986 | 34 | 0020 | .1052 | 18 | | | | |
| | | Panel B : | Without Lev | verage | | | | | | |
| - 75 | . 0006 | .0033 | 2.37** | .0003 | .0022 | 1.11 | | | | |
| - 50 | .0008 | .0050 | 2.36** | .0004 | .0033 | 1.10 | | | | |
| - 30 | .0010 | .0069 | 1.96* | .0006 | .0054 | 1.09 | | | | |
| -15 | .0003 | .0104 | . 36 | .0012 | .0105 | 1.05 | | | | |
| -10 | 0002 | .0150 | 15 | .0017 | .0152 | 1.02 | | | | |
| - 5 | 0005 | . 0246 | 31 | .0027 | .0264 | . 92 | | | | |
| - 2 | .0009 | .0358 | . 35 | .0051 | .0380 | 1.24 | | | | |
| 0 | .0021 | .0522 | . 57 | .0052 | .047 3 | 1.01 | | | | |
| . 5 | . 0009 | .0556 | . 24 | .0029 | .0566 | .47 | | | | |
| 1 | 0020 | .0555 | 51 | 0033 | .0586 | 52 | | | | |
| | - 75 - 50 - 30 - 15 - 10 - 5 - 2 0 . 5 1 - 75 - 50 - 30 - 15 - 10 - 5 - 2 0 . 5 1 - 15 - 10 - 5 - 2 0 . 5 1 | (simpl -75 .0006 -50 .0008 -30 .0010 -15 .0009 -10 .0003 -50000 -2 .0003 0 .0017 .5 .0015 10023 -75 .0006 -50 .0008 -30 .0010 -15 .0003 -100023 -100002 -50005 -2 .0009 0 .0021 .5 .0009 10020 | $(simple probabil) 1936 - 1984 \sigma(d)Panel A-75 .0006 .0033-50 .0008 .0050-30 .0010 .0069-15 .0009 .0114-10 .0003 .0157-50000 .0261-2 .0003 .03940 .0017 .0703.5 .0015 .096510023 ,0986Panel B :-75 .0006 .0033-50 .0008 .0050-30 .0010 .0069-15 .0003 .0104-100002 .0150-50005 .0246-2 .0009 .03580 .0021 .0522.5 .0009 .055610020 .0555$ | $(simple probability assessme} 1936 - 1986 agy d \sigma(d) tPanel A : With Leve-75 .0006 .0033 2.37**-50 .0008 .0050 2.36**-30 .0010 .0069 1.96*-15 .0009 .0114 1.07-10 .0003 .0157 .26-5 .0000 .0261 .01-2 .0003 .0394 .120 .0017 .0703 .34.5 .0015 .0965 .221 .0023 ,0986 .34Panel B : Without Leve-75 .0006 .0033 2.37**-50 .0008 .0050 2.36**-30 .0010 .0069 1.96*-15 .0003 .0104 .36-10 .0002 .0150 .15-5 .0005 .0246 .31-2 .0009 .0358 .350 .0021 .0522 .57.5 .0009 .0556 .241 .0020 .0555 .51$ | $\begin{array}{c c c c c c c c c c c c c c c c c c c $ | $\begin{array}{c c c c c c c c c c c c c c c c c c c $ | | | | |

Paired t-tests : 32 vs. 40 Quarter Estimating Period, 12 Value-Weighted Industry Indices (simple probability assessment approach)

Table 33b

* Significant at the 5% level.

** Significant at the 1% level.

3

ζ.

Table 34a

| | | L934 - 1986 | • • |] | .966 - 1986 | 1986 | |
|-----------|-------|--------------|-----------|---------|-------------|------|--|
| Strategy | d | <i>o</i> (d) | t | ā | σ(d) | t | |
| | | Panel A | : With Le | verage | | | |
| Power -75 | 0000 | .0022 | 23 | . 0000 | .0028 | . 10 | |
| Power -50 | 0001 | .0033 | 28 | .0000 | .00/2 | . 10 | |
| Power -30 | .0001 | .0049 | .16 | .0001 | .0069 | .10 | |
| Power -15 | 0001 | .0091 | 22 | .0001 | .0133 | .09 | |
| Power -10 | 0002 | .0125 | 23 | .0002 | .0193 | . 08 | |
| Power -5 | 0004 | .0225 | 29 | .0003 | .0348 | .09 | |
| Power -2 | 0008 | .0405 | 30 | .0010 | .0631 | .15 | |
| Power O | .0016 | .0735 | . 32 | .0068 | .1149 | . 54 | |
| Power .5 | .0002 | .0824 | .04 | 0008 | .1292 | 06 | |
| Power l | .0049 | .0950 | .75 | .0062 | .1458 | . 39 | |
| | | Panel B : | Without I | everage | | | |
| Power -75 | 0000 | .0022 | 23 | . 0000 | .0028 | . 10 | |
| Power -50 | 0001 | .0033 | 28 | .0000 | .0042 | .10 | |
| Power -30 | .0001 | .0049 | .16 | .0001 | .0069 | . 10 | |
| Power -15 | 0001 | .0087 | 12 | .0001 | .0133 | . 09 | |
| Power -10 | 0002 | .0125 | 23 | .0002 | .0193 | .08 | |
| Power -5 | 0004 | .0223 | 27 | , 0006 | .0343 | .15 | |
| Power -2 | .0003 | .0382 | .10 | .0032 | .0594 | .49 | |
| Power O | .0009 | .0567 | . 24 | .0034 | .0883 | . 35 | |
| Power .5 | .0019 | .0560 | .48 | . 0033 | .0879 | . 35 | |
| Power 1 | .0013 | .0546 | . 36 | .0011 | .0853 | . 12 | |

Paired t-tests : Simple Probability Assessment Approach vs.Inflation Adapter to Simple Probability Assessment Approach, 12 Equal-Weighted Industry Indices(32 quarter estimating period)

S.

Table 34b

| | | 1934 - 198 | 6 | <i>A</i> | L966 - 198 | 6 |
|-----------|-------|------------|------------|----------|--|-------|
| Strategy | ā | σ(d) | t | d | $\sigma(d)$ | t |
| | بو . | Panel A | : With Lev | erage | | |
| Power -75 | .0001 | .0023 | .42 | 0001 | .0027 | 39 |
| Power -50 | .0001 | .0034 | .41 | 0002 | .0041 | 37 |
| Power -30 | .0001 | .0051 | . 34 | 0003 | .0067 | 35 |
| Power -15 | 0000 | .0094 | 03 | 0004 | .0131 | 29 |
| Power -10 | .0000 | .0123 | °.05 | 0005 | .0177 | 24 |
| Power -5 | 0003 | .0215 | 22 | 0012 | .0329 | 34 |
| Power -2 | 0012 | .0291 | 61 | 0029 | .0453 | 59 |
| Power O | 0019 | .0501 | 54 | 0068 | .0774 | 81 |
| Power .5 | 0007 | .0619 | 16 | 0054 | . 0964 | 52 |
| Power 1 | 0089 | .0952 | -1.37 | 0229 | .1499 | -1.40 |
| | | Panel B : | Without Le | everage | ************************************** | |
| Power -75 | .0001 | .0023 | .42 | 0001 | .0027 | 39 |
| Power -50 | .0001 | .0034 | .41 | 0002 | .0041 | 37 |
| Power -30 | .0001 | ,0051 | . 35 | 0003 | .0067 | 35 |
| Power -15 | 0000 | .0091 | 02 | 0004 | .0131 | 29 |
| Power -10 | 0002 | .0116 | 19 | 0008 | .0167 | 41 |
| Power -5 | 0004 | .0169 | 38 | 0012 | .0256 | 42 |
| Power -2 | 0003 | .0258 | 17 | 0011 | .0400 | 25 |
| Power O | 0012 | .0410 | 43 | 0050 | .0631 | 72- |
| Power .5 | 0022 | .0497 | 64 | - , 0064 | .0772 | 76 |
| Power l | 0059 | .0619 | -1.39 | 0149 | .0973 | -1.41 |

Paired t-tests : Simple Probability Assessment Approach vs.Inflation Adapter to Simple Probability Assessment Approach, 12 Value-Weighted Industry Indices(32 quarter estimating period)

| | | 1934 - 198 | 6 | 1 | 196 <u>6</u> - 1986 | 5 | | | |
|-----------|----------|------------|----------------|--------|---------------------|-------------|--|--|--|
| Strategy | d | σ(d) | t | d | σ(d) | Ċ | | | |
| | | Panel A | : With Leve | rage | | | | | |
| Power -75 | 0005 | .0028 | -2.69** | 0002 | .0023 | 79 | | | |
| Power -50 | 0008 | .0041 | -2.67** | 0003 | .0035 | 78 | | | |
| Power -30 | 0011 | .0063 | -2.60** | 0004 | .0057 | 75 | | | |
| Power -15 | 0017 | .0116 | -2.10* | 0008 | .0111 | 69 | | | |
| Power -10 | 0021 | .0161 | -1.86* | 0011 | .0160 | 62 | | | |
| Power -5 | 0023 | .0267 | -1.26 | 0006 | .0272 | 21 | | | |
| Power -2 | 0028 | .0416 | 96 | .0009 | .0416 | . 20 | | | |
| Power 0 | .0054 | .0826 | .96 | .0118 | .0718 | 1.51 | | | |
| Power .5 | .0056 | .1006 | .81 | .0027 | .0785 | . 31 | | | |
| Power 1 | .0002 | .0947 | .02 | .0052 | .0920 | . 52 | | | |
| | ~ | Panel B | : Withou't Lev | verage | | ······· | | | |
| Power -75 | 0005 | .0028 | -2.69** | 0002 | .0023 | 79 | | | |
| Power -50 | 0008 | .0041 | -2.67** | 0003 | .0035 | 78 | | | |
| Power -30 | 0011 | .0063 | -2.60** | 0005 | .0057 | 75 | | | |
| Power -15 | 0015 | .0113 | -1.94* | 0008 | .0111 | 69 | | | |
| Power -10 | 0016 | .0159 | -1.49 | 0011 | .0160 | 62 | | | |
| Power -5 | 0022 | .0253 | -1.29 | 0007 | .0257 | 27 | | | |
| Power -2 | .0005 | .0375 | . 20 | .0046 | .0379 | 1.12 | | | |
| Power O | .0022 | . 0598 | 53 | .0058 | .0530 | 1.00 | | | |
| Power .5 | 0003 | .0558 | 06 | .0044 | .0573 | , 71 | | | |
| Power 1 | 0012 | .0530 | 33 | .0031 | .0574 | .49 | | | |

Paired t-tests : Simple Probability Assessment Approach vs. "Sum-ofthe-Digits" Probability Assessment Approach, 12 Equal-Weighted Industry Indices(32 quarter estimating period)

* Significant at the 5% level.

** Significant at the 1% level.

Table 35b

| | | 1934 - 198 | 6 | | 1966 - 1986 | | | |
|---------------------------------------|----------|---------------|-----------|----------------|-------------|--------|--|--|
| Strategy | đ | σ(d) | t | d d | σ(d) | t | | |
| · · · · · · · · · · · · · · · · · · · | | Banel A | : With L | everage | | | | |
| P | | - | | | | | | |
| Power -75 | - ,0004 | .0034 | -1.78* | 0001 | .0027 | 36 | | |
| Power -50 | 0006 | .0050 | -1.77* | 0002 | .0040 | 35 | | |
| Power -30 | 0011 | .0076 | -2.08* | 0002) | .0065 | 31 | | |
| Power -15 | 0015 | .0130 | -1.70* | 0003 | .0126 | 23 | | |
| Power -10 | 0021 | .0167 | -1.82* | 0004 | .0173 | 21 | | |
| Power -5 | 0022 | ·.0291 | -1.12 | .0001 | .0320 | .03 | | |
| Power -2 | 0007 | .0421 | 24 | .0019 | .0449 | . 39 | | |
| Power 0 | .0031 | .0713 | .64 | .0006 | ·.0717 | .77 | | |
| Power 5 | .0053 | . 1015 | .75 | 0005 | .0875 | 05 | | |
| Power 1 | 0005 | .1123 | .07 | 0187 | .1164 | -1.47 | | |
| | | Panel B | : Without | Leverage | | | | |
| Power -75 | 0004 | .0034 | -1.78* | 0001 | .0027 | 36 | | |
| Power -50 | 0006 | .0050 | -1.77* | 0002 | .0040 | 35 | | |
| Power -30, | 0011 | .0076 | -2.08* | 0002 | .0065 | 31 | | |
| Power -15 | 0014 | .0122 | -1.64 | 0003 | .0126 | 23 | | |
| Power -10 | - • 0018 | .0165 | -1,55 | 0005 | .0171 | 24 | | |
| Power -5 | 0011 | .0249 | 63 | .0007 | .0249 | .25 | | |
| Power -2 | .10006 | .0363 | . 24 | .0028 | .0416 | .61 | | |
| Power 0 | .0001 | .0463 | .02 | - 0037 | 0474 | - 72 | | |
| Power 5 | - 0037 | 0489 | -1 12 | - 0086 | 0 611 | -1 20 | | |
| Power 1 | - 0038 | 0594 | - 92 | - 0148 | 0776 | -1 75* | | |
| , on we a | .0050 | .0374 | | 0140 | .0770 | -1./// | | |

Paired t-tests : Simple Probability Assessment Approach vs. "Sum-ofthe-Digits" Probability Assessment Approach, 12 Value-Weighted Industry Indices(32 quarter estimating period)

* Significant at the 5% level.

Table 36a

| Strategy | | 5 | • | • | ····· | | | |
|----------|------|--------|-------------------------|--------------|---------|---------------------|--------|--|
| | | a d | 1934 - 1986 d σ(d) t | | | .966 - 1986 σ(d) | 5 t | |
| . · | | | Panel A | : With Lever | age | | | |
| Power | -75 | .0018 | .0098 | 2.62** | .0007 | .0035 | 1.74* | |
| Power | - 50 | .0026 | .0148 | 2.55** | .0010 | .0052 | 1.73* | |
| Power | - 30 | .0042 | .0233 | 2.62** | .0016 | .0086 | 1.72* | |
| Power | -15 | .0063 | .0347 | 2.66** | . 00 30 | .0165 | 1.68* | |
| Power | -10 | .0082 | .0395 | 3.04** | .0043 | .0238 | 1.64 | |
| Power | - 5 | .0105 | .0515 | 2.97** | .0067 | .0420 | 1.47 | |
| Power | - 2 | .0117 | .0688 | 2.48** | .0095 | .0696 | 1.24 | |
| Power | 0 | .0085 | .0757 | 1.63 | .0130 | .0895 | 1.33 | |
| Power | . 5 | .0013 | .0935 | .21 | .0063 | .0956 | . 60 | |
| Power | 1 | .0033 | .1348 | . 36 | .0053 | .1616 | . 30 | |
| | | | Panel B : | Without Leve | erage | 7 | | |
| Power | - 75 | .0018 | .0098 | 2.62** | . 0007 | .0035 | 1.74* | |
| Power | - 50 | .0026 | .0148 | 2.55** | .0010 | .0052 | 1.73* | |
| Power | - 30 | .0042 | .0233 | 2.62** | .0016 | .0086 | 1.72* | |
| Power | -15 | .0062 | . 0274 | 3.28** | .0030 | .0165 | 1.68* | |
| Power | -10 | .0073 | .0316 | 3.38** | .0043 | .0238 | 1.64 | |
| Power | - 5 | .0090 | .0442 | 2.97** | .0068 | .0418 | 1.49 | |
| Power | - 2 | .0081 | .0495 | 2.38** | .0077 | .0550 | 1.28 | |
| Power | 0 | .0012 | .0469 | . 38 | .0058 | .0509 | 1.05 | |
| Power | . 5 | . 0009 | .0551 | . 23 | .0087 | .0573 | 1.40 | |
| Power | 1 | .0027 | .0724 | .55 | .0050 | .860 | . 53 | |

Paired t-tests : Simple Probability Assessment Approach vs. "All-of-History" Probability Assessment Approach, 12 Equal-Weighted Industry Indices(32 quarter estimating period)

* Significant at the 5% level.

** Significant at the 1% level.

Į.

Table 36b

| | | 1934 - 1980 | 6 | | 1966 - 1986 | 4 |
|-----------|------------------------------|-------------|----------------|-------------|-------------|-------|
| Strategy | \overline{d} $\sigma(d)$ t | | đ | $\sigma(d)$ | t | |
| | | Panel A | : With Leve | rage | | |
| Power -75 | .0023 | .0115 | 2.91** | . 0006 | .0041 | 1.24 |
| Power -50 | .0034 | .0171 | 2.88** | .0008 | .0061 | 1.23 |
| Power -30 | .0047 | .0251 | 2.75** | .0013 | .0099 | 1.22 |
| Power -15 | .0066 | .0366 | 2.63** | .0024 | .0191 | 1.17 |
| Power -10 | .0072 | .0441 | 2.37** | .0034 | .0276 | 1.13 |
| Power -5 | .0086 | .0553 | 2.27* | .0044 | .0453 | . 89 |
| Power -2 | . 0099 | .0681 | ,2.13 * | .0046 | .0728 | . 58 |
| Power O | .0077 | .0907 | 1.24 | 0009 | .1075 | 08 |
| Power .5 | .0018 | .1188 | .22 | .0024 | .1162 | .19 |
| Power 1 | 0071 | .1312 | 79 | 0175 | .1519 | -1.06 |
| | | Panel B : | Without Lev | verage | | |
| Power -75 | .0023 | .0115 | 2.91** | . 0006 | .0041 | 1.24 |
| Power -50 | .0034 | .0171 | 2.88** | .0008 | .0061 | 1.23 |
| Power -30 | . 0047 | .0251 | 2.75** | .0013 | .0099 | 1.22 |
| Power -15 | .0056 | .0305 | 2.66** | .0024 | .0191 | 1.17 |
| Power -10 | .0063 | .0352 | 2.61** | .0034 | .0276 | 1.13 |
| Power -5 | .0077 | .0444 | 2.51** | .0042 | .0425 | . 90 |
| Power -2 | .0059 | .0491 | 1.74* | .0030 | .0569 | .48 |
| Power O | .0015 | .0633 | . 34 | 0032 | .0715 | 42 |
| Power .5 | 0025 | .0730 | 49 | 0031 | .0831 | 34 |
| Power 1 | 0042 | .0720 | 85 | 0116 | .0821 | -1.30 |
| | | | | | | |

Paired t-tests : Simple Probability Assessment Approach vs. "All-of-History" Probability Assessment Approach, 12 Value-Weighted Industry Indices(32 quarter estimating period)

* Significant at the 5% level.

** Significant at the 1% level.

 $\langle |$

Table 37a

| | _ | 1934 - 198 | 6 | _ 1 | .966 - 1986 | 5 | | | | |
|-----------|--------|------------|--------------|--------|-------------|--------------|--|--|--|--|
| Strategy | d | σ(d) | t | d | 0(d) | t | | | | |
| · · · · | • | Panel A | : With Leven | cage | | . | | | | |
| Power -75 | .0003 | .0013 | 3.38** | .0002 | ,0013 | 1.14 | | | | |
| Power -50 | .0004 | .0019 | 3.30** | .0002 | .0019 | 1.13 | | | | |
| Power -30 | .0007 | .0028 | 3.75** | .0004 | .0032 | 1.11 | | | | |
| Power -15 | .0013 | .0054 | 3.53** | .0007 | .0061 | 1.05 | | | | |
| Power -10 | .0016 | .0077 | 3.02** | .0010 | .0088 | . 99 | | | | |
| Power -5 | .0025 | .0135 | 2.66** | .0011 | .0156 | .65 | | | | |
| Power -2 | .0034 | .0200 | 2.46** | .0016 | .0251 | . 59 | | | | |
| Power O | .0027 | .0325 | 1.19 | .0007 | .0401 | .15 | | | | |
| Power .5 | 0013 | .0467 | 41 | 0075 | .0539 | -1.27 | | | | |
| Power 1 | .0007 | .0416 | .24 | .0050 | .0514 | .88 - | | | | |
| <u></u> | / | Panel B : | Without Lev | verage | | | | | | |
| Power -75 | .0003 | .0013 | 3,38** | . 0002 | .0013 | 1.14 | | | | |
| Power -50 | . 0004 | .0019 | 3.30** | .0002 | .0019 | 1.13 | | | | |
| Power -30 | .0007 | .0028 | 3.75** | .0004 | .0032 | 1.11 | | | | |
| Power -15 | .0012 | .0053 | 3.26** | . 0007 | .0061 | 1.05 | | | | |
| Power -10 | .0016 | .0077 | 2.99** | .0010 | .0088 | . 99 | | | | |
| Power -5 | .0024 | .0132 | 2.61** | .0011 | .0155 | . 68 | | | | |
| Power -2 | .0017 | .0162 | 1.50 | .0013 | .0214 | . 55 | | | | |
| Power O | .0007 | .0254 | .41 | 0004 | .0320 | 10 | | | | |
| Power .5 | 0010 | .0193 | 73 | 0018 | .0271 | 60 | | | | |
| Power 1 | 0014 | .0181 | -1.16 | 0001 | .0242 | 04 | | | | |

Paired t-tests : Simple Probability Assessment Approach vs. "Disaster States" Scenario When J - 1, 12 Equal-Weighted Industry Indices(32 quarter estimating period)

* Significant at the 5% level.

** Significant at the 1% level.

۰**۰**.

Table 37b

| States" Scenario When J = 1, 12 Value-Weighted Industry Indices(32 quarter estimating period) | | | | | | | | | |
|--|---------|------------|-------------|-------------|---------|-------|--|--|--|
| | | 1934 - 198 | 6 | 1966 - 1986 | | | | | |
| Strategy | d | σ(d) | t | <u>d</u> | σ(d) | t | | | |
| | | Panel A | : With Leve | rage | <u></u> | | | | |
| Power -75 | .0003 | .0013 | 3.28** | .0001 | .0014 | 1.00 | | | |
| Power -50 | .0004 | .0019 | 3.23** | .0002 | .0020 | . 99 | | | |
| Power -30 | .0005 | .0029 | 2.35** | .0003 | .0033 | .96 | | | |
| Power -15 | . 0008 | .0055 | 2.25* | .0006 | .0063 | . 90 | | | |
| Power -10 | .0011 | .0076 | 2.01* | .0008 | .0092 | .85 | | | |
| Power -5 | .0014 | .0127 | 1.65 | .0006 | .0160 | . 36 | | | |
| Power -2 | .0018 | .0205 | 1.32 | .0001 | .0264 | .05 | | | |
| Power O | .0013 | .0367 | . 52 | 0022 | .0502 | 40 | | | |
| Power .5 | .0006 | .0505 | .16 | .0016 | .0620 | . 23 | | | |
| Power 1 | 0040 | .0694 | 84 | 0015 | .0982 | -1.36 | | | |
| - | | Panel B : | Without Lev | erage | | | | | |
| Power -75 | .0003 | .0013 | 3.28** | .0001 | .0014 | 1.00 | | | |
| Power -50 | .0004 | .0019 | 3.23** | .0002 | . 0020 | . 99 | | | |
| Power -30 | · .0005 | .0029 | 2.35** | .0003 | .0033 | .96 | | | |
| Power -15 | . 0008 | .0051 | 2.28* | .0006 | .0063 | .90 | | | |
| Power -10 | .0010 | .0074 | 2.04* | .0008 | .0092 | .85 | | | |
| Power -5 | .0014 | .0123 | 1.67* | .0004 | .0156 | .23 | | | |
| Power -2 | . 0003 | .0178 | . 20 | .0003 | .0246 | .13 | | | |
| Power O | .0005 | .0301 | . 23 | 0029 | .0421 | 62 | | | |
| Power .5 | .0002 | .0352 | .09 | 0023 | .0481 | 44 | | | |
| Power 1 | 0012 | .0486 | 37 | 0060 | .0715 | 77 | | | |

Paired t-tests : Simple Probability Assessment Approach vs. "Disaster

* Significant at the 5% level.

۵

۴

** Significant at the 1% level.

Table 38

Summary Results of Jensen's Performance Index Regression

 $r_{pt} - r_{Lt} - \alpha_p + \beta_p (r_{mt} - r_L) + \epsilon_{pt}$

When the 10 Active Strategies Manage 12 Equal-Weighted Industries (32 quarter estimating period, simple probability assessment)

| | Mean | S.Dev | Min | Max | Pos 5% | Numbe itive 1% | or of Negative 5% 1% |
|--------------------------------|------------------|-----------------|------------------|-------------------|-----------|----------------------|----------------------------|
| | Pe | nel A : | 1934-86, | With Leverag | e | | |
| R ² | .444 | .153 | . 256 | .652 | _ | | |
| α (in %) (t-value) | .661 (1.649) | .417 (.427) | .098 (.704) | 1.333 (2.127) | 5 | 10 | •••• |
| β (t-value) | .872 (13.462) | .891 (4.256) | .063 (8.509) | 2.724 (19.841) | ^ | | |
| | Pe | anel B : | 1966-86, | With Leverag | e | | |
| R ² | . 589 | .045 | . 554 | .667 | | | |
| α (in %) (t-value) . | 1.108 (2.007) | 1.000 (.202) | .071 (1.661) | 2.731 (2.435) | 9 | 10 1 | |
| β (t-value) | .779 (10.918) | .775 (1.075) | .041 (10.097) | 2.724 (12.817) | | | |
| | Pan | el C : 19 | 934-86, W | ithout Lever | age | | |
| R ² | . 480 | . 174 | . 256 | .714 | | | |
| α (in %) (<i>t</i> -value) | .393 (1.548) | .202 (.606) | .098 (.378) | .743 (2.320) | 4 | 10 | |
| β (t-value) | .604 (14.800) | .501 (5.287) | .063 (8.509) | 1.388 (22.901) | × | | |
| | Pan | el D : 19 | 966-86, W | ithout Lever | age | | |
| R ² | .601 | . 059 | . 554 | . 691 | | | |
| a (in %) (t-value) | .786 (2.000) | .584 (.103) | .071 (1.739) | 1.545 (2.145) | 10 | 10 | · · · |
| β (t-value) | .568 (11.248) | .482 (1.473) | .041 (10.097) | 1.250 (13.544) | | | |

Note: t-values in parentheses.

.,e

Summary Results of Jensen's Performance Index Regression

$r_{pt} - r_{Lt} - \alpha_p + \beta_p (r_{mt} - r_{Lt}) + \epsilon_{pt}$

When the 10 Active Strategies Manage 12 Value-Weighted Industries (32 quarter estimating period, simple probability assessment)

| | | | | | Number of | | | |
|----------|--|--|--|---|--|---|---|--|
| | | | | Po | sitive | Neg | ative | |
| Mean | S.Dev | Min | Max | 5% | 18 | 58 | 18 | |
| P | anel A : | 1934-86, | With Lever | age | | | | |
| . 378 | . 157 | .186 | . 626 | | | | | |
| . 371 | .233 | 102 | . 676 | | 9 | | 1 | |
| (1.214) | (.550) | (089) | (1.753) | 3 | ••• | • • • | ••• | |
| .750 | . 749 | .063 | 2.420 | | | | | |
| (11.655) | (4.064) | (6.923) | (18.752) | | | | | |
| Pa | anel B : | 1966-86, | With Lever | age | | <u> </u> | | |
| .454 | .087 | . 386 | .633 | | فعسر | | | |
| . 177 | . 514 | -1.233 | .713 | | 9 | | 1 | |
| (.760) | (.624) | (914) | (1.153) | • • • | | • • • | • • • | |
| .637 | .638 | .035 | 1.848 | | | | | |
| (8.390) | (1,602) | (7.184) | (11.897) | | | | | |
| Pan | el C : 19 | 034-86, W | ithout Leve | erage | | | | |
| . 419 | . 186 | .186 | . 703 | | | | | |
| . 168 | . 243 | 368 | .424 | | 8 | | 2 | |
| (.965) | (.861) | (721) | (1.753) | 3 | • • • | • • • | • • • | |
| . 529 | .428 | .063 | 1.286 | | • | | | |
| (13.044) | (5.323) | (6.923) | (22.318) | | | | | |
| Pan | el D : 19 | 966-86, W | ithout Leve | erage | | | - | |
| . 479 | . 110 | . 386 | . 687 | | | | | |
| .025 | . 435 | -1.045 | .493 | | 7 | | 3 | |
| (.563) | (.854) | (-1.362) | (1.153) | • • • | • • • | ••• | • • • | |
| . 477 | .414 | .035 | 1.185 | | | | | |
| (8,923) | (2.167) | (7.184) | (13.415) | | | | | |
| | Mean P .378 .371 (1.214) .750 (11.655) Pa .454 .177 (.760) .637 (8.390) Pan .419 .168 (.965) .529 (13.044) Pan .479 .025 (.563) .477 (8.923) | MeanS.DevPanel A :.378.157.371.233 (1.214) $(.550)$.750.749 (11.655) (4.064) Panel B :.454.087.177.514 $(.760)$ $(.624)$.637.638 (8.390) (1.602) Panel C :Panel C :.419.186.168.243 $(.965)$ $(.861)$.529.428 (13.044) (5.323) Panel D :.479.110.025.435 $(.563)$ $(.854)$.477.414 (8.923) (2.167) | MeanS. DevMinPanel A : 1934-86,.378.157.186.371.233.102(1.214)(.550)(089).750.749.063(11.655)(4.064)(6.923)Panel B : 1966-86,.454.087.386.177.514.454.087.386.177.514.637.638.035(8.390)(1.602)(7.184)Panel C : 1934-86, W.419.186.168.243.368(.965)(.861)(721).529.428.063(13.044)(5.323)(6.923)Panel D : 1966-86, W.479.110.386.025.435.1.045(.563)(.854)(-1.362).477.414.035(8.923)(2.167).7.184) | MeanS. DevMinMaxPanel A :1934-86, With Lever.378.157.186.626.371.233 102 .676(1.214)(.550)(089)(1.753).750.749.0632.420(11.655)(4.064)(6.923)(18.752)Panel B :1966-86, With Lever.454.087.386.633.177.514-1.233.713(.760)(.624)(914)(1.153).637.638.0351.848(8.390)(1.602)(7.184)(11.897)Panel C :1934-86, Without Leve.419.186.186.703.168.243368.424(.965)(.861)(721)(1.753)S29.428.0631.286(13.044)(5.323)(6.923)(22.318)Panel D :1966-86, Without Leve.479.110.386.687.025.435-1.045.493(.563)(.854)(-1.362)(1.153).477.414.0351.185(8.923)(2.167)(7.184)(13.415) | MeanS.DevMinMaxPo.Panel A :1934-86, With Leverage.378.157.186.626.371.233102.676(1.214)(.550)(089)(1.753)3.750.749.0632.420(11.655)(4.064)(6.923)(18.752)Panel B :1966-86, With Leverage.454.087.386.633.177.514-1.233.713(.760)(.624)(914)(1.153).637.638.0351.848(8.390)(1.602)(7.184)(11.897)Panel C :1934-86, Without Leverage.419.186.186.703.168.243368.424(.965)(.861)(721)(1.753)3Sign529.428.0631.286(13.044)(5.323)(6.923)(22.318)Panel D :1966-86, Without Leverage.479.110.386.687.025.435-1.045.493(.563)(.854)(-1.362)(1.153)477.414.0351.185(8.923)(2.167)(7.184)(13.415) | MeanS.DevMinMaxPositive 5%Numb Positive 5%Panel A : 1934-86, With Leverage.378.157.186.626.371.233102.6769(1.214)(.550)(089)(1.753)3750.749.0632.420(11.655)(4.064)(6.923)(18.752)Panel B : 1966-86, With Leverage.454.087.386.633.177.514-1.233.7139(.760)(.624)(914)(1.153)637.638.0351.848(8.390)(1.602)(7.184)(11.897)Panel C : 1934-86, Without Leverage.419.186.186.703.168.243368.4248(.965)(.861)(721)(1.753)3529.428.0631.286(13.044)(5.323)(6.923)(22.318)Panel D : 1966-86, Without Leverage.479.110.386.687.025.435-1.045.493653)(.854)(-1.362)(1.153)477.414.0351.185477.414.0351.185 | MeanS.DevMinMaxNumber of PositiveNeg StPanel A : 1934-86, With Leverage.378.157.186.626.371.233.102.6769(1.214)(.550)(089)(1.753)3750.749.0632.420(11.655)(4.064)(6.923)(18.752)Panel B : 1966-86, With Leverage.454.087.386.633.177.514-1.233.7139(.760)(.624)(914)(1.153)637.638.0351.848(8.390)(1.602)(7.184)(11.897)Panel C : 1934-86, Without Leverage.419.186.186.703.637.638.0631.286(13.044)(5.323)(6.923)(22.318)Panel D : 1966-86, Without Leverage.479.110.386.687.025.435-1.045.4937(.563)(.854)(-1.362)(1.153)477.414.0351.185477.414.0351.185 | |

Note: t-values in parentheses.

ţ

Summary Results of Treynor-Mazuy's Market Timing Regression

$\mathbf{r}_{\text{pt}} - \mathbf{r}_{\text{Lt}} - \alpha_{\text{p}} + \beta_{1\text{p}}(\mathbf{r}_{\text{mt}} - \mathbf{r}_{\text{Lt}}) + \beta_{2\text{p}}(\mathbf{r}_{\text{mt}} - \mathbf{r}_{\text{Lt}})^2 + \epsilon_{\text{pt}}$

Without a Correction for Heteroscedasticity When the 10 Active Strategies Manage 12 Equal-Weighted Industries (32 quarter estimating period, simple probability assessment)

١

| | | | | Number of | | | |
|------------------|--|---|--|--|--|---|--|
| Мали | C Dest | M.L., | Maria | Pos | itive | Negative | |
| mean | 5.Dev | M1N | max | J K C | | | |
| Ĭ | Panel A : 1 | 1934-86, | With Leverage | | | | |
| .459 | . 146 | . 282 | . 652 | | | | |
| .004 | .010 | 024 | .012 | | 9 | L | |
| (1.875) | (1.362) | (-1.561) | (2.903) | 7 | 6 | ••• ••• | |
| .864 (13-595) | .864 (3.984) | .066 | 2.626 (19.546) | | | | |
| . 359 | 1,444 | 406 | 4.510 | | 3 | 7 | |
| (866) | (2.266) | (-2.711) | (4.614) | 1 | 1 | 5 4 | |
| P | anel B : 1 | 966-86, | With Leverage | | | | |
| . 603 | .040 | , 572 | . 674 | | | | |
| .006 | .006 | . 003 | .018 | | 10 | | |
| (.778) | (.192) | (.600) | (1.304) | ••• | • • • | | |
| .784 (11.107) | .779 (1.011) (| .041 | 2.152 (12.929) | | | | |
| . 695 | . 586 | .056 | 1.877 | j | 10 | | |
| (1.645) | (.306) | (.908) | (1.843) | 7 | • • • | •••• | |
| Par | nel C : 193 | 34-86, W | ithout Leverag | ,e | | | |
| . 492 | . 166 | . 282 | . 719 | | | | |
| . 004 | .005 | 007 | .010 | | 8 | 2 | |
| (1.805) | (1.375) (| (-1.010) | (3.018) | 7 | 6 | · · · · · · · · | |
| .603 (14.884) | .492 (4.993) | .066 | 1.361 | | | | |
| .039 | . 504 | - 306 | 1 211 | | 3 | , | |
| (~.911) | (1.904) (| -2.711) | (2.813) | 2 | 1 | 5 3 | |
| Pan | el D : 196 | 6-86, Wi | lthout Leverag | е | | · · ······ ··· ··· ··· ··· ··· ··· ··· | |
| . 613 | .051 | . 572 | . 691 | | | | |
| .006 | .005 | .003 | .014 | 1 | 0 | | |
| (.929) | (.303) | (.686) | (1.428) | | | | |
| . 570 | .483 | .041 | 1.252 | | | | |
| (11.300) | (1.555) (279 | 056 | 760 | 1 | 0 | | |
| (1.352) | (.668) | (.263) | (1.843) | 6 | | · · · · | |
| | Mean I .459 .004 (1.875) .864 (13.595) .359 (866) P .603 .006 (.778) .784 (11.107) .695 (1.645) Par .492 .004 (1.805) .603 (14.884) .039 (911) Pan .613 .006 (.929) .570 (11.380) .311 (1.352) | MeanS.DevPanel A : 1.459.146.004.010 (1.875) (1.362) .864.864 (13.595) (3.984) .359 1.444 (866) (2.266) Panel B : 1.603.040.006.006 $(.778)$ $(.192)$.784.779 (11.107) (1.011) .695.586 (1.645) $(.306)$ Panel C : 193.492.166.004.005 (1.805) (1.375) (1.805) (1.375) (1.805) (1.375) (1.884) (4.993) .039.504 (911) (1.904) Panel D : 196.613.051.006.005 $(.929)$ $(.303)$.570.483 (11.380) (1.335) $(.311)$.228 (1.352) $(.668)$ | MeanS.DevMinPanel A : 1934-86,.459.146.282.004.010.024(1.875)(1.362)(-1.561).864.864.066(13.595)(3.984)(8.933).3591.444.406(866)(2.266)(-2.711)Panel B : 1966-86,.603.040.572.006.006.003.778)(.192)(.600).784.779.041(11.107)(1.011)(10.331).695.586.056(1.645)(.306)(.908)Panel C : 1934-86, W.492.166.282.004.005.005.007(1.805)(1.375)(1-1.010).603.492.066(14.884)(4.993)(8.933).039.504.306(911)(1.904)(-2.711)Panel D : 1966-86, Wi.613.051.570.483.041(11.380)(1.335).311.228.056(1.352)(.668)(.263) | MeanS. DevMinMaxPanel A : 1934-86, With Leverage.459.146.282.652.004.010 024 .012(1.875)(1.362)(-1.561)(2.903).864.864.0662.626(13.595)(3.984)(8.933)(19.546).3591.644 406 4.510(866)(2.266)(-2.711)(4.614)Panel B : 1966-86, With Leverage.603.040.572.674.006.006.003.018(.778)(.192)(.600)(1.304).784.779.0412.152(11.107)(1.011)(10.331)(12.929).695.586.0561.877(1.645)(.306)(.908)(1.843)Panel C : 1934-86, Without Leverage.492.166.282.719.004.005007.010(1.805)(1.375)(-1.010)(3.018).603.492.0661.361(14.884)(4.993)(8.933)(22.550).039.5043061.211(911)(1.904)(-2.711)(2.813)Panel D : 1966-86, Without Leverage.613.051.572.691.006.005.003.014(.929)(.303)(.686)(1.428).570.483.0411.252(11.380)(1.335)(10.331) | MeanS. DevMinMaxFosPanel A :1934-86, With Leverage.459.146.282.652.004.010024.012(1.875)(1.362)(-1.561)(2.903)7.864.864.0662.626(13.595)(3.984)(8.933)(19.546).3591.4444064.510(866)(2.266)(-2.711)(4.614)1Panel B :1966-86, With Leverage.603.040.572.674.006.006.003.018(.778)(.192)(.600)(1.304).784.779.0412.152(11.107)(1.011)(10.331)(12.929).695.586.0561.877(1.645)(.306)(.908)(1.843)Panel C :1934-86, Without Leverage.492.166.282.492.166.282.603.492.066.613.051.572.691.006.003.0141(1.805)(1.375).039.504306.291.066.2813.205.003.0141.929.003.613.051.570.483.041.252.11.28.570.483.3041<.252 | MeanS.DevMinMaxPost Live 5818Panel A :1934-86, With Leverage.459.146.282.652.004.010 024 .0129(1.875)(1.362)(-1.561)(2.903)76.864.864.0662.626(13.595)(3.984)(8.933)(19.546)3.3591.4444064.5103(866)(2.266)(-2.711)(4.614)1Panel B :1966-86, With Leverage.603.040.572.674.006.006.003.01810(.778)(.192)(.600)(1.304)784.779.0412.152(11.107)(1.011)(10.331)(12.929).695.586.0561.87710(1.645)(.306)(.908)(1.843)7.004.005007.0108(1.805)(1.375)(-1.010)(3.018)7.603.492.0661.361(14.848)(4.993)(8.933)(22.550).039.5043061.2113(911)(1.904)(-2.711)(2.813)21Panel D :1966-86, Without Leverage.613.051.572.691006.005.003.01410(.929)(.303)(.686)(1.428)570 | |

Table 40b

Summary Results of Treynor-Mazuy's Market Timing Regression

$\mathbf{r}_{pt} - \mathbf{r}_{Lt} - \alpha_{p} + \beta_{1p}(\mathbf{r}_{mt} - \mathbf{r}_{Lt}) + \beta_{2p}(\mathbf{r}_{mt} - \mathbf{r}_{Lt})^{2} + \epsilon_{pt}$

With a Correction for Heteroscedasticity When the 10 Active Strategies Manage 12 Equal-Weighted Industries (32 quarter estimating period, simple probability assessment)

| | | | | | Numb | per of | |
|--|-----------------|------------------|-----------------|-------------------|-------------------|------------|-------------------|
| | Mean | S.Dev | Min | Max 🕴 | Posi 5% | tive 1% | Negative 5% 1% |
| | F | Panel A : | 1934-86, | With Leverage | | | |
| R ² | - . 996 | . 006 | . 983 | 1.000 | | | |
| a (t-value) | .004 (2.244) | .010 (1.807) | 024 (-1.897) | .012 (3.832) | 7 | 96 | 1 |
| βı (t-value) | .864 (8.166) | .864 | .066 | 2.626 | | | |
| β2 (t-value) | . 359 (323) | 1.444 (1.186) | 406 (-1.196) | 4.510 (2.722) | 1 | 3 | 7 |
| ************************************** | P | anel B : | 1966-86, | With Leverage | | | |
| R ² | . 994 | .007 | . 979 | 1.000 | | | |
| a (t-value) | .006 (.743) | .006 (.167) | .003 (.544) | .018 (1.202) | 10 |) | · · · · |
| βı (t-value) | .784 (7.133) | .779 (.781) | .041 (6.590) | 2.152 (8.708) | | | |
| βz (t-value) | .695 (.838) | .586 (.167) | .056 (.457) | 1.877 (.952) | 10 |) | |
| | Par | nel C : 1 | 934-86, W | ithout Leverag | je | | |
| R ² | . 998 | .003 | . 990 | 1.000 | | | |
| a (t-value) | .004 (2.121) | .005 (1.739) | 007 (-1.173) | .010 (3.832) | ٤ 7 | 3 5 | 2 |
| βı (t·valuo) | .603 (9.348) | .492 (4.601) | .066 (4.142) | 1.361 (17.174) | | | |
| β2 (t-value) | .039 (322) | .504 (1.026) | 306 (-1.196) | 1.211 (1.799) | 3 1 | • • • • | 7 |
| | Pan | el D : 19 | 966-86, W | ithout Leverag | e | | |
| R ² | . 997 | . 001 | . 995 | . 999 | | | |
| a (t-value) | .006 (.881) | .005 (.258) | .003 (.671) | .014 (1.316) | 10 . <i></i> . | | · · · · |
| βı (t-value) | .570 (7.456) | .483 (1.217) | .041 (6.590) | 1.252 (9.543) | | | |
| βı (t-value) | .311 (.692) | .228 (.342) | .056 (.145) | .760 (.952) | 10 | | |

Table 41a

Summary Results of Treynor-Mazuy's Market Timing Regression

$r_{pt} - r_{Lt} - \alpha_{p} + \beta_{1p}(r_{mt} - r_{Lt}) + \beta_{2p}(r_{mt} - r_{Lt})^{2} + \epsilon_{pt}$

Without a Correction for Heteroscedasticity When the 10 Active Strategies Manage 12 Value-Weighted Industries (32 quarter estimating period, simple probability assessment)

| | | | | | | Numb | er of |
|------------------------------|-------------------|-------------------|--------------------|-------------------|-----------|--|-------------------|
| | Mean | S.Dev | Min | Max | Pos 58 | itive 1% | Negative 58 18 |
| | | Panel A : | 1934-86, | With Leverag | 6 | | |
| R ² | . 387 | .154 | . 198 | . 640 | | | |
| α (t-value) | .004 (1.608) | .008 (1.145) | 018 (-1.442) | .011 (2.408) | 7 | 93 | 1 |
| β_1 (<i>t</i> -value) | .751 (11.761) | .737 | .065 | 2.368 (18.475) | | | |
| β2 (t-value) | 070 (-1.122) | .825) (1.423) | 547 (-2.064) | 2.364 (2.845) | 1 | 1 1 | 9 6 |
| | | Panel B : | 1966-86, | With Leverag | e | | ****** |
| R ² | .465 | .083 | .400 | .635 | | | |
| α (t-value) | 001 (.023) | .003 (.242) | 006 (406) | .001 (.226) | | 6 | 4 |
| βı (t-value) | .640 (8.466) | .638 (1.551) | .035 (7.282) | 1.842 (11.800) | | | |
| β2 (t-value) | .391 (1.106) | .589 (.615) | 807 (681) | 1.393 (1.488) | | 9 | 1 |
| | Pa | une1 C : 1 | 934-86, W | Ithout Levera | ige | ************************************** | |
| R ² | . 427 | . 182 | . 198 | . 704 | | | |
| α (t-value) | .004 (1.562) | .004 (1.079) | 004 (702) | .007 (2.408) | 7 | 8 3 | 2 |
| βı (t-value) | . 535 (13.159) | .428 (5.159) | .065 (7.132) | 1.284 (22.023) | | | |
| β2 (t-value) | 267 (-1.480) | .168 (.703) | 510 (-2.103) | .056 (.147) | | 1 | 9 7 |
| | Pa | nel D : 1 | 966-86, W | ithout Levera | ge | | |
| R ² | . 490 | . 107 | . 400 | . 703 | , | | |
| α (t-value) | 001 (.031) | .001 (.221) | 004 (401) | .001 (.226) | | 6 | 4. |
| βı (t-value) | .478 (8.998) | .412 (2.135) | .035 (7.282) | 1.175 (13.537) | | | |
| β2 (t-value) | .102 (.792) | .544 (1.075) | -1.355 (-2.055) | .812 (1.534) | | 8 | 2 1 |

Table 41b

Summary Results of Treynor-Mazuy's Market Timing Regression

$\mathbf{r}_{\text{pt}} - \mathbf{r}_{\text{Lt}} - \alpha_{\text{p}} + \beta_{1\text{p}}(\mathbf{r}_{\text{mt}} - \mathbf{r}_{\text{Lt}}) + \beta_{2\text{p}}(\mathbf{r}_{\text{mt}} - \mathbf{r}_{\text{Lt}})^2 + \epsilon_{\text{pt}}$

With a Correction for Heteroscedasticity When the 10 Active Strategies Manage 12 Value-Weighted Industries (32 quarter estimating period, simple probability assessment)

| | | | | | | Number of | | |
|-----------------|-----------------|-----------------|--------------------------|-------------------|---------------|--------------|-------------|------------|
| | Mean | S.Dev | Min | Max | Pos 5% | itive 1% | Negat 5% | tiye 1% |
| | P | anel A : | 1934-86, | With Leverage | | | <u></u> | |
| R ² | .987 | .015 | . 951 | . 999 | - • | | | |
| α (t-value) | .004 (1.887) | .008 (1.350) | 018 (-1.431) | .011 (3.076) | 7 | 9 5 | | |
| βı (t-value) | .751 (7.530) | .737 (2.932) | .065 (4.176) | 2.368 (13.515) | 2 | | | |
| β2 (t-value) | 070 (616) | .825 、(.790) | 547 (- <u>1.1</u> 71) | 2.364 (1.545) | ••• | 1 | 9 ··· | ; |
| | P | anel B : | 1966-86, | With Leverage | 1 | | <u> </u> | |
| R ² | . 990 | . 020 | . 932 | 1.000 | - | | | |
| a (t-value) | 001 (.030) | .003 (.237) | 006 (385) | .001 (.230) | •••• | 6 | 4 | ••• |
| βı (t-value) | .640 (5.979) | .638 (.954) | .035 (5.374) | 1.842 (8.485) | | | | |
| β2 (t·value) | .391 (.654) | .589 (.371) | 807 (395) | 1.393 (.841) | | 9 | 1 | • • • |
| | Pan | el C : 1 | 934-86, W | ithout Levera | ge | | | |
| R ² | . 994 | . 008 | .976 | 1.000 | | | | |
| a (t-value) | .004 (1.802) | .004 (1.289) | 004 (697) | .007 (3.076) | 6 | 8 5 | 2 | ••• |
| βı (t-value) | .535 (8.683) | .428 (4.378) | .065 (4.716) | 1.284 (18.587) | | | | |
| β2 (t-value) | 267 (795) | .168 (.399) | 510 (-1.171) | .056 (.095) | ••• | 1 | 9 | ••• |
| <u> </u> | Pan | el D : 1 | 966-86, W | ithout Leverag | ge | | | |
| R ² | . 998 | . 003 | . 990 | 1.000 | | | | |
| a (t-value) | 001 (.041) | .001 (.210) | 004 (357) | .001 (.230) | | 6 | 4 | ••• |
| βı (t-value) | .478 (6.362) | .412 (1.644) | .035 (5.374) | 1.175 (10.832) | | | | |
| 02 (t-value) | . 102 (.444) | .544 (.715) | -1.355 (-1.494) | .812 (.854) | | 8 | 2 | ••• |

Table 42a

Summary Results of Henriksson-Merton's Market Timing Regression

$\mathbf{r}_{pt} - \mathbf{r}_{Lt} - \alpha_{p} + \beta_{1p}(\mathbf{r}_{mt} - \mathbf{r}_{Lt}) + \beta_{2p}\mathbf{y}_{t} + \epsilon_{pt}$

Without a Correction for Heteroscedasticity When the 10 Active Strategies Manage 12 Equal-Weighted Industries (32 quarter estimating period, simple probability assessment)

| | Mean | S.Dev | Min | Max | Num Positive 5% 1% | oer of Negativo 5% 1% |
|-----------------|-----------------|------------------|--------------------|-------------------|--------------------------|---|
| ; | P | anel A : | 1934-86, | With Leverage | | n da an |
| R ² | .451 | .153 | , 266 | . 658 | | |
| α (t-value) | 288 (.908) | 1.759 (1.523) | -5,286 (-2,486) | .775 (2.215) | 7 5 | 3 1 |
| βı (t-value) | .998 (7.849) | 1.138 (3.849) | .046 (3.515) | 3.556 (13.342) | | |
| β2 (t-value) | 280 (.175) | .580 (1.778) | 101 (-1.635) | 1.846 (3.796) | 5 3 1 | 5 |
| | Pa | anel B : | 1966-86, | With Leverage | | |
| R ² | . 603 | . 040 | . 572 | . 676 | | |
| α (t-value) | .055 (106) | .269 (.209) | 186 (259) | .618 (.291) | ີ2 | 8 |
| βı (t-value) | .927 (7.121) | .890 (.334) | .053 (6.884) | 2.500 (7.938) | | |
| β2 (t-value) | .302 (1.635) | .246 (.308) | .025 (.908) | .733 (1.839) | 10 7 | • • • • • • • • • • • • • • • • • • • |
| | Pan | el C : 19 | 934-86, W | ithout Leverag | e | |
| R ² | . 485 | . 173 | . 26℃ | . 720 | | |
| α (t-value) | .059 (.924) | .694 (1.385) | -1.508 (-1.626) | .587 (2.215) | 7 5 | 3 |
| βı (t-value) | .648 (8.501) | .588 (4.198) | .046 (3.515) | 1.618 (14.837) | | |
| β2 (t-value) | .098 (.069) | .199 (1.434) | 081 (-1.635) | .510 (2.407) | 5 2 1 | 5 |
| | Pan | el D : 19 |)66-86, W | ithout Leverag | e | |
| R ² | . 613 | .051 | . 572 | . 692 | - | |
| a (t-value) | .277 (.094) | .485 (.450) | 184 (256) | 1.070 (.802) | 4 | 6 |
| βı (t-value) | .639 (7.053) | .503 (.191) | .053 (6.884) | 1.320 (7.338) | | |
| β2 (t-value) | .146 (1.376) | .098 (.621) | .025 (.403) | .332 (1.839) | 10 6 | • • • • • • • • • |

Note: t-values in parentheses.

Table 42b

Summary Results of Henriksson-Merton's Market Timing Regression

$\mathbf{r}_{pt} - \mathbf{r}_{Lt} - \alpha_{p} + \beta_{1p}(\mathbf{r}_{mt} - \mathbf{r}_{Lt}) + \beta_{2p}\mathbf{y}_{t} + \epsilon_{pt}$

With a Correction for Heteroscedasticity When the 10 Active Strategies Manage 12 Equal-Weighted Industries (32 quarter estimating period, simple probability assessment)

| | | · · · · · · · · · · · · · · · · · · · | | | | Num | Number of | | |
|----------------------|---|---------------------------------------|------------------|--------------------|-------------------|-------------------|-------------------|--|--|
| | 2 | Mean | S,Dev | Min | Max | Positive 5% 1% | Negative 5% 1% | | |
| | : | , I | Panel A : | 1934-86, | With Leverage | e . | | | |
| R ² | | . 996 | . 006 | . 982 | 1.000 | - | | | |
| a (t-value) | | 288 (.791) | 1.759 (1.320) | -5.286 (-2.125) | .775 (1.962) | 7 4 | 3 1 ≠ | | |
| βı (t-value) | | . 998 (4. 320) | 1.138 (1.383) | .046 (2.772) | 3.556 (6.494) | | | | |
| β2 (t-value) | | 280 (.172) | .580 (.983) | 101 (764) | 1.846 (2.238) | 5 1 | َ 5 ۰۰۰ ۰۰۰ | | |
| | | P | anel B : | 1966-86, | With Leverage | • | | | |
| R ² | | . 989 | .007 | . 976 | 1.000 | - | | | |
| α (t-value) | | .055 (095) | .269 (.175) | 186 (224) | .618 (.235) | 2 | 8 | | |
| βı (t-value) | | .927 (3.927) | .890 (.280) | .053 (3.747) | 2.500 (4.629) | с. | | | |
| β2 (t-value) | | .302 (1.002) | .246 (.195) | .025 (.557) | .733 、 (1.139) | 10 7 | * | | |
| | | Par | nel C : 1 | 934-86, W | ithout Levera | ge | | | |
| R ² | | . 998 | . 002 | .995 | 1.000 | | | | |
| a (t-value) | | .059 (.776) | .694 (1.222) | -1.508 (-1.445) | .587 (1.962) | 7 4 | 3 | | |
| βı (t-value) | | .648 (4.799) | .588 (1.938) | .046 (2.772) | 1.618 (8.710) | | | | |
| β2 (t-value) | | .098 (.134) | .1/99 (.836) | 081 (764) | .510 (1.552) | 5 | 5 | | |
| | | Pa. | el D : 1 | 966-86, W | ithout Levera | ge | • | | |
| .2 R ² | | . 992 | . 006 | . 978 | 1.000 | | | | |
| α (t-value) | | .277 (.070) | .485 (.375) | 184 (221) | 1.070 (.664) | 4 | 6 | | |
| βı (t-value) | | .639 (3.966) | . 503 (. 314) | .053 (3.747) | 1.320 (4.500) | | | | |
| β2 (t-value) | | .146 (.848) | .098 (.381) | .025 (.263) | .332 (1.139) | 10 | | | |

Note: t-values in parentheses.

3

Table 43a

Summary Results of Henriksson-Merton's Market Timing Regression

$\mathbf{r}_{pt} - \mathbf{r}_{Lt} - \alpha_{p} + \beta_{1p}(\mathbf{r}_{mt} - \mathbf{r}_{Lt}) + \beta_{2p}\mathbf{y}_{t} + \epsilon_{pt}$

Without a Correction for Heteroscedasticity When the 10 Active Strategies Manage 12 Value-Weighted Industries (32 quarter estimating period, simple probability assessment)

| • | Mean | S.Dev | Min | Мах | Numb Positive 5% 1% | oer of Negative 5% 1% |
|---------------------------------------|-----------------|--------------------------|--------------------|-------------------|---------------------------|-----------------------------|
| | P | anel A : | 1934-86, | With Leverag | je | |
| R ² | . 379 | . 157 | . 188 | . 629 | | હ |
| a (t-value) | .291 (.995) | .755 (.825) | -1.923 (-1.064) | .724 (1.753) | 9 3 | 1 |
| βı (t-value) | .760 (6.379) | .815 (2.792) | .054 (3.352) | 2.662 (11.753) | | |
| β2 (t-value) | .024 (296) | .175 (.634) | 082 (913) | .537 (1.299) | 3 | * 7 |
| · · · · · · · · · · · · · · · · · · · | P | anel B : | 1966-86, | With Loverag | çe | |
| R ² | .463 | . 085 | · . 396 | . 638 | | |
| α (t-value) | 289 (298) | .558 (.258) | -1.510 (775) | .669 (.296) | 1 | 9 |
| βı (t-value) | .702 (5.162) | .644 (.532) | .044 (4.729) | 1.651 (6.182) | | |
| β2 (t-value) | .134 (.935) | .288 (<i>.</i> 677) | 546 (-1.045) | .574 (1.382) | · 9 | 1 |
| nt - | Pai | nel C : 1 | 934-86, W | Ithout Lever | age | |
| R ² | . 421 | . 186 | . 188 | . 704 | | |
| α (<i>t</i> -value) | .429 (1.159) | .177 (.536) | .116 (.143) | .676 (1.753) | 10 、3 | · · · · |
| βı (t-value) | .494 (6.828) | .409 (3.034) | .054 (3.352) | 1.221 (12.049) | - 1 | |
| βz (t-value) | 077 (710) | .046 (.159) | 155 (913) | - 021 (- 363) | · · · · | 10 |
| | Pat | nel D : 1 | 966-86, W | ithout Lever | age | |
| R ² | . 488 | . 109 | . 396 | . 704 | | |
| α (t-value) | 030 (159) | .453 (.408) | 739 (728) | 1.160 (.920) | 2 | 8 |
| βı (t-value) | .485 (5.175) | . 363 (.49 <u>2</u>) | .044 (4.729) | .956 (6.161) | | ТО |
| β2 (t-value) | .016 (.614) | (1.096) (1.096) | 632 (-2.177) | .353 (1.510) | 8 | 2 1 |

Table 43b

Summary Results of Henriksson-Merton's Market Timing Regression

$\mathbf{r}_{pt} - \mathbf{r}_{Lt} - \alpha_p + \beta_{1p}(\mathbf{r}_{mt} - \mathbf{r}_{Lt}) + \beta_{2p}\mathbf{y}_t + \epsilon_{pt}$

With a Correction for Heteroscedasticity When the 10 Active Strategies Manage 12 Value-Weighted Industries (32 quarter estimating period, simple probability assessment)

| | | | | | Number of | | | |
|--------------------|---------|-----------------|-----------|------------------|------------|-------|-----------------|--|
| | Maan | S Dov | Min | May | Pos | itive | Negative | |
| | nean | 5.Dev | | nax | , J8 | | 58 18 | |
| | · P | anel A : | 1934-86, | With Leverag | e | | ч , т. | |
| R ² | . 992 | .007 | .979 | 1.000 | | | | |
| α | . 291 | .755 | -1.,923 | . 724 | | 9 | 1 | |
| (t-value) | (.971) | (.794) | . (828) | (1.778) | 3 | ••• | | |
| βι | . 760 | .815 | . 054 | 2.662 | | | • | |
| (t-value) | (3.742) | (.930) | (2.6/4) | (5.375) | | | | |
| β2 | .024 | .175 | 082 | . 53,7 | | 3 | 7 | |
| (t-value) | . (186) | (.3/0) | (200) | (./39) | ••• | ••• | ••• | |
| _ | Pa | anel B : | 1966-86, | With Leverage | e , | | - | |
| R ² | . 982 | .019 | . 928 | 1.000 | | | 5 × | |
| α | 289 | . 558 | -1.510 | .669 | | 1 | 9 | |
| (<i>t</i> -value) | (268) | (.224) | (644) | (.274) | | | | |
| βι | . 702 | . 644 | · .044 | 1.651 | | | | |
| (t-value) | (3.246) | (.167) | (3.097) | (3,512) | | | | |
| β2 | .134 | .288 | 546 | . 574 | 4 | 9 | 1 | |
| (t-value) | (.646) | (.473) | (728) | (.913) | ••• | • • • | · · · · · · · · | |
| <u></u> | Pan | el C : 1 | 934-86,-W | ithout Levera | ge | | | |
| R ² | , .994 | .006 | . 983 | 1.000 | | | | |
| a | . 429 | .177 | .116 | . 676 | | 10 | | |
| (t-value) | (1.081) | (.576) | (.129) | (1.778) | 3 | ••• | | |
| βι | . 494 | . 409 | .054 | 1.221 | × | | 1. | |
| (t-value) | (4.132) | (1.412) | (2.674) | (6.690) | | | | |
| β2 | - `.077 | .046 | 155 | - <i>.</i> , 021 | • | • • | 10 | |
| (t-value) . | (435) | (.116) | (566) | (-:193) | • • • | | | |
| | Pan | el D : 1 | 966-86, W | ithout Levera | ge | | | |
| R ² | . 989 | .005 | . 983 | . 998 | | 1 | | |
| a | 030 | . 453 | - 739 | 1.160 | | 2 | 8 | |
| (t-value) | (137) | (.389) | (623) | (.917) | . | | •••• | |
| βı | . 485 | . 363 | .044 | .956 | | | | |
| (t-value) | (3.293) | (.234) | (3,097) | (3.810) | | | | |
| β2 | .016 | . 249 | 632 | . 353 | | 8 | ື 2 | |
| (t-value) | (.403) | (. <u>83</u> 2) | (-1.807) | (.901) | | | 1 | |

Note: t-values in parentheses.

æ

Figure la

Geometric Means and Standard Deviations of Annual Portfolio Returns for 10 Power Policies with Twelve Equal-Weighted Industry Indices, 1934-1986 (Quarterly revision, with leverage, 32 quarter estimating period)



Geometric Means and Standard Deviations of Annual Portfolio Returns for 10 Power Policies with Twelve Equal-Weighted Industry Indices, 1966-1986 (Quarterly revision, with leverage, 32 quarter estimating period)


Geometric Means and Standard Deviations of Annual Portfolio Returns for 10 Power Policies with Twelve Equal-Weighted Industry Indices, 1934-1986 (Quarterly revision, without leverage, 32 quarter estimating period)



Geometric Means and Standard Deviations of Annual Portfolio Returns for 10 Power Policies with Twelve Equal-Weighted Industry Indices, 1966-1986 (Quarterly revision, without leverage, 32 quarter estimating period)



Geometric Means and Standard Deviations of Annual Portfolio Returns for 10 Power Policies with Twelve Value-Weighted Industry Indices, 1934-1986 (Quarterly revision, with leverage, 32 quarter estimating period)



Geometric Means and Standard Deviations of Annual Portfolio Returns for 10 Power Policies with Twelve Value-Weighted Industry Indices, 1966-1986 (Quarterly revision, with leverage, 32 quarter estimating period)



Geometric Means and Standard Deviations of Annual Portfolio Returns for 10 Power Policies with Twelve Value-Weighted Industry Indices, 1934-1986 (Quarterly revision, without leverage, 32 quarter estimating period)







Average Excess Returns and Betas of Quarterly Portfolio Returns for 10 Power Policies With Twelve Equal-Weighted Industry Indices and Equal- and Value-Weighted Benchmarks, 1966-86 (32 quarter estimating period, with leverage)



Figure 3

- Adler, Michael and Bernard Dumas. 1983. "International Portfolio Choice and Corporation Finance: A Survey." Journal of Finance, 38 (June): 925-984.
- Agmon, Tamir. 1972. "The Relations among Equity Markets: A Study of Share Price Co-Movements in the United States, United Kingdom, Germany and Japan." Journal of Finance, 27 (September): 839-855.
- Agmon, Tamir. 1973. "Country Risk: The Significance of the Country Factor for Share Price Movements in the United Kingdom, Germany and Japan." Journal of Business, 46 (January): 24-32.
- Arrow, Kenneth J. 1965. "Aspects of the Theory of Risk Bearing." In Yrjo Jahnsson Lectures. Helsinki: The Yrjö Jahnsson Foundation.
- Banz, Rolf. 1981. "The Relationship Between Return and Market Value of Common Stocks." Journal of Financial Economics, 9 (March): 3-18.
- Bawa, Vijay; Stephen Brown; and Roger Klein. 1979. Estimation Risk and Optimal Portfolio Choice. Amsterdam: North Holland.
- Best, Michael J. 1975. "A Feasible Conjugate Direction Method to Solve Linearly Constrained Optimization Problems." Journal of Optimization Theory and Applications, 16 (July): 25-38.
- Brigham, Eugene F. and James L. Pappas. 1969. "Rates of Return on Common Stock." Journal of Business, 42 (July): 302-316.
- Breeden, Douglas T.; Michael R. Gibbons; and Robert H. Litzenberger. 1986. "Empirical Tests of the Consumption-Oriented CAPM." Working Paper, Duke University, (March).
- CRSP Stock Files, Monthly and Daily Data User's Guide. 1987. Center for Research in Security Prices, University of Chicago.
- Dybvig, Philip and Stephen A. Ross. 1985. "The Analytics of Performance Measurement Using a Security Market Line." Journal of Finance, 40 (June): 401-415.

Ŷ

- Fama, Eugene F. 1972. "Components of Investment Performance." Journal of Finance, 27 (June): 551-567.
- Fama, Eugene F. and James D. MacBeth. 1974. "Long-Term Growth in a Short-Term Market." Journal of Finance, 29 (June): 857-885.
- Farrell, James L. Jr. 1974. "Analyzing Covariation of Return to Determine Homogeneous Stock Groupings." Journal of Business, 47 (April): 186-207.

241

- Farrell, James L. Jr. 1975. "Homogeneous Stock Groupings: Implications for Portfolio Management." *Einancial Analysts Journal*, 31 (May-June): 50-62.
- Grauer, Robert R. 1981a. "A Comparison of Growth Optimal and Mean Variance Investment Policies." Journal of Financial and Quantitative Analysis, 16 (March): 1-21.
- Grauer, Robert R. 1981b. "Investment Policy Implications of the Capital Asset Pricing Model." Journal of Finance, 36 (March): 127-141.
- Grauer, Robert R. 1986. "Normality, Solvency, and Portfolio Choice." Journal of Financial and Quantitative Analysis, 21 (September): 265-278.
- Grauer, Robert R. 1987. "Further Ambiguity When Performance is Measured by the Security Market Line." Working Paper, Simon Fraser University, (December).
- Grauer, Robert R. and Nils H. Hakansson. 1981. "Long-Run Investment Theory with Applications to Pension Funds." In Geld, Banken und Wersicherungen, Hermann Göppl and Rudolph Henn, eds. Band I, Königstein: Athenäum.
- Grauer, Robert R. and Nils H. Hakansson. 1982. "Higher Return, Lower Risk: Historical Returns on Long-Run, Actively Managed Portfolios of Stocks, Bonds and Bills, 1936-1978." Financial Analysts Journal, 38 (March-April): 39-53.
- Grauer, Robert R. and Nils H. Hakansson. 1984. "Historical Returns on Portfolios of Stocks, Bonds, and Bills, With and Without Small Stocks." In Geld, Banken und Versicherungen, Hermann Göppl and Rudolf Henn, eds. Band II, Königstein: Athenäum.
- Grauer, Robert R. and Nils H. Hakansson. 1985. "1934-1984 Returns on Levered, Actively Managed Long-Run Portfolios of Stocks, Bonds and Bills." *Financial Analysts Journal*, 41 (September-October): 24-43.
- Grauer, Robert R. and Nils H. Hakansson. 1986. "A Half-Century of Returns on Levered and Unlevered Portfolios of Stocks, Bonds, and Bills, With and Without Small Stocks." Journal of Business, 59 (April): 287-318.
- Grauer, Robert R. and Nils H. Hakansson. 1987. "Gains From International Diversification: 1968-85 Returns on Portfolios of Stocks and Bonds." Journal of Finance, 42 (July): 721-739.
- Grauer, Robert R. and Nils H. Hakansson. 1988. "On Timing the Market: The Empirical Distribution Approach With an Inflation Adapter." Working Paper.
- Grubel, Herbert G. 1968. "Internationally Diversified Portfolios: Welfare Gains and Capital Flows." American Economic Review, 58 (December): 1299-1314.

- Hakansson, Nils H. 1969. "Optimal Investment and Consumption Under Risk, an Uncertain Lifetime and Insurance." International Economic Review, 10 (October): 443-466.
- Hakansson, Nils H. 1969. "Risk Disposition and the Separation Property in Portfolio Selection." Journal of Financial and Quantitative Analysis, 4 (December): 401-416.
- Hakansson, Nils H. 1970. "An Induced Theory" of the Firm Under Risk: the Pure Mutual Fund." Journal of Financial and Quantitative Analysis, 3 (June): 155-178.
- Hakansson, Nils H. 1970. "Optimal Investment and Consumption Strategies Under Risk for a Class of Utility Functions." *Econometrica*, 38 (September): 587-607.
- Hakansson, Nils H. 1971a. "Capital Growth and the Mean-Variance Approach to Portfolio Selection." Journal of Financial and Quantitative Analysis, 6 (January): 517-557.
- Hakansson, Nils H. 1971b. "Multi-Period Mean-Variance Analysis: Toward a General Theory of Portfolio Choice." Journal of Finance, 26 (September): 857-884.
- Hakansson, Nils H. 1971c. "On Optimal Myopic Portfolio Policies, With and Without Serial Correlation of Yields." *Journal of Business*, 44 (July): 324-334.
- Hakansson, Nils H. 1974. "Convergence to Isoelastic Utility and Policy in Multiperiod Portfolio Choice." Journal of Financial Economics, 1 (September): 201-224.
- Henriksson, Roy D. 1984. "Market Timing and Mutual Fund Performance: An Empirical Investigation." *Journal of Business*, 57 (January): 73-96.
- Henriksson, Roy D. and Robert C. Merton. 1981. "On Market Timing and Investment Performance II: Statistical Procedures for Evaluating Forecasting Skills." Journal of Business, 54 (October): 513-533.
- Huberman, Gur and Stephen A. Ross. 1983. "Portfolio Turnpike Theorems, Risk Aversion, and Regularly Varying Utility Functions." *Econometrica*, 51 (September): 1345-1361.
- Ibbotson, Roger G. and Rex A. Sinquefield. 1982. Stocks, Bonds, Bills and Inflation: The Past and the Future. Charlottesville, Va.: Financial Analysts Research Foundation.
- Ibbotson Associates, Inc. 1988. Stocks, Bonds, Bills and Inflation: 1988 Yearbook. Chicago: Ibbotson Associates, Inc.

- Jagannathan, Ravi and Robert A. Korajczyk. 1986. "Assessing the Market Timing Performance of Managed Portfolios." *Journal of Business*, 59 (April): 217-235.
- Jensen, Michael C: 1968. "The Performance of Mutual Funds in the Period 1945-1964." Journal of Finance, 23 (March): 389-416.
- Jorion, Philippe. 1986. "Bayes-Stein Estimation for Portfolio Analysis." Journal of Financial and Quantitative Analysis, 21 (September): 279-292.
- Kallberg, J. G. and William T. Ziemba. 1983. "Comparison of Alternative Utility Functions in Portfolio Selection Problems." Management Science, 29 (November): 1257-1276.
- King, Benjamin F. 1966. "Market and Industry Factors in Stock Price Behavior." Journal of Business, 39 (January): 139-190.
- Kroll, Yoram; Haim Levy; and Harry M. Markowitz. 1984. "Mean-Variance versus — Direct Utility Maximization." Journal of Finance, 39 (March): 47-75.
- Latané, Henry A. and Donald L. Tuttle. 1968. "Industry Analysis Framework for Forming Probability Beliefs." Financial Analysts Journal, 24 (July-August): 51-61.
- Leland, Hayne. 1972. "On Turnpike Portfolios." In *Mathematical Methods in Investment and Finance*, G. P. Szego and K. Shell, eds. Amsterdam: North Holland.
- Lessard, Donald R. 1973. "International Portfolio Diversification: A Multivariate Analysis for a Group of Latin American Countries." Journal of Finance, 28 (June): 619-633.
- Levy, Haim and Marshall Sarnat. 1970. "International Diversification of Investment Portfolios." American Economic Review, 60 (September): 668-675.
- Levy, Haim and Harry M. Markowitz. 1979. "Approximating Expected Utility by a Function of Mean and Variance." American Economic Review, 69 (June): 308-317.
- Livingston, Miles. 1977. "Industry Movements of Common Stocks." Journal of Finance, 32 (June): 861-874.
- Logue, Dennis E. 1982. "An Experiment in International Diversification." Journal of Portfolio Management, 9 (Fall): 22-27.
- Markowitz, Harry M. 1952. "Portfolio Selection." Journal of Finance, 7 (March): 77-91.
- Markowitz, Harry M. 1959. Portfolio Selection: Efficient Diversification of Investments. New York: John Wiley.

- McDonald, John G. 1973. "French Mutual Fund Performance: Evaluation of Internationally Diversified Portfolios." Journal of Finance, 28 (December): 1161-1180.
- Merton, Robert C. 1969 "Lifetime Portfolio Selection Under Uncertainty: The Continuous Time Case." *Review of Economics and Statistics*, 50 (August): 247-257.

Merton, Robert C. 1981. "On Market Timing and Investment Performance I: An Equilibrium Theory of Value for Market Forecasts." Journal of Business, 54 (July): 363-401.

- Meyers, Stephen L. 1973. "A Re-examination of Market and Industry Factors in Stock Price Behavior." *Journal of Finance*, 28 (June): 695-705.
- Mossin, Jan. 1968. "Optimal Multiperiod Portfolio Policies." Journal of Business, 41 (April): 215-229.
- Pratt, John W. 1964, "Risk-Aversion in the Small and in the Large." Econometrica, 32 (January-April): 122,4136.
 - Pulley, Lawrence B. 1981. "A General Mean-Variance Approximation to Expected Utility for Short Holding Periods." Journal of Financial and Quantitative Analysis, 16 (September): 361-373.
 - Pulley, Lawrence B. 1983. "Mean-Variance Approximation to Expected Logarithmic Utility." Operations Research, 31 (July-August): 685-696.
- Reilly, Frank K. and Eugene Drzycimski. 1974. "Alternative Industry Performance and Risk." Journal of Financial and Quantitative Analysis, 9 (June): 423-446.
- Reinganum, Marc. 1981. "Misspecification of Capital Asset Pricing: Empirical Anomalies Based on Earnings Yields and Market Values." Journal of Financial Economics, 9 (March): 19-46.
- Roll, Richard. 1977. "A Critique of the Asset Pricing Theory's Tests: Part 1: On Past and Potential Testability of the Theory." Journal of Financial Economics, 4 (March): 129-176.
- Roll, Richard. 1978. "Ambiguity When Performance is Measured by the Securities Market Line." Journal of Finance, 33 (September): 1051-1069.
- Ross, Stephen A. 1974. "Portfolio Turnpike Theorems for Constant Policies." Journal of Financial Economics, 1 (July): 171-198.
- Samuelson, Paul A. 1969. "Lifetime Portfolio Selection by Dynamic Stochastic Programming." Review of Economics and Statistics, 50 (August): 239-246.
- Sharpe, William F. 1966. "Mutual Fund Performance." Journal of Business, 34 (January): 119-138.

- Sharpe, William F. 1982. "Factors in New York Stock Exchange Security Returns, 1931-1979." Journal of Portfolio Management, (Summer): 5-19.
- Sharpe, William F. 1985. Investments. New Jersey: Prentice-Hall.
- Solnik, Bruno H. 1974. "Why Not Diversify Internationally Rather than Domestically." Financial Analysts Journal, 30 (July-August):
- Solnik, Bruno H. and Bernard Noetzlin. 1982. Optimal International Asset Allocation." Journal of Portfolio Management, 9 (Fall): 11-21.
- Sorensen, Eric H. and Terry Burke. 1986. "Portfolio Returns from Active Industry Group Rotation." *Financial Analysts Journal*, 42 (September-October): 43-50
- Standard Industrial Classification Manual. 1967. Washington: Office of Statistical Standards of the Bureau of the Budget.
- Treynor, Jack L. 1965. "How to Rate the Management of Mutual Funds." Harvard Business Review, 43 (January-February): 63-75.
- Treynor, Jack L. and Mazuy, Kay K. 1966. "Can Mutual Funds Outguess the Market." Harvard Business Review, 44 (July-August): 131-136.
- Tysseland, Milford S. "Further Tests of the Validity of the Industry Approach to Investment Analysis." Journal of Financial and Quantitative Analysis, 6 (March): 835-847.
- White, Halbert. 1980. "A Heteroscedasticity-Consistent Covariance Matrix Estimator and a Direct Test for Heteroscedasticity." Econometrica, 48 (May): 817-838.
- Ziemba, William T; C. Parkan; and R. Brooks-Hill. 1974. "Calculation of Investment Portfolios with Risk-Free Borrowing and Lending." Management Science, 21 (October): 209-222.